# EFFECTS OF ANTI-OXIDATIVE AND CHOLESTEROL LOWERING CAPACITIES OF SELECTED EDIBLE-MEDICINAL MUSHROOMS TOWARDS AMELIORATION OF ALZHEIMER'S DISEASE

# MOHAMMAD AZIZUR RAHMAN

# FACULTY OF SCIENCE UNIVERSITY OF MALAYA KUALA LUMPUR

## EFFECTS OF ANTI-OXIDATIVE AND CHOLESTEROL LOWERING CAPACITIES OF SELECTED EDIBLE-MEDICINAL MUSHROOMS TOWARDS AMELIORATION OF ALZHEIMER'S DISEASE

## MOHAMMAD AZIZUR RAHMAN

## THESIS SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

FACULTY OF SCIENCE UNIVERSITY OF MALAYA KUALA LUMPUR

## UNIVERSITY OF MALAYA ORIGINAL LITERARY WORK DECLARATION

Name of Candidate:Mohammad Azizur Rahman

Registration/Matric No:SHC120054

Name of Degree: Doctor of Philosophy

Title of Project Paper/Research Report/Dissertation/Thesis ("this Work"):Effects of Anti-oxidative and Cholesterol Lowering Capacities of Selected Edible-Medicinal Mushrooms Towards Amelioration of Alzheimer's Disease

Field of Study:Biochemistry

I do solemnly and sincerely declare that:

- (1) I am the sole author/writer of this Work;
- (2) This Work is original;
- (3) Any use of any work in which copyright exists was done by way of fair dealing and for permitted purposes and any excerpt or extract from, or reference to or reproduction of any copyright work has been disclosed expressly and sufficiently and the title of the Work and its authorship have been acknowledged in this Work;
- (4) I do not have any actual knowledge nor do I ought reasonably to know that the making of this work constitutes an infringement of any copyright work;
- (5) I hereby assign all and every rights in the copyright to this Work to the University of Malaya ("UM"), who henceforth shall be owner of the copyright in this Work and that any reproduction or use in any form or by any means whatsoever is prohibited without the written consent of UM having been first had and obtained;
- (6) I am fully aware that if in the course of making this Work I have infringed any copyright whether intentionally or otherwise, I may be subject to legal action or any other action as may be determined by UM.

Candidate's Signature

Date:

Subscribed and solemnly declared before,

Witness's Signature

Date:

Name:

Designation:

#### ABSTRACT

Alzheimer's disease (AD) is a neuro-degenerative disorder affecting mainly the elderly people. Though more than 40 million people are suffering from AD complications up to present date, there is hardly any treatment effective in withstanding the progression of AD. Thus, demand for natural, safe and cost-effective AD therapeutics has got momentum. Oxidative stress (OS) and increased plasma cholesterol levels have been implicated among the numerous pathomechanistic factors of AD. Despite some discrepencies, strategies aimed at lowering OS and hypercholesterolemia seems promising as AD therapeutics. Mushrooms have been highly hailed for providing health benefits including anti-oxidative and hypocholesterolemic numerous potentialities. However, combined anti-oxidative and hypocholesterolemic effect of edible-medicinal mushrooms have not been reported though these two aspects have been studied separately. Also, as an AD therapeutic agent, mushroom has not been interpreted through their anti-oxidative and hypocholesterolemic properties. Thus, the present study has been designed to elucidate the anti-oxidative, hypocholesterolemic and AD ameliorating properties of the selected edible-medicinal mushrooms: H. erinaceus, L. edodes, F. velutipes and G. lucidum. Five solvent-solvent partitioned fractions (methanol:dichloromethane, dichloromethane, hexane, ethylacetate, aqueous and hot water extract (HWE) of each of the four mushroom species were screened through in vitro anti-oxidant tests including 2,2-diphenyl-1-picrylhydrazyl (DPPH) free radical scavenging, Folin-Ciocalteu assay, lipid peroxidation (LPO) and low density lipo-protein (LDL) oxidation inhibition tests. Among all, the HWE of G. lucidum had been found possessing the best in vitro anti-oxidative capacity and highest content of polyphenols, tri-terpenoids and sterols. Thus, this mushroom extract had been chosen for the rest of the studies. In vivo anti-oxidative and hypocholesterolemic studies upon rats showed increasing effect towards plasma and liver anti-oxidant enzymes (catalase,

glutathione peroxidase and superoxide dismutase) while total cholesterol (TC), triacylglycerol (TG), LDL-cholesterol were decreased and HDL-cholesterol was increasd. This mushroom extract demonstrated absence of adverse effect upon the organ function tests. As derangement of memory and learning abilities is the most notable complication associated with AD, the effect of the HWE of G. lucidum had been tested upon the AD model rats. Feeding of HWE of G. lucidum was found to improve memory and learning abilities of the AD rats. This cognitive improvement has been supported by elevated levels of memory related neurotransmitters in the respective rats. Transmission electron microscopic (TEM) studies demonstrated pronged neuronal dendrites in the G. lucidum HWE treated rats than those of the non-treated. Finally, brain comparative identified differentially proteomics have expressed proteins involved in neurotransmission, metabolism, cellular stress response and misfolding repairment. Improved functional network had been observed among the proteome of the G. lucidum HWE treated rats through STRING analysis. Ingenuity pathway analysis (IPA) identified nervous system development, cell-cell signaling and interaction, molecular transport and cell death and survival among the top-most functional networks among the experimental subjects. Thus, AD ameliorating effect of G. lucidum through antioxidative and hypocholesterolemic performances might be implicated in AD therapeutics.

#### ABSTRAK

Penyakit Alzheimer (AD) adalah gangguan degeneratif neuro yang mempunyai kesan ke atas orang tua lazimnya. Walaupun lebih daripada 40 juta orang yang menderita komplikasi AD sehingga kini, tiada rawatan yang berkesan yang wujud untuk menahan perkembangan AD. Oleh itu, keperluan untuk sebatian terapeutik bercirikan semulajadi, selamat dan kos efektif telah mendapat perhatian. Tekanan oksidatif (OS) dan peningkatan paras kolesterol plasma adalah faktor penting di antara pelbagai faktor-faktor yang menyebabkan AD. Walaupun terdapat beberapa perbezaan, strategi untuk menurunkan OS dan hiperkolesterolemia berpotensi sebagai terapeutik AD. Cendawan dikenali dengan pelbagai manfaat kesihatan termasuk anti-oksida dan potensi hipokolesterolemik. Walau bagaimanapun gabungan ciri anti-pengoksidaan dan kesan hipokolesterolemik cendawan yang boleh dimakan atau ubatan tidak dilaporkan sedangkan kedua-dua aspek telah dikaji secara berasingan. Juga, cendawan masih tidak dianggap sebagai agen terapeutik AD, berdasarkan ciri-ciri antioksida dan hipokolesterolemik. Oleh itu, kajian ini telah dilakukan untuk menjelaskan sifat-sifat anti-pengoksidaan dan hypocholesterolemia cendawan terpilih (H. erinaceus, L. edodes, F. velutipes dan G. lucidum) dalam mencegah AD. Lima pecahan pelarutpelarut (methanol : diklorometana, diklorometana, heksana, etil asetat, akueous dan ekstrak air panas (HWE) bagi setiap empat spesies cendawan telah disaring melalui ujian in vitro antioksida termasuk memerangkap radikal bebas 2,2-diphenyl-1picrylhydrazyl (DPPH), esei Folin-Ciocalteu, pengoksidaan lipid (LPO) dan ujian perencatan pengoksidaan lipoprotein ketumpatan rendah. Antara semua, HWE G. lucidum telah didapati memiliki potensi terbaik kapasiti anti-oksida in vitro dan kandungan tertinggi polifenol, tri-terpenoid dan sterol. Oleh itu, ekstrak cendawan ini telah dipilih untuk kajian selanjutnya. Kajian *in vivo* anti-oksida dan hipokolesterolemik ke atas tikus menunjukkan kesan peningkatan paras enzim anti-oksida plasma dan hati

(katalase, glutation peroksidase, superoxide dismutase) manakala jumlah kolesterol LDL-kolesterol berkurangan (TC), triasigliserol (TG), dan HDL-kolesterol meningkatkan kesan dari ekstrak cendawan ini tanpa memberi kesan buruk kepada ujian fungsi organ. Oleh kerana gangguan daya ingatan dan kebolehanan belajar adalah komplikasi yang paling ketara yang dikaitkan dengan AD, kesan HWE G. lucidum telah diuji ke atas tikus model AD. Pemakanan HWE G. lucidum telah didapati meningkatkan daya ingatan dan kebolehan pembelajaran tikus AD. Peningkatan kognitif telah disokong oleh peningkatan tahap neurotransmitter memori yang berkaitan dalam tikus masing-masing. Bukti lanjut dari kajian histopatologi (DAPI) yang menunjukkan peningkatan sel-sel neuron dan pemanjangan ketara dendrit dalam tikus yang dirawat dengan G. lucidum HWE (kajian TEM). Akhirnya, perbandingan proteomik otak telah mengenal pasti protein terzahir yang terlibat dalam penghantaran maklumat. metabolisme, tindak balas tekanan selular dan pembaikan pelipatan. Pemuliharaan rangkaian berfungsi telah meningkat di kalangan proteome tikus dirawat oleh G. lucidum HWE berdasarkan analisis STRING. Analisis laluan 'ingenuity' (IPA) mengenal pasti pembangunan sistem saraf, isyarat sel-sel dan interaksi, pengangkutan molekul dan sel mati dan hidup di kalangan rangkaian berfungsi paling atas di kalangan subjek eksperimen. Oleh itu, G. lucidum membendung kesan AD melalui ciri-ciri antioksida dan hipokolesterolemik mungkin berpotensi sebagai calon untuk terapeutik AD.

#### ACKNOWLEDGEMENTS

All praise to Almighty ALLAH Whose constant blessings have enabled me to be engaged in Alzheimer's disease (AD) research.

My sincere apathy to those AD patients whom I observed during my childhood but for the lack of appropriate measures I could not help them. But, they instilled the firm determination in my mind that in future, I would explore AD therapeutic strategy for the betterment of the mankind.

These are nobody other than Professor Dr. Noorlidah Abdullah and Dr. Norhaniza Aminudin, my revered supervisor and co-supervisor, respectively, who could realize my dormant dream and offered me with their every possible support, suggestions, constructive criticism and inspiration to prosecute the AD ameliorating effects of mushroom that has brought forth the present thesis. I am whole-heartedly grateful to them.

My sincere gratitude also gush forth to my fellow lab-members at the Mycology laboratory and Mushroom Research Centre, Institute of Biological Sciences, Faculty of Science, University of Malaya (UM), Malaysia. I am indebted to Dr. Ali Ashrafjadeh of the Proteomics branch of the Medical Biotechnology Laboratory, UM for helping me in proteomics and to Professor Dr. Shahdat Hossain of the Laboratory of Alternative Medicine and Behavioral Neuroscience, Jahangirnagar University, Bangladesh, for being an overseas expert of my studies.

I am thankful to the Bright Sparks unit, UM, for providing me with the financial supports and the Institute of Research Monitoring and Management, UM, for granting aid in conducting the current research.

My family members deserve special thanks for helping me at home and abroad.

Last but not the least, let we make a new world free of AD through mushroom-based therapeutic approaches.

## TABLE OF CONTENTS

Abstract	iii
Abstrak	v
Acknowledgements	vii
Table of Contents	viii
List of Figures	xix
List of Tables	xxii
List of Symbols and Abbreviations	xxiii
	3

CHA	APTER 1: GENERAL INTRODUCTION	1
1.1	Current trend and future challenges of Alzheimer's disease	1
1.2	Research gap and importance of the present work	2
1.3	Research questions	2
1.4	Research objectives	3
1.5	Outline of the thesis	3

CH	APTER 2: LITEI	RATURE REVIEW	5			
2.1	Oxidative stress	(OS) and its physiological complications	5			
2.2	Anti-oxidative defense and demand for natural anti-oxidant					
2.3	Unperturbed cel	lular signaling with anti-oxidant usage	7			
2.4	2.4 Hypercholesterolemia					
	2.4.1 Manage	ment of hypercholesterolemia	8			
	2.4.1.1	Statins	9			
	2.4.1.2	Fibrates	9			
	2.4.1.3	Bile acid sequestrants	10			
	2.4.1.4	Nicotinic acid derivatives	10			
	2.4.1.5	Estrogen replacement therapy (ERT)	10			

		2.4.1.6 Inhibitors of proprotein convertase subtilisin/ke	exin type 9
		(PCSK9)	11
2.5	Alzhein	er's disease (AD)	11
	2.5.1	General features	11
	2.5.2	Epidemiology and Prevalence of AD	
	2.5.3	Types of AD	
		2.5.3.1 Early onset and late onset	13
		2.5.3.2 Familial and sporadic	
	2.5.4	mpaired memory – the most prominent feature of AD	
		2.5.4.1 Memory and learning	
		2.5.4.2 Types of memory	
		2.5.4.3 Synapses at the ground of memory formation	
		2.5.4.4 Pre – and post – synaptic protein links in memory for	ormation 16
		2.5.4.5 Memory and learning ability disruption in AD	
	2.5.5	Causes of AD	
		2.5.5.1 Genetic predisposition	
		2.5.5.2 Amyloid plaque formation	
		2.5.5.3 Neurofibrillary tangle formation	
		2.5.5.4 Oxidative stress	
		2.5.5.5 Protein oxidation in AD	
		2.5.5.6 Mitochondrial cascade hypothesis	
		2.5.5.7 Inflammation	
		2.5.5.8 Synaptic dysfunction	
		2.5.5.9 Cholinergic hypothesis	
		2.5.5.10 Dopaminergic hypothesis	
		2.5.5.11 Excitotoxicity and Ca <sup>2+</sup> dyshomeostasis	

		2.5.5.12 Decreased level of neurotrophic factors	26
		2.5.5.13 Defective endocytosis	26
		2.5.5.14 Defected energy metabolism	26
		2.5.5.15 Vascular hypothesis	27
		2.5.5.16 ABSENT hypothesis	27
		2.5.5.17 Sink hypothesis	27
2.6	Influen	ce of cholesterol upon Aβ production	28
	2.6.1	Lipid raft based Aβ production	28
	2.6.2	Apo E4 driven Aβ production	30
2.7	Associa	ation among OS, hypercholesterolemia and AD	31
2.8	Mushro	oom – the functional food and traditional medicine	33
2.9	Curren	t trends in AD treatment	34
		2.9.1.1 Enhancement of neurotransmitter level and neurotransmis	sion 34
		2.9.1.2 Modulation of the secretases	35
		2.9.1.3 Immunotherapy	36
		2.9.1.4 Inhibition of Aβ fibrillation	36
		2.9.1.5 Inhibition of tau kinases	36
		2.9.1.6 Anti-inflammatory drugs	36
		2.9.1.7 Cholesterol lowering drugs	37
		2.9.1.8 Anti-oxidant supplementation	37
		2.9.1.9 Estrogen replacement therapy	
		2.9.1.10 Caregiving	
2.10	Therap	eutic suitability of edible – medicinal mushrooms against AD	
	2.10.1	Synaptic degeneration prevention	
	2.10.2	Reversion of neuronal apoptosis	39
	2.10.3	Aβ deposition lowering	40

	2.10.4 Neurite outgrowth and neurogeneration stimulation				
	2.10.5	Acetylcholine esterase inhibitory effects	42		
	2.10.6	Dopaminergic activity	42		
	2.10.7	BACE1 inhibitory effect	42		
2.11	Proteor	nics	43		
	2.11.1	Merits of proteomics studies	44		
	2.11.2	Importance of proteomics approach in AD therapeutics	44		

3.1	Introdu	iction	47
3.2	Materia	als and Methods	49
	3.2.1	Mushroom samples	49
	3.2.2	Preparation of Solvent-Solvent Partitioned Fractions of Mushrooms.	50
	3.2.3	Preparation of Hot Water Extracts	51
	3.2.4	DPPH Free Radical Scavenging Assay	51
	3.2.5	Folin-Ciocalteu Assay	52
	3.2.6	Lipid Peroxidaton Inhbition Test	53
	3.2.7	LDL Oxidation Inhibition Test	54
		3.2.7.1 Conjugated diene (CD) measurement	54
		3.2.7.2 Inhibitory effect upon the formation of Malondiald	ehyde
		(MDA)	55
3.3	Results 56		
	3.3.1	DPPH Free Radical Scavenging Performance	56
	3.3.2	Folin-Ciocalteu assay	57
	3.3.3	Inhibition of Lipid Peroxidation	58
	3.3.4	Inhibition of LDL Oxidation	59

		3.3.4.1 Lengthening of the Lag time of CD Formation	59
		3.3.4.2 Malondialdehyde (MDA) assay	60
3.4	Discus	ssion	61
	3.4.1	DPPH free radical scavenging effect	61
	3.4.2	Folin-Ciocalteu assay	62
	3.4.3	Inhibition of lipid peroxidation	62
	3.4.4	Inhibition of LDL oxidation	63
3.5	Conclu	usion	64
CH. SEI	APTER JECTEI	4: DETERMINATION OF BIO-ACTIVE COMPONENTS	IN THE
4.1	Introd	uction	66
4.2	Materi	als and methods	67
	4.2.1	GC-MS/MS Analysis	67
	4.2.2	LC-MS/MS Analysis	67
	4.2.3	Determination of β-D Glucan level	68
4.3	Result	s	68
	4.3.1	GC-MS/MS Analysis of the Hexane Fraction of G. lucidum	68
	4.3.2	GC-MS/MS Analysis of the Hexane Fraction of <i>H. erinaceus</i>	69
	4.3.3	GC-MS/MS Analysis of the Hexane Fraction of <i>L. edodes</i>	70
	4.3.4	LC-MS/MS Analysis of the MD Fraction of <i>F. velutipes</i>	70
	4.3.5	LC-MS/MS Analysis of the HWE of G. lucidum	71
	4.3.6	Determination of $\beta$ -D-Glucan Level in the HWE of <i>G. lucidum</i>	75
4.4	Discus	ssion	75
	4.4.1	GC-MS/MS analysis of the hexane fraction of <i>G. lucidum</i>	75
	4.4.2	GC-MS/MS analysis of the hexane fraction of <i>H. erinaceus</i>	78
	4.4.3	GC-MS/MS analysis of the hexane fraction of <i>L. edodes</i>	79

	4.4.4	LC-MS/MS analysis of the MD fraction of <i>F. velutipes</i>	83
	4.4.5	LC-MS/MS analysis of the HWE of <i>G. lucidum</i>	85
	4.4.6	Determination of $\beta$ -D glucan level in the HWE of <i>G. lucidum</i>	88
4.5	Conclu	ision	89

5.1	Introdu	uction		90
5.2	Materi	als and M	ethods	91
	5.2.1	Animals		91
	5.2.2	Measure	ement of <i>in vivo</i> anti-oxidative effect of <i>G. lucidum</i> HWE	92
		5.2.2.1	Catalase activity assay	93
		5.2.2.2	Glutathione peroxidase (GPx) activity assay	94
		5.2.2.3	Superoxide dismutase (SOD) activity assay	95
	5.2.3	Hypoch	olesterolemic study	97
		5.2.3.1	Measurement of body weight	97
		5.2.3.2	Measurement of triacylglycerol	97
		5.2.3.3	Measurement of total cholesterol	98
		5.2.3.4	Measurement of HDL-C	100
		5.2.3.5	Measurement of VLDL and LDL	101
		5.2.3.6	Measurement of atherogenic index (AI) and LDL/HDL ratio	o 101
	5.2.4	Organ fi	unction tests (Cardiac, Liver and Kidney Function Tests)	101
		5.2.4.1	Measurement of Aspartate aminotransferase/Glutamate ox	alate
			transaminase (AST/GOT) activity	101
		5.2.4.2	Measurement of Alanine aminotransferase/Glutamate pyre	uvate
			transaminase (ALT/GPT) activity	103
		5.2.4.3	Measurement of Alkaline phosphatase (ALP) activity	105

		5.2.4.4	Measurement of serum total bilirubin	
		5.2.4.5	Measurement of serum creatinine level	
		5.2.4.6	Measurement of blood urea level	
		5.2.4.7	Measurement of Protein concentration	
	5.2.5	Inhibitic	on of HMG Co-A reductase (HMGR) activity	109
5.3	Results	5		110
	5.3.1	Effect of	n enzymatic anti-oxidative defense	110
	5.3.2	Hypoch	olesterolemic effect of G. lucidum HWE	
		5.3.2.1	Effect of G. lucidum HWE on body weight change	111
		5.3.2.2	Effect on plasma TG level	
		5.3.2.3	Effect on plasma TC level	
		5.3.2.4	Effect on plasma HDL-C level	114
		5.3.2.5	Effect on plasma LDL-C and VLDL-C level	114
		5.3.2.6	Effect on atherogenic index (AI) and LDL/HDL ratio	114
		5.3.2.7	Inhibition of HMG Co-A reductase (HMGR) activity	114
	5.3.3	Organ fi	unction tests (Cardiac, liver and kidney function tests)	115
5.4	Discus	sion		116
	5.4.1	Effect of	n enzymatic anti-oxidative defense	116
	5.4.2	Hypoch	olesterolemic effect of G. lucidum HWE	117
		5.4.2.1	Effect on body weight change	117
		5.4.2.2	Effect on plasma TG level	
		5.4.2.3	Effect on plasma TC level	119
		5.4.2.4	Effect on plasma HDL-C level	
		5.4.2.5	Effect on plasma LDL-C and VLDL-C	
		5.4.2.6	Effect on AI and LDL/HDL ratio	121
		5.4.2.7	Inhibition of HMGR	

	5.4.3	Organ function tests	.123
5.5	Conclu	ision	.125

СНА <i>LUC</i> 6.1	APTER CIDUM Introdu	6: ALZHEIMER'S DISEASE AMELIORATING EFFECT OF 0 12 action	G. 26 26
6.2	Materia	als and methods	27
	6.2.1	Animals	27
	6.2.2	Preparation of AD model rats	27
	6.2.3	Post-operative care and recovery of the rats	29
	6.2.4	Memory and learning related behavioral tests	30
		6.2.4.1 Eight-armed radial maze test	31
		6.2.4.2 Novel object recognition (NOR) test	34
	6.2.5	Neuro - biochemical and neuro - immunological tests of memory an	nd
		learning related parameters	35
		6.2.5.1 Plasma and brain lysate preparation	35
		6.2.5.2 Antibodies and enzyme linked immunosorbent assays (ELISAs	s)
		13	5
		A. $A\beta$ (1-42) oligomers	35
		B. $A\beta$ (1-42) anti-fibrillating assay of <i>G. lucidum</i> HWE13	36
		6.2.5.3 Transmission electron microscopy (TEM) studies	37
	A.	Brain sample preparation for transmission electron microscope (TEM	Л)
		study	37
6.3	Results		)
	6.3.1	Effect of intra-cerebroventricular infusion of Aβ1-42	39
		6.3.1.1 Memory and learning related behavioral tests	40
		6.3.1.2 Eight-armed radial maze study14	40

		6.3.1.3 Novel object recognition (NOR) test
	6.3.2	Neuro - biochemical and neuro - immunological tests of memory and
		earning related parameters
	6.3.3	Effect of <i>G. lucidum</i> HWE on inhibition of A $\beta$ (1-42) fibril formation 144
		6.3.3.1 TEM studies of the hippocampus
6.4	Discus	on145
	6.4.1	Effect of intra-cerebroventricula $A\beta_{1-42}$ infusion
	6.4.2	Enhancement of memory and learning by <i>G. lucidum</i> HWE146
		6.4.2.1 Eight-armed radial maze study146
		6.4.2.2 Novel object recognition test
	6.4.3	Antibodies and enzyme linked immunosorbent assays (ELISAs)148
		6.4.3.1 A $\beta_{(1-42)}$ oligomer lowering effect of <i>G. lucidum</i> HWE
		6.4.3.2 Neurotransmission maintenance effect of <i>G. lucidum</i> HWE 149
		6.4.3.3 Maintenance of pre-synaptic membrane and long-term
		potentiation by G. lucidum HWE
		6.4.3.4 Maintenance of post – synaptic density by <i>G. lucidum</i> HWE.151
		6.4.3.5 Neuro-inflammation lowering effect of <i>G. lucidum</i> HWE152
		6.4.3.6 Enhancement of cholinergic neurotransmission by G. lucidum
		HWE153
	6.4.4	inhibition of A $\beta$ (1-42) oligomerization and fibrillation by <i>G. lucidum</i>
		HWE153
		6.4.4.1 TEM studies of the hippocampus154
6.5	Conclu	on

CHAPTER 7: PROTEOMICS-BASED ANALYSIS ON ALZHEIMER'S DISEASE AMELIORATING EFFECT OF <i>G. LUCIDUM</i>					
7.1	Introdu	action	.156		
7.2	Materia	als and methods	. 158		
	7.2.1	Brain sample preparation and protein quantification	. 158		
		7.2.1.1 Brain sample collection	. 158		
		7.2.1.2 Protein extraction	. 158		
		7.2.1.3 Delipidation procedure	. 158		
		7.2.1.4 Protein quantification	. 159		
	7.2.2	Protein separation through SDS-PAGE	. 159		
		7.2.2.1 SDS-PAGE assemblage	. 159		
		7.2.2.2 Loading sample preparation	.160		
		7.2.2.3 Running of SDS-PAGE	. 160		
		7.2.2.4 Staining of the gel	. 160		
		7.2.2.5 Destaining of the gel	. 161		
	7.2.3	Preparation of the gel for protein identification	. 161		
		7.2.3.1 In-gel tryptic digestion	. 161		
		7.2.3.2 Reduction and alkylation:	. 161		
		7.2.3.3 Digestion	. 162		
		7.2.3.4 Extraction	. 162		
		7.2.3.5 Desalting and Zip tipping procedure	. 163		
	7.2.4	LC-MS/MS Q-TOF quantification	. 163		
	7.2.5	Bioinformatics and analysis of protein-protein interaction (PPI)	.165		
7.3	Results	S	166		
7.4	Discus	sion	. 169		
		7.4.1.1 Functional classification of the significantly regulated prot	teins		

	7.4.1.2	Protein-protein interaction (PPI) findings	197				
	1.	PPI among the up-regulated proteins	197				
	A.	PPI among up-regulated proteins of AD vs control group	197				
	B.	PPI among upregulated proteins in AD vs AE group	200				
	2.	PPI among the downregulated proteins	204				
	A.	PPI among the down-regulated proteins of the AD vs of	ontrol				
		group	204				
	7.4.1.3	Identification of functional pathway interaction through IP	A 207				
7.5 Conclu	sion		223				
CHAPTER 8: GENERAL DISCUSSION							
CHAPTER 9	CHAPTER 9: CONCLUSION						
REFERENC	CES		233				
List of Public	ations an	d Papers Presented	288				
Appendix A -	– Anti-ox	iidative and hypocholesterolemic studies	292				
Appendix B – AD Tests							
Appendix C - Statistical analyses							
Appendix D – Preparation of solutions and gels							
Appendix E - Differentially regulated proteins							
Appendix F- ingenuity pathway analysis (ipa) summary							
Appendix G - Ethical permission letter							

## LIST OF FIGURES

Figure 2.1: Types of memory
Figure 2.2: Pre – and post – synaptic proteins involved in memory processing
Figure 2.3: Amyloid β production (Zhang, 2012)18
Figure 2.4: Photomicrograph of AD brain (Adapted from Robbins and Cotran Pathologic Basis of Disease, 7 <sup>th</sup> edition)
Figure 3.1: Selected edible-medicinal mushrooms of the present study
Figure 3.2: Preparation of the solvent-solvent partitioned fractions
Figure 3.3: DPPH free radical scavenging effect
Figure 3.4: Folin-Ciocalteu assay
Figure 3.5: Lipid peroxidation inhibitory effect
Figure 3.6: LDL oxidation inhibitory effect of the mushroom fractions and extracts through reduced production of MDA
Figure 4.1: Lipid peroxidation inhibitory effect of α-tocopherol80
Figure 4.2: Protocatechuic acid (PCA) – mediated scavenging mechanism of DPPH free radical
Figure 4.3: Protocatechuic acid – mediated metal ion chelating mechanism
Figure 5.1: Effect of <i>G. lucidum</i> HWE upon body weight change of the rats112
Figure 5.2 HMG Co-A reductase inhibitory effect of mushrooms
Figure 6.1: Preparation of AD model rats and collection of blood
Figure 6.2: Eight armed radial maze test
Figure 6.3: Novel object recognition test
Figure 6.4: Reference memory error of the rats140
Figure 6.5: Working memory error of the rats
Figure 6.6: Time spent in exploring the baited arms

Figure 6.7: Preference for novel object142
Figure 6.8: Fibrillation inhibitory effect of <i>G. lucidum</i> HWE144
Figure 6.9: TEM view of the CA1 region of the rat hippocampus145
Figure 7.1: A) SDS-PAGE image and B) Venn diagram of the identified proteins 166
Figure 7.2: Gene ontology (GO) analysis featuring biological processes of the identified hippocampal proteins
Figure 7.3: Gene ontology (GO) analysis featuring molecular functions of the identified hippocampal proteins
Figure 7.4: Gene ontology (GO) analysis featuring cellular localization of the identified hippocampal proteins
Figure 7.5: PPI network of the up-regulated proteins of the AD vs Control group 199
Figure 7.6: PPI network of the up-regulated proteins in the AD vs AE group201
Figure 7.7: PPI network of the up-regulated proteins in the C vs AE group
Figure 7.8: PPI network among the down-regulated proteins of the AD vs C group205
Figure 7.9: PPI among the down-regulated proteins of the AD vs AE group206
Figure 7.10: Top-scoring gene networks differentially expressed between AD and C group (top-most network 1)
Figure 7.11: Top-scoring gene networks differentially expressed between AD and C group (top-most network 2)
Figure 7.12: 14-3-3 protein-mediated signaling and associated proteins in the AD vs C group
Figure 7.13: Top-scoring gene networks differentially expressed between AD ane AE group (top-most network 1)
Figure 7.14: Top-scoring gene networks differentially expressed between AD and AE group (top-most network 2)
Figure 7.15: Top-scoring gene networks differentially expressed between AD and AE group (top-most network 3)
Figure 7.16: Top-scoring gene networks of the differentially expressed between C and AE group (top-most network 1)

Figure 7.17: Top-scoring gene networks	of the differentially expressed between C	VS
AE group (top-most network 2)	2	221

Figure	7.18:	Top-scoring	gene	networks	differentially	expressed	between	С	and	AE
group (	(top-m	ost network 3	)							222

### LIST OF TABLES

Table 3.1: LDL oxidation inhibitory effect of the mushroom fractions and e through lengthening the lag time of CD formation (minute)	extracts
Table 4.1: Identified bio-components in the hexane fraction of <i>G. lucidum</i>	68
Table 4.2: Identified bio-components in the hexane fraction of <i>H. erinaceus</i>	69
Table 4.3: Identified bio-components in the hexane fraction of <i>L. edodes</i>	70
Table 4.4: Polyphenolic compounds identified in the MD fraction of <i>F. velutipes</i> .	71
Table 4.5: Gano - components identified in the HWE of G. lucidum	72
Table 5.1: Plasma anti-oxidant enzyme levels	110
Table 5.2: Liver anti-oxidant enzyme levels	111
Table 5.3 Lipid profile, plasma atherogenic index and LDL/HDL ratio	113
Table 5.4 Ganoderma lucidum HWE fed rats' organ function tests	116
Table 6.1: Memory and learning related behavior tests performed	131
Table 6.2: Antibodies and enzyme linked immunosorbent assays (ELISAs)	143

## LIST OF SYMBOLS AND ABBREVIATIONS

Abbreviations	Definitions			
Αβ	Amyloid beta			
AD	Alzheimer's disease			
AI	Atherogenic index			
APP	Horseradish peroxidase			
BDNF	Brain derived neurotrophic factor			
BBB	Blood brain barrier			
BM	Barnes maze			
CD	Conjuated diene			
DPPH	2,2 –diphenyl 1 picrylhydrazil			
ELISA	Enzyme linked immunosorbent assay			
FC	Fold change			
GC-MS/MS	Gas chromatography tandem mass spectrometry			
HDL-C	High density lipo-protein cholesterol			
IC <sub>50</sub>	Concentration that produces 50% inhibition			
IPA	Ingenuity pathway analysis			
LC-MS/MS	Liquid-chromatographt tandem mass spectrometry			
LDL-C	Low-density lipo-protein cholesterol			
LPO	Lipid peroxidation			
LTP	Long-term potentiation			
MDA	Malondialdehyde			
NOR	Novel object recognition			
PSD 95	Post-synaptic density 95			
RME	Reference memory error			
Rt	Retention time			
SNAP 25	Synaptosomal associated protein 25			
TBARS	Thiobarbituric acid reactive substances			
TC	Total cholesterol			
TG	Triacylglycerol			
TNFα	Tumor necrosis factor alpha			
VAchT	Vesicular acetylcholine transporter			
WME	Working memory error			

#### **CHAPTER 1: GENERAL INTRODUCTION**

#### 1.1 Current trend and future challenges of Alzheimer's disease

Alzhemier's disease (AD), a neurodegenerative disorder and the most common form of dementia, poses grave threat towards the ever increasing lifespan of the humanity. By 2050, the global figure of the centenarians (people aged 100 years) is expected to reach up to 2 million (Alzheimer's Association, 2016). During their lifetime, one in every eight male and one in every four female has been predicted to develop AD (Alzheimer's Association, 2016). There are more than 40 million people worldwide suffering from AD (Selkoe & Hardy, 2016). Although AD is an age-onset physical complication, manifested usually after the age of 60, its initiation and progression occurs during early stages of life (Swerdlow, 2007).

According to the amyloid cascade hypothesis, AD is caused by the accumulation of amyloid beta peptide (A $\beta$ ), derived from the plasma membrane bound amyloid precursor protein (APP), inside the neurons called as neurofibrillary tangles or outside the neurons called as plaques (Hardy & Selkoe, 2002). A $\beta$  aggregates cause neuronal cell toxicity, cell function loss and/or cell death leading to AD complications (Hardy & Selkoe, 2002).

Oxidative stress (OS) and hypercholesterolemia (elevated level of blood cholesterol) are the most common etiological factors coinciding AD (Stampfer, 2006). Elevated level of plasma cholesterol, through the formation of the "lipid rafts" in the plasma membrane bilayer, facilitates the deposition of the neurotoxic A $\beta$  (Ehehalt *et al.*, 2003). Cholesterol lowering drugs, statins or 3-hydroxy-3-methylglutaryl coenzyme A (HMG-

CoA) reductase inhibitors have been reported to be protective against AD development (DeKosky, 2005; Sierra *et al.*, 2011).

#### 1.2 Research gap and importance of the present work

Although AD is going to plague the aged humanity worldwide, till today, there is no specific drug for the treatment of AD; the available medico-strategies just delay the worsening of the symptoms. Yet, the synthetic drugs are highly expensive and cause numerous side effects. Therefore, to aid the ever increasing global aged and AD prone populace, finding out of safe, less expensive and easy to achieve therapeutic agents has emerged as an urgent need.

As modern life-style and dietary pattern evoke OS and hypercholesterolemia highly, lifestyle modification involving functional food based therapeutic approaches seem imperative in withstanding the progression of AD. In this context, mushrooms spur high due to their anti-oxidative and hypocholesterolemic effects. In AD model animals, been found boosting neurogenesis mushrooms have up and improving pathophysiological complications (Mori et al., 2011; Zhou et al., 2010). Still today, there is scarcity of therapeutic approach against AD through combined management of OS and hypercholesterolemia. In this regard, evaluation of mushrooms as the AD ameliorating agent beacons promising and the present study is aimed at elucidating the anti-oxidative potentiality of edible (H. ericaneus, F. velutipes and L. edodes) and medicinal mushrooms (G. lucidum), followed by the cholesterol lowering and AD ameliorating effects of G. lucidum.

#### **1.3 Research questions**

Whether oxidative stress attenuating and cholesterol lowering capacities of selected mushroom species are effective in enhancing memory and learning abilities of the AD subjects?

#### **1.4 Research objectives**

Objectives of the present research are

1. to determine the *in vitro* anti-oxidative effect of the solvent – solvent partitioned fractions and of hot water extract (HWE) of *Ganoderma lucidum* (GL), *Hericium ericaneus* (Lion's mane, LM), *Lentinula edodes* (Shiitake, SK) and *Flammulina velutipes* (Enoki, EN) mushrooms.

2. to identify the bio-active component present in the selected mushroom fractions and extracts.

3. to evaluate the *in vivo* anti-oxidative and hypocholesterolemic effect of the HWE of *G. lucidum* on rat models.

4. to investigate memory and learning related behavioral improvements by *G*. *lucidum* HWE towards AD model rats.

5. to perform proteomics analysis and protein-protein interaction of the AD model rats' brain tissue.

## 1.5 Outline of the thesis

Comparing the *in vitro* anti-oxidative performance of different solvent – solvent partitioned fractions and HWE of *G. lucidum*, *H. erinaceus*, *L. edodes* and *F. velutipes*, this study selects HWE of *G. lucidum* as the potent extract (Chapter 3) to further with identification of bio-active components (Chapter 4) and *in vivo* studies. Chapter 5 describes the *in vivo* anti-oxidative and hypocholesterolemic performance of the HWE of *G. lucidum*. AD ameliorating behavioral, antibody and ELISA and transmission electron microscopic outcomes of the HWE of *G. lucidum* have been described in Chapter 6. Chapter 7 outlines the proteomics study of the selected animal models.

Chapter 8 deals with the general discussion and Chapter 9 contains the concluding remarks and future aspect of the present study.

#### **CHAPTER 2: LITERATURE REVIEW**

#### 2.1 Oxidative stress (OS) and its physiological complications

In normal physiological condition, reactive oxygen species (ROS) and reactive nitrogen species (RNS) are generated as a part of cellular metabolic, signaling and defensive actions (Finkel & Holbrook, 2000). As they are highly reactive, they undergo reactions with other reactive anions and/or cations and turn into neutral molecules or the bodily systems neutralize, removes or attenuates them through different mechanisms (Valko *et al.*, 2007). Thus, these reactive species have very short half-lives (Valko *et al.*, 2007). But, when the level of their generation outnumbers their dismissal through enzymatic or non-enzymatic processes, the state of oxidative stress (OS) emanates. Persistent oxidative stress (OS) mediates structural and functional alteration of biomolecules (carbohydrates, proteins, lipids, nucleic acids), poses threat towards normal functioning of the tissues and organs that result in numerous pathophysiological consequences such as cardiovascular diseases (CVD), neurodegenerative diseases including AD, diabetes and aging (Dalle-Donne *et al.*, 2006; Valko *et al.*, 2007).

#### 2.2 Anti-oxidative defense and demand for natural anti-oxidant

Traditionally an "anti-oxidant" is a substance that prevents oxidation. In biological sense, an "anti-oxidant" is a natural or synthetic substance capable of withstanding or prolonging the deteriorating effect of oxygen. Halliwell, (1990) defined biological antioxidant as a substance that at small concentration can oppose the oxidative modification of the biomolecules. Food science considers anti-oxidants as the substances that slow down or thwart the detrimental upshot of the reactive species upon foods (Huang *et al.*, 2005). From biochemical and medicinal viewpoint, antioxidants are

the enzymatic or non-enzymatic substances having the potential of reverting oxidation (Kedare & Singh, 2011). Most of the natural anti-oxidants possess polyphenolic structures.

Evolutionarily, cells have adapted themselves to cope with the constant exposure of ROS with multiple defense strategies (Kedare & Singh, 2011). Cells thwart OS through physical defense involving stabilization of plasma membrane and steric hindrance. Antioxidative enzymatic defense system includes superoxide dismutase (SOD), catalase (CAT) and glutathione peroxidase (GPx) counterbalance the production of ROS (Valko et al., 2007). SOD first converts superoxide into relatively less harmful hydrogen peroxide and later on, CAT and GPx turns hydrogen peroxide into water (Lobo et al., 2010; Valko et al., 2007). Also, non-enzymatic intracellular anti-oxidants act directly by scavenging free radicals or through breaking of the free radical generating chain reactions and this type of anti-oxidant include glutathione, flavonoids, pyruvate, vitamin C (ascorbic acid), vitamin E ( $\alpha$  tocopherol) and carotenoids (Lobo *et al.*, 2010). Cells synthesize a trivial amount of the direct-acting anti-oxidants and most of them must be supplied through diet for maintenance of the normal cellular functioning against OS (Lobo et al., 2010). Thus, the supply of the exogenous anti-oxidants (natural or synthetic) seemed promising as a preventive therapeutic stratagem against ROSmediated pathophysiology. Though, both natural and synthetic anti-oxidants fall under choice in this regard the second form of anti-oxidant suffers from numerous side effects. To cite some examples, synthetic anti-oxidant butylated hydroxyl toluene (BHT) has been associated with lung carcinogenesis, butylated hydroxyl anisole negatively regulate mitogen-activated protein kinase (MAPK) and cause stomach cancer (Botterweck et al., 2000; Kahl, 1984; Umemura et al., 2001). In short, adverse effects generated from the usage of the synthetic anti-oxidants outweigh their benefits and their usage has faced severe criticism both in terms of dosage and regulatory perspectives

(Shahidi & Zhong, 2005). Thus, demand for natural, dietary anti-oxidants has got momentum (Lobo *et al.*, 2010).

#### 2.3 Unperturbed cellular signaling with anti-oxidant usage

ROS and RNS participate as messengers in the physiological redox signaling processes (Brigelius-Flohé & Flohé, 2011; Forman *et al.*, 2014; Holmstrom & Finkel, 2014; Murphy *et al.*, 2011; Sawa *et al.*, 2013). ROS and RNS mediate post-translational modification to the thiol moieties of the proteins involved in cell signaling through formation of sulfoxides, disulphides, sulfenic acid and also by S-glutathionylation and S-nitrosylation (Naviaux, 2012; Sies, 2015). Thus, the redox signaling based concept of OS incorporates it as a signal that provokes oxidative type of reactions and/or impairs the redox harmony that results in two consequences: heightened cellular defense, which can be termed as eustress and/or cellular damage that can be termed as distress (Niki, 2009, 2016). At physiological level, low level of OS had been deemed as an adaptive strategy i.e. eustress that associates augmented accumulation of defensive components including antioxidants, proteins and enzymes (Niki, 2009; Noguchi, 2008; Surh *et al.*, 2009). Role of ROS as the metabolic and longevity signaling molecules had also been reckoned (Chen *et al.*, 2005; Ristow & Schmeisser, 2011).

As ROS and RNS include a diverse array of components differing in structure and function, their removal by anti-oxidative strategy do not directly affect the beneficial roles of ROS and RNS in signaling (Niki, 2016).

#### 2.4 Hypercholesterolemia

Cholesterol is an integral component of the animal cell membrane and a precursor of vitamin D3 and steroid hormones. Cholesterol and triacylglycerols are carried in the blood with the aid of the lipo-proteins. Based on the density, lipo-proteins are of five

types: very low density lipo-protein (VLDL), intermediate density lipo-protein (IDL), and lipo-protein low-density lipo-protein (LDL) high density (HDL). Hypercholesterolemia refers to the elevated level of cholesterol in the blood. It is a form of hyperlipidemia (elevated lipid level in blood) and also ascribed as dyslipidemia (Durrington, 2003). Cholesterol is generated both from dietary sources and endogenously. Cholesterol metabolism occurs in the liver. Under normal physiological system, there occurs a balance between its biosynthesis and removal in the form of bile acids and excretion through intestine. Hypercholesterolemia stems from the increased synthesis and/or decreased removal of endogenous cholesterol (due to defective clearance activity of VLDL as well as increased catabolism of VLDL into IDL and LDL) and also from the excessive supply of dietary cholesterol or cholesterol precursors. The rate-limiting step of cholesterol biosynthesis is mediated by the 3hydroxy 3-methyl glutaryl co-enzyme A (HMG Co-A) reductase (HMGR). Cellular level of cholesterol regulates its activity: higher cholesterol level reduces whereas lower cholesterol level stimulates its functioning.

#### 2.4.1 Management of hypercholesterolemia

The Fifth Joint Task Force of the European Society of Cardiology and Other Societies on Cardiovascular Disease Prevention in Clinical Practice had suggested lifestyle and dietary modification as the first line and pharmacological intervention through utilization of lipid lowering drugs as the second line of treatment strategy against hypercholesterolemia (Perk *et al.*, 2012). Life style modification involves loss of weight, consumption of diet containing less cholesterol and higher saturated fatty acid, giving up of cigarette smoking and alcohol drinking, intake of unsaturated fatty acids such as DHA and regular physical exercise (Brown *et al.*, 2009; Galani & Schneider, 2007; Klein *et al.*, 2004). However, combination of both life style modification and pharmacological treatment sound better in real life (Last *et al.*, 2011). Currently, seven types of drugs are available for treating hypercholesterolemia that include statins (inhibitors of 3-hydroxy-3-methyl glutaryl co-enzyme A reductase), fibrates, bile acid sequestrants, derivatives of nicotinic acid, estrogen replacement therapy and monoclonal antibodies to proprotein convertase subtilisin/ kexin type 9 inhibitor (Bergeron *et al.*, 2015). Following are the important features of these hypocholesterolemic agents

#### 2.4.1.1 Statins

Statins inhibit intracellular cholesterol biosynthesis by inhibiting the activity of the HMG Co-A reductase and also have HDL-C increasing, LDL receptor increasing and triacylglycerol lowering effects (Liao & Laufs, 2005). Its side effects include myalgia, release of muscle and liver enzymes in the blood, contraindication with other drugs, nausea, headache, dyspepsia, constipation, insomnia, diarrhea and abdominal pain (Sinzinger *et al.*, 2002).

#### 2.4.1.2 Fibrates

Fibrates stimulate the expression of the peroxisome proliferator-activated receptor (PPAR)- $\alpha$ , which later regulates the inhibition of hepatic synthesis and secretion of triglycerides. They also stimulate the degradation and clearance of triglyceride-rich lipoproteins through enhanced expression of lipoprotein lipase (Staels *et al.*, 1998). This increased clearance of triglycerides results from a stimulation of the expression of lipoprotein lipase (Zhang *et al.*, 2001).

Its side effects include increased level of liver enzymes in the blood, gastrointestinal disturbance, myalgia, myositis, dermatitis, insomnia and impotency (Elisaf, 2002).

#### 2.4.1.3 Bile acid sequestrants

Bile acid sequestrants are the anion-exchange resins. They bind the bile acids in the gut and inhibit their entero-hepatic recycling process. As a consequence, synthesis of hepatic LDL receptors are upregulated for obtaining cholesterol to compensate the production of bile acids that ultimately results in reduction of LDL-C level (Einarsson *et al.*, 1991).

They are very often unpalatable and are not absorbed. Their side effects include constipation, abdominal bloating, diarrhea, headache and dizziness (Insull, 2006).

#### 2.4.1.4 Nicotinic acid derivatives

Vitamin B<sub>3</sub> (nicotinic acid) derivatives had been suggested having suppressing effect up on the hormone sensitive lipase of adipocyte and thus causes reduced synthesis of non-esterified fatty acid, TG, VLDL and LDL as well as increased synthesis of HDL (Kamanna & Kashyap, 2000). Its inhibitory effect up on diacylglycerol acyltransferase-2, the key enzyme of hepatic TG synthesis, also aids in its anti-hypercholesterolemic effect (Kamanna & Kashyap, 2008).

Its side effects include hypotension, hyperuricemia, abdominal discomfort, flushing, rhinitis, nausea, rash, pruritus and headache (Gille *et al.*, 2008).

#### 2.4.1.5 Estrogen replacement therapy (ERT)

Decreased level of estrogen especially in the post-menopausal women had been associated with increased rate of atherosclerosis (Mikkola & Clarkson, 2002). ERT had been found beneficial in the younger post-menopausal women who are at the earlier stage of hyperlipidemia and atherosclerosis (Mikkola & Clarkson, 2002). But, older (more than 65 years old) post-menopausal women having pre-existing atherosclerosis seemed unresponsive towards ERT (Mikkola & Clarkson, 2002). ERT has contraindication with liver diseases, thromboembolic diseases and breast cancer. It also causes headache, bloating and breast tenderness (Ross *et al.*, 2000).

#### 2.4.1.6 Inhibitors of proprotein convertase subtilisin/kexin type 9 (PCSK9)

Binding with the heapatic LDL receptors, PCSK9 lowers LDL uptake and increases blood LDL levels. Recently developed inhibitors (monoclonal antibodies) to PCSK9, lowers the production of PCSK9 and reduces their binding with the LDL receptors and thus lowers production of LDL and CVD risk (Bergeron *et al.*, 2015).

However, due to the multiple adverse effects of the synthetic antihypercholesterolemic medications, food and dietary supplement based approaches have gained the alternative therapeutic attention. In this regard dietary fiber and complex polysaccharide rich foods such as mushrooms have highly been adored.

#### 2.5 Alzheimer's disease (AD)

#### 2.5.1 General features

Alzheimer's disease (AD) is the most common form of dementia, the deteriorated behavioral and mental state that disrupts the affected person's everyday life. AD had been named after German neuro-pathologist and psychiatrist Alois Alzheimer who first described this neurodegenerative disease in 1906 (Hippius & Neundorfer, 2003). AD symptoms start with gradual decrement of the capability learning and memory. This occurs due to the neuronal degeneration in the brain region associated with memory and learning (Binetti *et al.*, 1996). As a consequence the affected person becomes unable to perform daily normal activities, becomes confused about time and space, faces problem in planning and executing even errands (Waldemar *et al.*, 2007). At advanced stage, the AD patients suffer from difficulties in speaking and writing, sleeping and awakening and even cannot recall their own names (Waldemar *et al.*, 2007). They also face problem in remembering their personal history, recognizing relatives and family

members (Bäckman *et al.*, 2004). They need help in dressing and toileting properly and cannot control their bladder and bowel activities. They become irritated and suspicious even about their caregivers though they become dependent on their caregivers and family members. At the final stage, they become unresponsive towards conversation and their language becomes limited to little words and finally they fully lose tendency of speaking (Taler & Phillips, 2008). Their muscle becomes rigid and cannot swallow well or move by themselves. They also become vulnerable to infectious diseases (Förstl & Kurz, 1999).

#### 2.5.2 Epidemiology and Prevalence of AD

According to the Alzheimer's statistics 2016, more than 44 million people are suffering from AD worldwide. Prevalence of AD is highest in the Western Europe, followed very closely by the USA while lowest in the sub-Saharan Africa. AD is the prime cause of disability in later age of life. In 2016, global cost of AD is estimated to be 605 billion US dollar (<u>http://www.alzheimers.net/resources/alzheimers-statistics/</u>). Prevalence of AD increases up to 15 times during the age range of 65 – 80 years. AD prevalence is higher in the developed countries than those of the developing and least developed (Mayeux & Stern, 2012). This might be due to the increased life span of the people living in the developed countries. As the developing countries are also harboring increased number of aged people, AD prevalence trend is also upward in those countries (Prince *et al.*, 2015). AD is posing threat to the global economic policy as it impacts world economy negatively.

#### 2.5.3 Types of AD

Generally, AD occurs in the aged people. Sometimes, the younger and mid – aged people also suffer from it. Genetic and environmental factors also influence its causation and combined types of AD can be described based on the criteria
# 2.5.3.1 Early onset and late onset

Early onset AD (EOAD) develops before age 65 and is less common (about 5% of all the AD cases) while the late-onset AD (LOAD) is the most common form of AD and affects people over 65 years of age (Zetterberg & Mattsson, 2014).

Both the age-onset types can fall into either familial or sporadic forms.

#### 2.5.3.2 Familial and sporadic

Early onset familial form is most rare and found in about 1 percent of total AD cases (Rogaev *et al.*, 1995). These patients bear fault in either chromosome 1 (*PS2*gene) or in chromosome 14 (*PS1* gene) or in chromosome 21 (*APP* gene) (Rogaev *et al.*, 1995). Age range of these patients is 40-50 years and sometimes may arise even at 30 years of age. People with Down's syndrome (bearing extra copy of chromosome 21) are at increased risk of developing early onset familial AD (Munter *et al.*, 2010). The apolipoprotein E (APOE) gene, especially its APOE- $\varepsilon$  4 allelle is considered as the genetic risk factor of late onset familial AD (Sando *et al.*, 2008).

Sporadic form of EOAD is also uncommon (about 10% of total AD cases) and the people with Down's syndrome are at increased risk of it. Sporadic form of LOAD, dubbed as the sporadic AD, is the most common (about 90% of total AD cases) form that affects people over 65 years old (Zetterberg & Mattsson, 2014). In every five year, there increases two-fold risk of sporadic AD development after 65 years age (Bermejo-Pareja *et al.*, 2008).

#### 2.5.4 Impaired memory – the most prominent feature of AD

# 2.5.4.1 Memory and learning

Memory is the recalling process of any information learned previously (Squire & Kandel, 2000). Learning is related with the acquirement process of something while

memory denotes the acquired expression of something. Thus, learning is a process involved in memory. Memory process involves multiple phases: encoding (information acquisition), storage (recording of the encoded information into short - or long – term form) and retrieval (recalling of the stored memory) (Izquierdo & Medina, 1997). Hippocampus, cerebral cortex, basal ganglia, amygdala and cerebellum of the brain participate in memory and learning abilities.

From neurochemical point of view, memory and learning abilities are the outcomes of the synaptic plasticity, changing (either strengthened or weakened) ability of the synapses in response to the stimuli (Abbott & Nelson, 2000). It may be of short-term (milliseconds) or long-term (minutes to hours) duration. The short – term synaptic plasticity alters the strength of neurotransmission while the long – term one changes the number and structure of the synapses. If synaptic plasticity strengthens the synapses, the state is termed as synaptic potentiation and if weakens, synaptic depression (Teyler & DiScenna, 1987). Based on the duration, both potentiation and depression may of short – and long–term. The short short–term potentiation (STP) is reversible while the longterm potentiation (LTP) is irreversible. The former (STP) is the neuro-molecular analog of short-term memory (STM) and the latter of long–term memory (LTM).

# 2.5.4.2 Types of memory

Following the modified model of Atkinson and Shiffrin (1968), different types of memory has been illustrated in Figure 2.1.

A. Short – term memory (STM): It is the memory of the most recent events and lasts for only a few seconds to minutes. As a bio-electrical type of memory, it incorporates changes in synaptic neurotransmission, enzyme activation and protein phosphorylation but no *de novo* protein synthesis. It may be reclassified into sensory (ability to retain sensory information immediately after the original stimulus has elapsed) and working (organized or manipulated form of the STM) memory types. The executive form of working memory is an important aspect of behavior as it maintains guidance, decision-making and reasoning while the visual memory deals with the memories formed from visual experience.

B. Long – term Memory (LTM): Recalling of the permanent information ranging from days, months to years even throughout one's lifetime are of long – term memory (LTM) type. It is a biochemical type of memory that can be sub-divided into procedural and declarative forms. Procedural memory sub-type of LTM is the memory of motor skills such as learning of procedures or skills and the declarative sub-type refers to the memory of facts and figures. Declarative memory is also known as the explicit memory which may be either semantic (related to general knowledge such as word meaning, concept, ideas) or of spatial (related to spatial orientation and environment) and/or episodic (related with specific experience).



Figure 2.1: Types of memory

## 2.5.4.3 Synapses at the ground of memory formation

Synapses, the interconnections of two nerve endings, are the ground of neurochemical and neuro-electrical signal transmission (Kleim *et al.*, 2002). Neurotransmitters released from the pre-synaptic membranes bind with the post-synaptic membrane receptor and causes signal transduction mediated polarization and/or depolarization (Ferreira & Klein, 2011; Yan *et al.*, 2016). Heightened synaptic connection and transmission induce enhanced LTP and improved memory. On the other hand, weakened synaptic connections and lowered transmission cause LTD and poor memory (Ferreira & Klein, 2011; Yan *et al.*, 2016).

# 2.5.4.4 Pre – and post – synaptic protein links in memory formation

A number of pre – and post – synaptic proteins participate in memory consolidation (Figure 2.2). Without the functioning of these proteins, the proper consolidation and recall of memory processes become impaired. Pre – synaptic proteins involved in memory consolidation include synaptosomal – associated protein 25 KD (SNAP 25), syntaxin 1B, synaptogamin 1, synaptojanin I, synaptophysin, synapsin I and II, amphiphysin, Munc 13-1, Munc 18, dynamin 1, Rab 3A, SV 2A, CASK, VDCC  $\alpha 2\delta 1$ , VDCC (N type) $\alpha$ 1B, ELKS, Piccolo, Bassoon, neurexin – 1 and – 3 and Tomosyn b (Rosenberg *et al.*, 2014). The post – synaptic memory related proteins are post – synaptic density protein 95 KD (PSD 95), PSD 93, TrkB, Glu A2 and A3, Glu N1 and N2B, GABA<sub>B</sub>R – 1, ProSAP1/Shank2, SAP 97, SAP 102, Homer 1, CaMKII $\alpha$ , CaMKII $\beta$ , Neuroligin 3 and 3, Cortactin B,  $\alpha$  Actinin 1, Drebrin, N-CAM and Agrin (Rosenberg *et al.*, 2014).



Figure 2.2: Pre – and post – synaptic proteins involved in memory processing (Adapted from https://resources.rndsystems.com)

## 2.5.4.5 Memory and learning ability disruption in AD

Disruption of memory is the first among the six cognitive domains most commonly altered in AD. Other five domains perturbed in AD are associated with learning abilities that include executive functioning, language, visuo-spatial functioning, attention and upset (Ross, 2009). At the onset, memory disruption might be mistakenly treated as the forgetfulness associated with the normal aging processes. As the diseases progresses, worsen memory and personality change make the everyday life challenging to them. As memory loss is the most expressive hallmark of AD, improvement of disrupted memory has been the focus of the AD treatment strategies.

#### 2.5.5 Causes of AD

Like other chronic diseases, AD had been believed to be the result of multiple causative factors. Various hypotheses had been put forward to explain the cause and mechanism of AD. The most common ones are described below.

#### 2.5.5.1 Genetic predisposition

About 0.1% of the early-onset familial AD occurs due to autosomal dominant inheritance. Mutation in any of the gens *APP*, *PS1 and PS2* or their combined effect might be the cause of AD.

## 2.5.5.2 Amyloid plaque formation

Glenner and Wong (1988) first proposed that amyloidogenesis of a particular protein might be a causative agent of AD . According to the amyloid cascade hypothesis of Aβ formation, Aβ is formed from APP by the sequential cleavage of  $\beta$ - and  $\gamma$ -secretase (Hardy & Higgins, 1992) (Figure 2.3). Normally, APP is cleaved by the  $\alpha$ -secretase and does not produce the harmful Aβ; but in AD, cleavage by  $\beta$ -secretase is favored (Hardy & Higgins, 1992). Increased production and decreased clearance of Aβ leads to the deposition of Aβ plaques that cause neurodegneration and AD consequences (Figure 2.4) (Selkoe & Hardy, 2016).



Figure 2.3: Amyloid β production (Zhang, 2012).



Figure 2.4: Photomicrograph of AD brain (Adapted from Robbins and Cotran Pathologic Basis of Disease, 7<sup>th</sup> edition).

The last twenty-five years have observed the dominating prevalence of the amyloid cascade hypothesis amongst all the others (Hardy & Selkoe, 2002; Selkoe & Hardy, 2016b). Some discrepancy of this hypothesis is that amyloid plaques are not always linked with cognitive impairments and cognitive-normal subjects with increased amyloid deposition in the brain had been reported (Benilova *et al.*, 2012; Giannakopoulos *et al.*, 2003; Ingelsson *et al.*, 2004; Perez-Nievas *et al.*, 2013). Thus, a reappraisal of the amyloid cascade hypothesis had emerged stating that oligomeric A $\beta$  are neurotoxic and sets the ground for synaptic dysfunction involving tau hyperphosphorylation and neurofibrillary tangles (NFT) formation ultimately resulting in neuronal loss (Hardy, 2006; Selkoe & Hardy, 2016b; Spires-Jones & Hyman, 2014).

Another approach towards interpreting the role of  $A\beta$  in AD pathogenesis is the "modern amyloid cascade hypothesis" which states that at low (picomolar) level,  $A\beta$  is not detrimental, rather memory enhancer and only at higher level, it impairs cognition; its soluble forms rather than the plaques are neurotoxic (Morley & Farr, 2014).

However, studies relating mutations in the genes *APP*, *PS-1*, *PS-2* and *Apo E4* support the amyloid cascade hypothesis of AD (Tanzi, 2012). Thus, dyshomeostasis of Aβ remains as the prime reason behind AD pathogenesis (Selkoe and Hardy 2016b).

Although, some mid - and late – stage clinical trials have failed to prove the amyloid cascade hypothesis fully, it would not be legitimate to abandon this hypothesis as the discrepancies of the clinical trials might occur due to flawed experimental design, choice of the experimental subjects, toxic biochemical reactions such as liver toxicity evoked by BACE1 inhibitors and non-selective inhibition of the  $\gamma$  – secretase and immunospecificity (De Strooper, 2014; Saxena, 2010; Toyn & Ahlijanian, 2014; Yan *et al.,* 2016). Considering the ever-increasing number of AD subjects in the next two decades, much research emphasizing the amyloid cascade hypothesis in parallel with those of others seem promising.

# 2.5.5.3 Neurofibrillary tangle formation

Neurofibrillary tangles (NFT) are formed mainly of tau, the microtubule associated protein. Tau regulates neuronal polarity, connects axonal microtubule with neuromembrane and provides stability. It has about thirty sites potential for phosphorylation and in AD, tau is highly phosphorylated that disrupts its connectivity and stability providing function (Geschwind, 2003). Hyperphosphorylated tau becomes aggregated and tangled (Figure 2.4). It loses association with microtubules and cannot process axonal transportation, causes neurotoxicity and impairs mitochondrial respiratory chain that leads towards neuronal death (Figure 2.4) (Götz *et al.*, 2006; Santacruz *et al.*, 2005). Tau also aids in Aβ-mediated neurotoxicity and neuronal death (Rapoport *et al.*, 2002). Also, kinases involved in tau phosphorylation such as cyclin dependent kinase 5 (CDK 5), glycogen synthase kinase 3 beta (GSK 3β) and microtubule affinity regulatory kinase (MARK) link AD pathogenesis with OS, inflammation, depleted energy generation and disrupted cellular communication (Ballatore *et al.*, 2007).

A drawback of this hypothesis is that though NFT impairs cognitive functioning, mutation in tau showed only fronto - parietal dementia (FTD) and not AD (Giannakopoulos *et al.*, 2003; Goedert & Jakes, 2005; Ingelsson *et al.*, 2004).

## 2.5.5.4 Oxidative stress

Oxidative stress hypothesis underscores "oxidative stress" as the upstream of AD pathogenesis (Nunomura et al., 2006; Nunomura et al., 2001). As ageing continues, the brain encounters oxidative and metabolic stress of multiple sorts. Besides, the brain itself consumes the highest oxygen, contains the highest amount of oxidation-prone unsaturated fatty acids as well as relatively low amount of anti-oxidative enzyme especially catalase. Early-onset, familial AD pathogenesis has been exacerbated through oxidative stress as revealed by the germline mutation studies. In addition, late-onset, familial and sporadic AD has also been implicated to be associated with oxidative stress through apolipoprotein E (Apo E) E4 allele(s) (Tamaoka et al., 2000). AD risk factors involving hypertension, diabetes mellitus, stroke, smoking and intoxication as well as life-style-oriented mediators have been directly or indirectly linked with increased oxidative stress and decreased anti-oxidative defenses (Tamaoka et al., 2000). Oxidation of lipid and nucleic acids in biological fluids along with decreased plasma level of antioxidants and total anti-oxidative capacity have been found reported in AD patients (Nunomura et al., 2006). Studies involving cell lines also tally this (Misonou et al., 2000; Paola et al., 2000). This result was further established by the finding that antioxidant vitamin E ( $\alpha$  - tocopherol) decreased both A $\beta$  and Tau lesions in animal models (Nakashima et al., 2004). Compared to the normal subjects, Aß plaques in AD brains contain higher amount of redox active metal ions ( $Cu^{2+}$ ,  $Zn^{2+}$ ,  $Fe^{2+}$ ) that can induce localized OS (Lovell et al., 1998; Roberts et al., 2012).

In addition, OS stimulates several kinases such as glycogen synthase kinase -3 and mitogen activated protein kinases (MAPK) that are active participants of AD pathogenesis. They phosphorylate Tau protein, and once phosphorylated, Tau is more vulnerable for oxidation that leads to neurofibrillary tangle (NFT) formation (Liu *et al.*, 2005). Thus, NFTs are also resultant products of neuronal oxidative stress.

Increased biomolecular oxidation is an important hallmark of AD. Increased level of carbonyls (protein oxidation product); MDA, TBARS, 4-hydroxy-2-trans nonenal (HNE) and isoprostanes (lipid peroxidation products); 8-hydroxy guanosine (8 OHG) and 8-hydroxy-2 deoxy-guanosine (nucleic acid oxidation products); glycation and glycoxidation (sugar oxidation products) in AD brains bear testimony towards OS induced AD pathogenesis (Butterfield *et al.*, 2001). OS driven mitochondrial dysfunction is another aspect of AD pathogenesis. Mitochondrial DNA contribute to the production of 13 proteins needed for respiratory chain and ROS-mediated mutation in mitochondrial DNA leads towards stable neuronal dysfunction as the neurons do not divide (Santos *et al.*, 2013). More importantly, AD neuronal mitochondria had been detected as the potential site of A $\beta$  accumulation, where it induces OS and disrupts the redox activities of some enzymes (Caspersen *et al.*, 2005; Lustbader *et al.*, 2004; Manczak *et al.*, 2006; Petersen *et al.*, 2008). Thus, OS induced mitochondrial dysfunction linked with impaired metabolic activity affect neuronal function and memory.

## 2.5.5.5 Protein oxidation in AD

Proteins are highly vulnerable to oxidative modification due to presence of functional groups in their constituent amino acids. Oxidative modification of proteins often leads towards their functional impairment. Normally, proteasome degrades and removes the oxidized proteins from the cell. Some enzymes such as glutathione reductase, thioredoxin and methionine sulfoxide reductase also participate in the repairing and

removing activity towards the mildly oxidized proteins (Friguet, 2006). Free radicals either oxidatively modify the amino acid side chains of the proteins and hence alter the protein backbone or cause protein fragmentation. Mechanistically, carbonyl groups (C=O) are either introduced into the protein's structure or are produced as in case of the reaction between proteins with those of the lipid peroxidation products such as MDA (Berlett & Stadtman, 1997). Peroxynitrite (ONOO<sup>-</sup>) mediated nitration of the protein tyrosine side chain is another aspect of post - translational protein modification (PTM). The resultant 3-nitrotyrosine hampers phosphorylation of the tyrosine residue by tyrosine kinase (TK) that affects TK mediated signal transduction process including those of nerve growth factor (NGF) and brain derived neurotrophic factor (BDNF) resulting in apoptosis (Mangialasche *et al.*, 2009). Peroxynitrite can also oxidize proteins.

## 2.5.5.6 Mitochondrial cascade hypothesis

According to this hypothesis, aging causes accumulation of toxic components through oxidative damage of the mitochondria and these toxic accumulants lead to "the 3-R response": reset (production of beta amyloids as an adaptive response towards the oxidative damage of the neurons), remove (reduced synaptic communication and apoptotic removal of the neurons) and replace (production of neurofibrillary tangles) (Swerdlow & Khan, 2004).

#### 2.5.5.7 Inflammation

Inflammatory hypothesis suggests that AD pathologies emanate from the inflammatory response towards extracellular amyloid- $\beta$  deposits. At the outset, inflammatory response is neuroprotective as it is aimed at clearing A $\beta$  (Rogers *et al.*, 2002; Wyss-Coray & Mucke, 2002). However, continuous overload of A $\beta$  surpasses the beneficial effects and turns into deleterious (Wyss-Coray & Mucke, 2002). Inflammatory response is governed by the microglia and increases over time through

Tau. The inflammatory markers (TNF  $\alpha$ , IL-2, IL-6, CRP) have been implicated to be directly or indirectly linked with cognitive decline and AD (Bettcher & Kramer, 2014). A $\beta$  initiates inflammation by binding with complement C1, microglial toll-line receptor (TLR) and receptor for advance glycation end products (RAGE) (Tacnet-Delorme *et al.*, 2001; Tahara *et al.*, 2006). Inflammatory and transcription factors (AP<sub>-1</sub>, NF- $\kappa$ B) become activated that generate ROS and activate the cholinergic neurons and involve astrocytes in the inflammatory cascade (Parvathenani *et al.*, 2003; Suzuki *et al.*, 2004). Repeated induction by A $\beta$  and supply of ATP to the microglia add fuel to this vicious cycle.

## 2.5.5.8 Synaptic dysfunction

Aβ released from the presynaptic membrane can bind to the post-synaptic membrane and modulate post-synaptic functions (Kamenetz *et al.*, 2003). Low level of Aβ can induce synaptic plasticity and maintain LTP (Puzzo *et al.*, 2011). On the contrary, increased Aβ level; dimeric, trimeric and oligomeric toxic forms of Aβ facilitates towards impaired LTP and evokes LTD, neuronal and synaptic loss of activity (Palop & Mucke, 2010; Sheng *et al.*, 2012; Spires-Jones & Hyman, 2014; Tu *et al.*, 2014; Yan *et al.*, 2016). Also, abnormality in the expression and function of the pre – and post – synaptic proteins also facilitate impaired memory and AD pathogenesis (Ardiles *et al.*, 2012; Ferreira & Klein, 2011; Hernandez & Abel, 2008; Sutton & Schuman, 2006) . Thus, Aβ mediated synaptic dysfunction disrupts the memory forming events and results in AD consequences.

# 2.5.5.9 Cholinergic hypothesis

Acetylcholine is an excitatory neurotransmitter of the CNS, neuromuscular junction and is essential for neuronal functioning and for memory and learning abilities. According to the cholinergic hypothesis of AD, decreased availability of the cholinergic neurotransmitter acetylcholine (Ach) leads towards AD consequences (Terry & Buccafusco, 2003). Produced by acetylcholine transferase from acetyl-CoA and choline, it is released into the synaptic cleft and upon binding to the post – synaptic neuron, exerts signal transduction. Acetyl choline esterase (AchE) breaks down Ach and does not allow prolonged action of Ach into the post–synaptic neuron that affects memory and learning abilities. Thus, treatment strategies have been developed based on this that agents having anti-acetyl choline esterase (AchEI) activity would have ameliorating effects on AD. Since 1970s, different AchEIs have been developed namely donepezil, galantamine, rivastigmine and tacrine (Craig *et al.*, 2011). AchEIs have been reported to improve cognitive and behavioral performance of the AD subjects.

### 2.5.5.10 Dopaminergic hypothesis

Dopamine can act both as an excitatory and inhibitory neurotransmitter as well as the precursor of other neuro-transmitters, anti-oxidant and anti-amyloidogenic substance (Himeno *et al.*, 2011). Dopaminergic hypothesis of AD describes  $A\beta$  – induced impaired level of dopamine in the hippocampus leads towards disrupted cognitive functions (Guzmán-Ramos *et al.*, 2012; Martorana & Koch, 2013). However, detailed clinical studies are called for establishing this hypothesis.

# 2.5.5.11 Excitotoxicity and Ca<sup>2+</sup>dyshomeostasis

Hyperactivation of the NMDA (N-methyl D-aspartate) receptor by  $A\beta$  or by Lglutamate destroys the post-synaptic Ca<sup>2+</sup> homeostasis, increases intra-neuronal Ca<sup>2+</sup> concentration, leading towards neuronal death (Hynd *et al.*, 2004; Koh *et al.*, 1990). NMDA receptor excitotoxicity also stimulates tau expression and hyper phosphorylation (Amadoro *et al.*, 2006). Increased level of intracellular Ca<sup>2+</sup> and Ca<sup>2+</sup>– dependent protease calpin associated with tau and decreased level of Ca<sup>2+</sup> buffering protein calbindin correspond towards Ca<sup>2+</sup> dyshomeostasis in AD brains (LaFerla, 2002). Endoplasmic reticulum stress mediated  $Ca^{2+}$  dyshomeostasis may also cause neuronal death in AD (Katayama *et al.*, 2004).

# 2.5.5.12 Decreased level of neurotrophic factors

Human brain harbor mainly four types of neurotrophic factors such as brain derived neurotrophic factor (BDNF), nerve growth factor (NGF), neurotrophin-3 (NT-3) and neurotrophin-4 (NT-4). Among them, the role of BDNF had been vastly studied. In AD subjects, decreased level of BDNF had been observed in the hippocampus and cortex implying its role in memory and learning and in neuro-protection (Lee *et al.*, 2005).

# 2.5.5.13 Defective endocytosis

Neurotransmitters left at the synaptic cleft are endocytosed and re-used for further synaptic transmission. Endocytosis of the neurotransmitters is essential for maintenance of proper signal transmission. Defective endocytosis impairs constant signaling between the neurons that affects memory formation and recalling (Nixon, 2007). Defective endocytosis also affects neuronal survival as they fail to receive extracellular regulatory signals (Nixon, 2005, 2007). Endosomal alteration is an earlier event in AD pathogenesis (Cataldo *et al.*, 2004; Nixon, 2005, 2007).

## 2.5.5.14 Defected energy metabolism

Altered enzymatic activity of glucose, glycogen metabolism and impaired function of the mitochondrial respiratory chain result in reduced energy (ATP) generation that deplete the neurons of vital support. Insulin insufficiency and resistance in diabetes also pose threat to the neuronal survivality and parallels AD progression (Suzanne & Wands, 2008).

#### 2.5.5.15 Vascular hypothesis

Vascular hypothesis has been described as a two-hit process: vascular risk factors (such as injury, APOE4, hypercholesterolemia, hypertension, hypoglycemia) disrupts the blood brain barrier, decreases blood flow causing reduced amyloid- $\beta$  clearance and increased accumulation of amyloid- $\beta$ . Though highly activated, the microglia fail to remove excessive amyloid- $\beta$  burden that results in inflammation and oxidative stress that intensifies synaptic loss, neuronal dysfunction, neurodegeneration and accelerated dementia (Stanimirovic & Friedman, 2012). Thus, amyloid- $\beta$  leads towards self-propagation of AD.

#### 2.5.5.16 ABSENT hypothesis

The ABSENT (Amyloid Beta Synergistic Endothelial and Neuronal Toxicity) is hypothesis is based on the research findings that not any one hypothesis by itself can fully explain the mechanism underpinning AD and the consequent histopathological alteration. It explains the toxicity of A $\beta$  to both vascular and neuronal tissues along with their synergistic effect on the onset and progression of AD (Roy & Rauk, 2005). ABSENT hypothesis indicates A $\beta$  as the sole generator of the free radicals that later leads to AD pathologies (Roy & Rauk, 2005).

# 2.5.5.17 Sink hypothesis

According to the sink hypothesis of AD, there occurs a concentration dependent shift of A $\beta$  from the brain into the blood. Decreased level of blood A $\beta$  level is an indicator of its lowered clearance and increased deposition and increased risk of AD (Yamamoto *et al.*, 2014; Zhang, Yan & Lee, 2011). Capacity of the blood albumin for binding of A $\beta$ determines their level in the blood and increased level of blood albumin-bound A $\beta$  is an indicator of increased clearance of A $\beta$  from the brain as well as decreased brain A $\beta$ deposition (DeMattos *et al.*, 2002; Yamamoto *et al.*, 2014).

#### 2.6 Influence of cholesterol upon Aβ production

#### **2.6.1 Lipid raft based Aβ production**

In normal physiological system,  $A\beta$  is produced in neuronal and non-neuronal cells (Busciglio et al., 1993). It is a 4 KDa peptide, hydrophobic in nature, generated from the amyloidogenic processing of amyloid precursor protein (APP). APP is a type I transmembrane glycoprotein expressed in every cell and its gene is in chromosome 21. APP undergoes two sequential enzymatic cleavages: first with the membrane bound aspartyl protease,  $\beta$  site APP cleaving enzyme (BACE,  $\beta$  secretase) that generates Nterminal extracellular soluble  $\beta$ -secretase-cleaved APP (sAPP or sAPP $\beta$ ) and transmembraner carboxyl terminal fragment (CTFB or C99 fragment) (Figure 2.3) (Golde et al., 2000). In the second step, CTF $\beta$  is cleaved by  $\gamma$ -secretase that produces Aß peptides of whose two most common are 40 amino acid long, Aβ40 and 42 amino acid long, AB42 (Figure 2.3) (Golde et al., 2000). Between AB1-40 and AB1-42, the second one along with its insoluble oligomers are much neurotoxic and potent in forming senile plaques in AD pathogenesis (Walsh & Selkoe, 2007). Their neurodegenerative mode of action involves inter alia oxidative stress, inflammatory and synaptic dysregulation (Mattson, 2004; Oddo et al., 2003; Reddy & Beal, 2008). Also, a cognate CTFy is produced (Figure 2.3). In the amyloidogenic pathway, about 90% of the produced AB are of AB40 type. In the non-amyloidogenic (secretory) pathway,  $\alpha$ secretase cleaves APP to generate the secreted derivative, sAPPa and CTFa (Golde et al., 2000). Later,  $\gamma$ -secretase cleaves CTF $\alpha$  and produces CTF $\gamma$  (Figure 2.3) (Golde et al., 2000). Indeed,  $\gamma$ -secretase can act only after the activities of  $\alpha$  and  $\beta$  secretases and  $\alpha$  and  $\beta$  secretases very often compete for APP (Skovronsky *et al.*, 2000). Thus, stimulation of one pathway coincides with inhibition of the other. However, in normocellular condition, the non-amyloidogenic (secretory) pathway prevails.

Experimental evidence suggests that the amyloidogenic processes occur in the lipid rafts or in the membrane of trans Golgi network and endosomes while the non-amyloidogenics occur beyond the lipid rafts of the plasma membrane (Ehehalt *et al.*, 2003; Kojro *et al.*, 2001; Lee *et al.*, 1998; Vetrivel *et al.*, 2004). Lipid rafts are membrane microdomains that are cholesterol rich, gangliosides and glycosyl phosphatidyl inositol anchored proteins supposed to be participants of cell signaling (Simons & Ikonen, 1997).

Normally, membranes of endoplasmic reticulum and Golgi apparatus contain low level of cholesterol. Increased cholesterol level or altered cholesterol distribution in these membranes mediate increased lipid raft production which subsequently lead towards increased AB generation (Ehehalt et al., 2003; Schroeder et al., 2001). Increased intracellular cholesterol alter membrane lipid composition, augment  $\beta$ - and  $\gamma$ secretase activities and stimulate APP cleavage resulting in increased AB production (Burns & Duff, 2003; Burns et al., 2003; Puglielli et al., 2003; Shobab et al., 2005; Wahrle et al., 2002). Intracellular cholesterol, at elevated level, attaches with the transmembrane domain of APP that leads to APP's insertion into the lipid raft and AB generation (Beel et al., 2010). Not only production but also aggregation and fibrillogenesis of  $A\beta$  in AD brain is facilitated by the membrane cholesterol level through pooling of Aβ with lipid raft component GM1 ganglioside (Kakio *et al.*, 2002). On the contrary, lowered level of cholesterol decreases AB fibrillogenesis through reduced binding of AB with lipid raft component GM1 (Mizuno et al., 1999). Even, altered sub-cellular distribution of cholesterol affects Aß production (Jin et al., 2004). However, the amyloidogenic pathway is highly favored with the esterified cholesterol rather than the non-esterified one (Bhattacharyya & Kovacs, 2010; Puglielli et al., 2001).

#### 2.6.2 Apo E4 driven Aβ production

Apolipoprotein E is the main carrier of cholesterol in the central nervous system (CNS) and also an important constituent of very low density lipoproteins (VLDL). Among its three allelles ( $\epsilon_2$ ,  $\epsilon_3$  and  $\epsilon_4$ ), individuals carrying the  $\epsilon_4$  allele are at heightened risk of developing AD as the ApoE  $\epsilon_4/\epsilon_4$  genotype increases fibrinogenesis in the brains of Alzheimer's disease patients (Hultman *et al.*, 2013). However,  $\epsilon_3$  is less potent in causing AD while  $\epsilon_2$  allele has been associated with decreasing AD risk (Hooijmans & Kiliaan, 2008).

ApoE  $\epsilon$ 4 shows gender specific effect (women more vulnerable than men) and also gene dosage effect (homozygous carriers are more at risk than their heterozygous counterparts) (Hooijmans & Kiliaan, 2008). People carrying APOE  $\epsilon$ 4 usually experience elevated blood cholesterol levels that later on aids in A $\beta$  accumulation and AD pathology (Eichner *et al.*, 2002; Hooijmans & Kiliaan, 2008; Kivipelto *et al.*, 2002). Elevated cholesterol in the brain also triggers A $\beta$  deposition.

However, some schools of thought have put forward contradiction to this mechanism (Dietschy & Turley, 2001; Ledesma & Dotti, 2006). Thus, more explanatory mechanism might be that the APOE  $\varepsilon 4$  allele upsurges serum cholesterol levels through atherosclerosis followed by hypoperfusion of specific brain areas *en route* to A $\beta$  deposition and neuronal degeneration, the hallmarks of AD. Besides, ApoE4 has been associated with vascular diseases including coronary artery disease, atherosclerosis, diabetes, hypertension and obesity (Davignon *et al.*, 1988; Wellington, 2004). The underlying mechanism might be its diminished clearance role towards LDL and chylomicrons. ApoE4 has also been reported influencing cerebral hemodynamics such as leakage of the blood-brain barrier (Zipser *et al.*, 2007) and cerebral amyloid angiopathy (Kinnecom *et al.*, 2007).

### 2.7 Association among OS, hypercholesterolemia and AD

The association between hypercholesterolemia and increased risk of AD pathogenesis had been emanated from multiple studies. Hypercholesterolemia linked increased deposition of  $A\beta$  in the rabbit hippocampus was first demonstrated by Sparks et al. (1994). Later, Refolo et al. (2000) showed that transgenic mice fed hypercholesterolemic diet become much prone to development of AD (Refolo et al., 2000). In their following studies, Refolo et al. (2001) observed that feeding of cholesterol lowering drugs to the transgenic AD mice reduces the risk of AD development by about 50% (Refolo et al., 2001). Increased brain level of cholesterol has been found during AD progression (Wood et al., 2002). The role of cholesterol in AD pathogenesis becomes clearer from the opposite relationship between plasma HDL level and risk of developing AD (Wolf et al., 2004). HDL can slow down the aggregation of A $\beta$  and thus prevents A $\beta$  toxicity (Olesen & Dagø, 2000). Epidemiological studies suggest 2-3 times greater risk of late-age dementia and AD in people having mid-life hypercholesterolemia than the normo-cholesterolemics (Anstey et al., 2008). Heightened AD neuropathologies have been observed in experimentallyinduced hypercholesterolemic animals (Shie et al., 2002). Alike to these findings, decreased neuronal cholesterol levels has been found to be inhibitory towards AB formation (Buxbaum et al., 2001; Wolozin, 2001). Correlation between neuritic plaques and hypercholesterolemia had further been evidenced from autopsy studies of the young patients (Pappolla et al., 2003). Major cholesterol metabolite 24S-hydroxycholesterol, had been found in higher concentrations in the plasma and cerebrospinal fluid of the AD patients (Schönknecht et al., 2002). Thus, hypercholesterolemia plays pivotal role in the progression of AD (Mathew et al., 2011).

Brain is much prone to OS due especially to its high lipid and low anti-oxidative defense arsenal content and its enormous (one fourth of total respired) oxygen

utilization (Ansari & Scheff, 2010; Mazzetti *et al.*, 2015; Reed, 2011; Sottero *et al.*, 2009). Inside brain, neurons are more vulnerable to OS and direct association between OS and A $\beta$  generation had been observed in both animal models in human subjects (McLellan *et al.*, 2003). Transgenic AD model mice (Tg19959) having partially defective anti-oxidant enzyme MnSOD, demonstrated accelerated OS as well as A $\beta$  level in brain. Two to three-fold overexpression of the same enzyme in the same model animals showed lowered OS (representing 50% increased level of catalase along with 50% reduced level of protein oxidation), about 33% decreased level of A $\beta$  deposition and restoration of memory deficit (Dumont *et al.*, 2009). In line with these, deficiency of Cu/ZnSOD1 in Tg2576 AD model mice enhanced OS-driven A $\beta$  oligomerization and memory loss (Murakami *et al.*, 2011).

Increased lipid peroxidation and reduced anti-oxidant status had been linked *apriori* to the onset of AD symptoms (Chen *et al.*, 2008; Resende *et al.*, 2008; Su *et al.*, 2010). Indeed, Aβ itself acts as a pro-oxidant (Combs *et al.*, 2001; Ding *et al.*, 2007). Methionine35 within the Aβ sequence make it much vulnerable to OS and its replacement with cysteine lowers its tendency towards oxidation (Butterfield & Boyd-Kimball, 2005; Butterfield *et al.*, 2013). Thus, it is plausible that the Aβ oligomers interpolate in the lipid bilayers and generate lipid peroxidation whose sequel continues to other biomolecules (Butterfield *et al.*, 2001; Naylor *et al.*, 2008; Tamagno *et al.*, 2006). Mitochondrial dysfunction often generates OS that is detrimental towards cellular survival and aids in AD pathogenesis (Castellani *et al.*, 2002; Melov *et al.*, 2007; Reddy, 2011; Yao, Jia *et al.*, 2009). Amyloid precursor protein (APP) and Aβ act as the mitochondrial deregulators that further the state of OS (Anandatheerthavarada *et al.*, 2003; Caspersen *et al.*, 2005; Manczak *et al.*, 2006).

Cholesterol cannot cross the blood brain barrier (BBB) *per se* and to get rid of excessive cholesterol, the brain converts it into relatively polar oxysterol 24 - OH and oxysterol 27 - OH, aided by the 24-hydroxylase and 27-hydroxylase, respectively. Oxysterols are capable of crossing the BBB and is exposed to the systemic circulation and then eliminated (Björkhem *et al.*, 2009; Björkhem & Meaney, 2004; Meaney *et al.*, 2007). Oxysterol 27 - OH can re-enter into the brain, exert much hypercholesterolemic condition there and propel A $\beta$  production (Björkhem, 2006; Heverin *et al.*, 2005; Sharma *et al.*, 2008). Thus, OS along with hypercholesterolemia might be implicated in the production, deposition and oligomerization of A $\beta$  and thus an important antecedent in AD pathogenesis (Oda *et al.*, 2010).

Thus, strategies effective in lowering plasma cholesterol or inhibiting the biosynthesis of cholesterol sound imperative for withstanding OS, hypercholesterolemia and AD pathogenesis.

#### 2.8 Mushroom – the functional food and traditional medicine

A functional food possesses bio-active components capable of providing nutritional as well as disease preventive and therapeutic support to its consumer (Augustyniak *et al.*, 2010). Mushrooms have long been associated with multiple aspects of human civilization such as their utilization in human diet and medicine due to their delicacy, nutritional and therapeutic values (Valverde *et al.*, 2015; Wasser 2014). Usage of mushrooms for medicinal purposes dates back to the Paleolithic age (more than 2000 years ago) and their psychedelic functionality had placed them among the divine foods and in mythology (Sanodiya *et al.*, 2009). Though, traditionally being used for a long time for various culinary and ailing purposes, mushrooms seem befitting as the functional foods against OS, hypercholesterolemia and AD (Guillamón *et al.*, 2010;

Sirtori *et al.*, 2009; Wasser, 2014). From the nutritional point of view, mushrooms possess considerably higher amount of carbohydrate, protein, vitamins and minerals and lower amount of fats (Barros *et al.*, 2007; Cheung, 2010). Their content of unsaturated fatty acids predominate over the saturated ones (Barros *et al.*, 2007; Cheung, 2010). Their carbohydrate content varies from 35% to 70% of their total dry weight and they are rich sources of vitamins and minerals (Cheung, 2010; Furlani & Godoy, 2008; Mattila *et al.*, 2001). They are also rich sources of dietary fibers that aid in their CVD preventive actions (Cheung, 2013; Nile & Park, 2014). Traditional Chinese medicine had used mushrooms in various ailments such as *Cordyceps* against hepatic diseases, *Pleurotus* for gastro-intestinal disorders and *Ganoderma* for neuro-psychological problems and as an energizing agent (Aida *et al.*, 2009; Paterson, 2006). Their therapeutic usage still remains in practice and some mushrooms had been included in the Chinese Pharmacopoeia (National Pharmacopoeia Committee, 2010).

#### 2.9 Current trends in AD treatment

#### 2.9.1.1 Enhancement of neurotransmitter level and neurotransmission

As different neurotransmitters and their signaling processes are affected in AD, treatment strategies have focused on improving both the levels of neurotransmitters and neurotransmission. Food and drug administration (FDA, USA) had approved four drugs (galantamine, donepezil, rivastigmine and memantine) as neurotransmitter modulating in AD. Galantamine, donepezil and rivastigmine inhibit AchE activity and modulate cholinergic neurotransmitter signaling processes (Davis & Couch, 2014). Meta analysis of the epidemiological studies indicates protection of the drugs against cognitive, behavioral and personality aberration associated with AD (Lanctôt *et al.*, 2009). Unfortunately, these drugs can not alter AD pathogenesis rather they improve only the AD symptoms temporarily, their side effects are ample and can affect cholinergic system beyond the brain and they are suitable for the preliminary AD patients only

(Hansen *et al.*, 2008; Winblad & Jelic, 2004). AD pathogenesis involves glutamate – mediated enhanced excitotoxicity of the NMDA receptor and memantine blocks this excitotoxicity through non-competitive antagonism (Danysz & Parsons, 2012). Neuro-protective role of memantine from glutamate mediated excitotoxicity and behavioral improvement of the mild to moderate AD patients had been reported (Areosa *et al.*, 2005). It also has cholinergic system improving effect and the combined donepezil and memantine therapy showed donepezil like cognitive and behavioral improvement in the mild to moderate AD subjects (Tariot *et al.*, 2004). Lecozotan, the antagonist of 5-hydroxytryptamin (5-HT) or serotonin receptor, aimed at improving the level of 5 - HT in AD subjects, also suffers from excessive side effects (Booth *et al.*, 2012).

Thus, the current treatment strategies aimed at improving neurotransmitter signaling could improve AD pathogenesis very little and the improved behavioral attainments were of very short duration. Thus, treatment strategy capable of sustained AD modulating effects involving much protection against neurodegeneration and cognitive improvement is quintessential.

#### 2.9.1.2 Modulation of the secretases

As the  $\beta$ - and  $\gamma$  – secretases are involved in generation of A $\beta$ , their controlled activity would aid in maintaining AD pathogenesis. Transgenic mice deficient of the  $\beta$  – secretase (BACE1) had shown decreased accumulation of brain A $\beta$  level (Ghosh *et al.*, 2012). Inhibition of the  $\gamma$  – secretase also showed promising outcomes (Siemers *et al.*, 2005). Also, reduced A $\beta$  deposition has been observed in case of the  $\alpha$  – secretase stimulator bryostatin that induced the non-amyloidogenic pathway of APP processing (Etcheberrigaray *et al.*, 2004).

#### 2.9.1.3 Immunotherapy

Both active and passive immunization of the transgenic animals with the anti -  $A\beta$  antibodies showed microglia – mediated clearance of  $A\beta$  plaques from the AD model animals (Bard *et al.*, 2000; Schenk *et al.*, 2004). However, adverse side effects including T – cell mediated encephalopathy led towards  $A\beta$  truncation at the C-terminal end and development of an immune-conjugate with the N – terminal of  $A\beta$  peptide attached to a carrier protein and having reduced risk (Schenk *et al.*, 2004).

## **2.9.1.4 Inhibition of Aβ fibrillation**

Inhibitors designed towards the promoters of A $\beta$ -Apo E and A $\beta$  – A $\beta$  interaction halt the conversion of A $\beta$  in the  $\beta$  sheets and thus hinder the fibrillation process. Alzhemed (NC - 531) mimics glycosaminoglycan, the promoter of A $\beta$  fibrillation and thus under trial as an anti – AD drug (Geerts, 2004). Similarly, the metal chelators (such as clioquinol, PBT-1 against Cu<sup>2+</sup> and Zn<sup>2+</sup>) can lower A $\beta$  level and fibrillation by chelating metal ions (Cherny *et al.*, 2001). Longer A $\beta$  peptides (such as A $\beta$  42, 43) are more self-aggregating and much AD-causing than their shorter counterparts such as A $\beta$ 40. Even, A $\beta$  40 may be amyloidogenesis preventing (Dickson *et al.*, 2007)

# 2.9.1.5 Inhibition of tau kinases

Phosphorylation of tau by kinases like CDK5, GSK -  $3\beta$  leads towards NFT formation and inhibition of the kinases are target of AD therapeutics. However, tau phosphorylation is interplay of multiple kinases and phosphatases and thus development of anti – AD drug using this approach seems cumbersome.

# 2.9.1.6 Anti-inflammatory drugs

As AD is associated of neuro-inflammation, usage of anti-inflammatory drugs seemed pertinent in AD therapeutics. Epidemiologically, non-steroidal anti-inflammatory drugs (NSAIDs) could reduce the risk of AD and lower A $\beta$  deposition in

transgenic mice brain (Aisen, 2002; Aisen *et al.*, 2003). NSAIDs have been supposed to be effective if used prior to neuro-degeneration occurs and have been suggested to be use in the mid-age, much earlier than the AD symptoms appear (Aisen, 2002; Aisen *et al.*, 2003).

# 2.9.1.7 Cholesterol lowering drugs

Retrospective case control analyses showed that the cholesterol lowering drugs, statins, lead towards reduced production and deposition of A $\beta$  (Wolozin *et al.*, 2000). About 74% reduced risk of AD had been implicated with the usage of the statins (Wolozin *et al.*, 2000). Statins can act as the facilitator of A $\beta$  clearance from the brains (Fassbender *et al.*, 2001). They have also been praised to be neuro-protective (Sierra *et al.*, 2010). However, reports relating very little cognitive improvement and no amelioration of the plasma and CSF A $\beta$  levels of the statin – utilized AD subjects have also been produced (Höglund *et al.*, 2005; Simons *et al.*, 2002; Sparks *et al.*, 2005) . Thus, large scale clinical trials are imperative for elucidating the effect of statin therapy on AD (DeKosky, 2005; Refolo *et al.*, 2001). AD ameliorating effect of the statins is also subject to their types, dosages, time of treatment regimen and also the stage of AD (Sierra *et al.*, 2010).

# 2.9.1.8 Anti-oxidant supplementation

Usage of the anti-oxidants seemed pertinent in withstanding OS mediated AD pathogenesis (Engelhart *et al.*, 2002). Large observational studies have implicated the AD lowering effect of vitamin E as its anti-oxidative performance (Sano *et al.*, 1997). Also, mild to no-effect of vitamin E on AD amelioration have also been found in randomized controlled trials (Petersen *et al.*, 2005; Sano *et al.*, 2005).

### **2.9.1.9** Estrogen replacement therapy

Estrogen has been reported to be neuro-protective and has been added in AD therapeutics (Simpkins *et al.*, 1997). Although epidemiological studies had linked reduced risk of AD in the post-menopausal estrogen supplemented women, large scale randomized controlled studies are yet to confirm its AD ameliorating effect (Mulnard *et al.*, 2000; Tang *et al.*, 1996).

# 2.9.1.10 Caregiving

Caregiving to the AD patients is highly recommended even at the early stage of the disease as they become fully dependent on the family members and relatives at the severe stage. Supporting the AD patients with their daily activities such as aiding in feeding, brushing, toilet using, road crossing add to the safety of the patients and save family (https://www.caregiver.org/alzheimers-diseasefinancial burden of the caregiving). Lifestyle modification of the family members including maintenance of blood pressure, sugar and lipid levels in the normal range, intake of balanced diet, avoidance of stress, regular physical exercise and free discussion among the family members are among the mostly suggested caregiving techniques (https://www.caregiver.org/alzheimers-disease-caregiving).

# 2.10 Therapeutic suitability of edible – medicinal mushrooms against AD

## 2.10.1 Synaptic degeneration prevention

Synaptic degeneration (loss of synaptic connection)-driven neurodegeneration is an important step in AD pathogenesis. Loss of synaptic density proteins such as synaptophysin, synaptotagmin and PSD - 95 parallel A $\beta$ -induced synaptotoxicity during AD progression (Reddy, 2011). Thus, neurotransmission becomes severely weakened. Aqueous extract of *G. lucidum* at 500 µg/ml concentrations *in vitro*, contributed significantly to the restoration of synaptic density protein, synaptophysin and thus

attenuated Aβ-induced synaptotoxicity in rat cortical neurons (Lai *et al.*, 2008). Mechanisms involved attenuation of phosphorylation of c-Jun N-terminal kinase (JNK), c-Jun, and p38 MAP kinase (Lai *et al.*, 2008).

Aβ causes decreased synaptophysin immunoreactivity along with its increased accumulation and aggregation in AD neuritis (Stagi *et al.*, 2005). Thus, axonal transport of synaptic vesicles becomes blocked. Stress signaling pathway JNK participates in this mechanism where NO exacerbates the situation. Inhibition of the phosphorylation of the JNK can overcome the blockage of synaptophysin transport (Stagi *et al.*, 2005). Aqueous extract of *G. lucidum* (500 µg/ml, 14 days treatment *in vitro*) significantly inhibited the phosphorylation of JNK in Aβ-stressed rat cortical neurons that potentiated it to overcome the Aβ-induced blockage of synaptophysin transport (Lai *et al.*, 2008).

## 2.10.2 Reversion of neuronal apoptosis

Neuronal apoptosis is another phenomenon in AD pathogenesis where A $\beta$  stimulates the activity of caspase-3 (Harada & Sugimoto, 1999). Apoptotic signaling pathway involves the protein kinase pathways including the JNK c-Jun and p38 MAP kinase (Harada *et al.*, 1999). Aqueous extract of *G. lucidum* reverted back A $\beta$ -induced neuronal apoptosis by inhibiting these pathways (Lai *et al.*, 2008). Specifically, *G. lucidum* inhibited phosphorylation at serine 67 and serine 73 of c-Jun, the substrate for JNK (Lai *et al.*, 2008). Orally administered *G. lucidum* polysaccharide (GLPS) at 100, 200 and 400mg/kg body weight significantly lowered neuronal apoptosis in rats (Zhou *et al.*, 2010b). In cultured rat cortical neurons, (GLPS) at 0.1, 1.0 and 10 µg/ml, reduced neuronal apoptosis in a dose dependent manner. GLPS induced neuro-protective mechanism involves downregulation of the expression of caspases-3, -8 and -9 and Bax, and inhibition of the reduction of Bcl-2 expression resulting in altered Bcl-2/Bax ratio (Zhou *et al.*, 2010b). Lysophosphatidylethanolamine (LPE), derived from *G. frondosa*  stimulated the activation of MAP kinase of rat pheochromocytoma PC12 cells and showed anti-apoptotic effects such as suppressed cell condensation and DNA ladder generation (Nishina *et al.*, 2006). LPE also upregulated the expression of neurofilament M and thus promoted neuronal differentiation of PC 12 cells (Nishina *et al.*, 2006).

### 2.10.3 Aβ deposition lowering

Mouse diet supplemented with 0.3%, 0.6% and 1.8% of *G. lucidum* powder, had been found to significantly lower A $\beta$  deposition in their brain along with increased antioxidative enzymatic levels and improved memory-related learning abilities (Wang *et al.*, 2004). Studies involving SH-SY5Y neuroblastoma cell lines identified enhanced non-amyloidogenic protein secretion (sAPP $\alpha$ ) activity of *G. lucidum*. In this case, *G. lucidum* mycelial extract mimicked the nerve growth factor (NGF) activity and it stimulated the phosphorylation of ERK 1/2 and PKC and involved the signaling cascades of PI3K and ERK (Pinweha *et al.*, 2008). *Hericium erinaceus* has also been reported to lower A $\beta$  plaque burden in the APPswe/PS1dE9 transgenic mouse model of AD. Also, the number of plaque-activated microglia and astrocytes in cerebral cortex and hippocampus were diminished (Tsai-Teng *et al.*, 2016).

#### 2.10.4 Neurite outgrowth and neurogeneration stimulation

Normal maintenance and differentiation of the neurons require supportive assistance from the neurotrophic factors (neurotrophins) such as brain derived neurotrophic factor (BDNF), glia derived neurotrophic factor (GDNF), nerve growth factor (NGF) and neurotrophin 3 (NT-3). The NGFs are polypeptides and too large to cross the BBB. Positive role of *H. erinaceus* upon neurite outgrowth and NGF generation have been reported by several studies (Lai *et al.*, 2013; Mori *et al.*, 2008, 2011 and Phan *et al.*, 2014). Mori *et al.* (2008) found that among four mushroom species, *Hericium erinaceus*, *Pleurotus eryngii*, *Grifola frondosa* and *Agaricus blazei*, only *H. erinaceus* 

extract promoted NGF mRNA expression in a concentration-dependent manner in 1321N1 human astrocytoma cells. However, hericenones C, D and E, constituents of H. erinaceus, failed to promote NGF gene expression in 1321N1 cells. Also, ddY mice fed 5% H. erinaceus dry powder for 7 days showed an increase in the level of NGF mRNA expression in the hippocampus. Another study by Mori et al. (2011) showed that H. erinaceus prevented intracerebroventricularly injected Aß (25-35) peptide-induced learning and memory deficit in mice. Short-term memory relating to space and visual recognition was ameliorated (Mori et al. 2011). The proposed mechanism was through stimulating NGF activity (Mori et al. 2011). The bio-active component involved was identified to be hericinones C, D and E (Mori et al., 2008 and 2011). Increased ratio of nerve growth factor (NGF) to NGF precursor (proNGF) and hippocampal neurogenesis have also been reported in case of the H. erinaceus ethanolic extract fed APPswe/PS1dE9 AD model mice (Tsai-teng et al., 2016). In an attempt to identify the neuro-supportive component of *H. erinaceus*, Park et al. (2002) demonstrated that an exopolysaccharide from its culture broth had neurite outgrowth promotive effect in PC12 cells. Lipophilic fraction of G. lucidum at 125 and 500 mg/L was found to promote neurtie outgrowth, stimulate NGF activity and potentiate PC 12 neuronal differentiation (Zhang et al., 2005). Seow et al. (2013) reported the comparatively enhanced neuritogenic effect of the aqueous extract of G. neo-japonicum than those of the G. lucidum and G. frondosa in PC 12 cells. At 50 µg/ml concentration, the G. neojaponicum stimulated 14.22% neuritogenesis, whereas at 75 µg/ml concentration, the effect was 12.61% for the G. lucidum and 12.07% for the G. frondosa extract (Seow et al., 2013). Bio-components present in these mushroom extracts activated the MEK/ERK1/2 and PI3K/Akt signaling pathways to promote neuritogenesis (Seow et al., 2013). Similar pattern of neurite-outgrowth stimulatory effect along with restoration of impaired memory in rats have been observed for Tremella fuciformis (Kim et al.,

2007; Park *et al.*, 2012). At 20  $\mu$ g/ml concentration, the aqueous sclerotium of *Lignosus rhinocerotis* induced PC 12 cells' neurite outgrowth of 24.4% (Eik *et al.*, 2011). In scopolamine- induced learning and memory deficient rats, *Cordyceps militaris* reverted back memory loss of the rats and promoted dose dependent (5-20  $\mu$ g/ml) neuritogenesis in Neuro2A mouse neuroblastoma cells (Lee *et al.*, 2011).

# 2.10.5 Acetylcholine esterase inhibitory effects

Decreased availability of the cholinergic neurotransmitter acetylcholine leads towards AD consequences (Inestrosa *et al.*, 1996; Recanatini *et al.*, 2004). Acetylcholine esterase inhibitors (AChEIs) inhibit the enzyme acetylcholinesterase, which in turn inhibits the breakdown of acetylcholine into acetate and choline and prolongs its duration of action (Inestrosa *et al.*, 1996; Recanatini *et al.*, 2004). *Ganoderma lucidum* at 2 mg/ml concentration showed 57% inhibition of the acetylcholine esterase activity *in vitro* (Hasnat *et al.*, 2013). Lanostenoid tri-terpens of *G. lucidum* have been supposed to act as the AchEIs and thus protect the cholinergic neurons (Lee *et al.*, 2011; Zhang *et al.*, 2011). Restoration of impaired memory in rats by *T. fuciformis* had been reported to be through activation of the cholinergic system (Park *et al.*, 2012).

# 2.10.6 Dopaminergic activity

Shielding effect against dopaminergic neuronal loss through reduced production of microglial proinflammatory factors have been associated with *G. lucidum* (Zhang *et al.,* 2011). Methanolic extract of G. *lucidum* at 400  $\mu$ g/ml, significantly inhibited the dopaminergic neurodegeneration of MES 23.5 cell membranes (Zhang *et al.,* 2011).

#### 2.10.7 BACE1 inhibitory effect

Beta-site APP-cleaving enzyme 1 (BACE1) catalyzes the regulatory step for the production of toxic A $\beta$  whose inhibition seems promising in controlling AD

pathogenesis (Yan *et al.*, 2016). Methanolic extract of *L. edodes* was found to inhibit by 40.1% of the BACE1 activity followed by *P. eryngii* (33.7%), *F. velutipes* (22.3%) and *A. bisporus* (18.3%) (Seo *et al.*, 2008). Through characterization of the cell-free extracts of different bacteria, fungi and yeast, Lee *et al.*, (2007) identified the BACE1 inhibitory effects (percentage scale) of mushrooms second only to *Saccharomyces cerevisiae*. Mushroom species having anti-BACE1 effects were *F. velutipes*, *P. ostreatus*, *G. frondosa*, *Dictyophora echinovolvata*, *Pholiota adipose*, *Fomitella fraxinea* and *Inonotus obliquus* (Lee *et al.*, 2007). BACE1 inhibitory components identified in *A. polytricha* has been detected as heat stable (Bennett *et al.*, 2013). Polyphenolic component hispidin, isolated from *Phellinus linteus*, had been found to non-competitively inhibit the BACE1 (Park *et al.*, 2004). BACE1 inhibitory effect of *A. polytricha* has also been indicated to be hispidine mediated (Bennett *et al.*, 2013).

AD hallmarks of neurodegenerative and memory related behavioral impairment is intricately linked with different types of proteins – their expression, suppression, deactivation, hyperactivation – it seems plausible to explore the proteomics of the experimental subjects. Thus, the following section has been designed to review the literature featured with AD proteomics.

#### **1.1 Proteomics**

"Proteomics", as a term had first been used in 1997 and this approach has facilitated highly in prognosis, diagnosis and in validating therapeutic approaches towards multiple diseases (James, 1997). "Proteome" stands for the entire set of proteins of an organism and "proteomics" denotes towards the study of the proteome (Anderson & Anderson, 1998). Proteomics approaches seem promising both for deciphering the pathomechanism and for identifying the bio-markers associated with AD. Biomarkers detection beacons towards the presence or absence, severity, trend and predictive information of a pathological condition. Thus, biomarkers predicting the risk and providing support towards amelioration of AD pathogenesis, is of utmost importance for mitigating the ever increasing onslaught of AD.

### 2.10.8 Merits of proteomics studies

- A. Proteomics enable simultaneous study, identification and characterization of the specific protein itself, its isoforms and post-translational modified variants (Moya-Alvarado *et al.*, 2015).
- B. Proteomics provides *de novo* hypothesis based future direction of the research through less biased experimental design (Moya-Alvarado *et al.*, 2015).

Demerits of proteomics studies

- a. Highly abundant proteins may obscure the identification and quantitation of the less abundant ones.
- b. MS suffer from accuracy of providing reproducible results for the peptides (Gillette & Carr, 2013).

# 2.10.9 Importance of proteomics approach in AD therapeutics

Proteomics approach would aid much in high-throughput screening of the etiology and therapeutic approaches for AD. Cellular proteomics studies might shed light on AD biomarker development and elucidate the AD pathomechanism. Proteomics reveal that oxidative stress driven alteration of metabolic enzymes and mitochondrial functioning stand among the causative factors of AD pathogenesis (Moya-Alvarado *et al.*, 2015).

Neuropathological hallmarks of AD involves amyloid plaques and neurofibrillary tangles (NFT) (Serrano-Pozo *et al.*, 2011). Amyloid plaques consist mainly of extracellular deposition of A $\beta$  peptides (Serrano-Pozo *et al.*, 2011). Proteomics of the postmortem amyloid plaques including liquid chromatography combined with tandem mass spectrometry (LC-MS/MS) had shown that in addition to A $\beta$ , amyloid plaques

harbor about 488 types of proteins (Liao *et al.*, 2004). Of them, twenty six had been marked as plaque specific (Liao *et al.*, 2004).

Contrary to the amyloid plaques, NFTs are intra-neuronal aggregates of misfolded and/or hyperphosphorylated tau protein. Upon neuronal death, the NFTs appear as extracellular (Serrano-Pozo *et al.*, 2011). Proteomics approaches involving LC-MS/MS had identified sixty three NFT associated proteins in addition to tau (Wang *et al.*, 2005) . The most prominent were apo-lipoprotein E,  $\alpha$ -synuclein and glyceraldehyde 3 phosphate dehydrogenase (GAPDH) (Wang *et al.*, 2005).

Besides amyloid plaques and NFT, neuropil threads and dystrophic neurites containing hyperphosphorylated and misfolded tau proteins, loss of neuronal neuropils and synapses, astrogliosis and microgliosis are other hallmarks of AD. Very often, synaptic and axonal degeneration result in cognitive impairment as well as dendritic atrophy, the retrograde degeneration of axons, and the eventual atrophy of dendrites and perikarya (Serrano-Pozo *et al.*, 2011). Loss of memory remains as the top-notch symptom of AD.

Proteomics endeavor towards AD pathogenesis and amelioration goes beyond amyloid plaque and NFT pathology and search for AD biomarkers and biochemical analyses that would be helpful in early diagnosis and treatment of AD. AD diagnosis through biomarkers detection is at upper hand than those of the cognitive function tests as subtle change in cognitive tests may impair the accuracy of the neuropsychological performances (Jack *et al.*, 2011). AD standard biomarkers involve positron emission tomography (PET) that detects brain functioning through measurement of glucose metabolism and magnetic resonance imaging (MRI) that exposes brain atrophy and CSF analysis for tau and phosphorylated tau proteins (Moya-Alvarado *et al.*, 2015). As mitigation of OS and hypercholesterolemia have been found to be linked with decreased A $\beta$  generation and AD pathogenesis, therapeutic approaches targeted towards combined anti-oxidative and hypocholesterolemic outcomes seem pertinent in withstanding AD pathogenesis. Still today, there is hardly any report dealing with the combined anti-oxidative and cholesterol lowering approach for AD amelioration. Likewise, though numerous health benefits of *G. lucidum* have been reported since hoary past, its AD ameliorating effect through anti-oxidative and cholesterol lowering strategy has remained unexplored. Thus, in the present study, the anti-oxidative and cholesterol lowering performance of *G. lucidum* has been incorporated for evaluating AD ameliorating potentiality of this mushroom. Following chapters vividly describe the processes applied, results obtained and their justification that answer to the research questions outlined in the previous chapter.

# CHAPTER 3: *IN VITRO* ANTI-OXIDATIVE INVESTIGATION OF SELECTED CULINARY-MEDICINAL MUSHROOMS

# 3.1 Introduction

Over the last few decades, natural substances possessing anti-oxidative effects have received immense therapeutic importance against OS and associated physiological disorders such as AD (Mancuso *et al.*, 2012). Mushrooms abound with numerous anti-oxidants (Valverde *et al.*, 2015). The most notable among them are the polyphenolics, polysaccharides such as  $\beta$ -D glucan, tri-terpenoids and sterols (Agarwal *et al.*, 2012; Guillamón *et al.*, 2010; Kao *et al.*, 2011). Mushrooms' possession of nutraceuticals, vitamins, minerals, other bio-components and comestibility has enabled to classify them into edible and medicinal types. Thus, comparative studies among mushrooms of both types seem apt in determining the most potent one followed by furthering its biomedical importance of any sort. In the present study, *H. erinaceus, F. velutipes* and *L. edodes* have been chosen as the edible mushrooms while *G. lucidum* as the medicinal to compare their *in vitro* anti-oxidative potentiality *en route* to associating the best one's content of bio-active components, *in vivo* anti-oxidative and cholesterol lowering capacities with those of AD ameliorating performances.

Conventionally, *in vitro* anti-oxidative tests are preferred *apriori* to *in vivo*. This may be due to the ease and to preliminarily screen the anti-oxidative potentiality of any compound. When a compound fulfills the criteria of *in vitro* anti-oxidative pursuit, then comes the context of *in vivo* experimentation. Thus, the first attempt of the present study was to test the *in vitro* anti-oxidative potentiality of the selected edible-medicinal mushrooms that would seek answer of the first objective stated in chapter 1 (Section

1.5).

*In vitro*, non-enzymatic antioxidant assays can broadly be classified into two major groups based on their mode of action:

a. Electron Transfer (ET) based and

b.. Hydrogen Atom Transfer (HAT) based.

ET measures the reducing capacity of an antioxidant through donation of electron to the oxidant. The oxidant becomes reduced that is manifested by the changed color. The intensity of the color developed corresponds to the antioxidant capacity and is measured. Examples include 2,2-diphenyl-1-picrylhydrazyl (DPPH) free radical scavenging, Folin-Ciocalteu (FC) assay, ferric reducing ability of plasma (FRAP), trolox equivalent anti-oxidant capacity (TEAC) assay, total antioxidant assay using Cu<sup>2+</sup> as the oxidant.

HAT measures the hydrogen atom donating capacity of an antioxidant (reductant). The process is based on the principle that both the antioxidant and the substrate competitively compete for the free radical (usually a peroxyl radical generated upon heating) while they decompose an azo compound. Examples include inhibition of lipid peroxidation, oxygen radical absorbance capacity (ORAC) assay, total radical trapping anti-oxidant parameter (TRAP) assay and crocin bleaching assay (CBA). As the transfer of hydrogen atom is the key step in chain breaking reactions, the HAT type assays are suitable for measuring radical chain breaking anti-oxidant capacities.

Beyond these two broad groups, there are other tests such as total oxidant scavenging capacity (TOSC) assay, the chemiluminescence and the electrochemiluminescence assays (Prior *et al.*, 2005). Properties of anti-oxidant activity is highly governed by the structure of the antioxidant its solubility in the given reaction system and partition co-efficient, ionization potentiality and bond dissociation (Prior *et al.*, 2005). As part of the
*in vitro* anti-oxidative assessment of the present study, DPPH free radical scavenging, Folin-Ciocalteu assay and inhibitory effects of the mushroom extracts and fractions towards lipid peroxidation and LDL oxidation were performed.

# **3.2 Materials and Methods**

# **3.2.1 Mushroom samples**

All the mushroom samples (*G. lucidum, H. erinaceus, F. velutipes and L. edodes*) were purchased from the local market (Figure 3.1). Purchased as the dried and large slices, the *G. lucidum* was cut into fine pieces and ground with the grinder (Polymix, PX-MFC 90D, Kinematica, Switzerland) into powder. The other three species were fresh and delicate. These were then cut into small pieces, dried in the oven until became moisture-free (1-3 days) and ground into powder.

	Mushroom species	Common name	Photo
	Ganoderma lucidum	Ling Zhi/Reishi	
	Hericium erinaceus	Lion's mane	
	Flammulina velutipes	Enoki	
	Lentinula edodes	Shiitake	

Figure 3.1: Selected edible-medicinal mushrooms of the present study

### 3.2.2 Preparation of Solvent-Solvent Partitioned Fractions of Mushrooms

Bio-functionality of the bio-components is highly affected by the extraction procedure and extracting solvents. We used five extracting solvents such as methanol, dichloromethane, ethyl acetate, hexane and water. Based on their density gradient and polarity, we performed gradual fractionation of the four mushroom species following an established method in our lab (Rahman et al., 2014) (Figure 3.2). In brief, four litres of methanol:dichloromethane (2:1) mixed with each 200 gm powder of every mushroom species were stirred with the benchtop shaker (SK-300, Lab Companion, USA) at 150 rpm for 3 days at room temperature. Extraction and filtration using Whatman no. 1 filter paper were repeated thrice and at every time, the collected organic solution was evaporated using a rotary evaporator (Büchi Rotavapor R-114, Switzerland) that produced the M:DCM fraction. Dissolving the M:DCM fraction in 90% aqueous methanol, we partitioned it with hexane  $(3 \times 100 \text{ mL})$ . From the two partitioned layers (hexane fraction forming the top layer), hexane fraction was separated using a separatory funnel and then rotary evaporated till dryness to form the aqueous methanolic layer (M). It is a semisolid fraction whose successive dissolution in distilled water (100 mL) and three times partitioning with dichloromethane (DCM, 3×100 mL) turned out the DCM fraction and the aqueous fraction. Through rotary evaporation, we apportioned the DCM fraction and further partitioning of the aqueous part with ethyl acetate (EA, 3×100 mLs) accrued the EA fraction afloating the aqueous residue. We obtained the aqueous fraction through freeze-drying (Labconco) of the aqueous fraction (Figure 3.2).



Figure 3.2: Preparation of the solvent-solvent partitioned fractions

### **3.2.3 Preparation of Hot Water Extracts**

For reparing the hot water extracts (HWE) of the selected edible-medicinal mushrooms, the fruiting bodies were powdered and boiled in distilled water at the ratio of 1: 20 (w/v) at 100 °C for 45 minutes. After cooling, boiled mushrooms were removed by using Whatman No. 1 filter paper and the remaining aqueous part underwent freezedrying (Labconco) that yielded the hot water extract of each mushroom species. We preserved the extracts in desiccator at room temperature for biochemical studies.

# 3.2.4 DPPH Free Radical Scavenging Assay

The principle of this assay is that the purple colored free radical 2,2-diphenyl-1picrylhydrazyl (DPPH') becomes reduced by the reducing agent or anti-oxidant present in the reaction mixture and is converted to the pale yellow colored hydrazine that corresponds to the decrease in absorbance at around 515-517 nm. For determining the DPPH free radical scavenging effects of the mushroom fractions, we followed the modified method of Abdullah *et al.* (2011) . Each of the solvent fractions (0.1 mL volume of 1.0 mg/mL concentration) was mixed with each 3.9 mL of 0.06 mM DPPH dissolved in methanol. The mixture was shaken in darkness for 30 minutes at room temperature. The absorbance of the mixture was recorded at 515 nm. Using methanol as the blank and quercetin as the positive control (100  $\mu$ g/mL), DPPH free radical scavenging percentage was calculated using the following equation:

Radical scavenging activity (%) =  $(A_0-A_s) \times 100/A_0$ 

where  $A_0$  stands for the absorbance of 0.06 mM DPPH in methanol, and  $A_s$  denotes the reaction mixture's absorbance. The IC<sub>50</sub> value (fraction concentration required for 50% scavenging) of the most potent solvent fraction was determined from the graph of the scavenging activity against respective mushroom fraction concentration.

### **3.2.5 Folin-Ciocalteu assay**

Folin and Ciocalteu first developed a method to measure protein based on Folin-Looney reagent's ability to determine tyrosine and tryptophan residues present in the protein structure (Folin & Ciocalteu, 1927). Later, Singleton *et al.* (1999) unfurled its utility towards determining content of phenol in wine and thenceforth this method had been used for measuring total phenol (phenolics) content in different biological samples.

In the present study, the modified methodology of Slinkard and Singleton (1977) was used to detect the reducing capacity of the mushroom fractions. In succinct, 250  $\mu$ L of 10% Folin-Ciocalteu reagent was added with equal volume of each of the fraction (100  $\mu$ g/mL). Shaking of the mixture continued for 3 minutes in darkness. Addition of 10% sodium carbonate (500  $\mu$ L) to the mixture followed 60 minutes incubation in the darkness. Then, absorbance was measured at 750 nm. Pursuance of the Folin-Ciocalteu

assay has been expressed as gallic acid equivalents (GAE) per gram of fraction utilizing the calibration curve of gallic acid (2-10  $\mu$ g/mL).

### 3.2.6 Lipid Peroxidaton Inhbition Test

Oxidative modification of lipid structures especially in the carbon-carbon double bonds is referred to as lipid peroxidation. It generates various by products such as malondialdehyde (MDA) (Dianzani *et al.*, 2008). MDA and other aldehydes emanating from lipid peroxidation react with thiobarbituric acid (TBA) to form the adduct "TBA-MDA" that later on yield pink colored thiobarbituric acid reactive substances (TBARS) (Moselhy *et al.*, 2013). Spectroscopic measurement of TBARS at its absorption maximum of 532-535 nm indicates the extent of lipid peroxidation (Moselhy *et al.*, 2013).

The modified method of Daker *et al.* (2008) was used to determine the inhibitory effect of each of the mushroom fraction upon buffered egg yolk lipid peroxidation. Succinctly, emulsification of fowl egg yolk was performed with 0.1M phosphate buffer (pH 7.4) to achieve the final volume of 25 g/L. As an inducer of lipid peroxidation 100  $\mu$ L of 1M ferrous sulphate (1M, 100  $\mu$ L) was added to the mixture. Mushroom fractions (100  $\mu$ L, 100  $\mu$ g/mL) were introduced to the mixture and shaken vigorously. Following 60 minutes incubation at room temperature, freshly prepared 15% trichloroacetic acid (TCA, 0.5 mL) and 1% thiobarbituric acid (TBA, 1 mL) were added. The tubes containing the reaction mixtures were kept on the boiling water bath for 10 minutes, followed by cooling at room temperature. Precipitating the proteins through centrifugation at 3,500 g for 10 minutes, the supernatant (100  $\mu$ L) was taken and TBARS formation was analyzed through measuring absorbance at 532 nm. As control, we used the buffered egg yolk with Fe<sup>2+</sup> only. Percentage inhibition of lipid

peroxidation i.e. percentage inhibition of TBARS formation was calculated using the following equation:

Inhibition (%) = 
$$(A0-As) \times 100/A0$$

Where, A0 is the absorbance of the control, and As is the absorbance of the reaction mixture containing the fraction. The  $IC_{50}$  value (fraction concentration required for 50% inhibition) of the most potent solvent fraction was determined from the graph of the scavenging activity against respective mushroom fraction concentration.

### 3.2.7 LDL Oxidation Inhibition Test

We studied two aspects of LDL oxidation for evaluating the individual mushroom fractions inhibitory effects towards LDL oxidation: lag time extension during conjugated diene (CD) formation and inhibition of malondialdehyde (MDA) production.

# 3.2.7.1 Conjugated diene (CD) measurement

Conjugated dienes (CD) are formed during the initial stage of lipid peroxidation due to abstraction of hydrogen atom from the methylene group (CH<sub>2</sub>) of PUFA followed by the molecular rearrangement. Thus, anti-oxidative studies through measurement of CDs indicate the initial lipid peroxidation effect of any anti-oxidant. Exerting lipid peroxidation through Fenton reagent, quantification of CDs are done usually by measuring the lag phase and/or calculating the gradual increased rate of absorbance and/or percentage of inhibition of CD production (Antolovich *et al.*, 2002).

Experimentally, direct oxidation of LDL molecules is performed through low level of transition metal ions in acidic conditions. This approach of LDL oxidation at cell-free system by redox-active metal ions ( $Fe^{2+}$ ,  $Cu^{2+}$ ) has biochemical similitude with that of cellular systems (Yoshida & Kisugi, 2010). Mediated by transition metal ions, PUFAs present in the LDL molecule undergo peroxidative modification and molecular

rearrangement producing CDs. Endogenous antioxidants ( $\alpha$  tocopherol) present in LDL molecule withstands the initial oxidizing process and this suppressed oxidative stage is known as the lag time of oxidation. The potency of anti-oxidant activity is measured through measuring the extended lag time of CD formation at 234 nm.

Method of Rahman *et al.* (2014) was followed to measure the mushroom fractions' potency in extending the lag time of CD formation. In short, human LDL underwent treatment with freshly prepared 0.1 mM FeSO4 solution at room temperature so that the final concentration of LDL becomes 150 µg protein/mL at pH 7.4 and final volume adjusted to 200 µL. Mushroom fractions' and HWE's anti-oxidative performance at 1 µg/mL concentration were studied at 20 minutes intervals for a period of 180 minutes at 234 nm. Distilled water, 0.1mM FeSO4 in ultrapure water (pH 7.4) and  $\alpha$ -tocopherol LDL (10 µg/mL, Calbiochem) were used as the blank, negative -and positive control, respectively.

# 3.2.7.2 Inhibitory effect upon the formation of Malondialdehyde (MDA)

Propagation phase of rapid LDL oxidation quickly follows the lag phase where lipid peroxides are produced. Lastly, decomposition phase ensues wherein double bonds are broken and MDAs are formed. For evaluating MDA formation inhibitory effect of the mushroom fractions, we used the method of Rahman *et al.* (2014) . Freshly prepared 1M FeSO4 at room temperature was used to generate oxidative stress to human LDL (final concentration 150 µg protein/mL, final volume 200 µL, pH 7.4). Mushroom fractions were used at 1 mg/mL concentration and Distilled water, 0.1mM FeSO4 in ultrapure water (pH 7.4) and  $\alpha$ -tocopherol LDL (100 µg/mL, Calbiochem) were used as the blank, negative -and positive control, respectively. All the reaction mixtures were incubated at 37 °C for 180 minutes. Then, tricholoroacetic acid (TCA, 15%, 500 µL) and thiobarbituric acid (TBA, 1%, 1 mL), both freshly prepared, were added and

incubated at 100 °C for 10 minutes followed by cooling at room temperature. An aliquot of the mixture taken to the ELISA plate was read at 532 nm. Using malondialdehyde dimethyl acetal as the standard at concentration range of  $0 - 100 \mu$ M, a calibration curve was prepared (TBARS assay kit, item no. 10009055, Cayman Chemicals, USA). Exploiting the calibration curve, we determined the level of MDA produced and expressed as  $\mu$ mol MDA/mg protein LDL.

### 3.3 Results

# 3.3.1 DPPH Free Radical Scavenging Performance

As shown in Figure 3.3, all the mushroom extracts (1 mg/mL) and the positive control, quercetin (100  $\mu$ g/mL) scavenged DPPH free radical and their rate of scavenging differed from each other (Raw data: Appendix A, Tables A - 1.1, 1.2, 1.3, 1.4, 1.5 and 1.6; Figures A - 1.1 and 1.2; Statistical data: Appendix C, Tables C- 1.1 and 1.2). Even, within the same mushroom species, different extract and solvent fractions had different scavenging activity.





Data are expressed as mean $\pm$ SE (Bar charts with different lower case superscripts are indicators of statistically significant difference at P $\leq$ 0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3). Here, MD stands for methanol:dichloromethane; HEX for hexane; DCM for dichloromethane, EA for ethylacetate, AQ for aqueous and HWE for hot water extract. Mushroom fractions and HWE were used at 1 mg/mL while the positive control querecetin (Q) at 100 µg/mL concentrations.

#### **3.3.2 Folin-Ciocalteu assay**

The HWE of all the mushroom species showed the best reducing capacity as shown in Figure 3.4 (Raw data: Appendix A, Tables A - 2.1, 2.2, 2.3, 2.4, 2.5 and Figure A 2.1; Statistical data: Appendix C, Tables C - 2.1 and 2.2). Reducing capacity of each gram of the HWE of *G. lucidum*, *H. erinaceus*, *L. edodes and F. velutipes*, was 81.06, 73.63, 63.03 and 56.97 mg gallic acid equivalent (GAE), respectively. The values were significantly different among each other at 5% level of confidence. EA fraction of *G. lucidum* was second in terms of reducing capacity (75.3 mg GAE/g), following very closely by those of DCM (52.4 mg GAE/g) and MD (51.3 mg GAE/g). Compared to each of the fraction of other three species, significantly higher ( $P \le 0.05$ ) reducing capacity in the MD, DCM and EA fraction of *G. lucidum* was also observed.



#### Figure 3.4: Folin-Ciocalteu assay

Data are expressed as mean $\pm$ SE. Bar charts with different lower case superscripts are indicators of statistically significant difference at P $\leq$ 0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3). Here, MD stands for methanol:dichloromethane; HEX for hexane; DCM for dichloromethane, EA for ethylacetate, AQ for aqueous and HWE for hot water extract. Mushroom fractions and HWE were used at 1 mg/mL while the positive control querecetin (Q) at 100 µg/mL concentrations.

### 3.3.3 Inhibition of Lipid Peroxidation

As depicted in Figure 3.5, the hexane fraction (at 1 mg/mL) of each mushroom species had the highest inhibitory effect towards lipid peroxidation (Raw data: Appendix A, Tables A - 3.1, 3.2, 3.3 and 3.4; Statistical data: Appendix C, Tables C - 3.1 and 3.2). Among the four mushroom species, the hexane fraction of *H. erinaceus* mostly inhibited (72.59%), followed sequentially by that of *L. edodes* (67.07%), *G. lucidum* (58.56%) and *F. velutipes* (55.39%). DCM fractions of all the mushrooms stood second to those of hexane in this inhibitory concert: *H. erinaceus* (66.59) > *L. edodes* (64.23) > *G. lucidum* (55.47) > *F. velutipes* (45.58). At 100 µg / mL, the positive control quercetin showed 80.54% inhibition of LPO and its IC<sub>50</sub> value was 0.056 mg/mL. IC<sub>50</sub> value of the hexane fraction of *H. erinaceus* was found to be 0.76 mg/mL.



#### Figure 3.5: Lipid peroxidation inhibitory effect

Data are expressed as mean±SE. Bar charts with different lower case superscripts are indicators of statistically significant difference at P $\leq$ 0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3). Here, MD stands for methanol:dichloromethane; HEX for hexane; DCM for dichloromethane, EA for ethylacetate, AQ for aqueous and HWE for hot water extract. Mushroom fractions and HWE were used at 1 mg/mL while the positive control querecetin (Q) at 100 µg/mL concentrations.

# 3.3.4 Inhibition of LDL Oxidation

### **3.3.4.1** Lengthening of the Lag time of CD Formation

At our experimental milieu, all the extracts of all the mushroom species showed lag time lengthening effect (Table 3.1) (Raw data: Appendix A, Tables A - 4.1, 4.2, 4.3 and 4.4; Statistical data: Appendix C, Tables C - 4.1 and 4.2) . However, their performance differed from each other. In case of every mushroom, the hexane fraction exerted the most lengthening effect (*G. lucidum* 127.67 minutes, *H. erinaceus* 120.43 minutes, *L. edodes* 115.78 minutes and *F. velutipes* 110.34 minutes) compared with the other fractions. Like that of lipid peroxidation inhibition (Figure 3.5), the DCM fractions of all the mushroom species stood second in lag time lengthening of CD formation (Table 3.1). Lengthening time of HWE of all the mushrooms was in the range of 80-86 minutes and third in the category. Effect of MD and EA was close to each other and their timing ranged from 40 to 50 minutes.

Extract/	<i>G</i> .	H.	L.	F. velutipes
Fraction/	lucidum	erinaceus	edodes	_
Standard				
MD	43.33 <sup>a</sup>	37.67 <sup>b</sup>	33.33 <sup>c</sup>	26.00 <sup>d</sup>
HEX	127.67 <sup>a</sup>	120.43 <sup>b</sup>	115.78 <sup>c</sup>	110.34 <sup>d</sup>
DCM	80.67 <sup>a</sup>	67.00 <sup>b</sup>	113.33 <sup>c</sup>	101.67 <sup>d</sup>
EA	53.33 <sup>a</sup>	62.33 <sup>b</sup>	42.33 <sup>c</sup>	62.33 <sup>d</sup>
AQ	39.33 <sup>a</sup>	36.00 <sup>b</sup>	24.00 <sup>c</sup>	60.67 <sup>d</sup>
HWE	43.33 <sup>abd</sup>	41.67 <sup>abd</sup>	36.67	44.00 <sup>abd</sup>
α-tocopherol		13	8.33	

 Table 3.1: LDL oxidation inhibitory effect of the mushroom fractions and extracts through lengthening the lag time of CD formation (minute)

Data are expressed as mean±SE. Mean values with different lower case superscripts are indicators of statistically significant difference at P $\leq$ 0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3). Here, MD stands for methanol:dichloromethane; HEX for hexane; DCM for dichloromethane, EA for ethylacetate, AQ for aqueous and HWE for hot water extract. Mushroom fractions and HWE were used at 1 mg/mL while the positive control,  $\alpha$ -tocopherol at 100  $\mu$ g/mL concentrations.

### 3.3.4.2 Malondialdehyde (MDA) assay

As shown in the figure 3.6, compared to the positive control,  $\alpha$ -tocopherol, the hexane fractions of all the mushroom species were most potent among the six fractions. Level of MDA generation ( $\mu$ M/mg protein-LDL) was comparable (*L. edodes* 13.39, *G. lucidum* 14.64, *H. erinaceus* 15.90 and *F. velutipes* 15.48  $\mu$ M/mg protein-LDL) with that of  $\alpha$ -tocopherol (10.46  $\mu$ M/mg protein-LDL) since the values were not statistically significant (Raw data: Appendix A, Tables A - 4.5, 4.6, 4.7, 4.8 and 4.9, Figure A 4.1; Statistical data: Appendix C, Tables C 4.3 and 4.4).



# Figure 3.6: LDL oxidation inhibitory effect of the mushroom fractions and extracts through reduced production of MDA

Data are expressed as mean±SE. Mean values with different lower case superscripts are indicators of statistically significant difference at P $\leq$ 0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3). Here, MD stands for methanol:dichloromethane; HEX for hexane; DCM for dichloromethane, EA for ethylacetate, AQ for aqueous, HWE for hot water extract and MDA for malondialdehyde. Mushroom fractions and HWE were used at 1 mg/mL while the positive control,  $\alpha$ -tocopherol at 100 µg/mL concentrations.

### **3.4 Discussion**

### **3.4.1 DPPH free radical scavenging effect**

Anti-oxidants scavenge DPPH free radicals by donating hydrogen atom with changing of the purple color to yellow with concomitant decrease in absorbance at 515 nm. Among six fractions, the hot water extract (HWE) of *G. lucidum* mostly (55.13%, significant at  $P \le 0.05$  level) scavenged the DPPH free radical (Figure 3.3). The MD fraction of this mushroom stood second with 42.36% (significant at  $P \le 0.05$  level) scavenging effect. DPPH free radical scavenging effect of the MD fraction of *G. lucidum*, *H. erinaceus*, *L. edodes* and *F. velutipes* were significantly different from each other at 5% confidence level. Moderate scavenging effect (18.88-28%) was shown by the DCM, EA and AQ fractions of each of the mushroom species. IC<sub>50</sub> of the most potent fraction, the HWE of *G. lucidum*, was 1.47 mg/mL compared to that of quercetin with 0.069 mg/.

DPPH free radical scavenging effect observed in the present study was in line with those reported by other researchers (Abdullah *et al.*, 2011; Mau *et al.*, 2002b; Mau *et al.*, 2005; Yildiz *et al.*, 2015). The methanolic extracts of *G. lucidum*, *G. lucidum* antler and *G. tsugae* have been reported to scavenge DPPH-free radical more than 70% (at 0.64 mg/mL concentration) and hydroxyl radical more than 50% (at 16 mg/mL concentration) (Mau, Lin & Chen, 2002). At 2.4 mg/mL concentration, ferrous ion chelating activity was 44.8% (*G. tsugae*), 55.5% (*G. lucidum*) and 67.7% (*G. lucidum* antler) (Mau *et al.*, 2002a). Similar anti-oxidative prowess has been reported for the ethanolic extract of *G. lucidum* (Rajasekaran & Kalaimagal, 2011). The best *in vitro* anti-oxidative effect of the HWE of *G. lucidum* among different solvent (hot water, hydro-alcoholic, chloroform and petroleum ether) extracts of have also been observed by Agarwal *et al.* (2012). Abdullah *et al.* (2011) also ranked the HWE of *G. lucidum* first on the basis of anti-oxidant index (30.1%) among fourteen culinary-medicinal mushrooms.

#### **3.4.2 Folin-Ciocalteu assay**

Among all, the HWE of *G. lucidum* had been found possessing the best reducing capacity as evident by the Folin-Ciocalteu assay (Figure 3.4) (Raw data: Appendix A, Tables A - 2.1, 2.2, 2.3, 2.4, 2.5 and Figure A 2.1; Statistical data: Appendix C, Tables C - 2.1 and 2.2). Observed Folin-Ciocalteu assay results are compatiable with those of Yildiz *et al.* (2015). However, compared to some other reports, the current findings are higher and indicate better reducing capacities of the respective mushroom species (Abdullah *et al.*, 2011; Heleno *et al.*, 2015; Kim *et al.*, 2008; Mau *et al.*, 2002; Rawat *et al.*, 2013; Reis *et al.*, 2012; Wong *et al.*, 2009). Mau *et al.* (2002) also observed about four times higher reducing capacities of the *Ganoderma* species than those of the *Coriolus versicolor*.

### 3.4.3 Inhibition of lipid peroxidation

In the present study, inhibitory performance of the mushroom extracts upon the lipid peroxidation (LPO) of the buffered egg yolk was assessed through measurement of TBARS. As TBARS generation is linked with the extent of lipid peroxidation, the lower the level of TBARS indicates the better anti-oxidative capacity of the mushroom extract. In this sense, the hexane fractions of all the selected mushrooms stood first in inhibiting LPO (Figure 3.5). The other fractions and extracts varied among each other in the context of TBARS formation inhibition. Present findings are compatible with those of Mau *et al.* (2002) who regarded the better LPO inhibitory effect of *G. lucidum* with its higher content of polyphenolics. In addition, the hexane fractions of the present study might have possessed lipo-philic bio-components that provided with the enhanced anti-oxidative support in lowering LPO.

### **3.4.4 Inhibition of LDL oxidation**

Based on established methods, *in vitro* LDL oxidation of human LDL was performed using transition metal ion  $Fe^{2+}$  (Lynch & Frei, 1993; Morgan & Leake, 1995; Visioli *et al.*, 2000). The proposition was such that redox-active metal ion ( $Fe^{2+}$ ,  $Cu^{2+}$ ) – mediated *in vitro* LDL oxidation is mechanistically alike to that of *in vivo* systems (Lapeyre *et al.*, 2005; Miranda *et al.*, 2000; Steinberg *et al.*, 1989; Turchi *et al.*, 2009). These ions participate in the peroxidative modification and molecular rearrangement of the poly unsaturated fatty acids (PUFAs) present in the LDL molecules and produce CD (Lynch & Frei, 1993; Morgan & Leake, 1995). At the initiation, LDL contained endogenous antioxidants guard against oxidative stress and thus there occurs a little OS, which is known as the lag phase of LDL oxidation. Upon addition of different mushroom extracts, the lag time had been extended (Table 3.1). Noteworthy, mushroom fractiontreated samples showed a gradual decrease in absorbance that is indicative of the decreased LDL oxidation as well as anti-oxidative potencies of the mushroom fractions.

As LDL oxidation is a step by step process, initial lag phase ends up quickly and rapid LDL oxidation phase (propagation phase) ensues with production of the lipid peroxides). At the third (decomposition) step, the double bonds break up with the generation of aldehydes such as malondialdehydes (MDA). In the present experimental arrangement, nucleophilic addition reaction of MDA with 2-thiobarbituric acid (TBA) produced thiobarbituric acid reactive substances (TBARS) (Lefevre *et al.*, 1997). Current investigation was to find out whether different solvent fractions of different mushrooms could inhibit or lower the production of MDA (in case of LDL oxidation) and TBARS (incase of lipid peroxidation). Obtained data suggest that the hexane fractions of all the mushrooms were most potent in inhibiting the formation of both MDA and TBARS (Figures 3.5 and 3.6). Lipophilic bio-components present in the non-polar hexane might impart in inhibiting lipid peroxidation and LDL oxidation in the

present experiment. Anti-oxidative mushroom bio-components might themselves become oxidized through quenching the free radicals and prevented lipid peroxidation and LDL oxidation (Turchi *et al.*, 2009). Probably, they had acted as electron donors towards  $Fe^{2+}$  and lowered the oxidative stress towards lipid and LDL molecules.

Conjugated dienes (CD) are formed during the initial phase of LDL oxidation. Antioxidant activity is usually determined based on the lengthening capacity of a substance upon the lag time of CD formation. Principally, the enhanced the lag time of any CD forming reaction, the lower the extent of LDL oxidation and the higher the antioxidative prowess of the anti-oxidant. In this context, the hexane fractions of all the mushrooms surpassed the rest of the fractions and the extracts. This trend was similar to those of LPO inhibition.

Malondialdehyde (MDA) is an intermediary product of LDL oxidation and we measured its level to test the potency of the mushroom extracts in withstanding the *in vitro* oxidation of human LDL. Among the four mushroom species, the *L. edodes* showed the best inhibition of LDL oxidation as indicated by the least production of MDA (Figure 3.6). DCM fractions of all the mushrooms stood next to those of hexane (Figure 3.6). The pattern of LDL oxidation inhibition tallied with those of lipid peroxidation inhibition (Figure 3.5). Aqueous fraction of all the mushrooms showed the least inhibitory potential while those of MD, EA and HWE were moderate. This may be due to the presence and outperformance of the lipo-philic anti-oxidants in the hexane fractions of the mushroom species studied.

# **3.5** Conclusion

The above mentioned *in vitro* anti-oxidative test results reveal that the HWE of *G*. *lucidum* excels most in scavenging DPPH free radicals and possesses the highest reducing capacity as assessed by the Folin-Ciocalteu outcomes. However, the hexane fractions of all the mushrooms surpassed the respective HWEs in inhibiting lipid peroxidation and LDL oxidation. Though the hexane fraction of *H. erinaceus* and *L. edodes* mostly inhibited the formation of TBARS and MDA, respectively, the hexane fraction of *G. lucidum* lengthened the lag time of CD formation most. Thus, among the selected edible-medicinal mushrooms, the *in vitro* anti-oxidative performance of *G. lucidum* was the best. Among the five solvent-solvent partitioned fractions and HWEs of each of the four selected species, the HWE of *G. lucidum* possessed the highest free radical scavenging and reducing capacities. Thus, the HWE of *G. lucidum* was selected for assessing its *in vivo* anti-oxidative (chapter 5), cholesterol lowering (chapter 6) and AD ameliorating potentiality (chapter 6). Prior to those studies, bio-active components present in the HWE of *G. lucidum* and other potential extracts of the selected mushrooms have been determined whose description have been appended in the next chapter (chapter 4).

# CHAPTER 4: DETERMINATION OF BIO-ACTIVE COMPONENTS IN THE SELECTED MUSHROOM FRACTIONS AND EXTRACTS

# 4.1 Introduction

Through *in vitro* tests described in the previous chapter, differential anti-oxidative performance of different fractions and extracts of different mushroom species have been observed. Stated objectives of *in vivo* anti-oxidative, cholesterol lowering and AD ameliorating effect of the selected mushrooms depend on the presence of the responsible bio-active components. Thus, it is imperative to identify the presence of the bio-active components in the potential mushroom fractions and extracts. This would fulfill the second objective of the present study.

Due to solvent-solvent partitioning, the selected mushroom fractions and extracts had been fractionated either into lipophilic or into hydrophilic types. As a consequence, determination of the bio-active components in the respective fractions and extracts are largely dependent on the solubility-based identification processes. Based on this principle, gas chromatography - tandem mass spectrometry (GC-MS/MS) for the lipophilic and liquid chromatography – tandem mass spectrometry (LC-MS/MS) for the hydrophilic fractions and extracts, respectively, were used to identify the bio-active components. The hexane fractions of *G. lucidum*, *H. erinaceus*, *L. edodes* were selected for GC-MS while the MD fraction of *F. velutipes* and the HWE of *G. lucidum* were selected for LC-MS analyses. Content of  $\beta$ -D glucan in the HWE of *G. lucidum* was also determined using commercially available kits.

### 4.2 Materials and methods

### 4.2.1 GC-MS/MS Analysis

For identifying bio-active components in the hexane fractions of *G. lucidum*, *H. erinaceus* and *L. edodes*, GC-MS/MS analysis was performed utilizing gas chromatography directly coupled to a mass spectrometer system (Agilent 7000 C triple quadruple GC/MS system, USA). The system was equipped with a HP-5ms silica capillary column (30 m x 250  $\mu$ m, 0.25  $\mu$ m film) with injector temperature 250°C, split less injection technique, injection volume of 1.5  $\mu$ L, flow rate 1 mL/min, oven temperature 70°C – 300°C, helium gas as the carrier, electronic ionization mode with 70 eV ionization energy, ion source temperature 200 °C and interface temperature 300°C and masses scanned in the range of 50–1200 m/z. The peaks were identified comparing with the mass spectra library of the National Institute of Standards and Technology (NIST 14)/Environmental Protection Agency (EPA)/National Institutes of Health (NIH), USA.

### 4.2.2 LC-MS/MS Analysis

The LC/MS/MS analysis of the MD fraction of *F. velutipes* and the HWE of *G. lucidum* were performed using the Applied Bioscience's Sciex 6500 QTRAP tandem mass spectrometer (model 5017226-H, serial BL21931304) equipped with the UHPLC FX15 system of Perkinn Elmer. A Turbo V spray source connected the chromatography system with the ionization chamber. Mobile phase A consisted of 5mM ammonium formate and water with 0.1% formic acid v/v while mobile phase B consisted of 5mM ammonium formate and acetonitrile with 0.1% formic acid v/v. The process was accomplished with injection volume of 10  $\mu$ L, flow rate of 400  $\mu$ L / min using the Phenomenex aqua column (particle size 3  $\mu$ M, 50 mm x 1.2 mm). A diode array detector recorded the LC spectra in the range of 50-1200 m/z.

# 4.2.3 Determination of β-D Glucan level

The level of  $\beta$ -D glucan in the HWE of *G. lucidum* was measured using the megazyme  $\beta$  glucan assay kit (yeast and mushroom, K YBGL 01/2016) (Megazyme, USA). Manufacturer's instructions supplied with the kit package were followed during experimentation and the calculation was performed using the calculating software provided by the manufacturer (K-YBGL\_CALC, Megazyme, USA).

# 4.3 Results

# 4.3.1 GC-MS/MS Analysis of the Hexane Fraction of G. lucidum

The GC-MS/MS chromatogram of the hexane fraction of *G. lucidum* has been demonstrated at Figure A 5.1 of appendix A and the information relating the most abundant gano-components in Table 4.1 of this chapter.

No.	Name	Structure	Retention time	Peak area (%)	M. formula	M. t (g/mol)
1	11-Octadecenoic acid, methyl ester (Vaccenic acid)	~~~~ <sup>1</sup> он	16.591	16.81	C <sub>18</sub> H <sub>34</sub> O <sub>2</sub>	282.46
2	9-Octadecenoic acid (Oleic acid)		17.054	12.57	C <sub>18</sub> H <sub>34</sub> O <sub>2</sub>	282.46
3	n-Hexadecanoic acid (Palmitic acid)	~~~~~~	15.355	6.90	C <sub>16</sub> H <sub>32</sub> O <sub>2</sub>	256.42
4	cis-9, cis-12 Octadecadienoic acid (Linoleic acid)	ОН	14.886	5.57	C <sub>18</sub> H <sub>32</sub> O <sub>2</sub>	280.45
5	Tetradecanoic acid/ Myristic acid	Он Он	13.226	1.18	C <sub>14</sub> H <sub>28</sub> O <sub>2</sub>	228.37
5	5,6-Dihydro ergosterol		25.837	0.68	C <sub>28</sub> H <sub>46</sub> O	398.66

Table 4.1: Identified bio-components in the hexane fraction of G. lucidum

As found from the GC-MS analyses, the hexane fraction of *G. lucidum* abounds with 11-Octadecenoic acid, methyl ester (vaccenic acid), 9-Octadecenoic acid (oleicacid), n-Hexadecanoic acid (palmitic acid), cis-9, cis-12 Octadecadienoic acid (linoleic acid), tetradecanoic acid/myristic acid and 5,6-dihydro ergosterol. Some prior studies also identified the palmitic, oleic and linoleic fatty acids as the principal fatty acids present in the *G. lucidum* (Lv *et al.*, 2012; Stojković *et al.*, 2014).

# 4.3.2 GC-MS/MS Analysis of the Hexane Fraction of H. erinaceus

The GC-MS/MS chromatogram of the hexane fraction of *H. erinaceus* has been provided in Figure A 5.2 of appendix A and the identified bio-components are listed in Table 4.2 of this chapter. Based on the peak area percentage, the relative retention time (Rt) and respective quality, ten components were identified.

No	Name	Structure	Rt	Deak	м	M wt
110.	Ivanie	Structure	Kt	Area	formula	(g/
				(%)		mol)
1	Ergosterol	HIC CHS HIC CHS	25.72	5.20	C <sub>28</sub> H <sub>44</sub> O	396.65
2	Octadecanoic acid/Stearic acid	но	17.541	6.28	$C_{18}H_{36}O_2$	284.48
3	9,12-Octadecadienoic acid (Z,Z/ <i>cis</i> -Linoleic acid)	ОН	17.35	27.31	C <sub>18</sub> H <sub>32</sub> O <sub>2</sub>	280.45
4	n-Heptadecanoic acid/Margaric acid		16.83	2.15	C <sub>17</sub> H <sub>34</sub> O <sub>2</sub>	270.45
5	n-Hexadecanoic acid /Palmitic acid	ОН	14.91	2.64	C <sub>16</sub> H <sub>32</sub> O <sub>2</sub>	254.41
6	Pentadecanoic acid	С С С С С С С С С С С С С С С С С С С	14.37	1.57	$C_{15}H_{30}O_2$	242.39
7	Tetradecanoic acid/Myristic acid, methyl ester	Он Он	13.24	0.58	$C_{14}H_{28}O_2$	228.37
8	Cetene (1-Hexadecene)	$\qquad \qquad $	11.31	2.04	C <sub>16</sub> H <sub>32</sub>	224.43
9	Phenol	OH	10.42	2.33	C <sub>6</sub> H <sub>6</sub> O	94.11
10	3-Tetradecene	$\sim \sim $	8.86	1.23	C <sub>14</sub> H <sub>28</sub>	196.37

Table 4.2: Identified bio-components in the hexane fraction of *H. erinaceus* 

# 4.3.3 GC-MS/MS Analysis of the Hexane Fraction of L. edodes

As shown in Table 4.3 in this chapter and in Figure A 5.3 of appendix A, the most abundant bio-active components present in the hexane fraction of *L. edodes* were  $\alpha$ -tocopherol (vitamin E), oleic acid, linoleic acid, ergosterol and butyric acid.

No.	Bio-	Structure	М.	M.wt
	component		formula	(g/mol)
1	α-tocopherol	$HO \qquad \qquad$	C <sub>29</sub> H <sub>50</sub> O <sub>2</sub>	430.71
2	Oleic acid	0	C <sub>18</sub> H <sub>34</sub> O <sub>2</sub>	282.46
3	Linoleic acid	ОН	C <sub>18</sub> H <sub>32</sub> O <sub>2</sub>	280.45
4	Ergosterol	H <sub>3</sub> C CH <sub>3</sub> CH <sub>3</sub> CH <sub>3</sub> CH <sub>3</sub> CH <sub>3</sub> CH <sub>3</sub> CH <sub>3</sub>	C <sub>28</sub> H <sub>44</sub> O	396.65
5	Butyric acid	ОН	$C_4H_8O_2$	88.11

Table 4.3: Identified bio-components in the hexane fraction of L. edodes

# 4.3.4 LC-MS/MS Analysis of the MD Fraction of F. velutipes

Polyphenolic substances namely protocatechuic acid, p-coumaric acid and ellagic acid in the MD fraction of *F. velutipes* (Table 4.4; Appendix A, Figures A-5.4, 5.5, 5.6).

No.	Rt	Mode (+/-	Compound name and structure	Molecular	M. wt.
-	(m)	)		Iormula	(g/mol)
1	3.4	-	НО	$C_7H_6O_4$	154.12
			HO		
			HO′ ≦O		
			Protocatechuic acid		
2	5.2	-	но	C <sub>9</sub> H <sub>8</sub> O <sub>3</sub>	164.16
			p-coumaric acid		
3	7.68	_	HO HO HO Ellagic acid	$C_{14}H_6O_8$	302.197

Table 4.4: Polyphenolic compounds identified in the MD fraction of F. velutipes

# 4.3.5 LC-MS/MS Analysis of the HWE of G. lucidum

As shown in the (Table 4.5, Appendix A, Figure A 5.7) LC-MS/MS analysis of the HWE of *G. lucidum* revealed the presence of phenolics (protocatechuic acid, cinnamic acid, p-hydorxy cinnamic acid, gallic acid, caffeic acid), monogalloyl glucose, caffeoyl glucose, Lucidenic acid A and F, Elfvingic acid and tri-terpenoids (ganoderic acids A, B, C, D, F, AM1, LM2 and hydroxy ganoderic acid). However, some of the peaks could not be related (834, 835, 838.5, 839, 1034.6 m/z, Da) with the known bio-components and those remained unidentified (Table 4.5).

No.	m/z	Gano-component	Structure
1	108.0	Protocatechuic acid	HO HO HO
2	125.0	Gallic acid	СООН НО ОН ОН
3	164.0	p-Hydroxy cinnamic acid	ОН
4	192.0	Caffeic acid	OH HO O O O O
5	332.0	Monogalloyl glucose	HO HO HO HO HO HO HO HO HO HO HO HO HO H
6	338.0	Cinnamic acid	OH Cinnamic acid
7	341.1	Caffeoyl glucose	HO HO HO Caffeoyl glucose
8	349.0	D-Alanine	O OH I I NH <sub>2</sub> D Alanine

# Table 4.5: Gano - components identified in the HWE of G. lucidum

No.	m/z	Gano-component	Structure
9	496.2	Lucidenic acid A	о с с с с с с с с с с с с с с с с с с с
10	510.2	Ganoderic acid A	о н он
11	513.2	Ganoderic Acid AM1	
		Ganoderic acid LM2	о с но с
12	517.3	Ganoderic acid B	но он он
13	529.2	Ganoderic acid C6	

# Table 4.5 (continued): Gano - components identified in the HWE of G. lucidum

No.	m/z	Gano-component	Structure
14	533.2	Ganoderic acid G	
15	535.2	12-Hydroxyganoderic acid C2	HO, VIII, OH O, O, T, OH O, O, T, OH
16	549.2	Ganoderic Acid D	о с с с с с с с с с с с с с с с с с с с
17	552	Ganoderic Acid H	
18	554.3	Ganoderic Acid J	О ОН
19	555.3	Ganoderic Acid K	

# Table 4.5 (continued): Gano - components identified in the HWE of G. lucidum

No.	m/z	Gano-component	Structure
20	559	Ganoderic Acid S	ОН
21	571.2	Ganoderic acid Z	но
22	834.0	Unidentified	
23	835.0	Unidentified	
24	838.5	Unidentified	
25	839.0	Unidentified	
26	1034.6	Unidentified	

# Table 4.5 (continued): Gano - components identified in the HWE of G. lucidum

### 4.3.6 Determination of β-D-Glucan Level in the HWE of G. lucidum

The best *in vitro* anti-oxidative performance and content of the highest anti-oxidative and other bio-components in the HWE of *G. lucidum* intrigued the determination of its  $\beta$ -D-glucan content. Beyond its anti-oxidative performance,  $\beta$ -D-glucan has been revered for anti-hyerprlipidemia, anti-tumor, immunomodulating and other health giving potentials (Kanagasabapathy *et al.*, 2013; Wasser, 2002). In the present study, as much as 44.15% (w/w)  $\beta$ -D glucan was detected in the HWE of *G. lucidum* (Raw data: Appendix A, Figure A 5.8, calculation using K-YBGL\_CALC, Megazyme, USA).

### 4.4 Discussion

### 4.4.1 GC-MS/MS analysis of the hexane fraction of G. lucidum

11-Octadecenoic acid also called vaccenic acid (VA) was among the most available gano-components present in the hexane fraction (Table 4.1). It is an omega-7 fatty acid and finding from this study is novel in respect of being available in the mushroom

world, specifically in G. lucidum. Its cholesterol lowering, anti-atherosclerotic and anticarcinogenic effects have been reported (Kritchevsky et al., 2000; Lock et al., 2004). Next to VA, oleic acid (9-Octadecenoic acid) was the second most abundant ganocomponent in the hexane fraction of G. lucidum. Anti-oxidative and anti-atherosclerotic functionality of its monoacylglycerol (MAG) moieties had been noticed both in vitro and in cellular models (Cho et al., 2010). It is much potent than the palmitic-MAG in scavengeing free radicals, inhibiting Cu<sup>2+</sup>- induced LDL oxidation, inhibited lowering the activity against LDL-associated phospholipase A(2) along with potent activation of paraoxonase activity, that maintains cellular anti-oxidative defense arsenal (Cho et al., 2010). Similar mode of free-radical scavenging and anti-oxidative protection might have been employed in the present study. Diet rich-in oleic acid aids in lowering blood LDL level (Allman-Farinelli et al., 2005). Besides, incorporation of oleic acid in the culture media had been reported lowering the uptake of oxidized LDL molecules into the THP-1 cells as well as decreased generation of malondialdehyde and lipid hydroperoxide (Cho et al., 2010). At the molecular level, oleic acid reduces the expression of vascular cell adhesion molecule-1 (VCAM-1), intercellular adhesion molecule-1 (ECAM-1) and E-selectin and interrupts the activation of nuclear factorkappa B (NFkB) and thus aids in controlling atherogenesis (Carluccio et al., 1999; Cho et al., 2010). Epidemiological studies also reveal that oleic acid, through an alternative mode of action, modulates vascular response to atherogenic precursors (Carluccio et al., 1999).

n - Hexadecanoic acid (palmitic acid) had been found as the third most abundant gano-component (Table 4.1). It is a saturated fatty acid having anti-oxidant activities (Keawsa-Ard *et al.*, 2012). Another novel anti-oxidant present in the hexane fraction of *G. lucidum* was the tetradecanoic acid/myristic acid (Keawsa-Ard *et al.*, 2012). Among others, linoleic acid (LA, cis-9, cis-12 octadecadienoic acid) and its different

constitutional and stereoisomers had been detected that belong to the group "conjugated linoleic acid (CLA)". Both *in vitro* and *in vivo* anti-oxidative performance of CLA had been found (da Silva Marineli *et al.*, 2012; Palacios *et al.*, 2003). In rat plasma, CLA had been found increasing catalase level by 3.7 fold and in liver, CLA supplemented with phytosterol, decreased lipid peroxidation by 58% (da Silva Marineli *et al.*, 2012). Its anti-oxidative potential against mitochondrial and peroxisomal lipid peroxidation had been found better than that of vitamin A (Palacios *et al.*, 2003). Multiple strands of research had demonstrated hypocholesterolemic, LDL lowering and anti-atherosclerotic effect of CLA (Kritchevsky *et al.*, Lee *et al.*, 1994; 2004; Toomey *et al.*, 2003). Proposed anti-atherosclerotic effect of CLA involves stimulation of the nuclear transcription factor peroxisome proliferator activated receptor  $\gamma$  (PPAR $\gamma$ ) expression in the vascular smooth muscle cells (Toomey *et al.*, 2003). Reduced risk of coronary heart diseases (CHD) has been linked with increased consumption of linoleic acid (Rudel, 1999).

5, 6 - Dihydro ergosterol was another important gano-component detected in the present study (Table 4.1). Ergosterol is the most common fungal sterol and among numerous bio-functionalities, its role as the membrane antioxidant has also been regarded (Wiseman, 1993; Yaoita *et al.*, 2002; Zhang *et al.*, 2003). Its level varies depending on the cultivation and extraction techniques of mushrooms (Barreira *et al.*, 2013; Lv *et al.*, 2012; Stojković *et al.*, 2014). Kim *et al.* (1999) extracted ergosterol peroxide from the mushroom *Armillariella mellea* and reported its anti-oxidant capacity to be far exceeding that of  $\alpha$ -tocopherol and thiourea. Supplementation of ergosterol and oleic acid in the medium containing *Saccharomyces cerevisiae* had been reported to lower the production of reactive oxygen species, oxidative stress and the resultant damage towards cellular and biomolecular levels (Landolfo *et al.*, 2010). Sterols from *G. lucidum* had been found inhibitory towards generation of reactive oxygen species

(ROS) and protective against hypoxia/reoxygenation-mediated oxidative stress (Zhao *et al.*, 2005).

Mushrooms' anti-oxidant effects relating to DPPH free radical scavenging and Folin-Ciocalteu assay (reducing capacity test) has been attributed towards distinct groups of bio-components (Kalogeropoulos *et al.*, 2013). Phenolic and polyphenolic group of reducing substances have been regarded providing DPPH free-radical scavenging and Folin-Ciocalteu assay performance while triterpenoids have been linked with metal chelating activities (Kalogeropoulos *et al.*, 2013). Thus, 5,6-dihydroxy ergosterol and structurally relevant components such as 7, 22 - ergostadienol, ergosta 7, 22- dien-3-ol present in the hexane fraction might have been involved in chelating transition metal ion, Fe<sup>2+</sup> and thus mediated reduced oxidative stress and consequent lowered LDL oxidation.

# 4.4.2 GC-MS/MS analysis of the hexane fraction of *H. erinaceus*

Ergosterol was among the most abundant bio-components present in the hexane fraction of *H. erinaceus* (Table 4.2). Indeed, ergosterol is the principal sterol in the edible mushrooms that provide anti-oxidative defense (Yaoita *et al.*, 2002; Zhang *et al.*, 2003). 5,6-Dihydroergosterol has potent inhibitory role against the expression of iNOs and production of nitric oxide (NO) that manifest it as an OS retarding and inflammation preventing agent (Park *et al.*, 2013). Total sterol extracted from mushrooms had been suggested to have ameliorating effect on hypoxia/reoxygenation – induced oxidative stress and inhibitory effect on the formation of reactive oxygen species (ROS) (Zhao *et al.*, 2005).

Fatty acids of both saturated (octadecanoic acid/stearic acid, n-heptadecanoic acid/margaric acid, n-hexadecanoic acid/palmitic acid, pentadecanoic acid and tetradecanoic acid/myristic acid methyl ester) and unsaturated types (9, 12-

octadecadienoic acid/Z,Z/*cis*-linoleic acid) were also detected (Table 4.2). These components had been reported having preventive role against oxidative stress induced physiological complications (Guillamón *et al.*, 2010; Kalač, 2009). For instance, compared to other saturated fatty acids, the LDL lowering capacity of octadecanoic acid is higher and the anti-oxidative effect of 9, 12-octadecadienoic acid (linoleic acid) is praiseworthy (Hunter *et al.*, 2010; Peyrat-Maillard *et al.*, 2003).

# 4.4.3 GC-MS/MS analysis of the hexane fraction of *L. edodes*

The most abundant bio-component in the hexane fraction of *L. edodes* was  $\alpha$ -tocopherol (Table 4.3). Indeed,  $\alpha$ -tocopherol is a chain breaking anti-oxidant whose participation might be related towards the LDL oxidation inhibitory effect of this mushroom. Its free radical scavenging mode involves hydrogen atom transfer (HAT). Reacting with a lipid peroxyl molecule, it becomes converted to a radical form of  $\alpha$ -tocopherol with concomitant generation of a hydroperoxide and thus halts the propagation of peroxidative reaction (Figure 4.1). The intermediates produced are either unreactive or less harmful than the lipid peroxyl molecules. In the next phase, the  $\alpha$ -tocopherol radical form can either revert back to its previous form or combine with a new peroxyl molecule to be converted into an inactive form (Figure 4.1).



Figure 4.1: Lipid peroxidation inhibitory effect of α-tocopherol

In the physiological system,  $\alpha$ -tocopherol is apt in reconstituting the free radicals into inactive molecules based on its capacity to recycle from the chromanoxyl form, interaction with lipid radicals and its distribution pattern in the membrane (Serbinova *et al.*, 1991).  $\alpha$ -tocopherol had been found lessening OS in both human plasma and *in vitro* LDL oxidation generated through Cu<sup>2+</sup> and/or 2,2'-azobis-(2-amidinopropane) hydrochloride (AAPH) supplementation (Kontush *et al.*, 1996). However, its *in vivo* anti-oxidative potency depends on the presence of co-antioxidants such as ascorbate that aid in its regeneration (Kontush *et al.*, 1996). Intake of  $\alpha$ -tocopherol (400 IU/day) in human had caused decreased LDL oxidation, increased lag time of LDL oxidation and increased serum oxygen radical scavenging capacity (O'Byrne *et al.*, 2002).  $\alpha$ -tocopherol – mediated increament of lag time from 59.6 to 98.9 minutes had been reported at its physiological level of 10–30  $\mu$ M (Baldi *et al.*, 2003). Besides, its anti-oxidative turned anti-atherosclerotic effect had been found in several animal studies (Peluzio *et al.*, 2001; Schwenke *et al.*, 2002). Reis *et al.* (2012) reported the content of highest amount of  $\alpha$ -tocopherol in *L. edodes* (0.92  $\mu$ g/100 g) whole powder than those

of *P. ostreatus* (0.59 μg/100 g), *P. eryngii* (0.25 μg/100 g) and *F. velutipes* (0.19 μg/100 g) (Reis *et al.*, 2012).

Like that of the hexane fraction of *G. lucidum*, ergosterol was found in the hexane fraction of *L. edodes*. Its anti-oxidative effect had been reported by several studies and even better than those of thiourea and  $\alpha$ -tocopherol (Kim *et al.*, 1999; Wiseman *et al.*, 1993). Its inhibitory activity towards phosphorylation of MAPK, C/EBPb transcriptional activity and NF-kB and thus suppression of inflammatory responses had also been documented (Kim *et al.*, 1999).

Anti-oxidative performance of mushrooms varies from species to species and nature of the extracting solvent and procedure applied highly affects the respective mushroom's potency. When extraction process involves polar solvents such as methanol, phenolic and polyphenolics provide scavenging effect towards DPPH free-radicals and reducing capacities in Folin - Ciocalteu assay while triterpenoids participate in metal ion chelation (Kobori *et al.*, 2007; Landolfo *et al.*, 2010). In case of non-polar solvents such as hexane, ergosterol dominates over other anti-oxidative mushroom bio-components (Kalogeropoulos *et al.*, 2013). Thus, in congruence with the previous findings, ergosterol; 5, 6 - dihydroxy ergosterol; 7, 22 – ergostadienol; ergosta 7, 22- dien-3-ol present in the hexane fraction might be implicated in chelating Fe<sup>2+</sup>. This may be the putative mechanism of mushroom-mediated reduced LDL oxidation as observed in the present study. Anti-oxidative performance of ergosterol is so much prominent that in the edible mushroom *Grifola frondosa*, it stands as the second-most anti-oxidat among a number of bio-components (Nieto & Chegwin, 2008).

Among fatty acids present in the hexane fraction of *L. edodes*, were oleic, linoleic and butyric acid. Anti-oxidative and anti-atherosclerotic excellence of oleic acid had been observed both *in vitro* and in cellular models (Cho *et al.*, 2010). Its free radical

scavenging and inhibitory potency towards LDL oxidation far exceeds that of the palmitic acide moiety (Cho et al., 2010). Its importance also lies in maintenance of cellular anti-oxidative state through activation of paraoxonase activity (Cho et al., 2010). In addition to reduced LDL oxidation, oleic acid has blood LDL lowering effect (Allman-Farinelli et al., 2005). Its anti-atherosclerotic mode of action involves reduced expression of VCAM-1, ECAM-1, E-selectin and interrupted activaty of NFkB resulting in lowered adherence of monocytes to endothelial wall and dampened proinflammatory and pro-atherogenic activity (Carluccio et al., 1999; Massaro et al., 1999). Similar mode of action had been reported for butyric acid, another short chain fatty acid found in the hexane fraction of L. edodes (Massaro et al., 1999; Zapolska-Downar et al., 2004). Linoleic acid (LA, cis-9, cis-12 octadecadienoic acid) stands among the most abundant unsaturated fatty acids present in culinary-medicinal mushrooms (Kalogeropoulos et al., 2013). Linoleic acid in its conjugated form had been found having anti-oxidative as well as blood LDL and triacylaglyceride lowering effects (Carluccio et al., 1999; Palacios et al., 2003). CLA alone had been found increasing plasma level of catalase by 3.7 fold and decreasing lipid peroxidation by 58% in rat hepatic tissues (da Silva Marineli et al., 2012). Its protective role against oxidation of mitochondrial and peroxisomal PUFAs had been found far exceeding that of vitamin A (Palacios et al., 2003). Compared to P. ostreatus, P. eryngii and F. velutipes, increased level of PUFA (82%) and lowered level of saturated fatty acids (SFAs) (15.1%) in the whole powder of L. edodes had been detected by Reis et al. (2012). In terms of most abundancy, linoleic acid was top among the PUFAs while oleic acid among the monounsaturated fatty acids (MUFAs) and palmitic acid among the SFAs (Reis et al., 2012).

As discussed in case of the hexane fraction of *G. lucidum* (section 4.3.1), content of ergosterol in the same fraction of *L. edodes* might contribute to the HMGR inhibitory effects (Gil-Ramírez *et al.*, 2013; Gil-Ramírez *et al.*, 2011; Gil-Ramírez *et al.*, 2013).

Among 26 edible-medicinal mushrooms, *L. edodes* had been reported to be the best in inhibiting the activity of HMGR (Gil-Ramírez *et al.*, 2013).

### 4.4.4 LC-MS/MS analysis of the MD fraction of F. velutipes

Protocatechuic acid, p-coumaric acid and ellagic acid were the most abundant polyphenolics present in the MD fraction of *F. velutipes* (Table 4.5).

### A. Protocatechuic acid

i.

Protocatechuic acid (PCA), an important member of the polyphenolic anti-oxidants possesses numerous health benefits (Kakkar & Bais, 2014). Its presence in both wild and edible-medicinal mushrooms have been documented (Barros *et al.*, 2009; Mattila *et al.*, 2002). Its *in vitro* free radical scavenging (DPPH, ABTS, superoxide and hydroxyl ion), metal ion chelating ( $Fe^{2+}$ ) capacity and content of reducing power had been found better than those of the synthetic anti-oxidant and vitamin E analogue, trolox. Its *in vitro* anti-oxidative excellence remains unaltered in both aqueous and lipid media (Li *et al.*, 2011). Its anti-oxidative mode of action entails either free-radical scavenging and/or metal ion chelation.

- a. PCA- mediated scavenging of DPPH free-radical involves the following steps (Figure 4.2):
  - Acceptance of a hydrogen atom (H•) by the PCA from DPPH free radical (DPPH•) and formation of the unstable anion PCA• (A) and stable DPPH-H.
- ii. Formation of a stable quinone by withdrawing a hydrogen atom (H•) from PCA• (A).



Figure 4.2: Protocatechuic acid (PCA) – mediated scavenging mechanism of DPPH free radical

b. Metal ion chelation by PCA: Metal ion  $(Fe^{2+})$  is chelated by the orthodihydroxyl group of PCA (Figure 4.3).



Figure 4.3: Protocatechuic acid – mediated metal ion chelating mechanism
# B. p-Coumaric (p-Hydroxy cinnamic acid moeiey)

Prior to our findings, identification and quantification of p-coumaric acid had been performed in wild mushrooms collected from northeast Portugal (Barros *et al.*, 2009; Vaz *et al.*, 2011). Inhibitory effect towards both *in vitro* and *in vivo* LDL oxidation by p-coumaric acid had been found (Zang *et al.*, 2000). The probable mechanism is <sup>•</sup>OH scavenging (Zang *et al.*, 2000). It has been found superior to curcumin in decreasing oxidative stress in the human colon cell culture (HT-29) (Ferguson *et al.*, 2005).

## C. Ellagic acid

Ellagic acid, a flavonoid possesses anti-oxidant capacity better than that of melatonin and ascorbic acid (Atta-Ur-Rahman *et al.*, 2001; Festa *et al.*, 2000). Its anti-oxidative protection against colon cell lines (HCT 16) had also been found (Seeram *et al.*, 2005).

Lipid hydroperoxides are generated from PUFA peroxidation during LDL oxidation. When hydroperoxides decompose, the process becomes a chain reaction. Phenolic antioxidants, through chanin breaking mechanism, reduce the alkoxyl and/or peroxyl radicals to alkoxides and/or hydroperoxides, respectively (Rice-Evans *et al.*, 1997). Potency of the phenolic antioxidants depends on the position of the phenolic group in the structure, ability to stabilize the unstable phenoxyl radical and redox potential which is inversely related to the anti-oxidative capacity (Rice-Evans *et al.*, 1997; Teixeira *et al.*, 2013). Other factors such as presence of pro- and anti-oxidative enzymes and other anti-oxidants also affect the anti-oxidative performance of the phenolic anti-oxidants (Rice-Evans *et al.*, 1997; Teixeira *et al.*, 2013).

#### 4.4.5 LC-MS/MS analysis of the HWE of G. lucidum

Content of phenolics and their contribution towards anti-oxidative potentiality of *G*. *lucidum* had been reported previously (Abdullah *et al.*, 2011; Hasnat *et al.*, 2013; Mau

*et al.*, 2002a; Stojković *et al.*, 2014). Phenolics stand among the most potent antioxidative mushroom bio-components. Their content vary from species to species, part to part of the same mushroom species and their activities depend on the process of extraction and mode of application as well as to the oxidant they encounter (Heleno *et al.*, 2012; Mau *et al.*, 2002b; Stojković *et al.*, 2014). Protocatechuic acid, cinnamic acid, p-hydroxy cinnamic acid and p-hydroxy benzoic acid are among the most commonly available phenolics in the *Ganoderma* mushrooms (Heleno *et al.*, 2012; Mau *et al.*, 2005). Ganoderic acids of more than 140 categories had been reported to be present *G. lucidum* (Ye *et al.*, 2010). Anti-oxidative performance of the ganoderic acids stand among their immense bio-functionalities (Smina *et al.*, 2011).

Based on the source of mushroom production, extraction procedures and anti-oxidant content, different anti-oxidative performance have been observed among different varieties of *Ganoderma* and *G. lucidum* itself. For example, *G. lucidum* collected from China, Korea, Portugal and Serbia all had shown anti-oxidative performance but their extent differed from each other with respect to reducing capacity and free radical scavenging (Heleno *et al.*, 2012; Kim *et al.*, 2008; Stojković *et al.*, 2014). Phenolic extracts of the fruiting bodies of *G. lucidum* had higher anti-oxidant performance and higher content of total phenolics while the polysaccharide extracts of the spores had higher anti-oxidative capacity (Heleno *et al.*, 2012). Similarly, content of ganoderic acids were found to be variable among 36 different samples of *Ganoderma* isolated from different parts of China and content of *G. lucidum* was 10 times higher than those of *G. tsugae* (Wang *et al.*, 2006).

In the present study, the HWE containing the polyphenolic and tri-terpenoids outperformed in annihilating the DPPH free-radicals and possessing the total phenolics while the non-polar fractions surpassed the polar fractions in demolition of lipid

peroxidation and LDL oxidation activities. The modus operandi might be the higher content and availability of the phenolics in the HWE of the respective mushroom species. Total phenolic assay measures the reducing capacity of the Folin reagent in the alkaline media. Anti-oxidants containing higher phenolics are supposed to reduce the oxidant at higher extent based on the higher reducing power (hydrogen atom) provided by the hydroxyl group(s) attached with the benzene structure. The positive control quercetin contains five hydroxyl groups (at 3, 3', 4', 5, 7 positions) and one carbonyl group (at 4'position) and showed the highest reducing capacity. Compared to quercetin, single or multiple hydroxyl groups present in the phenolic or polyphenolic acids (protocatechuic acid, p-hydroxy coumaric acid, ellagic acid, gallic acid, caffeic acid, cinnamic acid, p-hydorxy cinnamic acid) contained in the HWE of the respective mushroom species conferred its best reducing capacity and DPPH free radical annihilating efficiency. Since DPPH and Folin-Ciocalteu reactions were performed in aqueous microenvironment, perhaps the lipo-philic anti-oxidants (ergosterol, saturated and unsaturated fatty acids) could not access at all or very little to exert their free radical annihilating and Folin reagent reducing actions. Besides, complexity of the total phenolics, nature of individual phenolic component, steric hindrance and accessibility, sample preparation, ionic potentiality of the reagents used and color interference highly affect the relationship between DPPH free radical scavenging and Folin-Ciocalteu assays (Apak et al., 2013; Huang et al., 2005). This assumption is supported by the previous finding that the extent of anti-oxidative performance of the mushrooms depends not only on its content of phenolics but also on the nature of the phenolics (free or bound to polysaccharide) (Heleno et al., 2012).

In case of lipid peroxidation and LDL oxidation inhibition, the higher performance of the HEX of the mushroom species might be attributed to their content of lipophilic (ergosterol and fatty acids) and/or amphiphilic anti-oxidants ( $\alpha$ -tocopherol).

Supposedly, suitable alignment of  $\alpha$ -tocopherol in the lipid peroxidation solvent facilitated its superb anti-lipid peroxidation effect than those of the phenolics of the HWE (Maqsood & Benjakul, 2010). Another apprehension is that the antioxidant activities of mushrooms determined by chemical tests surpass those of the biochemical assays (Barros *et al.*, 2008). As inhibition of lipid peroxidation of the buffered egg yolk is a bio-chemical test, the discrepancy in the anti-oxidative performance between the hexane and HWE is natural and might be mediated by the different bio-components present in different amount and differing functionality in the respective mushroom fractions. Similar proposition might be applied in case of the LDL inhibitory effect of the hexane extracts of different mushrooms.

# 4.4.6 Determination of β-D glucan level in the HWE of G. lucidum

Anti-oxidative potentiality of the HWE of *G. lucidum* described in the previous chapter might partly be attributed to the content of this higher amount (44.15%) of  $\beta$ -D glucan in the extract (Raw data: Appendix A, Figure 5.8, calculation using K-YBGL\_CALC, Megazyme, USA). This claim is supported by some other observations that have linked this polysaccharide's free radical scavenging potential even at lower level than that of us (Khan *et al.*, 2014; Kofuji *et al.*, 2012; Kogan *et al.*, 2005; Kozarski *et al.*, 2011; Maity *et al.*, 2014; Patra *et al.*, 2013; Şener *et al.*, 2006). Hydroxyl radical scavenging capacity of  $\beta$ -D-glucan had been found intermediary between the hydrophilic anti-oxidant mannitol and the lipo-philic anti-oxidant  $\alpha$ -tocopherol (Babincova *et al.*, 2002). Its mode of action involves hydrogen atom transfer mechanism and termination of the chain reaction of free radical generation. Its level varies from mushroom species to species and also based on the extraction process (Ahmad *et al.*, 2014). Its protective role against OS induced neurotoxicity has also been linked with its anti-oxidative properties (Kaya *et al.*, 2016). In conjunction with its *in vivo* anti-oxidative effect through improving CAT, GPX and SOD, lipid lowering

efficiency of  $\beta$ -D glucan from *P. sajor-caju* has been found to be achieved through stimulating the expression of hormone sensitive lipase and adipose triglyceride lipase along with lowering the expression of lipoprotein lipase (LPL), peroxisome proliferatoractivated receptor gamma (PPAR- $\gamma$ ) and sterol regulatory binding protein-1c (SREBP-1c) (Kanagasabapathy *et al.*, 2013). As the present study is aimed at evaluating the antioxidative, cholesterol lowering and AD ameliorating effect of selected culinarymedicinal mushrooms, the observed anti-oxidative effect of the *G. lucidum* HWE along with its content of higher amount of  $\beta$ -D glucan has been of greater importance in choosing this mushroom and extract for furthering *in vivo* studies.

# 4.5 Conclusion

Both culinary (*H. erinaceus, L. edodes* and *F. velutipes*) and medicinal (*G. lucidum*) mushrooms possess anti-oxidative potentialities. Anti-oxidative performance of different solvent-partitioned fractions varies from each other. The reason might be differing content of bio-active components in different fractions. Among four mushroom species (*G. lucidum, H. erinaceus, L. edodes* and *F. velutipes*) of the present study, the *G. lucidum* showed the top-notch *in vitro* anti-oxidative performance. Under the current experimental set up, the HWE of *G. lucidum* outperformed the solvent-partitioned fractions. Phenolics and tri-terpenoids gano-components identified in the HWE of *G. lucidum* might contribute to this superb performance that beacons towards this mushroom extract's therapeutic potentiality. Based on these, *in vivo* studies of *G. lucidum* HWE had been outlined in the following chapters.

# CHAPTER 5: *IN VIVO* ANTI - OXIDATIVE AND HYPOCHOLESTEROLEMIC EFFECT OF THE HOT WATER EXTRACT OF *G. LUCIDUM*

# 5.1 Introduction

As the HWE of *G. lucidum* showed the best *in vitro* anti-oxidative potentiality (chapter 3) and contained higher amount of bio-active components (chapter 4), it was selected in the present study for assessing its *in vivo* anti-oxidative and cholesterol lowering performance *en route* to deciphering its AD ameliorating effect. The experimental paradigm was based on the fact that the brain is much vulnerabale to OS due to its higher rate of oxygen utilization (one-fourth of total respired oxygen) and lipid content in contrast to possession of lower anti-oxidative defense mechanism (Ansari & Scheff, 2010; Mazzetti *et al.*, 2015; Reed, 2011; Sottero *et al.*, 2009).

Oxidative stress directly affects the the neuronal cells and A $\beta$  itself exacerbates the condition (McLellan *et al.*, 2003). Even partial defect in the SOD (Mn) could not provide anti-oxidative support to the Tg19959 AD model mice (Dumont *et al.*, 2009). On the other hand, only 2 - 3 times overexpression of this enzyme in the same animals could lower OS and also decrease A $\beta$  deposition and improve memory (Dumont *et al.*, 2009). Similarly, enhanced AD symptoms such as excellerated rate of OS - driven A $\beta$ oligomerization and memory loss have been observed in case of the Tg2576 AD model mice deficient of SOD1(Cu/Zn) (Murakami *et al.*, 2011). High amount of redox - active metal ions (Cu<sup>2+</sup> and Fe<sup>2+</sup>), phospholipids and poly unsaturated fatty acids (PUFAs) also make the brain cells much susceptible to OS (Garbarino *et al.*, 2015; Patel, 2016). As a result, compromised *in vivo* anti-oxidative defense and abrupt lipid peroxidation greatly affect AD pathogenesis (Chen *et al.*, 2008; Resende *et al.*, 2008; Su *et al.*, 2010). Even, A $\beta$  itself acts as a pro-oxidative agent (Combs *et al.*, 2001; Ding *et al.*, 2007). Proteomics observation reveals that methionine35 of the A $\beta$  sequence raises its OS causing prowess while its replacement with cysteine reduces the oxidative risk (Butterfield & Boyd-Kimball, 2005; Butterfield *et al.*, 2013). Besides, OS generated from mitochondrial dysfunction also adds injury to AD pathogenesis (Castellani *et al.*, 2002; Melov *et al.*, 2007; Reddy, 2011; Yao, Jia *et al.*, 2009). The situation is further exacerbated through APP and A $\beta$  as they act as mitochondrial deregulators (Anandatheerthavarada *et al.*, 2003; Caspersen *et al.*, 2005; Manczak *et al.*, 2006).

Oxidative stress associated with hypercholesterolemia heightens the risks of AD (Oda *et al.*, 2010). Even, hypercholesterolemia itself is a risk factor of AD as it provides with the lipid rafts for A $\beta$  generation (Björkhem, 2006; Heverin *et al.*, 2005; Sharma *et al.*, 2008). Hypercholesterolemic animals have been found vulnerable towards AD development including enhanced A $\beta$  deposition in the hippocampus (Sparks *et al.*, 1994; Refolo *et al.*, 2000). On the other hand, cholesterol lowering approaches seem promising in reducing AD symptoms (Refolo *et al.*, 2001; Matthew *et al.*, 2011). Conceptually, AD therapeutic approaches entailing improved anti-oxidative defense along with cholesterol lowering potential seem imperative. Thus, the *in vivo* anti-oxidative and hypcholesterolemic effects of the selected HWE of *G. lucidum* have been determined whose description has been put forward in the following sections.

# 5.2 Materials and Methods

### 5.2.1 Animals

Ninety wistar male rats (weight range  $120 \pm 5$  gm) were divided into six groups: control (C), *G. lucidum* HWE fed control (CE), hypercholesterolemic (H), *G. lucidum* HWE fed hypercholesterolemic (HE), Alzheimer's diseased (A in case of anti-oxidative, hypochoesterolemic, memory and learning related behavioral and biochemical and histopathological tests. AD in case of proteomics study) and *G. lucidum* HWE fed

Alzheimer's diseased (AE) each group containing 15 rats. In vivo anti-oxidative and hypocholesterolemic studies of each of the six groups were performed. Hypercholesterolemia to the H rats was evoked by adding 1% cholesterol and 1% cholic acid (for intestinal better absorption of cholesterol) with the basal diet. Alzheimer's diseased model rats were prepared through infusion of A $\beta$  (1-42) in the right ventricle. The extract fed groups (CE, HE and AE) received 200 mg/kbw G. lucidum HWE. No specific dosage regimen of mushrooms had been found in the existing literature for *in* vivo experimentation. For the mushroom powder, the mostly utilized dosages range from 3% to 20% (w/w) (Hossain et al., 2003; Mori et al., 2008; Talpur et al., 2002). In case of extracts, 100-400 mg/kbw of the animals had been found (Anandhi et al., 2013; Jeong et al., 2010; Omar et al., 2015; Yamac et al., 2010). As the species of mushrooms and the strain, type, age, sex and habitat of the experimental animals greatly influence the experimental output, the intermediary dosage regimen of the published literature (200 mg/kbw) was used for evaluating the hypocholesterolemic effect of G. lucidum HWE. Animals had been housed in a 12 hr day night cycle at 25±2 °C temperature. Twenty four hours following the last treatment and test, the rats were kept in fasting overnight. Then, the rats were anesthesized with intra-peritoneal injection of sodium pentobarbital (35 mg/kbw) and sacrificed. Blood was collected from the inferior vena cava immediately, centrifuged at 1000 rpm, collected plasma and serum and preserved at -80 °C). All the experimental protocols had been approved by the ethical permission committee, University of Malaya Institutional Animal Care and Use Committee (UMIACUC) [Ethics reference no. ISB/25/04/2013/NA (R)] (Appendix G).

# 5.2.2 Measurement of in vivo anti-oxidative effect of G. lucidum HWE

In the present sudy, levels of catalase (CAT), glutathione peroxidase (GPX) and superoxide dismutase (SOD) in the plasma and liver homogenates of the controlled and

experimental rats were determined to evaluate th *in vivo* anti-oxidative potentiality of the *G. lucidum* HWE.

#### 5.2.2.1 Catalase activity assay

a. Principle

Catalase (CAT) activity was assayed using the assay kit purchased from Cayman Chemical, USA (item no. 707002) and following the instruction provided. CAT possesses both catabolic (conversion of  $H_2O_2$  into water and molecular oxygen) and peroxidative activities (conversion of  $H_2O_2$  into water and formaldehyde in presence of low molecular weight alcohol such as methanol).

b. Reaction

$$2 H_2O_2 \longrightarrow 2 H_2O + O_2 \quad \text{(catalytic activity)}$$
$$2 H_2O_2 + CH_3OH \longrightarrow 2 H_2O + HCOOH \text{(peroxidative activity)}$$

The assay protocol utilizes the peroxidative action of CAT in presence of methanol and  $H_2O_2$  that produces formaldehyde. Chromogen purpald (4-amino-3-hydrazino-5mercapto-1, 2, 4-triazole) is allowed to react specifically with formaldehyde that produces a hetero-bicyclic compound whose subsequent oxidation is reflected through changing of the reaction mixture from colorless to purple and measurement of the absorbance at 540 nm.

c. Procedure

First of all, 100  $\mu$ l assay buffer (100 mM potassium phosphate, pH 7.0) was added to the wells of 96-well ELISA plate. In the standard wells, methanol (30  $\mu$ l), each standard diluted with sample buffer (20  $\mu$ l) and formaldehyde (final concentration 0, 5, 15, 30, 45, 60, and 75  $\mu$ M) were added. In the control wells, 20  $\mu$ l of CAT control were added. Then, samples (20  $\mu$ l) (five times diluted serum sample and ten times diluted tissue homogenate diluted with sample buffer containing 25 mM potassium phosphate, pH 7.5, having 1 mM EDTA and 0.1% BSA) were added to the sample wells. Addition of 20  $\mu$ l of H<sub>2</sub>O<sub>2</sub> to the wells started the peroxidative reaction of CAT and the initiation time was recorded. The plate, been covered, had undergone occasional shaking in a shaker for 20 minutes at room temperature. Addition of 30  $\mu$ l of potassium hydroxide terminated the reaction. Then, 30  $\mu$ l of purpald was added and incubated for 10 minutes at room temperature. To each well, 10  $\mu$ l of potassium periodate was added and incubated for 5 minutes while covered and shaking. Finally, absorbance was read at 540 nm using a plate reader.

#### 5.2.2.2 Glutathione peroxidase (GPx) activity assay

a. Principle

Glutathione peroxidase (GPx) assay was performed using the kit purchased from Cayman Chemical, USA (item no. 703102) and following the company's instructions. Indirect measurement of GPX activity was followed through a coupled reaction with glutathione reductase (GR) based on the principle that GPx is involved in catalyzing the reduction of  $H_2O_2$  and other hydroperoxides. In the first stage of the experiment, GPx had been involved in reducing cumene hydroperoxide (ROOH) by reduced glutathione (GSH) and production of oxidized glutathione (GSSG). In the second phase, GSSG is converted to its reduced form (GSH) by NADPH with the catalysis of GR. As a result, NADPH itself is oxidized to NADP<sup>+</sup> along with decrease in absorbance at 340 nm. At rate limiting state, this absorbance change is directly proportional to the GPX activity of the sample and is calculated accordingly. b. Reaction:

Phase I : 2 GSH + ROOH 
$$GPx$$
 GSSG + ROH + H<sub>2</sub>O  
Phase II : GSSG + NADPH<sup>+</sup> + H<sup>+</sup>  $GR$  2 GSH + NADP<sup>+</sup>

c. Procedure

For blank (non-enzymatic/background study, we used 3 wells of the 96-well ELISA plate and added 120  $\mu$ L of assay buffer and 50  $\mu$ L of co-substrate mixture (as provided by the manufacturer). Bovine erythrocyte GPx was used as the positive control. For sample and positive control, 20  $\mu$ L of each was added with 100  $\mu$ L of assay buffer and 50  $\mu$ L of co-substrate mixture in triplicate. Reaction was initiated by adding 20  $\mu$ L of cumene hydroperoxide in each of the wells and initiation time noted. The plate was shaken for 10 seconds inside the plate reader and the absorbance read at 340 nm at every one minute for total 5 time points. The decrease in absorbance was recorded and the obtained data used for calculating GPx activity.

# 5.2.2.3 Superoxide dismutase (SOD) activity assay

a. Principle

Using the superoxide dismutase (SOD) activity assay kit purchased from Cayman Chemical, USA (item no. 706002) and following the company's instructions, the activity of SOD was assessed. SOD usually dismutates superoxide anions into hydrogen peroxide and molecular oxygen.

$$2 O_2^- + 2 H^+$$
 SOD  $H_2O_2 + O_2$ 

In the first phase of the present experiment, superoxide radicals are generated from xanthine using xanthine oxidase (XO). In the second phase, superoxide radicals are

reduced by 2-(4-iodophenyl)-3-(4-nitrophenol)-5-phenyltetrazolium chloride (INT) and produces a formazen dye (red colored). Rate of reduction is directly proportional to XO activity and inhibited by SOD present in the sample. SOD activity is measured from the extent inhibition of reduction (1 unit of SOD causes 50% inhibition of reduction by INT). Thus, increased SOD level in the sample causes decreased superoxide production and lowered formazen generation.

b. Reaction

Phase I:

Xanthine + Xanthine oxidase (XO)  $O_2^-$  + Uric acid

Phase II:

 $O_2^- + INT \longrightarrow$  Formazen dye

Or,

 $2 O_2 + 2 H^+$  SOD  $H_2O_2 + O_2$ 

c. Procedure

In each well of the 96-well ELISA plate, 200  $\mu$ L of INT was used followed by 10  $\mu$ L of either diluted sample and/or standard. Reaction was initiated by adding 20  $\mu$ L of xanthine oxidase in each of the wells and initiation time noted. The plate was covered, shaken for 10 seconds and incubated for 30 minutes at room temperature. Then, absorbance was read at 450 nm, values plotted and calculated to obtain the SOD activity.

#### 5.2.3 Hypocholesterolemic study

#### 5.2.3.1 Measurement of body weight

Using a digital scale (Kent scientific, SCL 66110, USA) body weight of each of the rats was measured at every 24 hours and recorded up to the end of the experiment. Weekly average body weight (gm) of rats belonging to different groups was compared with each other to find the effect of *G. lucidum* HWE feeding up on hypercholesterolemia and hyperlipidemia.

#### 5.2.3.2 Measurement of triacylglycerol

a. Principle

Following the method of Werner *et al.*, (1981), the level of triacylglycerol (TG) was measured through a couple of enzymatic reactions. Firstly, lipase action generated fatty acid and glycerol from TG. Secondly, glycerokinase with ATP converts glycerol to glycerol-3-phosphate. Thirdly, glycerol-3-phosphate is oxidized by glycerophosphate oxidase to dihydroxyacetone phosphate and hydrogen peroxide. In the fourth step, peroxidase mediated reaction of hydrogen peroxide with 4-aminoantipyrine and N-Ethyl-N-sulfopropyl-n-anisidine (ESPAS) forms a coloured complex whose intensity is proportional to the TG concentration and is measured spectrophotometrically at 546 nm.

b. Reaction:

Triglycerides + 3H<sub>2</sub>O ------ Lipase -----> glycerol + fatty acids

Glycerol + ATP ------ glycerokinase -----> glycerol-3-phosphate + ADP

Glycerol-3-phosphate + O<sub>2</sub> ------ glycerophosphate oxidase ----->

dihydroxyacetone phosphate + H<sub>2</sub>O<sub>2</sub>

peroxidase

# c. Reagents

Reagent 1: Pipes Buffer (pH 7.5) 50 mmol/L, ESPAS 1 mmol/L, Magnesium Salt 15 mmol/L, ATP 0.3 mmol/L.

Reagent 2: Enzymes such as Lipoprotein lipase (1100 U/L), Glycerokinase (800 U/L), Glycerol – 3 – Phosphate Oxidase (5000 U/L), Peroxidase (350 U/L).

Reagent 3: Standard Glycerol (Triglycerides equivalent) 200 mg/dl.

d. Method

10  $\mu$ L of plasma and/or standards were mixed with 1 mL of TG working reagent. We mixed the contents well and incubated at 37 °C for 10 minutes. Absorbance of the standard and the samples were measured against a reagent blank at 546 nm.

e. Calculation

# 5.2.3.3 Measurement of total cholesterol

a. Principle

The enzymatic method of Allain *et al.*, (1974) was followed to determine the level of TC in the plasma and liver homogenates of the rats . Using cholesterol esterase, cholesterol esters were hydrolyzed into free cholesterol and fatty acids. Free cholesterol underwent oxidation by cholesterol oxidase and produced cholest-4en-3-one and hydrogen peroxide. A colored quinone complex was produced through

reaction of hydrogen peroxide with 4-amino antipyrene in presence of phenol. Intensity of the colored quinone complex is directly proportional to the concentration of cholesterol present and is measured at 500 nm spectro-photometrically.

b. Reaction

Cholesterol esters	cholesterol esterase	cholesterol +	- Fatty acids
Cholesterol	cholesterol oxidase	Cholest-4-en-	3-one + H <sub>2</sub> O <sub>2</sub>
$H_2O_2 + Phenol + 4-Amin$	oantipyrine peroxic	lase >	Red Quinone + H <sub>2</sub> O
c. Reagents			

- A. 1. Buffer (50 mmol/L pH 6.7)
  - 2. Cholesterol oxidase <sup>3</sup> 50 U/L
  - 3. Cholesterol esterase <sup>3</sup> 100 U/L
  - 4. Peroxidase <sup>3</sup> 3 IU/L
  - 5. 4-amino antipyrine (0.4 mmol/L)

B. Cholesterol standard (200 mg/dL).

d. Method

10  $\mu$ L of plasma and/or standards were mixed with 1 mL of cholesterol reagent. We mixed the contents well and incubated at 37°C for 10 minutes. Absorbance of the standard and the samples were measured against a reagent blank at 500 nm.

e. Calculation

Cholesterol concentration (mg/dL) =  $\frac{(OD \text{ of test sample} - OD \text{ of blank})}{(OD \text{ of standard} - OD \text{ of blank})} \times 200$ 

# 5.2.3.4 Measurement of HDL-C

a. Principle

The method of Izzo *et al.*, (1981) was followed to determine the level of HDL-C in plasma and liver homogenates of the rats. Precipitation of the apo-B containing lipo-proteins (chylomicrons, VLDL and LDL) using polyethylene glycol and centrifugation let HDL-C free for measurement whose concentration is measured spectrophotometrically.

- b. Reagents
- A. Precipitating reagents: Glycine buffer pH 10.2, polyethylenge glycol (10 g/dL)
- B. HDL-C standard: 50 mg/dL.
  - c. Procedure

Chylomicrons were precipitated, VLDL and LDL adding precipitating reagent (0.3 ml precipitating solution added to 0.3 mL of sample), mixing thoroughly and centrifuging at 4000 rpm for 10 minutes. We collected the supernatant and to 100  $\mu$ L, 500  $\mu$ L of HDL-C working solution added. Mixing them well we incubated for 5 minutes and then absorbance read at 500 nm.

d. Calculation

 $HDL-C \text{ concentration } (mg/dL) = \frac{(OD \text{ of test sample} - OD \text{ of blank})}{(OD \text{ of standard} - OD \text{ of blank})} \times 200$ 

#### 5.2.3.5 Measurement of VLDL and LDL

The levels of VLDL and LDL were measured using the Friedewald equation (Friedewald *et al.*, 1972). Calculations were performed as follows

VLDL-C (mg/dL) = 
$$TG/5$$

LDL-C (mg/dL) = TC - [(TG/5) + HDL-C]

#### 5.2.3.6 Measurement of atherogenic index (AI) and LDL/HDL ratio

Atherogenic index of plasma (AIP) was measured following the method of Harnafi *et al.* (2008), and the formula used was

$$AIP = (TC - HDL - C) / HDL - C.$$

We calculated the ratio of plasma LDL-C to HDL-C for measuring the LDL/HDL ratio.

#### 5.2.4 Organ function tests (Cardiac, Liver and Kidney Function Tests)

5.2.4.1 Measurement of Aspartate aminotransferase/Glutamate oxalate transaminase (AST/GOT) activity

a. Principle

Based on the method of Reitman and Frankel (1957), the activity of AST was measured using AST assay kit (Sigma-Aldrich, MAK 055) . AST reversibly transfers an amino group from L-aspartate (an  $\alpha$  amino acid) to 2-oxoglutarate/  $\alpha$  ketoglutarate (an  $\alpha$  keto acid) and forms oxaloacetate (an  $\alpha$  keto acid) and glutamate (an  $\alpha$  amino acid).

L-Aspartate +  $\alpha$  ketoglutarate -----AST-----Oxaloacetate + Glutamate

Generation of gluatamate is proportional to the activity of AST and is expressed as U/L (1 unit of AST is defined the amount of enzyme that generates 1.0  $\mu$ mole of glutamate per minute at pH 8.0 at 37 °C). Rate of conversion was measured at 450 nm.

b. Procedure:

Glutamate standard curve was prepared using 10  $\mu$ L of glutamate standards (concentration 0, 2, 4, 6, 8 and 10 nmole/well) in a 96-well ELISA plate. In the other wells, 10  $\mu$ L of sample to the 100  $\mu$ L of the AST master mix in the ELISA plate well, mixed thoroughly and took initial absorbance (A450<sub>initial</sub>) within 2 minutes (T<sub>initial</sub>) at 37 °C. At every 5 minutes, we took the absorbance (A<sub>450</sub>) until value of the sample surpassed the highest value of the standard (10 nmol / well). Penultimate value before that of the highest (nearest or crossing the most concentrated standard) corresponds to the final absorbance (A450<sub>final</sub>) and the corresponding penultimate time is the final time (T<sub>final</sub>). It was observed whether both the A450<sub>initial</sub> and A450<sub>final</sub> remain within the range of the standard curve.

# c. Calculation

Background values were corrected by subtracting the value obtained for 0 concetration of glutamate from each of the  $A450_{final}$  of standard and sample value.

Change in measurement from T<sub>initial</sub> to T<sub>final</sub> was calculated as follows:

$$\Delta A_{450} = A450_{\text{final}} - A450_{\text{initial}}$$

Comparing the  $\Delta A_{450}$  of each sample with the standard curve, the amount of glutamate (nmole) produced was determined.

AST activity was calculated following the formula:

AST activity =  $(B \times Dilution factor) / (T \times V)$ 

Where,

B = Amount of glutamate (nmole) produced

 $T = T_{final}$  -  $T_{initial}$ 

V = Sample volume (mL) used.

# 5.2.4.2 Measurement of Alanine aminotransferase/Glutamate pyruvate transaminase (ALT/GPT) activity

a. Principle

Following the method of Reitman and Frankel (1957), the activity of ALT/GPT was measured using the kit purchased from Sigma-Aldrich (MAK 052) . Normally, ALT transfers an amino group from alanine to  $\alpha$  ketoglutarate and forms oxaloacetate and pyruvate.

L-Alanine + a ketoglutarate -----ATL-----Oxaloacetate + Pyruvate

Generation of pyruvate is proportional to the activity of AST and is expressed as U/L (1 unit of ALT is defined as the amount of enzyme that generates 1.0  $\mu$ mole of pyruvate per minute at 37 °C). Rate of conversion was measured at 570 nm.

b. Procedure

Pyruvate standard curve was prepared using 10  $\mu$ L of pyruvate standards (concentration 0, 0.2, 0.4, 0.6, 0.8 and 1 nmole/well) in a 96-well ELISA plate. In the other wells, 10  $\mu$ L of sample to the 100  $\mu$ L of the ALT master mix in the ELISA plate well, mixed thoroughly and took initial absorbance (A570<sub>initial</sub>) within 2 minutes (T<sub>initial</sub>) at 37 °C. At every 5 minutes, we took the absorbance (A<sub>570</sub>) until value of the

sample surpassed the highest value of the standard (10 nmol/well). Penultimate value before that of the highest (nearest or crossing the most concentrated standard) corresponds to the final absorbance (A570<sub>final</sub>) and the corresponding penultimate time is the final time ( $T_{final}$ ). It was observed whether both the A570<sub>initial</sub> and A570<sub>final</sub> remain within the range of the standard curve.

c. Calculation

Background values were corrected by subtracting the value obtained for 0 concetration of glutamate from each of the  $A570_{final}$  of standard and sample value.

Change in measurement from T<sub>initial</sub> to T<sub>final</sub> was calculated as follows:

$$\Delta A_{570} = A570_{\text{final}} - A570_{\text{initial}}$$

Comparing the  $\Delta A_{570}$  of each sample with the standard curve, the amount of pyruvate (nmole) produced was determined.

ALT activity was calculated following the formula:

ALT activity =  $(B \times Dilution factor) / (T \times V)$ 

Where,

B = Amount of pyruvate (nmole) produced

 $T = T_{final}$  -  $T_{initial}$ 

V =Sample volume (mL) used.

#### 5.2.4.3 Measurement of Alkaline phosphatase (ALP) activity

a. Principle

Following the principle that in alkaline medium, ALP hydrolyzes the organic monophosphate esters into phenol/alcohol and inorganic phosphate, the ALP activity was measured using the assay kit (Abcam, ab83369) (Tietz *et al.*, 1983). In the applied experimental condition, ALP dephosphoralizes p-nitrophenyl phosphate (pNPP) into p-nitrophenol and phosphate at pH 10. Intensity of the yellow colored p-nitrophenol is proportional to the ALP activity at 405 nm.

p-nitrophenyl phosphate (pNPP)------ALP----- p-nitrophenol + phosphate

b. Procedure

20  $\mu$ L of sample and/or standards were mixed with 20  $\mu$ L of 0.67M pNPP and 960  $\mu$ L of reaction buffer. Adding 20  $\mu$ L of ALP, we mixed the contents well and read the increase in absorbance of the standard and the samples against a reagent blank at 405 nm.

c. Calculation

 $ALP activity (U/L) = \frac{(OD of test sample - OD of blank)}{(OD of standard - OD of blank)} \times Dilution factor$ 

#### 5.2.4.4 Measurement of serum total bilirubin

a. Principle

Following the modified method of Mori (1978), the level of total bilirubin in the rat serum serum was measured. According to the principle, total bilirubin reacts with diazotized sulfanilic acid in presence of caffeine to produce a colored complex

whose intensity at 530 nm is directly proportional to the bilirubin present in the sample.

Bilirubin + Diazotized sulfanilic acid ----- Azobilirubin

b. Procedure

At first, water/blank (250  $\mu$ L) and reconstituted calibrator (200  $\mu$ L water + 50  $\mu$ L calibrator) was used in the ELISA plate wells. Then, in the separate wells, 50  $\mu$ L of each of the samples were added with 200  $\mu$ L of the working reagent and mixed well. Keeping the plate at 37°C for 10 minutes, absorbance was read at 530 nm.

c. Calculation

Bilirubin concentration  $(mg/dL) = \frac{(OD \text{ of test sample} - OD \text{ of blank})}{(OD \text{ of calibrator} - OD \text{ of blank})}$ 

Where, 5 mg/dL = equivalent bilirubin concentration of the calibrator.

# 5.2.4.5 Measurement of serum creatinine level

a. Principle

Based on the principle of Jaffe reaction, the level of rat serum creatinine level was measured (Bonsnes & Taussky, 1945; Schirmeister *et al.*, 1964). A coupled enzymatic reaction converts creatinine into creatine which in presence of alkaline picrate solution forms a yellow/orange colored complex and intensity of the complex at 570 nm is proportional to the creatinine present.

b. Procedure

In ELISA plate well, creatinine standards of 0, 0.2, 0.4, 0.6, 0.8 and 1.0 nmole/well were prepared (total volume 50  $\mu$ L/well, using the creatinine assay buffer). Aliquot of sample (2-50  $\mu$ L) were added in the respective wells and total volume made up to 50  $\mu$ L/well using the creatinine assay buffer. Then, 50  $\mu$ L of the reaction master mix (44/42  $\mu$ L creatinine assay buffer + 2  $\mu$ L creatininase + 2  $\mu$ L creatinase + 2  $\mu$ L creatinine enzyme mix + 2  $\mu$ L creatinine probe) were added to each well. After mixing thoroughly and keeping the plate at 37°C for 10 minutes, absorbance was read at 530 nm.

c. Calculation

Creatinine concentration  $(mg/dL) = \frac{(OD \text{ of test sample} - OD \text{ of blank})}{(OD \text{ of standard} - OD \text{ of blank})}$ 

Creatinine concentration (mg/dL) = Sa / Sv

Where, Sa = Amount of Creatinine in unknown sample (nmole) from standard curve

 $Sv = Sample volume (\mu L).$ 

#### 5.2.4.6 Measurement of blood urea level

a. Principle

Based on the principle of Jung *et al.* (1975), the level of rat blood urea was measured. A coupled enzymatic reaction converts urea into a colored complex whose intensity at 570 nm is proportional to the urea present.

b. Procedure

In ELISA plate well, urea standards of 1, 2, 3, 4 and 5 nmole/well were prepared (total volume 50  $\mu$ L/well, using the urea assay buffer). Aliquot of sample (2-50  $\mu$ L) were added in the respective wells and total volume made up to 50  $\mu$ L/well using the urea assay buffer. Then, 50  $\mu$ L of the reaction master mix (44/42  $\mu$ L urea assay buffer + 2  $\mu$ L peroxidase substrate + 2  $\mu$ L developers + 2  $\mu$ L enzyme mixes + 2  $\mu$ L convertase) were added to each well. After mixing thoroughly, we kept the plate at 37 °C for 60 minutes and read absorbance at 570 nm.

c. Calculation

Urea concentration (mg/dL) = Sa / Sv

Where, Sa = Amount of Creatinine in unknown sample (nmole) from standard curve

 $Sv = Sample volume (\mu L).$ 

# 5.2.4.7 Measurement of Protein concentration

a. Principle

Based on the principle of the biuret reaction that in alkaline media, proteins (actually the amino acids cysteine, cystine, tyrosine and tryptophan) reduce  $Cu^{2+}$  into  $Cu^{+}$ , total protein concentration in the samples was estimated using the Pierce BCA protein assay kit (Thermo scientific, 23225). Bicinchoninic acid present in the sample chelates  $Cu^{+1}$  and produces purple colored product whose absorbance at 562 nm is proportional to the protein concentration present in the sample. b. Materials

Pierce BCA protein assay kit, BCA working reagent, bovine serum albumin (BSA) as the standard.

c. Procedure

BSA standards were prepared maintaining the concentration range of 20-2000  $\mu$ g/mL and working reagent with mixing BCA reagents A and B in the ratio of 50:1. 2 mL of working reagent and 100  $\mu$ L each of the sample and/or standard were added, mixed well and incubated at 37 °C for 30 minutes. Then, absorbance was measured at 562 nm.

# 5.2.5 Inhibition of HMG Co-A reductase (HMGR) activity

a. Principle

The *in vitro* inhibitory effect of the HWE of *G. lucidum* was measured following the methods of Gholamhoseinian *et al* (2010). Commercially available HMGR assay kit was utilized in the experimentation (Sigma-Aldrich, CS 1090, St. Louis, USA). HMGR catalyzes the conversion of HMG Co-A into mevalonate and Co-ASH with concomitant oxidation of NADPH into NADP<sup>+</sup> and decrease in absorption at 340 nm. Spectrophotometric measurement of absorbance related to the inhibition of the *G. lucidum* HWE up on HMGR activity against the pravastatin.

HMG Co-A + 2 NADPH + H<sup>+</sup> ------ HMGR----- Mevalonate + CoA-SH + 2NADP<sup>+</sup>

#### b. Procedure

Following the manufacturer's recommendation, in the 96-well ELISA plate well, mixing of 181  $\mu$ L assay buffer, 1  $\mu$ L *G. lucidum* HWE/pravastatin, 4  $\mu$ L NADPH, 12  $\mu$ L HMG Co-A and finally 2  $\mu$ L HMGR were performed. The plates were mixed gently and subjected to absorbance measurement (Biotek H1 multimode plate reader associated with Gen 5 software) at every 20 seconds for a period of 20 minutes. Percentage inhibition of the HMGR activity was calculated following the equation below:

c. Calculation

Inhibition of HMGR activity (%) = 
$$(A0-As) \times 100/A0$$

Where, A0 is the absorbance of the control, and As is the absorbance of the reaction mixture containing the extract.

#### 5.3 Results

# 5.3.1 Effect on enzymatic anti-oxidative defense

As shown in Tables 5.1 and 5.2, feeding of *G. lucidum* HWE increased enzymatic anti-oxidative defense of the rats (Raw data: Appendix A, Tables A - 4.10 - 4.47; Figures A - 4.1, 4.2 and 4.3; Statistical data: Appendix C, Tables C - 5.1-5.4).

Plasma anti-	С	CE	Н	HE	Α	AE
oxidant						
enzyme level						
(nmol/mL)						
Catalase	9.83	14.66	7.34	11.78	7.33	10.57
(CAT)	$\pm 0.10^{a}$	$\pm 0.30^{b}$	$\pm 0.30^{c,e}$	$\pm 0.23^{d}$	±	$\pm 0.15^{f}$
					$0.20^{c,e}$	
Glutathione	29.53	42.23	26.03	32.37	25.40	31.20
peroxidase	$\pm 0.70^{a}$	$\pm 0.22^{a}$	$\pm 0.5^{c,e}$	$\pm 0.34^{d,f}$	±	±
(GPX)					$0.50^{c,e}$	0.30 <sup>d,f</sup>
Superoxide	59.96	67.42	48.20	50.27	44.80	49.42
dismutase	$\pm 0.80^{a}$	$\pm 0.42^{b}$	$\pm 0.6^{c,f}$	$\pm 0.40^{d,f}$	±	±
(SOD)					0.95 <sup>a</sup>	$0.75^{c,d,f}$

Table 5.1: Plasma anti-oxidant enzyme levels

Data are expressed as mean $\pm$ SE. Mean values with different lower case superscripts are indicators of statistically significant difference at P $\leq$ 0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3). Where, C stands for the control group; CE for the *G. lucidum* HWE fed control group; H for the hypercholesterolemic group; HE for *G. lucidum* HWE fed H group; A for the AD group and AE for the *G. lucidum* HWE fed AD group.

Table 5.2: Liver anti-oxidant enzyme levels	Table	5.2: ]	Liver	anti-oxidant	enzyme	levels
---	-------	--------	-------	--------------	--------	--------

Liver anti-	С	CE	Н	HE	Α	AE
oxidant						
enzyme level						
(nmol/mL)						
Catalase	59.64	67.88	46.55	52.4	46.81	53.89
(CAT)	$\pm 0.53^{a}$	$\pm 0.37^{b}$	$\pm 0.52^{c,e}$	$\pm 0.54^{d}$	±	±
					$0.45^{c,e}$	$0.47^{\mathrm{f}}$
Glutathione	55.80	59.93	49.26	54.15	50.93	53.07
peroxidase	$\pm 0.32^{a}$	$\pm 0.12^{b}$	$\pm 0.68^{\circ}$	$\pm 0.37^{d,f}$	±	±
(GPX)					0.65 <sup>e</sup>	0.28 <sup>d,f</sup>
Superoxide	85.46	90.55	78.92	82.34	77.73	80.80
dismutase	$\pm 0.53^{a}$	$\pm 0.16^{b}$	$\pm 1.19^{c,e}$	$\pm 0.54^{d,f}$	±	±
(SOD)					0.96 <sup>c,e</sup>	0.42 <sup>d,f</sup>

Data are expressed as mean $\pm$ SE. Mean values with different lower case superscripts are indicators of statistically significant difference at P $\leq$ 0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3). Where, MD stands for methanol:dichloromethane; HEX for hexane; DCM for dichloromethane, EA for ethylacetate, AQ for aqueous and HWE for hot water extract.

#### 5.3.2 Hypocholesterolemic effect of G. lucidum HWE

#### 5.3.2.1 Effect of G. lucidum HWE on body weight change

During the eight week long observation, gradual increase in body weight (ranging from  $140 \pm 5$  g up to  $280 \pm 2$  g) of the rats of all the groups was noticed (Figure 5.1) (Raw data: Appendix A, Table A 5.1; Statistical data: Appendix C, Tables C - 6.1 and 6.2). However, there was distinct body weight change among different groups. The hypercholesterolemic rats (H) gained the maximum weight and their rate of becoming weighty surpassed that of the others.



Figure 5.1: Effect of G. lucidum HWE upon body weight change of the rats.

Data are expressed as mean $\pm$ SE (n=3). Data were analyzed with one-way ANOVA and post-hoc Tukey's HSD test (P $\leq$ 0.05). Where, C=Control, CE= *G. lucidum* HWE fed control, H=hypercholesterolemic, HE=*G. lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= *G. lucidum* HWE fed AD rats, respectively.

Weight gain tendency of the normo-cholesterolemic rats was moderate and feeding of *G. lucidum* HWE resulted in decreased weight gain as time passed by. Similarly, *G. lucidum* HWE fed AD rats experienced relatively lower body weight growth than their non-fed counterparts. These findings indicate towards body weight lowering potencies of the *G. lucidum* HWE up on the experimental animals.

# 5.3.2.2 Effect on plasma TG level

Feeding of hypercholesterolemic diet to the rats resulted in their increased plasma TG levels (Table 5.3) (Raw data: Appendix A, Table A - 5.2; Statistical data: Appendix C, Tables C - 6.3 and 6.4). The hypercholesterolemic (H) rats had 1.58 times higher plasma TG level compared to the controlled (C) rats indicating the increased atherogenic propensity of the H rats. Feeding of *G. lucidum* HWE lowered plasma TG level in both the control, hypercholesterolemic and AD rats. Plasma TG lowering effect of the *G. lucidum* HWE was in the order of controlled (15%) > hypercholesterolemic

(13%) > AD (5%) rats (Table 5.3). TG lowering effect of *G. lucidum* HWE was statistically significant in each group compared to their respective controls (Table 5.3).

Plasma parameter	С	CE	Η	HE	Α	AE
(mg/uL)						
TG	122.73	103.93	192.0	167.33	$140.93 \pm$	$133.33 \pm$
	$\pm 0.63^{a}$	$\pm 0.85^{b}$	$\pm 0.880^{c}$	$\pm 0.90^{d}$	$0.70^{e}$	$0.52^{\mathrm{f}}$
TC	90.6	81.9	135.2	109.4	$114.67 \pm$	$101.87 \pm$
	$\pm 0.74^{a}$	$\pm 0.91^{b}$	$\pm 0.84^{c}$	$\pm 1.0^{d}$	0.85 <sup>e</sup>	0.56 <sup>f</sup>
HDL – C	29.06	36.47	26.94	33.33	26.73 ±	32.4 ±
	$\pm 0.42^{a}$	$\pm 0.55^{b}$	$\pm \ 0.58^{c,e}$	$\pm 0.40^{d,f}$	0.40 <sup>c,e</sup>	1.35 <sup>d,f</sup>
LDL – C	36.97	24.68	69.86	42.60	59.75 ±	42.8
	$\pm 0.58^{a}$	$\pm 0.10^{b}$	$\pm 0.90^{\circ}$	$\pm 1.34^{d,f}$	0.91 <sup>e</sup>	$\pm 0.57^{d,f}$
VLDL – C	24.54	20.78	38.40	33.47	28.19 ±	$26.67 \pm$
	$\pm 0.13^{a}$	$\pm 0.17^{b}$	$\pm 0.18^{c}$	$\pm 0.18^{d}$	0.14 <sup>e</sup>	0.10 <sup>f</sup>
AI	2.12	1.25	4.05	2.29	2.4	1.54
	$\pm 0.04^{a,d}$	$\pm 0.04^{b}$	$\pm 0.10^{c}$	$\pm 0.05^{a,d,e}$	$\pm 0.05^{d,e}$	$\pm 0.05^{f}$
LDL/HDL	1.28	0.68	2.6	1.29	2.25	1.32
	$\pm 0.02^{a,d,f}$	$\pm 0.04^{b}$	$\pm 0.08^{c}$	$\pm 0.05^{a,d,f}$	$\pm 0.06^{e}$	$\pm 0.027^{a,d,f}$

Table 5.3 Lipid profile, plasma atherogenic index and LDL/HDL ratio

Where, C=Control, CE= G. lucidum HWE fed control, H=hypercholesterolemic, HE=G. *lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= G. *lucidum* HWE fed AD rats, respectively. Data are expressed as mean $\pm$ SE. Mean values containing different lower case superscripts are statistically significant at P $\leq$ 0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3).

#### 5.3.2.3 Effect on plasma TC level

Feeding of hypercholesterolemic diet to the rats resulted in 1.5 times increased plasma TC levels (Table 5.3) (Raw data: Appendix A, Table A 5.3; Statistical data: Appendix C, Tables C - 6.3 and 6.4). Later on, treating the rats with *G. lucidum* HWE lowered plasma TC level significantly in all the rat groups (Table 5.3). Unlike that of TG lowering pattern, plasma TC lowering effect *G. lucidum* HWE on the hypercholesterolemic rats was higher than those of the normo-cholesterolemics and the pattern was H (19.08%) > A (11.16%) > C (9.60%) rats (Table 5.3).

#### 5.3.2.4 Effect on plasma HDL-C level

In the present study, the hypercholesterolemic and AD rats had lowered plasma HDL levels compared to the normo-cholesterolemic controls (Table 5.3) (Raw data: Appendix A, Table A 5.4; Statistical data: Appendix C, Tables C - 6.3 and 6.4) . Feeding of *G. lucidum* HWE increased plasma HDL-C levels significantly (P $\leq$ 0.05) in all the rat groups (Table 5.3). However, the rate of increasing was highest in the normocholesterolemics and the increasing trend was C (25.5%) > H (23.80%) > A (21.20%) rats (Table 5.3).

# 5.3.2.5 Effect on plasma LDL-C and VLDL-C level

Plasma LDL-C level increased 1.89 times in the hypercholesterolemic and 1.62 times in the AD rats, compared to the controlled ones (Table 5.3) (Raw data: Appendix A, Tables A - 5.5 and 5.6; Statistical data: Appendix C, Tables C - 6.3 and 6.4). Similarly, plasma VLDL-C level increased by 1.56 times in the H and by 1.15 times in the A rats (Table 5.3).

# 5.3.2.6 Effect on atherogenic index (AI) and LDL/HDL ratio

As depicted in Table 5.3 (Raw data: Appendix A, Tables A - 5.7 and 5.8; Statistical data: Appendix C, Tables C - 6.3 and 6.4), the hypercholesterolemic rats had increased atherogenic index (AI) compared to their normo-cholesterolemic counterparts. However, the *G. lucidum* HWE fed rats had lowered atherogenic index of plasma (AIP). On the other hand, the hypercholesterolemic and the AD rats showed higher LDL/HDL ratio compared to the *G. lucidum* HWE fed rats (Table 5.3).

#### 5.3.2.7 Inhibition of HMG Co-A reductase (HMGR) activity

As shown in the Figure 5.2, the HEX and the HWE of *G. lucidum* showed the most potent inhibitory effect towards HMGR activity. HMGR inhibitory effect of both of the fractions was very close to each other (52.12 for HWE and 53.3 for HEX) and

statistically not significant ( $p \le 0.05$ ) (Raw data: Appendix A, Tables A - 6.1-6.4; Statistical data: Appendix C, Tables C - 6.5 and 6.6).



Figure 5.2 HMG Co-A reductase inhibitory effect of mushrooms Where, C=Control, CE= G. *lucidum* HWE fed control, H=hypercholesterolemic, HE=G. *lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= G. *lucidum* HWE fed AD rats, respectively. Data are expressed as mean±SE. Bar charts with different lower case superscripts are statistically significant at P $\leq$ 0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3).

# 5.3.3 Organ function tests (Cardiac, liver and kidney function tests)

Feeding of *G. lucidum* HWE at the dosage of 200 mg/kbw did cause neither any visible adverse effect nor death of the experimental animals. Rather, the rats' body weight increased and their demand for intaking *G. lucidum* HWE increased as they grew up. On the other hand, significantly ( $P \le 0.05$ ) increased levels of serum AST/GOT, ALT/GPT and ALP enzymes in the hypercholesterolemic and AD rats compared with their controlled counterparts were observed (Table 5.4) (Raw data: Appendix A, Tables A - 7.1-7.8, Figures A - 4.1-4.3 and 5.1; Statistical data: Appendix C, Tables C - 6.7 and 6.8).

-	·					
Organ function	С	CE	Н	HE	А	AE
tests						
SGOT/AST	50.8	52.33	74.67	63.33	69.4	64.13
(U/L)	$\pm 0.42^{a}$	$\pm 0.44^{b}$	$\pm 0.73^{\circ}$	$\pm 0.40^{d,f}$	$\pm 0.55^{e}$	$\pm 0.58^{d,f}$
SGPT/ALT	19.93	21.66	28.67	22.87	29.6	24.8
(U/L)	$\pm 0.50^{a}$	$\pm 0.37^{b}$	$\pm 0.5^{c,e}$	$\pm 0.276^{d}$	$\pm 0.36^{c,e}$	$\pm 0.38^{\mathrm{f}}$
ALP (U/L)	65.73	60.06	106.07	94.06	87.73 ±	81.33
	$\pm 0.63^{a}$	$\pm 0.52^{b}$	$\pm 0.51^{\circ}$	$\pm 0.61^{d}$	0.54 <sup>e</sup>	$\pm 0.43^{f}$
Bilirubin	0.21	0.22	0.53	0.33	0.48	0.40
(mg/dL)	$\pm 0.006^{a,b}$	$\pm 0.005^{a,b}$	$\pm 0.01^{\circ}$	$\pm 0.005^{d}$	$\pm 0.008^{e}$	$\pm 0.008^{\mathrm{f}}$
Creatinine	0.25	0.27	0.66	0.53	0.5	0.35
(mg/dL)	$\pm 0.005^{a,b}$	$\pm 0.023^{a,b}$	$\pm 0.01^{\circ}$	$\pm 0.01^{d}$	$\pm 0.007^{e}$	$\pm 0.007^{\mathrm{f}}$
Urea (mg/dL)	15.80	15.67	24.80	19.87	$20.67 \pm$	17.53
	$\pm 0.48^{a,b,f}$	$\pm 0.31^{a,b}$	$\pm 0.44^{c}$	$\pm 0.73^{d,e,f}$	0.30 <sup>d,e,f</sup>	±
						0.33 <sup>a,d,e,f</sup>
LPO	1.33	1.42	2.67	1.85	2.46	2.27
(nmol/mg	$\pm 0.044^{a,b}$	$\pm 0.05^{a,b}$	$\pm 0.04^{\circ}$	$\pm 0.06^{d}$	$\pm 0.009^{e}$	$\pm 0.02^{\mathrm{f}}$
protein)						

Table 5.4 Ganoderma lucidum HWE fed rats' organ function tests

Where, C=Control, CE= *G. lucidum* HWE fed control, H=hypercholesterolemic, HE=*G. lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= *G. lucidum* HWE fed AD rats, respectively. Data are expressed as mean±SE. Mean values containing different lower case superscripts are statistically significant at P≤0.05 level with one-way ANOVA and post-hoc Tukey's HSD test (n=3).

#### 5.4 Discussion

#### 5.4.1 Effect on enzymatic anti-oxidative defense

Compared to the controlled rats, the hypercholesterolemics had lowered levels of plasma and liver anti-oxidant enzymatic (CAT, GPX and SOD) activities (Tables 5.1 and 5.2). Lowered levels of the anti-oxidative enzymes indicate deleterious effects of hypercholesterolemia such as increased OS generation and loss of cell membrane function and integrity. Hypercholesterolemia led to increased OS generation and associated reduced anti-oxidative defense had been noticed by multiple studies. This is due to increased lipid peroxidation, increased LDL oxidation in the arterial wall and decreased anti-oxidative enzymatic activity (Stocker & Keaney, 2004). Importantly, elevated cholesterol can stimulate superoxide production and inhibit nitric oxide generation (Landmesser *et al.*, 1999). Feeding of *G. lucidum* HWE increased enzymatic anti-oxidative defense of the rats as shown in Tables 5.1 and 5.2. Current findings are also in compatible with those of Wang *et al.* (2004) who showed that addition of 0.3%,

0.6% and 1.8% of *G. lucidum* powder with the basal diet significantly increased the levels of anti-oxidative enzymes (glutathione reductase, GPX and SOD) in different organs of the mice. Like the present *in vitro* (chapter 3) and *in vivo* anti-oxidative tests, Hasnat *et al* (2013) also reported the *in vitro* and *in vivo* anti – oxidative defense uplifting effect of the aqueous extract of *G. lucidum* grown on germinated brown rice.

Administration of *Ganoderma* tri-terpenoids and polysaccharide (PSG-1) to animals had been found to increase blood and liver CAT, GPX and SOD activity (Jang *et al.*, 2014; Lakshmi *et al.*, 2006; Smina *et al.*, 2011; Sudheesh *et al.*, 2012; Zhu *et al.*, 2016). Actually the anti-oxidative enzymes (CAT, GPX and SOD) participate in mutual antioxidative defense of the organisms. In the mitochondria of the eukaryotes, MnSOD is the first anti-oxidative enzyme that lowers the steady state of superoxide ( $O_2^-$ ). GPX acts mainly in the liver and reduce the hydroperoxides. Polyphenolics, tri-terpenes and polysaccharide present in the *G. lucidum* HWE might be involved in providing antioxidative defenses (Smina *et al.*, 2011; Zhu *et al.*, 2016).

# 5.4.2 Hypocholesterolemic effect of G. lucidum HWE

#### 5.4.2.1 Effect on body weight change

In the present study, *G. lucidum* HWE fed rats experienced lowered rate of body weight (Figure 5.1). Previous studies concerning effect of food intake on the animal body weight had presented conflicting results (Harnafi *et al.*, 2009; Ramachandran *et al.*, 2003). Supposedly, hypercholesterolemic diet had hindering effect upon intestinal absorption of proteins and other nutrients whose net effect was negative weight balance in those findings. This may be the cause why Wang *et al.* (2004) had not noticed change in body weight of the mice whose basal diet was supplanted with 0.3%, 0.6% and 1.8% of *G. lucidum* powder. Current findings are in agreement with several other findings

who reported a gradual increased body weight in the cholesterol fed rats (Amin *et al.*, 2011; Nwozo *et al.*, 2011; Otunola *et al.*, 2010).

Body weight change of the experimental animals occurs in two successive stages that overlap with each other to some extent. Firstly, during the maturing period, all the body parts (head, trunk, tail, internal organs) grow and gain relative weight. Secondly, during post-maturation, the organs do not gain relative weight so much and increased body fat corresponds towards increased body weight. As the present experimental animals were almost matured from their beginning of the experiment, decreased body weight gain of the *G. lucidum* HWE fed rats might occur due to decreased fat (triacylglycerol) deposition, decreased cholesterol biosynthesis or increased lipolysis and/or the combined effect of all.

#### 5.4.2.2 Effect on plasma TG level

TG regulates lipoprotein interactions and thus plays important role in maintaining lipid homeostasis. Its own level also varies depending on the lipoproteins especially LDL-C. Increased TG levels in the hypercholesterolemic rats might be due to the decreased clearance of TG owing to the lowered lipoprotein lipase (LPL) activity or due to increased deposition of LDL-C (Nofer *et al.*, 2002) (Table 5.3). Previously, Kabir and Kimura (1988) reported the TG lowering effect in the 5% dried *G. lucidum* fed spontaneously hypertensive rats. Human intervention studies supplemented with *G. lucidum* had also documented its TG lowering effect (Chu *et al.*, 2012; Wachtel-Galor *et al.*, 2004).

TG lowering effect of the HWE *G. lucidum* might be mediated by different strategies. First of all, decreased cholesterol biosynthesis through increased inhibition of the HMGR might be involved. Secondly, the phenolic content of the HWE of *G. lucidum* might form complexes with micelles and impair further metabolism (Bose *et* 

*al.*, 2008). Thirdly, the phenolics might inhibit pancreatic lipase activity resulting in lowered TG biosynthesis or stimulate the plasma lipoprotein lipase that restored the catabolism of TG (Ikeda *et al.*, 2005). Overall, the HWE of *G. lucidum* could withstand plasma elevation of TG level and could aid in guarding against hypercholesterolemia and atherogenic propensity.

# 5.4.2.3 Effect on plasma TC level

Significant reduction (P $\leq$ 0.05) in TC levels of the *G. lucidum* HWE-treated rats were observed in the present study (Table 5.3). The reason for increased rate of TC lowering in the hypercholesterolemic rats compared to those of the normo-cholesterolemics might be that the basal rate of cholesterol metabolism remained unaffected with the mushroom dosage applied during the experimental time-course (Hossain *et al.*, 2003). Controlled human intervention studies had also documented such effect (Chu *et al.*, 2012; Wachtel-Galor *et al.*, 2004).

TC lowering effect of *G. lucidum* might be mediated by several mechanisms. Importantly, the ganoderic acids share the structural features of the active site of HMG Co-A and might bind in the active site of HMGR that contributed its competitive inhibition of HMGR and lowered cholesterol biosynthesis. Besides, ganoderic acids might inhibit 14 $\alpha$  demethylase and inhibit the conversion of lanosterol to lathosterol and thus inhibit the last step of cholesterol biosynthesis (Hajjaj *et al.*, 2005; Li *et al.*, 2006).

Also, gano-polyphenolics might bind with bile acids and lower solubility of cholesterol in micelles (Ngamukote *et al.*, 2011). Gano-polyphenolics might also inhibit the activity of pancreatic cholesterol esterase and thus reduced intestinal cholesterol absorption (Heidrich *et al.*, 2004; Ngamukote *et al.*, 2011). In addition,  $\beta$ -D glucan present in the *G. lucidum* HWE might also form viscous gel and impair the formation of micelles that would lower intestinal cholesterol absorption (Cheung, 2013). Similarly,

chitin and chitosan might also aid in this concert (Neyrinck *et al.*, 2009). Moreover, increased fecal excretion of cholesterol in the form of bile acids might also be involved in the TC lowering performance of *G. lucidum*. In a similar study, Berger *et al* (2004) reported that feeding of the dried powder of *G. lucidum* to the hamsters at 2.5 % dosage, increase fecal total bile acids, chenodeoxycholate and corpostanol-3-one.

# 5.4.2.4 Effect on plasma HDL-C level

Hyperlipidemia is associated with decreased HDL level and increased susceptibility towards OS (Hansel *et al.*, 2004). Lowered level of HDL-C in the hypercholesterolemic rats of the present study might be due to the accelerated clearance of apo A1 from the plasma following hypercholesterolemia in the H and A rats (Baba *et al.*, 2007; Sorci-Thomas *et al.*, 1989). On the other hand, significantly increased (P $\leq$ 0.05) plasma HDL-C level in the *G. lucidum* HWE fed rats indicates increased clearance of TC from the peripheral tissue to the liver for excretion that points towards CVD ameliorating effect of *G. lucidum* HWE. Among multiple strategies, decreased biosynthesis and increased catabolism of TG and TC might result in increased plasma HDL-C level of the rats.

# 5.4.2.5 Effect on plasma LDL-C and VLDL-C

Increased levels of LDL-C and VLDL-C in the hypercholesterolemic rats might arise from their intake of hypercholesterolemic diet that might have downregulated the hepatic LDL receptors of the H rats (Roach *et al.*, 1993). *Ganoderma lucidum* HWE supplementation caused significant lowering effect upon plasma level of both LDL-C and VLDL-C in all the rat groups (Table 5.3). LDL-C lowering effect was highest for the hypercholesterolemic rats (39.02%), moderate in the controlled (33.27%) and lowest in the AD rats (28.36%) (Table 5.3). *Ganoderma lucidum* HWE caused significantly lowering effect upon the plasma LDL-C level of both H and AD rats as compared to the controlled (Table 5.3). *Ganoderma lucidum* HWE mediated decreased cholesterol
absorption and biosynthesis might cause decreased availability of hepatic cholesterol for lipo-protein biosynthesis in the extract fed rats (CE, HE, AE). As a consequence, decreased VLDL secretion in the plasma along with decreased conversion of VLDL into LDL may end into lowered plasma LDL level (Burnett *et al.*, 1997, 1999). Mechanistically, *G. lucidum* HWE induced increased LDL receptor in the rat hepatocytes may contribute to the LDL lowering effect that resulted in lowered secretion of LDL in the rat plasma (Bursill *et al.*, 2007). Increased clearance of LDL from the blood of the mushroom-fed rats may also have been involved (Burnett *et al.*, 1999). In case of *P. ostreatus* fed hypercholesterolemic rats, increased plasma clearance of VLDL had been noticed (Bobek & Ozdin, 1993). Feeding of *G. lucidum* HWE might have increased the fractional turnover of VLDL-C and lipoprotein lipase activity that resulted in decreased VLDL-C and TG level in the mushroom fed rats (Bobek *et al.*, 1993; Bobek & Ozdin, 1993; Kimball *et al.*, 1983).

# 5.4.2.6 Effect on AI and LDL/HDL ratio

The increased atherogenic index (AI) of plasma of the hypercholesterolemic rats observed in the present study (Table 5.3) might be emanated from the increased level of the atherogenic lipoprotein fractions (LDL-C) associated with hyperlipidemia (Dobiasova, 2006; Dobiášová & Frohlich, 2001). Like other hypolipidemic natural products, lowered AI values in the *G. lucidum* HWE fed rats also was expected (Acay *et al.,* 2014). Significantly lowered AI values in the *G. lucidum* HWE fed rats was observed in the present study (Table 5.3). Lowered AI in the *G. lucidum* HWE fed rats corresponds towards this mushroom's lipid lowering, anti-atherogenic, cardioprotective and AD ameliorating performance (Dobiasova, 2006; Dobiášová & Frohlich, 2001).

LDL/HDL ratio is another indicator of hyperlipidemia led physiological complications including atherosclerosis and AD (Granholm *et al.*, 2008). Higher HDL

level, lower LDL level and lowered LDL/HDL ratio is expected for the hyperlipidemia modulating agents (Granholm *et al.*, 2008). *Ganoderma lucidum* HWE feeding could lower the LDL/HDL ratio in every rat group (Table 5.3). These results further substantiate the hypolipidemic, hypocholesterolemic, anti-atherogenic and AD modulating potentiality of the *G. lucidum* HWE (Granholm *et al.*, 2008).

# 5.4.2.7 Inhibition of HMGR

All the mushroom fractions and extracts that underwent in vitro anti-oxidative assessments in chapter 3, also were analyzed for theing inhibitory effect towards HMGR. Compared to others, the hexane and the HWE of G. lucidum showed better inhibitory effects towards HMGR activity (Figure 5.2). As the presence of statin in the G. lucidum fractions had not been searched in the present study, it is difficult to convincingly indicate that natural statins were present in the respective fractions. However, ergosterol present in the hexane fraction might be regarded as the inhibitor of HMGR activity. Our proposition is supported by Gil-Ramirez et al (2013a, 2013b) who reported the HMGR inhibitory effects of edible mushrooms to be mediated by sterol (ergosterol) and polysaccharide (β-D glucan) (Gil-Ramírez et al., 2013; Gil-Ramirez et al., 2011; Gil-Ramírez et al., 2013). Thus, contrary to statins, a novel in vitro inhibitory mechanism of HMGR activity might have been functional in case of G. lucidum fractions (Gil-Ramírez et al., 2013; Gil-Ramírez et al., 2013). Gano-components (especially ergosterol) might have scavenged the HMGR substrate, HMG Co-A and itself bound to the enzyme that resulted in lowered HMGR activity. Ergosterol peroxide from the edible mushroom Sarcodon aspratus had been found to down-regulate the expression of low-density lipoprotein receptor (LDLR) and HMG-CoA reductase genes in RAW264.7 cells (Kobori et al., 2007). Feeding of 5% G. lucidum dried powder to the hamsters resulted in 1.5-fold decreased HMGR activity in the hepatic microsomes (Berger et al., 2004). Berger et al. (2004) also identified sterols and triterpenes, rather

than statins in *G. lucidum* fractions, as inhibitors to the HMGR activity *ex vivo* (Berger *et al.*, 2004). Also, 26-oxyenosterols (ganoderols A, B, ganoderal A and ganoderic acid Y) derived from *G. lucidum* have been found inhibitory towards cholesterol biosynthesis in human hepatic cell lines T9A4 (Hajjaj *et al.*, 2005).

Ganoderic acids (7-oxo-ganoderic acid Z and 15-hydroxy-ganoderic acid S) isolated from the lipophilic fraction of *G. lucidum* had been found inhibitory towards HMGR activity (Li *et al.*, 2006). Ganoderol A at 7  $\mu$ M concentration, lowered cholesterol biosynthesis up to 40% in the T9A4 cell lines and the mechanism of inhibition was either through conversion of acetate and mevalonate as the cholesterol biosynthetic precursors or by inhibiting 14 $\alpha$  demethylase in the conversion of lanosterol to lathosterol (Hajjaj *et al.*, 2005). As outlined in chapter 4, different ganoderic acids detected in the HWE of *G. lucidum* might also have contributed towards HMGR inhibitory effects (Li *et al.*, 2006).

# 5.4.3 Organ function tests

Compared with the control and the *G. lucidum* HWE fed rats, the AD and the hypercholesterolemic rats were detected with elevated levels of serum AST/GOT, ALT/GPT and ALP enzymes in the present study. Usually, these are the enzymes of diagnostic importance whose level rise in blood as they leak out of the damaged cell membranes following CVD complications. Thus, cholesterol affected the membrane integrity and permeability of the rats. In the similar fashion, significantly (p  $\leq 0.05$ ) increased bilirubin, creatinine and urea level in the hypercholesterolemic rats also indicate the cholesterol-mediated perturbed metabolism of the respective animals (Table 5.4). Elevated bilirubin and creatinine levels in the hypercholesterolemic rats indicate their abnormal hepatic and renal functions, respectively (Muntner *et al.*, 2000). Besides, increased serum bilirubin remains as a individual risk factor of CVD (de Sauvage

Nolting *et al.*, 2011). Cholesterol-rich diet might cause excessive protein catabolism and amino acid deamination for gluconeogenesis that resulted in elevated urea level. A number of studies also reported high fat diet induced impaired organ function and lipid profile in the experimental animals (Frantz *et al.*, 2012; Saki *et al.*, 2011; Vaziri, 2003). ALT is much specific for hepatocellular injury than AST as ALT is predominantly present in the hepatic cytosol and AST is available in liver, heart and in kidney (Nikolaou *et al.*, 2012).

Hypercholesterolemia-induced oxidative stress leads towards hepatocyte membrane lipid peroxidation and disrupts the membrane integrity allowing the AST, ALT and ALP to be leaked into the serum (Arhan et al., 2009). However, feeding of G. lucidum HWE had ameliorating effect upon both the enzymatic and non-enzymatic biomarkers (Table 5.4). These findings indicate the cellular membrane integrity maintaining and leakage restoring capability of the G. lucidum. Our findings are in agreement with those of Lakshmi et al., (2006), who reported that G. lucidum was able in reverting the AST, ALT and ALP levels back to their normal levels in the benzo[a]pyrene-induced hepatic injured rats. Similar protective role of G. lucidum had been reported for the ethanol and carbon tetrachloride-injured hepatotoxicity (Ali et al., 2014; El Shawi et al., 2015; Jang et al., 2014; Lin & Lin, 2006; Sudheesh et al., 2012). Free-radical scavenging effect of G. lucidum extract had been suggested providing hepato-protective support against hepatocellular necrosis (Lin & Lin, 2006). In case of G. atrum, polysaccharide (PSG-1) had been found providing hepato-protective effect through lowering AST and ALT levels along with increasing anti-oxidative potentiality and stimulating increased synthesis of short chain fatty acids (SCFA) by the liver (Zhu et al., 2016). SCFAs (acetic acid, propionic acid, butyric acid) are fermented products of complex carbohydrates. Ganoderma lucidum β-D glucan might be fermented by the rat gut microbiota into SCFAs that later on might be reabsorbed and provide cholesterol

lowering effect in the liver (Bhakta & Kumar, 2013; Hara *et al.*, 1999; Hu *et al.*, 2013) . Peptide isolated from *G. lucidum* had also been found potent in ameliorating carbon tetrachloride induced hepatotoxicity through anti-oxidative mode with corresponding lowered AST, ALT levels (He *et al.*, 2008). Even, tri-terpenes isolated from *G. lucidum* had been found safe and having no adverse effect upon hepatic and renal functioning as evidenced by the maintenance of the normal value of AST, ALT, ALP, urea and creatinine in the treated rats (Smina *et al.*, 2011). As evidenced from the current findings, feeding of *G. lucidum* HWE did not cause any adverse effect to the rats rather evoked anti-oxidative defense and plasma hypolipidemic status (Tables 5.1 - 5.4). Thus, *G. lucidum* HWE could be considered as safe and non-toxic to the experimental animals.

# 5.5 Conclusion

Feeding of the HWE of *G. lucidum* enhanced *in vivo* anti-oxidative defenses and also provided hypocholesterolemic effects to the rats. Heightened anti-oxidant status of the animals had been evidenced by the increased plasma levels of the anti-oxidant enzymes (CAT, GPX and SOD) and by the decreased plasma levels of TG, TC, LDL-C, LDL/HDL ratio, AI and increased HDL-C. The hypocholesterolemic effect of the HWE of *G. lucidum* has further been proved through its lowering effect upon the cholesterol biosynthetic regulatory enzyme, HMG Co-A reductase. Moreover, feeding of the HWE of *G. lucidum* did not cause any detrimental effect to the experimental animals (Table 5.4). In conjunction with the previous chapter's (chapter 3) *in vitro* anti-oxidative findings with those of current chapter's *in vivo* anti-oxidative and hypocholesterolemic effects, the HWE of *G. lucidum* could be interpreted in AD amelioration.

#### CHAPTER 6: ALZHEIMER'S DISEASE AMELIORATING EFFECT OF G.

## **LUCIDUM**

# **6.1 Introduction**

Alzheimer's disease (AD) is neuro-degenerative disorder affecting mainly the elderly people. People afflicted with AD suffer from declined memory and learning abilities, problem in speaking and performing daily activities by him or her. The patient becomes solely dependent on the family members and the care-givers and ultimately bed-bound. This occurs because the neurons of the AD patients become damaged due to the formation of A $\beta$  plaques and/or NFTs. Currently, more than 40 million people are suffering with AD round the globe and the alarmingly increased rate of this fatal disorder adds extra burden to the ever increasing economical and societal sectors and demands immediate medico-healthcare oriented management (Alzheimer's Association, 2016).

Despite tremendous efforts in combating this global epidemic, there is hardly any AD medication available upto present date. The available therapeutic strategies are aimed at symptom – modifying targets rather than preventing the neurons from damages. For example, the six drugs approved by the United States Food and Drug Administration (USFDA) can improve AD symptoms only through modulating brain neurotransmitter release. The failure to achieve the ultimate goal in slowing AD progression might remain hidden into multiple causative factors encompassing OS, metabolism, hypertension, genetics, epigenetics as well as adaptive response to some stressors. Thus, effective management of the co-existing factors of AD has been highly regarded in the most recent recommendation from the Alzheimer's association (Alzheimer's Association, 2016). As the HWE of *G. lucidum* showed most potent *in vitro* (chapter 3) and *in vivo* anti-oxidative and cholesterol lowering effects (chapter 5) as well as possessed the highest amount of bio-components (chapter 4), this was selected for evaluating its AD ameliorating effect. Another rationale for choosing *G. lucidum* also emanates from the fact that despite its immense medicinal values, there is hardly any study available that incorporates anti-oxidative, hypocholesterolemic and AD ameliorating effect of this mushroom. Thus, the present chapter is aimed at determining the AD ameliorating effect of this mushroom through memory related learning and behavioral tests, neuro - biochemical and histopathological assessments.

#### 6.2 Materials and methods

#### 6.2.1 Animals

Ninety wistar male rats were divided into six groups, each group containing 15 rats, based on their weight range (120  $\pm$  5 gm). The groups were control (C), *G. lucidum* HWE fed control (CE), hypercholesterolemic (H), *G. lucidum* HWE fed hypercholesterolemic (HE), Alzheimer's diseased (A in case of anti-oxidative, hypochoesterolemic, memory and learning related behavioral and biochemical and histopathological tests. AD in case of proteomics study) and *G. lucidum* HWE fed Alzheimer's diseased (AE). The AE group received 200 mg/kg body weight *G. lucidum* HWE. All the experimental protocols had been approved by the ethical permission committee, University of Malaya Institutional Animal Care and Use Committee (UMIACUC) [Ethics reference no. ISB/25/04/2013/NA (R)] (Appendix G).

# 6.2.2 Preparation of AD model rats

Male wistar rats were prepared by injecting A $\beta_{1-42}$  (ab120959, abcam, USA) in the right cerebral ventricles (Figure 6.1). Compared with the transgenic AD model rats, this

process is economical and the progression of AD symptoms and amyloidogenesis have been reported to be like those of human subjects (Lecanu & Papadopoulos, 2013). Also, Drugs found effective in rat models have already been translated to human therapeutic purposes and rat models provide opportunities towards pharmacodynamic studies of candidate drug or bio-component involved in AD therapeutics (Shineman *et al.*, 2011). Anesthetizing the rats with intra-peritoneal injection of sodium pentobarbital (40 mg/kg body weight), hair on the rats' heads was shaved and fixed the rats in the stereotaxic frame (SR-5R-HT, Narishige, Japan) by using the locks associated with it. Povidoneiodine (6%, USP) was used as antiseptic to rub the shaved portion of the head and placed the head at the midpoint of the stereotaxic frame. The skull was opened through incision, clipped the skin and sterile cotton and ice cold saline (0.9% NaCl) were used to wash bleeding. Removing skull-linked muscles, the skull was kept open and dried for a while so that the bregma became visible. Paralleling the skull to the frame, the ventricular points were stereotaxically spotted at 1.2 mm lateral and 0.8 mm posterior distant from the bregma following the brain atlas of Paxinos and Watson (1998). Marking of the spotted points followed microdrilling to make two pin holes at the two ventricles. For infusion of A $\beta_{1-42}$ , a vehicle consisting of 35% acetonitrile (v/v) and 0.1% trifluoroacetic acid (v/v, pH 2.0) was used. A $\beta_{1-42}$  (5 µL, 2.5 nmol at 1 µL/min flow rate) was infused in the right ventricle and 5 µL of 1% Alcl<sub>3</sub> in the left ventricle (for  $A\beta_{1-42}$  aggregation) using a Hamilton microsyringe (Neuros model 7001 KH, Hamilton, USA). After infusion, the rat was left untouched for a while so that adequate absorption could take place and then sealed the pin holes with sterile bone foam and cement. After drying of the bone cement, the skin clips were removed, wetted skin with saline (0.9% saline) and stitched the skin using sterile needle and thread. Thus, the AD positive controls were prepared. A sham control group was prepared using the same solvent injection. In order to be certain about  $A\beta_{1-42}$  infusion in the cerebral ventricles,

another trial surgery of rats maintaining the same procedure and using coomassie brilliant blue instead of A $\beta_{1-42}$  was performed in parallel with the original. After about 1 hour, rats of the trial group were sacrificed, the brains collected and the ventricles observed. Oberving the dye in the right cerebral ventricle following the same procedure, reaching of  $A\beta_{1-42}$  to the right cerebral ventricles could be ascertained (Figure 6.1).



A. Intra-peritoneal injection of sodium pentobarbital for anesthesizing the rat.



B. A $\beta$  infusion to rat's right cerebral ventricle.





D. Collection of blood.

C. Coomassie brilliant blue showing the evidence of  $A\beta$  reaching to the right cerebral ventricle. Figure 6.1: Preparation of AD model rats and collection of blood

# 6.2.3 Post-operative care and recovery of the rats

As post-operative care to the rats, antibiotic ciprofloxacin was injected intramuscularly to avoid infection and put the rats in a temperature-regulatory chamber for maintaining their body temperature at 37°C. After their anesthesia period was over, food and saline were provided to the rats and transferred those to the sterile bedding for recovering of the surgical stress. During recovery phase (about 7 days), repeated injection of antibiotic (cephalexin, 15 mg/kg body weight, thrice daily) associated with careful feeding and drinking assistance to the rats. After the rats recovered from surgical stress and became healthy, they were adapted for the behavioral tests. All the experimental protocols had been approved by the ethical permission committee, University of Malaya Institutional Animal Care and Use Committee (UMIACUC) [Ethics reference no. ISB/25/04/2013/NA (R)] (Appendix G).

# 6.2.4 Memory and learning related behavioral tests

Evolutionarily, humans and rodents (mice, rats, guinea pigs, hamsters) have very close root of origin and they share much common behavioral and survival strategies. Thus, behavioral studies on rodents exemplify human behavioral traits whose analyses aid greatly in identifying and managing pathophysiological alterations as well for establishing therapeutic approaches. Animal models of behavioral abnormalities, especially those for neurodegenerative diseases such as AD have been in vogue for the last few decades. Capacity of an animal to orient itself in its environment is essential for its very existence as its feeding, mating and escaping from the enemies depend on this feature. Interestingly, orientation is an important criterion of spatial memory and AD patients gradually suffer from disorientation of time and place. At extreme cases, the AD patients cannot even locate their own houses though they had been living there since their birth. In memory and learning related behavior tests, both spatial and nonspatial performances of experimental animals are studied and through comparison with the controlled ones, alteration of the affected animals and amelioration of the drug treated ones are measured. In the present study, behavioral tests for both spatial and non-spatial memory and learning were performed whose descriptions have been appended below (Table 6.1).

Tests for spatial memory and	Tests for non-spatial memory		
learning	and learning		
Eight-armed radial maze test.	Novel object recognition (NOR) test.		

# Table 6.1: Memory and learning related behavior tests performed

# 6.2.4.1 Eight-armed radial maze test

# A. Principle

Eight-armed radial maze test is based on the principle of win-shift task: the experimental animal is supposed to achieve the reward (food pellet, as it is kept hungry) from different locations within the environment and it must forage among different goals. While foraging, it uses landmarks and visual cues for memorizing spatial location of the reward and later on utilizes its acquired memory for further navigation. Its repetitive search in the same arm of the maze and failure to achieve the reward in a single trial are considered as poor or impaired memory. Developed by Olton and Samuelson (1976), the eight armed radial maze had successfully been used in deciphering the spatial learning abilities of the rats. There are different procedural attempts for data collection and memory measurement in radial arm maze (RAM). Matching-to-sample (MTS) free-choice system measures the long-term memory, where only a few arms contain food always and all the arms are kept open. In non-match to sample (NMTS) free-choice strategy, the animals are initially allowed to enter all the arms and receive food without delay. As trial progresses, delay in receiving food is applied and short-term memory is measured. In delayed match to sample (DMTS) freechoice process, the animals are allowed to receive food from the baited arms, they are provided with delay to visit the baited arms. Subsequent trials involve alteration of the baited arms. Long-term memory performance is the resultant effect of animals' visiting

to the baited and un-baited arms, food pellet receiving and maze latency (delay). All these are free-choice selection processes and there is also forced-choice type of DMTS, where some of the arms of the RAM are blocked and the animals are forced to explore the un-blocked ones only. In the present study, the DMTS free-choice procedure was performed to measure the memory and learning related behavior ameliorating effect of *G. lucidum* HWE upon the rats.

#### B. Eight-armed maze set up

An eight - armed radial maze made of polyvinyl chloride (0.5 cm width) was used in the present experiment (Figure 6.2). The maze was placed in a closed room and raised 38 inches above from the floor with arms prolonged from a central octagonal arena with a diameter of 30 cm. Each arm of the maze was of 50 cm long, 11 cm wide and 4 cm height. Additional arms (12 cm height) extended from the octagonal arena prevented rats jumping from one arm to another. Two centimeter before the end of each arm, there were food cups of 2.5 cm diameter and 1.5 cm depth (Figure 6.2). Visual cues (wall posters, chair and tables) and the position of the experimenter behind a canopy were kept constant during the entire period of the experiment.

# C. Behavioral experiment

Behavioral experiments of 8-armed radial maze consisted of three phases: conditioning, acquisition and final experimentation

I. Conditioning: Each rat was handled for 5 minutes 7 days to habituate them to the experimenter and to the maze environment. The went up to gradual deprivation of food up to 15% of normal food intake to induce them to search for food as the food pellets (10 mg) were placed in the food cups of the maze.

II. Acquisition: Rats placed at the center of the maze explored the arms to find the food reward (Figure 6.2). They were trained for 7 days with one trial per day. Food pellets placed in any arm was kept constant across the whole session and trial.



A. 8-armed radial maze.

B. Rat in search of food pellet.



D. Rat found food pellet and looks at visual cue for memory formation.







E. Rat uses visual cues and memory to find the baited arm.

F. Improved memory leads the rat to the baited arm.

Figure 6.2: Eight armed radial maze test

- III. Experimentation phase: Each rat was examined to collect food reward from the food cups of the four of the eight arms (Figure 6.2). A trial was terminated after either all the bait had been consumed or after 5 minutes had elapsed. Following the memory error definition of Jarrard *et al.* (1984), we measured three parameters of memory function:
  - a. Reference memory error (RME): entry into the unbaited arms (arms containing no foods).
  - b. Working memory error (WME): repeated entry into the arms that had already been visited.
  - c. Maze latency: time required for completing the entry into the four baited arms.

# 6.2.4.2 Novel object recognition (NOR) test

Novel object recognition (NOR) test is used to test the recognition (working, shortand/or long-term) memory of the animals. The test is based on the natural tendency of the rodents to explore the new objects. Thus, animals with better recognition memory will explore the newly oriented objects for long time compared with the exploration timing of the familiar objects (Bevins & Besheer, 2006). A polypropylene box with dimension of  $120 \text{cm} \times 120 \text{cm} \times 120 \text{cm}$  and plastic toys were used and followed the procedure described by Bevins and Besheer (2006) (Figure 6.7). The processes involved habituation (2 days), training (2 days) and testing (4 days). Each rat was allowed to explore two identical objects for 5 minutes followed by recognition and exploration of the novel object for 5 minutes. Activities of the rats were monitored using video camera (HDR CX130E, Sony, Japan) and arcsoft showbiz software and the video files were tracked with the tracking software kinovea.

The results are expressed as percentage of preference for novel object exploration:

$$T (\%) = (T_{novel} / [T_{novel} + T_{familiar}] X 100).$$

Where,  $T_{familiar}$  = Time spent in exploring the familiar object.

 $T_{novel}$  = Time spent in exploring the novel object.



Figure 6.3: Novel object recognition test

# 6.2.5 Neuro - biochemical and neuro - immunological tests of memory and learning related parameters

#### 6.2.5.1 Plasma and brain lysate preparation

Twenty four hours following the last treatment and behavioral test, the rats were kept in fasting overnight. Then, the rats were anesthesized with intra-peritoneal injection of sodium pentobarbital (35 mg/kg body weight) and sacrificed. The brains were perfused through the left ventricle with ice-cold saline immediately after blood collection. Heads were separated from the body followed by immediate collection of the brains from the skulls and preserved at - 80 °C. However, rats whose brains were preserved in glutaraldehyde and/or formalin for histopathological studies were not perfused. Hippocampus was separated from the cerebral cortex followed by homogenization of the hippocampus using Dounce glass homogenizer (D8938, Sigma-Aldrich, USA) and ice cold phosphate buffer (25mM, pH 7.4). Tissue remnants were removed by centrifuging the homogenates at 800 g and collected the supernatant. An aliquot of the supernatant was used for preparing cytosolic fraction (through centrifuging at 1000 rpm), detergent soluble fraction (DSF) and detergent insoluble fraction (DIF) of hippocampus. DSF was prepared following the modified method of Hashimoto et al. (2005). Briefly, homogenization of the hippocampus was performed in 1% Triton X-100 containing ice cold phosphate buffer (25 mM, pH 7.4), followed by centrifugation at 800 g for 10 min and collection of the supernatant. Agitation of the supernatant for 2 hr over ice-cold temperature followed by centrifugation at 10, 000 rpm for 1 hr produced the supernatant denoted as DSF and the pellet denoted as the DIF.

# 6.2.5.2 Antibodies and enzyme linked immunosorbent assays (ELISAs)

#### A. $A\beta$ (1-42) oligomers

The levels of  $A\beta_{(1-42)}$  oligomers in DSF of hippocampus homogenates and brain derived neurotrophic factor (BDNF), synaptosomal associated protein 25 KD (SNAP 25), post-

synaptic density protein 95 KD (PSD 95), vesicular acetylcholine transporter (VAchT) and tumor necrosis factor  $\alpha$  (TNF $\alpha$ ) in the cytosolic fractions of the rat hippocampus homogenates were detected utilizing sandwich ELISA. Briefly, ELISA microplates (Cellstar, Greiner-bio one, USA) were coated with 20 µl of the DSF and or cytoplasmic fraction of hippocampus using 0.1M carbonate buffer (pH 9.6). Blocking was done using 3% BSA in PBS followed by addition of primary antibody [rabbit polyclonal anti-AB, Abcam, ab 10148; mouse monoclonal anti-BDNF, Abcam, ab 10505; mouse anti-SNAP 25, Abcam, ab 31281; mouse anti-PSD 95, Abcam, ab 18258; rabbit polyclonal anti-VAchT, Abcam, ab 68984 and mouse monoclonal anti-TNFa, Abcam, ab 1793, USA, respectively] with dilution of 1:1000 and incubated at room temperature for 2 h. HRP-coupled anti-rabbit IgG (Abcam, ab 6721, USA) was used as the secondary antibody and incubated for 2 h at 37°C. Then, tetramethyl benzidine substrate (TBS) (Sigma-Aldrich, USA) was added and incubated in dark for 30 min at 25°C followed by addition of stop solution (0.1 N HCl). Wells containing 0.1 M carbonate buffer (pH 9.6) only were used as the negative controls. The wells were analyzed with a multiwell plate reader (Synergy H1, Gen 5 BioTek multimode plate reader, USA) at 450 nm.

#### B. Aβ (1-42) anti-fibrillating assay of G. lucidum HWE

As the conversion of oligomeric A $\beta$  (1-42) into its fibrillar form is an important aspect of AD pathogenesis and withstanding the fibrillation process stands as an important AD preventing stratagem, the inhibitory effect of the *G. lucidum* HWE upon the *in vitro* fibrillation of A $\beta$  (1-42) oligomers was measured (Ahmed *et al.*, 2010). Thioflavin T test was used following the modified method of Ma *et al.* (2014). Briefly, A $\beta$ (1-42) was dissolved in the aggregation buffer (20 mM Tris, 100 mM NaCl, 1 mM EDTA, pH 7.4) and final concentration of A $\beta$ (1-42) made into 50  $\mu$ M with or without the addition of 20  $\mu$ M *G. lucidum* HWE and incubated at 37 °C for 24 h. Then, 40  $\mu$ L of the solution was mixed with 210  $\mu$ L of 10  $\mu$ M ThT and glycine-NaOH buffer (50mM, pH 8.0) and fluorescence measured with excitation wavelength of 448 nm and emission wavelength of 487 nm (FLx 800 fluorescence reader, Gen 5 Biotek, USA).

# 6.2.5.3 Transmission electron microscopy (TEM) studies

# A. Brain sample preparation for transmission electron microscope (TEM) study

As hippocampus is involved in memory and learning effects mostly and degenerative hippocampal neurons are hallmarks of AD and impaired memory, hippocampi of the experimental animals were chosen for TEM studies to view the effect of *G. lucidum* HWE upon the experimental animals. Following procedures have been performed to view the hippocampal neurons through TEM:

- a. Primary fixation, trimming and post fixation: In order to preserve the structures in their native state, chemical and irreversible fixation of the hippocampus were performed by immersing them in 2.5 % glutaraldehyde immediately after collection, kept overnight and trimmed the tissues into 1-2 mm sizes on the second day. Washing with cacodylate buffer 2-3 times, the trimmed tissues were post-fixed with the osmium tetroxide: cacodylate buffer (1:1) for 2 hours and then stored O/N at 4 °C after washing with cacodylate buffer.
- b. Washing and dehydration: After a series of washing 2 3 times each with doubled distilled water (5 min), 35% ethanol (10 min), 50% ethanol (10 min), 70% ethanol (10 min), 95% ethanol (15 min), 100% ethanol (2 times, each for 15 min), propylene oxide (15 min), propylene oxide : Epon [containing agar, dodecenyl succinic acid anhydride, methyl nadic anhydride and 2,4,6 tridimethyl amino ethyl phenol] (1 :1, 1 hour) and propylene oxide : Epon (1 : 3) (2 hour), we kept the samples in Epon O/N.

- c. Embedding: The samples were embedded using fresh resins and allowed to polymerize first at 37 °C for 5 hours and then at 60 °C O/N.
- d. Block trimming: By clamping the block firmly into the block holder and placing under the stereomicroscope (Leica, M 80), the excess embedding media was trimmed until the tissue was exposed. Through trimming, the block was made into a trapezium shape.
- e. Semithin sectioning: The trimmed block was clamped into a microtome chuck, adjusted the angle of the glass knife (prepared using Leica EM KMR3 and Reichert knifemaker) to use its "E" edge and trimmed the block surface (Leica ICC50 HD). Removing the debris, fine sections were collected with forceps and transferred them onto a drop of double distilled water on a glass slide. Glass slides containing the sections were collected upon bench top to let the section fixed.
- f. Toluidene blue staining: Putting a drop of filtered toluidine blue stain onto dried semithin sections, the slides were kept on the hot plate for 1 minute. Washing out the stain into a beaker, differentiation was performed with 95% ethanol. After washing two times with water, the slides were dried by putting on the hot plate (Electrochemical slide drying bench).
- g. Ultra-thin sectioning: Clamping the trimmed block into a ultramicrotome chuck securely and looking through microscope (Leica EM UC6), ultra-thin sections of 75-85 nm were prepared using a diamond knife and then collected onto copper grids.
- h. Uranyl acetate staining: The ultra-thin sections were stained facing side of the copper grids with millpore filtered and centrifuged uranyl acetate solution (on a piece of dental wax put on a petri dish) for 10 minutes while remaining it covered. Washing three times with doubled distilled water and wiping with filter

paper, the dental wax was surrounded with a pellet of NaOH and added a drop of millpore filtered and centrifuged lead citrate solution to the ultrathin sections facing side of the copper grids. After 10 minutes, the grids were washed three times and let the grid drying with a filter paper so that the grid was ready for viewing.

# B. Viewing of the prepared samples by TEM

The prepared samples were viewed using the TEM (Leon Libra 120, Zeiss, Germany) equipped with the soft imaging viewer v5.0 software, images captured and produced for reporting. All the experimental protocols had been approved by the ethical permission committee, University of Malaya Institutional Animal Care and Use Committee (UMIACUC) [Ethics reference no. ISB/25/04/2013/NA (R)] (Appendix G).

#### 6.3 Results

# 6.3.1 Effect of intra-cerebroventricular infusion of Aβ1-42

Infusion of soluble  $A\beta 1-42$  to the rat cerebral ventricles affected their memory and learning related behavioral tasks indicating the effectiveness of the current model of AD studies. In addition to the quantitative data described below under specific sub-sections, distinct behavioral discrepancies in the AD model rats compared with those of the agematched controls were observed. Among the mostly noted altered behavioral responses were such as the AD rats did not explore the rearing cages frequently, they remained aloof and rarely social and rarely tried to cross-over the cages.

#### 6.3.1.1 Memory and learning related behavioral tests

#### 6.3.1.2 Eight-armed radial maze study

As depicted in Figure 6.4 and 6.5, feeding of *G. lucidum* HWE lowered RME [ $F_{(5,42)}$  = 11.31, P $\leq$  0.05] and WME [ $F_{(5,42)}$  = 16.78, P $\leq$  0.05] in all the rat groups (Raw data: Appendix B, Tables B 1.1 - 1.15; Statistical data: Appendix C, Tables C -7.1, 7.2, 7.3 and 7.4). Tukey post-hoc multiple comparison tests revealed that WME lowering effect of *G. lucidum* HWE was statistically significant (P  $\leq$  0.05). Also the maze latency (time required to explore the baited arms) gradually decreased in the extract fed rats (Figure 6.6). In addition to the extract fed AD (AE) rats, the extract fed hypercholesterolemics (HE) showed better performance in the RAM test (Figures 6.5). These findings correspond towards this extract's memory and learning enhancement as well as AD ameliorating effect.



#### **Figure 6.4: Reference memory error of the rats**

Data of every 6 trials have been averaged over a block and expressed as mean $\pm$ SE (n=3). Data were analyzed with one-way ANOVA and post-hoc Tukey's HSD test (P $\leq$ 0.05). Here, C=Control, CE= *G. lucidum* HWE fed control, H=hypercholesterolemic, HE=*G. lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= *G. lucidum* HWE fed AD rats, respectively.





Data of every 6 trials have been averaged over a block and expressed as mean $\pm$ SE (n=3). Data were analyzed with one-way ANOVA and post-hoc Tukey's HSD test (P $\leq$ 0.05). Here, C=Control, CE= *G. lucidum* HWE fed control, H=hypercholesterolemic, HE=*G. lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= *G. lucidum* HWE fed AD rats, respectively.



# Figure 6.6: Time spent in exploring the baited arms

Data of every 6 trials have been averaged over a block and expressed as mean $\pm$ SE (n=3). Data were analyzed with one-way ANOVA and post-hoc Tukey's HSD test (P $\leq$ 0.05). Here, C=Control, CE= *G. lucidum* HWE fed control, H=hypercholesterolemic, HE=*G. lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= *G. lucidum* HWE fed AD rats, respectively.

# 1.1.1.1 Novel object recognition (NOR) test

In novel object recognition test, the *G. lucidum* HWE fed rats showed higher preference for recognizing and exploring the novel objects than each of the controlled counterparts (Figure 6.7) (Raw data: Appendix B, Table B 1.16 – B 1.21; Statistical data: Appendix C, Tables C - 7.5 and 7.6). Preference of the extract fed controlled rats (CE) was the highest (84.83%), followed by those of the extract fed hypercholesterolemic (HE, 71.67%) and extract fed AD rats (AE, 64%) (Figure 6.7). One way ANOVA with Tukey post-hoc test showed that although the values of each extract fed group did not significantly differ from those of the non-fed rats, there was significant group difference [ $F_{(5,30)}$ = 19.27, P ≤ 0.05].



#### Figure 6.7: Preference for novel object

Data are expressed as mean $\pm$ SE (n=3). Data were analyzed with one-way ANOVA and posthoc Tukey's HSD test (P $\leq$ 0.05). Here, C=Control, CE= *G. lucidum* HWE fed control, H=hypercholesterolemic, HE=*G. lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= *G. lucidum* HWE fed AD rats, respectively.

# 6.3.2 Neuro - biochemical and neuro - immunological tests of memory and learning related parameters

As depicted in the Table 6.2, the HWE of *G. lucidum* had ameliorating effect upon the memory and learning related neuro-biochemical and neuro-immunological markers of the rats such as BDNF, SNAP 25, PSD 95 and VAchT. On the other hand, decreased levels of anti-A $\beta$  (1-42) oligomers and TNF $\alpha$  levels were observed in all the rat groups treated with *G. lucidum* HWE (Table 6.2) (Raw data: Appendix B, Table B 8.1; Statistical data: Appendix C, Tables C - 8.1 and 8.2).

% of control	CE	Α	AE	Н	HE
Anti-Aβ	85.30 ±	$141.85 \pm$	119.80 ±	$107.03 \pm$	$102.56 \pm$
(1-42)	0.49 <sup>a</sup>	$0.60^{b}$	0.44 <sup>c</sup>	0.35 <sup>d,e</sup>	$0.50^{d,e}$
oligomer		C.			
BDNF	$109.57 \pm$	84.00 ±	90.43 ±	95.72 ±	$103.60 \pm$
	0.91 <sup>a,e</sup>	1.78 <sup>b,c</sup>	1.74 <sup>b,c,d</sup>	$1.03^{b,c,d}$	0.29 <sup>a,e</sup>
SNAP 25	$104.80 \pm$	85.33 ±	89.16 ±	93.25 ±	$94.09 \pm$
	0.71 <sup>a</sup>	$0.69^{b}$	0.57 <sup>c</sup>	$0.63^{d,e}$	$0.78^{d,e}$
PSD 95	$102.44 \pm$	91.56 ±	94.23 ±	94.83 ±	95.57 ±
	1.25 <sup>a</sup>	0.94 <sup>b,c,d,e</sup>	$0.42^{b,c,d,e}$	$0.83^{b,c,d,e}$	0.98 <sup>b,c,d,e</sup>
TNFα	90.02 ±	$120.50 \pm$	$111.05 \pm$	$103.62 \pm$	97.56 ±
	0.80 <sup>a</sup>	1.34 <sup>b</sup>	0.83 <sup>c</sup>	1.00 <sup>d</sup>	0.90 <sup>,e</sup>
VAchT	$109.12 \pm$	91.98 ±	$106.04 \pm$	96.48 ±	$111.89 \pm$
	$1.27^{a,c}$	0.68 <sup>b,d</sup>	$0.57^{a,c,e}$	$0.87^{b,d}$	$1.13^{a,c,e}$

Table 6.2: Antibodies and enzyme linked immunosorbent assays (ELISAs)

Data are expressed as mean±SE (n=3). Data were analyzed with one-way ANOVA and posthoc Tukey's HSD test (P $\leq$ 0.05). Here, C=Control, CE= *G. lucidum* HWE fed control, H=hypercholesterolemic, HE=*G. lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= *G. lucidum* HWE fed AD rats, respectively. BDNF stands for brain derived neurotrophic factor, SNAP 25 for synaptosomal associated protein 25 KD, PSD 95 for postsynaptic density protein 95 KD, TNF $\alpha$  for tumor necrosis factor  $\alpha$  and VachT for vesicular acetylcholine transporter, respectively.

# 6.3.3 Effect of G. lucidum HWE on inhibition of Aβ (1-42) fibril formation

As depicted in the Figure 6.8, addition of 20  $\mu$ M of *G. lucidum* HWE could significantly (one sample student's t test, P  $\leq$  0.05) lower the A $\beta$  (1-42) fibrillation process *in vitro* (Statistical data: Appendix C, Table C 9.1).



**Figure 6.8: Fibrillation inhibitory effect of** *G. lucidum* **HWE** Data are expressed as mean $\pm$ SE (n=3). Data were analyzed with one sample t test (P $\leq$ 0.05). Here, A $\beta$ = A $\beta$  oligomers and A $\beta$ +HWE= *G. lucidum* HWE treated A $\beta$  oligomers, respectively.

# 6.3.3.1 TEM studies of the hippocampus

As depicted in Figure 6.9, infusion of A $\beta$  (1-42) caused profuse alteration in the AD rats' hippocampal neuronal structure and organization as compared with the controls. Prolonged dendritic projections in the hippocampal neurons of the control and those of the *G. lucidum* HWE fed rats (AE) had been observed while the AD rats had mostly degenerative, plaque like structures.



Figure 6.9: TEM view of the CA1 region of the rat hippocampus Where, C=Control, CE= G. *lucidum* HWE fed control, H=hypercholesterolemic, HE=G. *lucidum* HWE fed hypercholesterolemic, A=AD model rats and AE= G. *lucidum* HWE fed AD rats, respectively.

# 6.4 Discussion

# 6.4.1 Effect of intra-cerebroventricula Aβ<sub>1-42</sub> infusion

Impaired behavioral responses of the AD rats demonstrate that A $\beta$ 1-42, even at soluble form *apriori* towards aggregation, plaque formation and neurodegneration. May be, A $\beta$ 1-42 disrupted short term and/or long-term potentiation (LTP) in the hippocampus and caused impaired memory and learning functions (Freir *et al.*, 2001; Stepanichev *et al.*, 2003; Stepanichev *et al.*, 2006). Also, the soluble form of A $\beta$ <sub>1-42</sub> itself might evoke impaired neuronal activities and neurodegneration that negatively

affected the A $\beta_{1-42}$  infused rats' learning and memory related behavioral performance (Townsend *et al.*, 2006).

#### 6.4.2 Enhancement of memory and learning by G. lucidum HWE

# 6.4.2.1 Eight-armed radial maze study

Feeding of *G. lucidum* HWE to the AD rats improved their memory and learnin abilities (Figures 6.4, 6.5 and 6.6). The stated claim is substantiated by the fact that DMTS free-choice is an important aspect of memory performance in the RAM (Stolberg, 2005). Gradual increased win-shift behavior of the *G. lucidum* HWE fed rats while the win-stay behavior in their non-fed counterparts was noted. These findings represent the improved associative memory of the extract fed rats as they preferred the food pellet associated arms to those of non-associated. Perhaps, the extract fed rats developed their cognitive maps better and extrapolated the location swifter and much precisely than those of the non-fed ones. As a whole, ingestion of *G. lucidum* HWE effected in improving spatial memory of the rats as evidenced by the decreased rate of RME, WME and maze latency in the extract fed rats effect (Figures 6.4, 6.5 and 6.6).

Perhaps, this is the first RAM report entailing *G. lucidum* HWE mediated memory and learning related behavioral amelioration of the AD model rats. Even, RAM reports involving other mushroom species are also not available so far. Thus, comparative discussion with other RAM studies involving *G. lucidum* and/or edible-medicinal mushrooms could not be incorporated in this section. However, RAM based AD ameliorating studies of multiple natural products have been conducted in different laboratories of the world.

Sweeney *et al.* (2007) conducted the delayed matching-to-sample (DMTS) freechoice type of RAM studies on intra-hippocampal  $A\beta_{1-40}$  infused rats and found that the animals' short-term memory and other memory related performance had been disrupted. Non-matching to sample (NMTS) type of RAM studies by Stepanichev *et al.* (2003) reported that intra-cerebrovascular infusion of A $\beta_{25-35}$  impairs short-term memory of the rats while the long-term memory remains unaffected. Their subsequent studies linked A $\beta_{25-35}$  driven neuro-degeneration in the CA1 region of the hippocampus with impaired memory in the experimental animals (Stepanichev *et al.*, 2006). Matching-to-sample (MTS) type of RAM-based memory tests by Holscher *et al.* (2007) had identified the memory related behavioral disruption in the A $\beta_{25-35}$ -infused rats. This group remarked that though intra - cerbroventricular infusion of A $\beta_{25-35}$  disrupted short-term memory and working memory in the rats, this status could be ameliorated and even reverted back to the normal (Holscher *et al.*, 2007). RAM based studies of AD amelioration by docosahexaenoic acid (DHA) and madecassoside had been reported by Hashimoto *et al.* (2005) and Hossain *et al.* (2014), respectively.

Hypercholesterolemia decreases cholinergic neurons and acetylcholine in the basal nucleus of brain and stimulates A $\beta$  formation and plaque deposition (Ullrich *et al.*, 2010). Hypercholesterolemia has also been implicated with neuro-inflammation and abnormal APP processing along with increased WME in mice (Thirumangalakudi *et al.*, 2008). In the current study, increased RME and WME in the hypercholesterolemic (H) rats might be due to this reason and compatible with other RAM studies (Ehrlich *et al.*, 2012; Granholm *et al.*, 2008). In line with this, gradual decreased rate of RME and WME in the *G. lucidum* HWE fed hypercholesterolemic (HE) rats indicate this mushroom's potentiality in reducing hypercholesterolemia - led memory dysfunction. In this connection, observed *in vitro* and *in vivo* hypocholesterolemic effect of this mushroom (chapter 5), in combination with memory enhancing effects, conform to the AD ameliorating potency of *G. lucidum*.

# 6.4.2.2 Novel object recognition test

Novel object recognition (NOR) is an important test for evaluating both short – and long-term memory. Present findings of higher preference and exploration of the *G. lucidum* HWE fed rats towards the novel objects indicate this mushroom's memory and learning ability improving effects (Figure 6.7). To the best of our knowledge, this is the first study incorporating *G. lucidum* in NOR test. However, studies incorporating *H. erinaceus* upon the AD mice models has reported improved spatial recognition and visual memory of the AD mice models prepared like those of the present study (Mori *et al.,* 2011). Memory improving effect has been deduced to be *inter alia* through increasing mossy fiber CA3 hippocampal neurons in the mice (Brandalise *et al.,* 2017). Similar mode of action might have been prevalent in case of the *G. lucidum* HWE fed rats showing higher preference for the novel objects and higher exploration of the environment. NOR is analogous to human declarative (spatial or episodic) memory (Figure 2.1) and based on the current memory improving findings, therapeutic potentiality of G. *lucidum* HWE seems promising in treating human AD subjects.

### 6.4.3 Antibodies and enzyme linked immunosorbent assays (ELISAs)

# 6.4.3.1 Aβ (1-42) oligomer lowering effect of G. lucidum HWE

As  $A\beta_{(1-42)}$  is the culprit for memory disruption, its measurement in the brain homogenate is apparent. Thus, the level of  $A\beta_{(1-42)}$  oligomer was measured using antibody against it in the DSF of the rat hippocampus as  $A\beta_{(1-42)}$  oligomer is a soluble protein. Significantly increased level of  $A\beta_{(1-42)}$  oligomer in the AD rats (41% higher than those of the controls) were found in the present study (Table 6.2) (Raw data: Appendix B, Table 8.1; Statistical data: Appendix C, Tables 8.1 and 8.2). This may be due to the combined effect of increased generation *per se* and deposition of those of through cerebro-ventricular infusion. On the contrary, significantly decreased (P  $\leq$  0.05) level of  $A\beta_{(1-42)}$  oligomer was observed in the *G. lucidum* HWE fed rats (lowered up to 19%) (Table 6.2). Although, hypercholesterolemia led 7% increased generation of  $A\beta_{(1)}$ 42) oligomer, feeding of the G. lucidum HWE could lower this level up to 2% (Table 6.2). This lowering of A $\beta_{(1-42)}$  oligomer burden in the AD and hypercholesterolemic rats' brain are indication of AD ameliorating effect of G. lucidum. Among multiple strategy, G. lucidum mediated decreased biosynthesis of cholestersol might be responsible for generating comparatively lowered level of  $A\beta_{(1-42)}$ . Content of phenolics in the HWE of G. lucidum might also contribute to this effect through their inhibitory effect upon BACE1 (Leow et al., 2013; Wu et al., 2015). Ganoderma lucidum mediated stimulation of the P13K and ERK signaling cascades resulting in stimulation of the nonamyloidogenic pathway and enhanced secretion of sAPPa might also have been involved (Pinweha et al., 2008). Current findings are also compatible with those of Wang et al (2004), who found that G. lucidum powder at 0.3%, 0.6% and 1.8% of diet can significantly lower A $\beta$  burden in the mouse brain, improve memory and learning abilites along with enhanced anti - oxidative enzymatic levels. Thus, lowering of brain A $\beta$  burden along with increasing anti – oxidative defense might confer the AD ameliorating effect of the G. lucidum HWE.

# 6.4.3.2 Neurotransmission maintenance effect of G. lucidum HWE

Brain derived neurotrophic factor (BDNF) is a neuroprotectin group of growth factor involved in neuronal survival and functioning. Significantly lowered level of BDNF in the AD rats compared to the controls (84% compared with 100% of control value) was found in the present study (Table 6.2) (Raw data: Appendix B, Table 8.2; Statistical data: Appendix C, Tables 8.1 and 8.2). Feeding of *G. lucidum* HWE resulted in increased BDNF level both in the AD (from 84%, raised up to 90.43%) and the hypercholesterolemic (from 95.72%, incrased up to 103.60%) rats and the increasing effect was significantly high in the latter ( $P \le 0.05$ ) (Table 6.2).

BDNF participates in both pre- and post-synaptic neurotransmission by binding with the tyrosine kinase receptor B (TrkB) and thus plays important role in memory and learning activities (Lu, 2003). Its impairment has been found to affect memory related learning and behavioral performances in different organisms and in different conditions (Tyler et al., 2002). AD patients possess considerably lower amount of BDNF (Shin et al., 2015). Deteriorated memory emanating from intra-cerebroventricular infusion of Aβ had been reported to be due to the lowering level of BDNF (Balducci et al., 2010). Learning activities increases BDNF level in the rodent hippocampus and cortex (Cotman & Berchtold, 2002). As described in the behavioral test section (6.4.2), the G. lucidum HWE fed rats performed better than the non-fed ones in the learning tests and the mushroom fed rats were supposed to contain higher level of BDNF. Observed neuro - biochemical findings are compatible with those of the behavioral test findings of the present study. This claim is substantiated by the finding that BDNF not only forms spatial memory but also regulates retention and recalling of it (Mizuno et al., 2000). Previous studies had shown that the triterpenoids present in G. lucidum provide BDNF enhancing effect (Zhang et al., 2011). In the current study, several tri-terpenoids have been detected in the HWE of G. lucidum (chapter 4, Table 4.5) that might have been involved in increasing the BDNF level of the mushroom fed rats' hippocampus and thus improved their memory and learning abilities. In addition, phenolics present in the HWE of G. lucidum might also confer effects towards increasing the hippocampal BDNF level (Leow et al., 2013; Zhao et al., 2013).

# 6.4.3.3 Maintenance of pre-synaptic membrane and long-term potentiation by *G*. *lucidum* HWE

Synaptosomal - associated protein 25 KD (SNAP 25) is a pre-synaptic membrane protein necessary for long – term potentiation (LTP) and working memory (Söderqvist *et al.*, 2010). Its decreased level had been noticed in AD subjects (Greber *et al.*, 1999).

Compared with the controls, significantly lowered ( $P \le 0.05$ ) level of SNAP in the hippocampus of the AD rats was observed (Table 6.2) (Raw data: Appendix B, Table 8.3; Statistical data: Appendix C, Tables 8.1 and 8.2). In case of *the G. lucidum* HWE fed rats, SNAP 25 level increased significantly ( $P \le 0.05$ ). Previous studies indicate that spatial recognition is mediated by SNAP 25 and rats' performance in MWM had been found to be negatively affected with the anti-SNAP 25 antisense oligonucleotide (Hou *et al.*, 2004). A $\beta$  leads towards loss of synaptophysin, another pre – synaptic protein, in the caspase – dependent way (Liu *et al.*, 2010). Thus, improved spatial recognition of the *G. lucidum* HWE fed rats in different spatial memory and learning related behavior tests (RAM and NOR) in the present study, might be attributed by the enhanced SNAP 25 level in the respective rats.

# 6.4.3.4 Maintenance of post – synaptic density by G. lucidum HWE

Post-synaptic density protein 95 KD (PSD – 95) is involved in maturation of the excitatory synapses and in maintenance of post – synaptic density (Chen *et al.*, 2011). It is a scaffold protein that takes part in regulating NMDA receptor mediated signaling processes (Sheng, 2001). Increased A $\beta$  accumulation and impaired memory have been reported to be associated with its disruption and AD patients exhibit lowered level of PSD 95 in the hippocampus (Sultana *et al.*, 2010a). Decreased level of PSD 95 in the hippocampus (Sultana *et al.*, 2010a). Decreased level of PSD 95 in the hippocampus (Sultana *et al.*, 2010a). Decreased level of PSD 95 in the AD and hypercholesterolemic rats (94.43% and 95.57%, respectively) compared with the control (100%) were noted in the present study (Table 6.2) (Raw data: Appendix B, Table 8.4; Statistical data: Appendix C, Tables 8.1 and 8.2). This might be due to A $\beta$  guided loss of PSD 95 in the caspase – dependent way (Liu *et al.*, 2010). However, feeding of *G. lucidum* HWE to the rats resulted in increased level of PSD-95 in the hippocampus. Thus, the improved memory and learning abilities of the *G. lucidum* HWE fed rats in the present study might be accrued from *inter alia* increased PSD 95

level in the hippocampi of the mushroom fed rats. Among multiple strategies, the antioxidative defense mechanism might be involved (Gray *et al.*, 2016).

# 6.4.3.5 Neuro-inflammation lowering effect of G. lucidum HWE

Neuro-inflammatory mode of AD pathogenesis involves increased level of tumor necrosis factor alpha (TNF $\alpha$ ) as evidenced by cellular and animal model studies as well as in AD patients (Álvarez et al., 2007; Bhaskar et al., 2014; Lourenco et al., 2013). TNF $\alpha$  acts as a neurotoxin and thus AD amelioration strategy points towards its reduced level (McAlpine et al., 2009). As depicted in Table 6.2, the AD rats had significantly (P  $\leq 0.05$ ) higher level of TNFa (120.5% compared to 100% of the controls) (Raw data: Appendix B, Table 8.5; Statistical data: Appendix C, Tables 8.1 and 8.2). Feeding of G. lucidum HWE significantly decreased (up to 1116.05%) TNFα level. Similar effect was observed in case of the mushroom fed hypercholesterolemic rats (lowered from 103.62% to 97.60%). At 400 µg/mL, the methanolic extract of G. lucidum had been found to significantly lower the production of microglial  $TNF\alpha$  and prevents the dopaminergic neurons (Zhang et al., 2011) . At the same dosage, it could lower the expression of TNFa mRNA up to 90% (Zhang et al., 2011) . Presence of phenolic antiinflammatory and anti-oxidant substances in the G. lucidum HWE might also cause TNF $\alpha$  lowering effect (Frautschy *et al.*, 2001). Content of  $\beta$ -D glucan, triterpenoids and polysaccharides had been supposed to confer this protecting effect of G. lucidum (Zhang et al., 2011) . Ganoderma lucidum polysaccharide (GLPS) administered to the traumatic spinal cord injured rats also showed neuroprotective effect through lowering TNFα level along with reduced production of MDA (Gokce et al., 2015). In addition to the TNFa lowering effect, MDA lowering effect of the HWE of G. lucidum was also observed in the present investigation (chapter 3, section 3.2.6 and 3.2.7). Thus, the current findings are compatible with those published ones as similar lowering effect of *G. lucidum* HWE upon TNF $\alpha$  level (Table 6.2) and presence of similar bio-components have been observed in the present study (Table 4.5).

# 6.4.3.6 Enhancement of cholinergic neurotransmission by G. lucidum HWE

Vesicular acetylcholine transporter (VAchT) transports acetylcholine (ACh) from the presynaptic membrane of the cholinergic neurons into the secretory vesicles (SV), from where ACh is released into the synaptic cleft (Chen et al., 2011). Activity of VAChT highly regulates cholinergic neurotransmission and VAChT deficiency leads towards neuromuscular abnormalities (Rodrigues et al., 2013). Impaired cognitive function in AD subjects had been linked with reduced VAChT level (Chen et al., 2011). In the present study, significantly lower (P  $\leq 0.05$ ) level of VAChT was noted in the AD model rats (91.98% in AD, compared to 100% for the controls) while the G. lucidum HWE fed rats had been observed with significantly higher level of VAChT (106.04%) (Table 6.2) (Raw data: Appendix B, Table 8.6; Statistical data: Appendix C, Tables 8.1 and 8.2). Reduced level of AChT in the hypercholesterolemic rats (96.48%) brings testimony towards hypercholesterolemia induced impairment of cholinergic neurons and VAChT activity (Ullrich et al., 2010). Thus, memory and learning related poor performances of the AD and the hypercholesterolemic rats may be due to inter alia disrupted VAChT activity and diminished cholinergic neurotransmission (Prado et al., 2006). In line with this, the ameliorated memory and learning related performances of the G. lucidum HWE fed rats might be the offshoot of ameliorated VAChT activity of the cholinergic neurons in the respective rats.

# 6.4.4 Inhibition of A $\beta$ (1-42) oligomerization and fibrillation by *G. lucidum* HWE

Agents capable of lowering and/or inhibiting A $\beta$  fibrillation seem promising as the therapeutics against AD pathogenesis (Stefani & Rigacci, 2013). As evidenced by the improved memory and learning abilities and neuro-biochemical and immunological

tests, it is plausible that the HWE of *G. lucidum* would also have inhibitory effect upon  $A\beta$  fibrillation. This surmise is based on the findings that the natural polyphenols are potent in withstanding  $A\beta$  fibrillation (Porat *et al.*, 2006; Stefani & Rigacci, 2013). The anti-oxidative mode of action has been assumed to be capable of lowering  $A\beta$  fibrillation process also (Porat *et al.*, 2006). In this context, the phenolics and polyphenolics present in the HWE of *G. lucidum* (chapter 4, Table 4.5) might have shielded  $A\beta$  (1-42) fibrillation in the present study through their anti-oxidative mechanism. In addition, structural similarities of the polyphenolics with that of the amyloid  $\beta$  sheet might also confer resistance towards fibrillation (Porat *et al.*, 2006). Catechol types of flavonoids (protocatechuic acid, p-hydroxy cinnamic acid, ellagic acid) (chapter 4, Table 4.5) present in the *G. lucidum* HWE might inhibit  $A\beta$  (1-42) fibrillation of the lysine residues (Rawat *et al.*, 2013; Sato *et al.*, 2013). However, other biochemical processes might also have been involved whose cumulative effect bear testimony to inhibitory effect of *G. lucidum* HWE up on  $A\beta$  fibrillation *en route* to the amelioration of AD pathogenesis.

# 6.4.4.1 TEM studies of the hippocampus

TEM studies featuring hippocampal neurons in their native state and pronged dendritic projections as well as absence of degenerative neuronal plaque like structures in the *G. lucidum* HWE-fed rats compared with those of the control and AD subjects are of immense importance (Figure 6.9). This is because neuronal degeneration is an important hallmark of AD. The memory and learning related outcomes of the *G. lucidum* HWE presented in the current study might be attributed *inter alia* to the prevention of neuronal degeneration also. Previous studies have shown the *G. lucidum* spore could reduce neuronal apoptosis in the hippocampus and thus prevent neuronal loss and aid in cognitive performance (Zhou *et al.*, 2012). Antagonization of the A $\beta$  – induced neurotoxicity of the aqueous extract of *G. lucidum* had also been reported (Lai

*et al.*, 2008). Similar mode of action might have been engaged with the outcomes of the current experiment. Also, anti-oxidative mode of neural protection against A $\beta$  and hypercholesterolemia – induced oxidative stress might be functional (Zhao *et al.*, 2004). Content of polyphenolic substances in the HWE of *G. lucidum* might be implicated in this function (chapter 4, Table 4.5) (Lakey-Beitia *et al.*, 2015). Besides, anti-inflammatory effect of *G. lucidum* tri-terpernoids might also confer the neuro-protective effect of this mushroom (De Silva *et al.*, 2013; Dudhgaonkar *et al.*, 2009). Present findings are also compatible with those of Yongpan *et al.* (2016), whose TEM based histopathological studies showed that the *G. lucidum* tri – terpenoids at 0.25, 0.5 and 1 g/kg body weight could improve neuronal degeneration in a dose dependent manner Yongpan *et al.* (2016).

### 6.5 Conclusion

Feeding of *G. lucidum* HWE to the rats improved their memory and learning abilities. Improved cognitive performance of the *G. lucidum* HWE fed rats in the behavioral (eight armed radial maze and novel object recognition tests) and antibody and ELISA based tests of neurotransmitters were observed. TEM studies also provide support towards neuro-protective effect of *G. lucidum* HWE. All of these findings and those described in the previous chapters bear testimony towards the AD ameliorating effects of the *G. lucidum* HWE. As diverse range of proteins are involved in the maintenance of memory and learning abilities of an organism, studies concerning the status (up regulated / down regulated) of different proteins seem imperative in deciphering its memory and learning related amelioration. Thus, proteomic analysis, study and protein-protein interaction of the relevant rat groups (controlled, AD and the *G. lucidum* HWE fed AD rats) were performed whose description has been put forward in the next chapter.

# CHAPTER 7: PROTEOMICS-BASED ANALYSIS ON ALZHEIMER'S DISEASE AMELIORATING EFFECT OF *G. LUCIDUM*

# 7.1 Introduction

As proteins are involved in many aspects of memory and learning processes and their derangements affect Alzheimer's disease (AD) pathogenesis highly, proteomics approaches seem pertinent in elucidating the state of AD and also of its amelioration. Separation of proteins from the complex biological mixture is the foremost part of proteomics approaches. Later, comes the identification of the individual protein and/or peptide for establishing it as a biomarker. For the last few years, two dimensional gel electrophoresis (2DGE) had been most extensively used for separation of proteins (Görg *et al.*, 2004). Separation principle is based on the molecular weight (MW) and electric charge (pI) of individual protein. Under electric current, migration of each of the proteins in polyacrylamide gel depends on its MW and pI. Separated proteins are then visualized by staining with silver and identified using the mass spectrometry. Mass spectrometry identifies the mass (m) and electric charge (z) carried by each protein and peptide and produces mass-to-charge (m/z) data. Using standard software and databases, the obtained m/z values are compared and the protein and peptides are characterized and identified.

Currently, liquid chromatography coupled with tandem mass spectrometry (LC-MS/MS) is the most attractive technique for proteomics studies of human and animal organs and tissues. In this approach, after separating the proteins in SDS-PAGE, protease (very often with trypsin) catalyzed digestion produces peptides that are loaded onto capillary column for reverse phase liquid chromatography. Peptides are eluted sequentially with buffer of increasing gradient and high voltage applied (2.0-2.5 KV).
High temperature evaporates the acidic liquid into gas and H<sup>+</sup> ionizes the peptides (electrospray ionization, ESI) and the ionized peptides move in the electromagnetic field which is scanned and measured by the detector and their m/z ratio is measured. Besides ESI, matrix - assisted laser desorption/ionization (MALDI) is also applied that uses a dry matrix instead of the liquid solvent for peptide ionization by laser pulses. While MALDI is suitable for pure, large peptides and protein, ESI can be applied for complex peptides and has upper hand in usage (Cravatt *et al.*, 2007). In the present study, SDS-PAGE followed by LC-MS/MS strategy have been applied for identifying the hippocampal proteins associated with AD pathogenesis and amelioration in the control (C), AD and AE rat groups.

Proteomic approaches utilizing 2DE and western blots can trap only the intact proteins and the proteolytically cleaved peptides are left that affects entire analyses. Procedure applied in the current research can grab both the proteolytically cleaved peptides and the intact proteins and minimizes the loss of the low molecular weight peptides. For accuracy, precision and reliability of proteomics, incorporation of both biological and technical replicates are essential. However, in this study, the biological samples were pooled to average the protein level and thus control the sample-sample variation. Technical replicate was maintained at 3 and frequency at 9 to represent independent measurement of noise signal and instrument error as well as to substantiate the accuracy and confidence of protein identification and to minimize the sample variation.

#### 7.2 Materials and methods

#### 7.2.1 Brain sample preparation and protein quantification

#### 7.2.1.1 Brain sample collection

Rats were anaesthesized with pentobarbital, sacrificed and head removed followed by collection of brain on ice bath. Brains were frozen in liquid nitrogen and stored at – 80 °C. All the experimental protocols had been approved by the ethical permission committee, University of Malaya Institutional Animal Care and Use Committee (UMIACUC) [Ethics reference no. ISB/25/04/2013/NA (R)] (Appendix G).

#### 7.2.1.2 Protein extraction

Protein extraction from the brain samples was performed following homogenization of the brain sample (50 mg) with lysis buffer (1ml) using a homogenizer (Polytron PT 1200, Kinematica). To avoid protein degradation, we added 10  $\mu$ L of protease inhibitor cocktail during homogenization followed by centrifugation at 10000× g at 4 °C for 10 minutes. The supernatant was collected and proceeded towards delipidation.

# 7.2.1.3 Delipidation procedure

As the brain is a fatty tissue, lipids may hamper protein separation and quantification steps and thus delipidation is an essential prerequisite for brain tissue proteomics. Prior to delipidation, 100  $\mu$ L of the brain supernatant was dried using speedvac. Delipidation procedures were conducted following the methods of Shevchenko *et al.* (2010). Briefly, 1.4 mL of ice-cold tri-n butyl phosphate / acetone / methanol mixture (1:12:1) was mixed with 100  $\mu$ L of extract and incubated for 90 min at 4°C. Centrifugation of the mixture at 2800× g at 4 °C was performed for 15 minutes that pelleted the precipitate. Later, sequential washing was performed with 1 ml each of TBP, acetone and methanol. Last of all, the pellet was air dried and proceeded for protein quantification and protein separation through SDS-PAGE.

### 7.2.1.4 Protein quantification

Content of total protein in the delipidated brain samples was determined using the Pierce BCA protein assay kit (Thermo scientific, 23225). Bovine serum albumin (BSA) was used as the standard and manufacturer's directions followed for protein quantification (Section 5.3.4.7 and 6.3.5.2 B; Appendix A: Figure A 5.1; Table A 7.8).

## 7.2.2 Protein separation through SDS-PAGE

In SDS-PAGE, proteins are separated based on their charge and molecular weight. Proteins become denatured by SDS and attain negative charges proportionately with their masses that render them to move to the anode end. Their movement also is imparted by their respective weight: the less weighty proteins moving faster than those of the higher. The mini-PROTEAN tetra cell (165-8000, BIO-RAD, USA) was used according to the manufacturer's instructions for running SDS-PAGE in the current study.

#### 7.2.2.1 SDS-PAGE assemblage

A set of vertical glass plate gel units were assembled with 0.75 mm spacer strips at the edges. Solution leakage was prevented by clamping the glass plates tightly and poured separating gel solution into the space by micropipete. Ultra-pure water was overlaid the gel top with ultrapure water and allowed the gel to polymerize for about two hours. After polymerization, the overlaying water was discarded and the stacking gel was casted. The separating gel surface was cautiously rinsed with the stacking gel for removing unpolymerized and residuous gel and poured the stacking gel (Appendix D, section 1: Preparation of solutions, buffers and gels). A comb from top-side was inserted immediately into the gel. Care was taken not to trap any bubble at the inner tip of the comb and we let the gel to polymerize for about one hour. After polymerization, the gel was used immediately.

#### 7.2.2.2 Loading sample preparation

The sample buffer (4×), previously prepared and stored at 4 °C, was allowed to equilibrate to room temperature (Appendix D, section 1: Preparation of solutions, buffers and gels). Mixing the sample buffer and rat brain homogenate (1:3) was followed by heating the mixture at 100 °C for 5 minutes adding about 5  $\mu$ L of trypan blue.

#### 7.2.2.3 Running of SDS-PAGE

The shorter glass plates had been clamped inward of the chamber and placed the glass units into the vertical electrophoretic tank. Running buffer was poured at both upper and lower chambers and loaded the (cooled down to room temperature and mixed well) samples ( $12 \mu L$ ,  $7 \mu g$ ) and protein marker ( $4 \mu L$ ) into the SDS-PAGE wells (Appendix D, section 1: Preparation of solutions, buffers and gels). Placing the lid onto the tank and plugging it properly, electrophoresis turned on. Electrophoresis was allowed at 80 V for the initial 20 minutes, at 100 V for the second 20 minutes and for the rest of the time at 120 V so that excess heat could be avoided. Movement of the blue dye was observed while electrophoresis going on and when the dye had crossed about 1 cm from the gel's bottom-layer, electrophoresis was turned off and the apparatus disassembled.

## 7.2.2.4 Staining of the gel

Removing from the glass plate, the gel was fixed by immersing it into fixing solution (mixture of 10% acetic acid and 40% methanol) and letting occasional stirring for 30 minutes. Separating the gel from the fixing solution, it was immersed in another container containing 0.1% coomassie brilliant blue with shaking for about 20 minutes (Appendix D, section 1: Preparation of solutions, buffers and gels).

### 7.2.2.5 Destaining of the gel

For destaining, 10% acetic acid solution (aqueous) was used. The gel immersed into the destaining solution contained in a covered box underwent occasional stiring until the entire gel was fully destained. Then, individual bands was cut and if any gel plug still contained stain, repeated shaking of the gel plugs in 50  $\mu$ L of 50% acetonitrile (ACN) in 50 mM ammonium bicarbonate was continued (Appendix D, section 1: Preparation of solutions, buffers and gels).

# 7.2.3 Preparation of the gel for protein identification

#### 7.2.3.1 In-gel tryptic digestion

Effective protein identification and characterization by MS requires several preparatory steps following separation of the proteins. Digestion of the large proteins into smaller fractions and peptides is the initial step that was performed for in-gel plugs. Digestion involved preparatory phases of reduction, alkylation and dehydration of the proteins in gel plugs.

# 7.2.3.2 Reduction and alkylation:

Under non-denaturing conditions, most of the proteins are resistant to enzymatic proteolysis. Thus, solubilization of the hydrophobic proteins is performed using detergent-chaotrope mixture added lysis buffer with sonication and vigorous vortexing. For disrupting the tertiary structures of the solubilized proteins, reduction and alkylation are applied to them so that the disulfide linkages are broken and cannot be re-formed. Reducing agents (dithiothreitol, 2-mercepatoethanol) convert disulfide bond of cysteine into free sulfhydryl groups. Reacting with the free sulfhydryl groups of cysteine, alkylating agents (iodoacetamide, iodoacetic acid) form S-carboxyamidomethyl-cysteine that cannot be reoxidized and disulfide bonds are not formed. For reduction, the gel plugs were incubated in 150  $\mu$ L of 10 mM dithiothreitol (DTT) in 100 mM

ammonium bicarbonate buffer at 60 °C for 30 minutes (Appendix D, section 1: Preparation of solutions, buffers and gels). After cooling at room temperature, the gel plugs were alkylated by incubating in 150  $\mu$ L of 55 mM iodoacetamide (IAA) in 100 mM ammonium bicarbonate for 20 minutes in the dark chamber. The gel plugs were washed in triplicate with 500  $\mu$ L of 50% ACN in 100 mM ammonium bicarbonate for 20 minutes. For dehydration of the gel plugs, shaking in 50  $\mu$ L of 100% ACN for 15 minutes was performed followed by drying the gel plugs in speed vacuum for 30 minutes at 4 °C.

### 7.2.3.3 Digestion

Usually, the denatured, reduced and alkylated proteins are subjected to enzymatic or chemical digestion. For enzymatic digestion, trypsin is most commonly used as it specifically cleaves at the carboxyl terminal end of arginine and lysine and produces peptide fragments containing suitable length and charge for MS analysis. The gel plugs were incubated with 25  $\mu$ L of 6 ng/  $\mu$ Ltrypsin in 50 mM ammonium bicarbonate at 37 °C overnight.

## 7.2.3.4 Extraction

After overnight digestion, the digested products were spun down by vortexing and transferred liquid to the fresh tubes. Adding 50  $\mu$ L 50% ACN to the tubes, shaking was continued for 15 minutes. Then, the gel plugs were incubated with 50  $\mu$ L of 100% ACN and shook for another 15 minutes. Transferring the liquid to the previous tubes, the digested samples were completely dried using the speed vacuum at 1000 rpm. The dried tubes were stored at - 80 °C for desalting and zip tip procedures (Appendix D, section 1: Preparation of solutions, buffers and gels).

## 7.2.3.5 Desalting and Zip tipping procedure

Desalting and Zip-tipping of the samples were performed using the respective solutions mentioned in (Appendix D, Section 2: Preparation of desalting and zipping solutions) and followed the steps as follows

**Reconstitution of protein samples:** The protein samples were reconstituted adding 10  $\mu$ L of 0.1% formic acid, vortexing and spinning down the samples.

Wetting: 10 µL of wetting solution was aspirated and dispensed three times.

**Equilibration:** For equilibrating the tips,  $10 \ \mu L$  of equilibrium solution was aspirated and dispensed for three times.

**Protein binding:** By pipetting in and out, 10  $\mu$ L of protein samples was passed through the zip-tips for 10 cycles. Repeatation of the step was performed without letting the tip to be dried.

Washing: For three cycles,  $10 \ \mu L$  of washing solution was aspirated and dispensed.

**Elution:** Adding 4  $\mu$ L of elution solution into zip-tip and dispension was performed for three times. Finally, the content was dispensed in the protein sample solution.

# 7.2.4 LC-MS/MS Q-TOF quantification

All the MS/MS instruments and software used in the present section of the study were from Agilent (Agilent, Santa Clara, CA, USA). Eluted sample obtained from zip tip procedure was dried and  $10\mu$ L of the lyophilized samples was reconstituted in the first LC mobile phase (0.1% formic acid) in triplicate. The peptides with a Nano-LC 1260 linked directly with an Accurate Mass Q-TOF 6550 containing a Chip-Cube interface Nano-ESI ion source. Polaris High Performance Chip was utilized and enriched the peptides using 360 nl enrichment column followed by their separation using the separation column (C18 reverse phase, 150mm x 75ŵm, 5 µm) with solvent A (0.2% formic acid in water) and a 5-80% gradient of solvent B (0.1% formic acid in acetonitrile) for 34 min with a flow rate of 0.35 µL/min. Mass data acquisition was undertaken at 8 spectra/second in the range of 100-200 m/z and subsequent collision induced dissociation (CID) of the twenty most intense ions. Setting the mass-tolerance of precursor and product ions at 20, MS/MS data acquisition was performed in the range of 200-3000m/z. In order to identify the proteins, the acquired MS/MS data were compared against the UniProtKB/Swiss Prot rat (Rattus norvegicus) database using the Spectrum Mill and X! Tandem. The differentially expressed proteins in the different groups were identified using their canonical sequence and proteins having fold change of at least 1.5 times were considered as the deregulated proteins. For validation of the identified proteins, the data were exported to the Scaffold database (version 4.5.1, Portland, USA). Proteins were grouped together if they would share at least two peptides and maintained their threshold level at 95.0% and <1% false discovery rate (FDR) by the Peptide Prophet algorithm with Scaffold delta - mass correction for the matched peptide-spectra. Proteins that contained similar peptides and could not be differentiated based on MS/MS analysis alone were grouped to satisfy the principles of parsimony. Proteins sharing significant peptide evidence were grouped into clusters. Proteins were annotated with GO terms from NCBI.

In the present experiment, label-free relative quantification was performed depending on the regulation of the peptides. For statistical analyses, the data were exported to the Mass Profiler Professional (MPP) software that analyzed depending on the MPP entities, the intensity of the total spectra of the proteins. Setting the baseline of the spectra to the median of the samples, frequencies of the entities were filtered minimally at all the replicates of each treatment. To overcome the complications of false discovery associated with multiple test analyses, ANOVA (P<0.05) was performed.

164

# 7.2.5 Bioinformatics and analysis of protein-protein interaction (PPI)

Most of the proteins do not work singly rather they participate in complex network or scaffold and interact with others. Thus, analysis of the relevant protein-protein networks provides important information in deciphering any bio-molecular system. Functional interaction networks of the proteins were identified using the STRING (Search Tool for the Retrieval of Interacting Genes/Proteins) database (version 10.0; <u>http://string-db.org/</u>). STRING displays protein-protein interactions in a large network of connectivity and protein hubs. Active prediction methods that we used were experiments, neighborhood, databases, gene fusions, coexpression, cooccurrence and text mining, using high confidence (0.7).

For further identifying over-representing pathways and biological functions, the ingenuity pathway analysis (IPA), build version: 389077M, content version: 27821452, (Release date: 2016-06-14) was used (<u>https://www.ingenuity.com/wp-content/themes/ingenuity-qiagen</u>). Datasets of the proteins significantly expressed (P<0.05) and having log fold change of 1.5 and higher were uploaded (AD versus C, AD versus AE and C versus AE). Analysis setup was as follows

Criteria	Set up
Reference set	Ingenuity Knowledge Base (Genes Only)
Relationship to	Direct and Indirect. Includes Endogenous Chemicals.
include	
Optional	My Pathways My List
Analyses	
Filter Summary	Consider only molecules and/or relationships where
	(species = Human) and (data sources =
	ClinicalTrials.gov or ClinVar or Ingenuity Expert
	Findings or Ingenuity ExpertAssist Findings)

#### 7.3 Results

Total 822 proteins with protein threshold at 95.0%, minimum peptide of 2 and peptide threshold at 0.1% FDR were identified in all the three groups (AD versus C, AD versus AE and AE versus C). Number of commonly expressed proteins among the three groups was 361 (Figure 7.1B). Among all the identified proteins (822), 329 were differentially expressed with statistical significance (P<0.05). Among the significantly regulated (P<0.05) 329 proteins, 289 met the criteria of fold change (LogFC of 1.5) cut off value (Appendix E, Table E 10.1). Number of proteins linked with AD, OS and hypercholesterolemia was 59, 20 and 12, respectively.



A. SDS-PAGE image: L1 - Control, L2 - AD and L3 – AE.



## Figure 7.1: A) SDS-PAGE image and B) Venn diagram of the identified proteins.

The differentially expressed (significantly up – and/or down – regulated at P<0.05) proteins were grouped with respect to their established functional links such as response to stimulus, memory and learning related neuro-synaptic function, metabolic processes, maintenance of cytoskeleton, immunological response, transportations and apoptosis.



Figure 7.2: Gene ontology (GO) analysis featuring biological processes of the identified hippocampal proteins.

The highest amount of proteins differentially expressed in the AD and mushroom fed AD (AE) rats were those involved in metabolic processes (26% increase in AD and 29% in the AE rats) (Figure 7.2). Extent of differential expression in case of the proteins involved in response to stimulus, multicellular organismal processes and developmental processes was very close together ( $20\pm2\%$ ) (Figure 7.2). Molecular function based categorization showed that proteins involved in the transporter and enzyme regulatory activities were of the most differently expressed types followed by those of molecular transducing and anti-oxidant activities (Figure 7.3). Cellular component based analyses identified the proteins located at the cytoplasm, cellular organelle and membrane as the most differentially expressed ones (Figure 7.4).



Figure 7.3: Gene ontology (GO) analysis featuring molecular functions of the identified hippocampal proteins



Figure 7.4: Gene ontology (GO) analysis featuring cellular localization of the identified hippocampal proteins

#### 7.4 Discussion

#### 7.4.1.1 Functional classification of the significantly regulated proteins

As the significantly regulated proteins are involved in diversified functions, discussions related to them have been appended below under sub-sections of their commonly known functions.

#### A. Proteins involved in neuronal structure and function

Differential expression of the proteins involved in neurotransmission, synaptic plasticity, neurogenesis, memory and learning related proteins such as neurochondrin, synaptophysin, synapsin-1, synapsin-2, synaptogyrin 3 (Syngr3), 4-aminobutyrate aminotransferase and 14-3-3 protein gamma were observed in the different rat groups. Associative learning and long-term memory related proteins glutathione-S-transferase 3 and tenascin R were up regulated. Synaptic plasticity promoting Ras related protein Rab 5a and nerve growth factor (NGF) signaling Rap-1A were also among the significantly up-regulated group. Similar was the case for the heat shock proteins (HSP) involved in the regulation of neuronal migration (HSP 90-alpha) and apoptosis (HSp 60). Up-regulation was also observed for the proteins involved in post-synaptic excitatory potential (serine/threonine-protein phosphatase, syntaxin 1B), pre- and post-synaptic density (isoform 2 of clathrin coat assembly protein AP180) and tyrosine phosphorylation (hemopexin) in the AD compared to the mushroom-treated (AE) group.

Proteins involved in synaptic organization (neurofascin) and synaptic vesicle budding (ADP-ribosylation factor 1), vesicle mediated transport (syntaxin 1A), neuronal differentiation and development (Dihydropyrimidinase-related protein 1 and 2), axonogenesis (2', 3'-cyclic-nucleotide 3'-phosphodiesterase), axonal choice point recognition (neuromodulin) and axonal transport (neurofilament light polypeptide) were also differentially expressed in the hippocampus of the three rat groups. Beta-soluble NSF attachment protein (SNAP- $\beta$ ) involved in the regulation of glutamatergic synaptic transmission, disassemble of SNARE complex and synaptic vesicle priming was also up regulated. In addition to these proteins, differentially up regulated expression of glial fibrillary acidic protein (GFAP) was also observed. GFAP is involved in long-term synaptic potentiation, neurotransmitter uptake, neurogenesis, glial and Schwan cell proliferation.

In the present study, down-regulated expression of memory and learning related proteins in the AD group was observed when compared with both C and AE group. However, there was inter-group variation in the extent of fold change in case of the down-regulated proteins. In the AD versus control group, down regulated expression of the proteins involved in dopamine decarboxylation, clusterin (stimulator of A $\beta$  and NFT), neuromodulin, neurofascin and NCAM 1 was observed. In the AD versus AE groups,  $\alpha$ -synuclein, synaptogyrin 1 and transgelin-3 were among the most important down regulated proteins. In addition to these, neuromodulin, excitatory amino acid transporter and park 7 were the mostly up regulated proteins in the AE versus C groups.

Following are the AD related proteins differentially expressed in the present study

**Syntaxin-1A:** Syntaxin-1A regulates vesicular trafficking during exocytosis and trans-membranal protein insertion (Sudhof, 2004). Decreased expression of syntaxin-1 A in the AD rats might have affected synaptic functions (Shevchenko *et al.*, 2012). Mushroom treatment might have synaptic function improving effect as up-regulated expression of this protein has been observed in the mushroom-treated group.

**Synaptogyrin-1:** Synaptogyrin-1 is involved in maintaining short- and long-term synaptic plasticity. Level of hippocampal syntaxin-1 A and synaptogyrin-1 had been found to be reduced in line with AD progression (Saetre *et al.*, 2011). Its lowered

expression in the AD and increased level in the AE rats reveals the ameliorating effect of *G. lucidum*.

**Neuromodulin (GAP-3):** Neuromodulin is a neuronal growth and neurite forming protein whose level decreases in AD brains. As an CSF biomarker, its lowered level has been found in other studies also (Babić *et al.*, 2014; Blennow, 2004; Bogdanovic *et al.*, 2000). However, in the present study, mushroom treatment (AE) has been found to increase the abundancy of this protein as is evidenced by the up-regulated expression.

**Neural cell adhesion molecule (NCAM):** NCAM plays important role in brain development and increased level of NCAM 1 in transgenic AD mouse model (Tg2576) and of NCAM 2 in human AD patients have been reported (Shevchenko *et al.*, 2012; Todaro *et al.*, 2004). Current findings were compatible with the previous ones as *G. lucidum* treatment helped increase the abundancy of NCAM.

**Endophilin A1:** Endophilin A1 is a membrane bending protein involved in CNS development, apoptosis, signal transduction and microtubule based movement. AD rats' hippocampi showed decreased expression while the control and AE rats experienced increased expression of endophilin A1 in the present study. In the temporal neocortex of the AD patients, decreased level of endophilin A1 has been observed (Musunuri *et al.*, 2014).

**Clathrin:** Clathrin group of proteins are involved in neuronal secretory functions and synaptic maintenance (Cao *et al.*, 2010). AD pathogenesis involves altered clathrin-associated membrane trafficking resulting in neurodegeneration (Cao *et al.*, 2010). Between the light and the heavy chains of clathrin, impaired distribution of the former has been linked with the AD pathogenesis (Cao *et al.*, 2010; Nakamura *et al.*, 1994).

Similar pattern was observed for the AD rats in the current experiment and an increasing trend following *G. lucidum* treatment (AE).

Septin: Septins are GTP-binding proteins found to be co-localized with the NFT in the AD brains (Kinoshita et al., 1998). Differential expression of septins have been observed in the present study. Contrary to the findings of Shin et al. (2004) and Musunury et al. (2014), down-regulated expression of septin-2 had been observed in the AD versus AE group of the present study. As septin-2 is involved in synaptic plasticity, its down-regulation in the AD versus AE group bears supports to the synaptic dysfunction associated with the AD pathogenesis. However, its down-regulated expression in the CE versus AE group is of intriguing. Interestingly, the isoform-2 of septin-5 had also been found down-regulated in both the AD versus AE and AE versus C groups which is compatible with the findings of Musunury et al. (2014), who found similar expression status in the temporal brain neo-cortex of the AD patients. Thus, differential expression even of the different isoforms of the same protein might be implicated in the AD pathogenesis and corresponding modulation demands differential therapeutic strategy. Current observation of the G. lucidum HWE upon differential expression of different isoforms of septin is a unique finding that demands further studies.

**UCH L1:** Ubiquitin carboxyl-terminal hydrolase L1 (UCH L1) is an important enzyme for maintenance of cognitive and synaptic functions (Gong *et al.*, 2006). Conflicting information regarding its expression has been documented in different AD cases. Though most of the researchers have noticed decreased and oxidatively modified form of UCH L1 in AD subjects, Sultana *et al* (2007), reported its 1.31-fold increase in the AD brain hippocampi (Castegana *et al.*, 2002; Choi *et al.*, 2004, Minjarez *et al.*,

2013). Increased expression of UCH L1 was observed in the *G. lucidum*-treated (AE) group.

**Soluble NSF-attachment protein beta (SNAP-\beta):** N-ethylmaleimide sensitive fusion proteins (NSF) are the part of APP and overexpressed in AD (Cottrell *et al.*, 2005). Soluble NSF-attachment proteins are involved in intracellular membrane fusion and vesicular trafficking. Among  $\alpha$ -,  $\beta$ - and  $\gamma$ - SNAPs,  $\alpha$  - and  $\gamma$ - SNAPs are expressed in different tissues while the  $\beta$ -SNAP is brain specific. In AD brain, differential expression and oxidized form of SNAP- $\beta$  had been detected through redox proteomics (Butterfield *et al.*, 2006b).

**Neuropolypeptide h3:** Neuropolypeptide h3 is a cholinergic neuro-stimulating peptide that falls in the phosphatidyloethanolamine binding protein group and is also known as Raf-kinase inhibitor protein (RKIP) and/or hippocampal cholinergic neurostimulating peptide (HCNP) (Castegna *et al.*, 2003). Our finding of down-regulated neuropolypeptide h3 is in agreement with those of Butterfield (2004). Oxidatively modified loss of function of neuropolypeptide h3 impairs phospholipid asymmetry that might be involved in extrusion of phosphatidyl serine to the outer membrane of neuron and signal for apoptosis and cause neuronal death (Daleke & Lyles, 2000). Also, neuropolypeptide h3 mediated stimulation of acetylcholine esterase (AchE) becomes compromised and this effect is heightened when HNE interacts with AChE in presence of A $\beta$  (1-42) in synaptosome (Butterfield & Lauderback, 2002; Morris, 2002). Thus, in AD brains, neuropolypeptide h3 is linked with cholinergic abnormalities and altered lipid metabolism that are the early events in AD pathogenesis (Butterfield, 2006).

**Annexin:** AD rats showed increased expression of annexin in the hippocampi. Previous studies have linked increased plasma annexin5 with increased AD risk (Sohma *et al.*, 2014). Transgenic AD mice (Tg2576) also expresses increased annexin in the brain cortex (Yamaguchi *et al.*, 2010).

**Glycogen synthase kinase 3**  $\beta$  (**GSK3** $\beta$ ): Glycogen synthase kinase 3  $\beta$  (GSK3 $\beta$ ) is a serine/threonine kinase having diversified regulatory functions ranging from glycogen metabolism to gene transcription. Overactivity of GSK-3 $\beta$  has been linked with elevated A $\beta$  production, tau hyperphosphorylation and impaired memory and learning activities (Kremer *et al.*, 2011). Memory affecting mechanism of GSK-3 $\beta$  involves interruption of intra-neuronal anterograde mitochondrial transportation and causation of "mitochondrial traffic jam" (Correia *et al.*, 2016; Zempel *et al.*, 2010).

**Serine/Threonine protein phosphatase:** Serine/Threonine protein phosphatase negatively regulate memory and learning abilities by impairing synaptic plasticity and LTP (Jouvenceau *et al.*, 2006). Up-regulation of serine/threonine protein phosphatase in the AD rats might contribute towards impaired memory and learning performance in the present study.

Serine protease inhibitors (serpins): Serine protease inhibitors (serpins) regulate proteolytic processing of proteins. Previous studies indicated their increased level in plasma and CSF of AD patients (Nielsen *et al.*, 2007). We also observed increased expression of serpins ( $\alpha$ 1-antitrypsin) in the AD rats' hippocampi. Alpha 1-antitrypsin (A1AT) has been reported to be co-localize with A $\beta$  plaques and NFTs (Gollin *et al.*, 1992).

## B. Proteins involved in metabolism

In the present study, the highest number of the identified up-regulated proteins was involved in the metabolic activities for all the three groups (C, AD and AE) (Appendix E, Table 10.1). Most of them were enzymes involved in both anabolism and catabolism of carbohydrates, proteins and lipids (Appendix E, Table 10.1). Enzymes associated with glucose metabolism (glycolysis, gluconeogenesis and citric acid cycle) were among the most abundantly up-regulated proteins in the present study (Appendix E, Table 10.1). This may be due to the exclusive utilization of glucose by the brain cells as the energy source. Apolipo-protein E (ApoE), the carrier of lipo-proteins, cholesterol and lipo-philic vitamins, was also up-regulated. Indeed, defective ApoE is an important genetic risk factor of hyperlipoproteinemia type III that results in increased plasma TC and TG levels along with impaired lipo-protein (VLDL, LDL and chylomicron) clearance. ApoE4 isoform had been linked even up to 20 times greater risk of late-onset sporadic AD (Hauser & Ryan, 2013).

Glucose is the main source of energy for the brain cells and alteration in energy production has been linked with AD pathogenesis (Ferreira et al., 2010; Parihar & Brewer, 2007). Present findings of the increased glycolytic enzymes (Hexokinase-1, Glucose-6-phosphate isomerase, Glyceraldehyde-3-phosphate dehydrogenase (GAPDH), L-lactate dehydrogenase B chain), pyruvate metabolizing enzymes (Pyruvate dehydrogenase E1 component subunit alpha and beta), lactate metabolizing enzyme (Llactate dehydrogenase B chain) are indicative of decreased utilization of glucose for energy production in the AD brains. Glycolytic enzyme GAPDH has other activities such as OS sensor of apoptosis and AD (Butterfield et al., 2010). Minjarez et al (2013) reported differential expression of GAPDH, ubiquitin carboxyl-terminal hydrolase isozyme L1 (UCHL-1) and transferrin as the AD biomarker. UCHL-1 is linked with OS and transferrin with iron regulatory processes. Findings of the present study are in agreement with those of Boyd-Kimball et al., (2005), David et al., (2006), Perluigi et al. (2009), Shin et al., (2004), Tilleman et al., (2002) and Musunuri et al., (2014). Differential expression of ATP synthase  $\alpha$ ,  $\beta$  and  $\gamma$  chain has been reported by Shin *et* al., (2004).

Brain utilizes glucose as the main source of energy and about 20% of total glucose metabolism and 30% of inspired oxygen is utilized by the brain (Vannucci & Vannucci, 2000). Neurons are solely dependent on glucose for energy production and even minuscule impairment of glucose metabolism could disrupt neuronal activities including memory (Butterfield *et al.*, 2012). Altered expression of the aforementioned proteins, mostly enzymes involved in glucose metabolism, might have affected energy generation, ionic balance, membrane asymmetry and transporters in the current experimental subjects. Consequently, impaired neuronal connectivity, transportation, signal transduction and neurotransmission might have resulted in declined cognitive and memory functions of the AD rats (Butterfield *et al.*, 2012).

Oxidative modification of the energy metabolism related proteins such as creatine kinase,  $\alpha$ -enolase and triosephosphate isomerase in the AD brains have been noticed by Butterfield *et al.*, (2004). Enolase, one of the regulatory enzymes of glycolysis, is among the differentially expressed proteins in the AD subjects in the present study. Besides its effect upon conversion of 2-phosphoglycerate to phosphoenolpyruvate through dehydration in glycolysis,  $\alpha$ -enolase acts as neurotrophin, stress protein and plasminogen binding protein (Butterfield & Lange, 2009). It is among the most commonly upregulated and early oxidized proteins in the brains of both MCI and AD patients (Butterfield & Lange, 2009; Butterfield & Sultana, 2007; Takano *et al.*, 2013). Impaired glucose metabolism and ATP generation in the AD brain has been linked with the nitration of the glycolytic enzymes glyceraldehyde-3-phosphate dehydrogenase (GAPDH) and  $\alpha$  enolase (Castegna *et al.*, 2002; Vanhanen & Soininen, 1998). Also, oxidation directed loss of GAPDH activity had been associated with increased A $\beta$  (1-42) level in AD brain (Boyd-Kimball *et al.*, 2005).

Carbonic anhydrase II (CAH II) catalyzes the reversible hydration of CO<sub>2</sub> and important in glycolysis, lipid metabolism, CSF production, maintenance of pH and fluid balance (Sly & Hu, 1995). In CNS, CAH II is linked with oligo-dendrocytes and protoplasmic astrocytes (Sly & Hu, 1995). Its deficiency had been reported to cause cognitive decline, mental retardation, cerebral calcification, renal tubular dysfunction and osteoporosis (Sly & Hu, 1995). Its oxidative modification mediated impaired pH balance in the neurons had been linked with AD pathogenesis (Sultana *et al.*, 2006).

Beyond energy metabolism, decreased rate of glycolysis in the AD subjects has been linked with reduced generation of acetylcholine that impairs memory and learning abilities (Poon *et al.*, 2004).

Among the down-regulated enzymatic proteins, phosphoglycerate mutase 1 and acyl coenzyme A thioesterase 1 and 2 were most notable.

**Phosphoglycerate mutase 1:** Phosphoglycerate mutase 1 interconverts 3phosphoglycerate and 2-phosphoglycerate in glycolysis and gluconeogenesis. In the AD rats, phosphoglycerate mutase 1 was found to be down-regulated. Decreased expression of this enzyme in the AD hippocampi and in neurons and astrocytes have been noticed by some previous researchers (Díez-Vives *et al.*, 2009; Sultana *et al.*, 2007).

Acyl coenzyme A thioesterase 1 and 2: Downregulation of acyl coenzyme A thioesterase 1 and upregulation of the acyl coenzyme A thioesterase 2 in the AD rats' hippocampi were observed in the present study. Similar type of observation has been reported for the SMP8 mouse model of AD (Díez-Vives *et al.*, 2009). The isoform 1 is cytoplasmic and 2 is mitochondrial though both are involved in the hydrolysis of acyl-coenzyme As to the free fatty acids (FFAs) and coenzyme A. FFAs themselves and their activated acyl co-A forms participate in numerous metabolic processes.

## C. Proteins involved in mitochondrial respiratory chain and ATP generation

In the AD brains, lowered expression of five proteins involved in mitochondrial respiratory chain and ATP generation was observed. They were either structural subunits of ATP synthase or proton-gradient maintaining functional units. The most notable was the vacuolar (V) ATPases. V-ATPases sub-units are involved in regulation of cellular transduction processes and pH. V-ATPases sub-expressed in the AD rats were V0-subunit A (ATP6V0A1), V1-subunit D (ATP6V1D) and V1-subunit F (ATP6V1F). Our findings are relevant with those of Minjarez *et al.*, (2016).

Mitochondrial abnormality including dysfunctional and lowered level of mitochondrial enzymes is another important feature of AD pathogenesis (Bosetti et al., 2002; Hirai et al., 2001). APP and Aβ can affect mitochondrial respiratory chain through decreased activity of some enzymes and increasing ROS production (Suzanne & Tong, 2014). Proteins and enzymes involved in energy metabolism are highly oxidized and functionally altered in AD brains that correspond to the reduced glucose metabolism; mitochondrial reduced functioning and OS interplay in AD pathogenesis (Hoyer, 2004; Sultana et al., 2006). Aß induced lowered activity of the mitochondrial respiratory chain in the neurons had also been noticed (Hirai et al., 2001; Lovell et al., 2005). Defect in one of the respiratory chain complexes affect another and finally the entire process of energy production become hampered (Morán et al., 2012). Comparatively, complex I is much prone to OS due to its content of Fe - S cluster. Redox proteomics studies had identified nitrated form of mitochondrial membrane associated voltage dependent anion channel (VDAC) and ATP synthase alpha chain in the AD brains (Sultana et al., 2006c). Supposedly, ATP production and normal functioning of the neurons become affected and as a consequence the brain cells die. ATP synthase  $\alpha$  is a part of the complex V of the mitochondrial respiratory chain and is located in the inner membrane. Its increased accumulation in the cytosol linked with

NFT had been observed in AD subjects (Sergeant *et al.*, 2003). Mitochondrial abnormality affects APP cleavage leading towards enhanced intra-neuronal A $\beta$  accumulation (Busciglio *et al.*, 2002).

Selective defect in the cytochrome c oxidase of the mitochondrial respiratory chain had been linked with reduced energy production and AD complications (Maurer *et al.*, 2000; Ojaimi *et al.*, 1999). Decreased expression was also observerd for cytochrome oxidase that transfers electron from ferrocytochrome c to oxygen in the respiratory chain.

In the normal neurons, about 30% of the neuronal mitochondrial pool is involved in anterograde and retrograde shifting in response to different stimuli (Correia *et al.*, 2016; Misgeld *et al.*, 2007). Overrated mitochondria are transported from neuronal other parts to the pre-synaptic neuronal termini to fulfill increased energy demand. But, in the AD neurons, this mitochondrial transportation becomes affected due to intra-neuronal Ca2+dyshomeostasis and microtubular disassembly (Correia *et al.*, 2016).

# D. Proteins involved in cellular stress responses

In the AD and the mushroom fed rat groups, proteins of both enzymatic and nonenzymatic types were significantly regulated and differentially expressed. The proteins were of overlapping functionalities against oxidative, nutritional and environmental stressors. Heat shock proteins (HSPs) of variable molecular weight and functions were among the mostly regulated ones.

#### a. Role of HSP in AD

Proteins performing the role of the molecular chaperones such as the heat shock proteins (HSPs) are involved in different stress responses. Altered expression of different HSPs in the present study demonstrate their altered levels in the AD pathogenesis and corresponding amelioration in the respective subjects. As AD pathogenesis involves protein misfolding, increased chaperones are expressed in AD brains in response to A $\beta$  and tau proteins (Blair *et al.*, 2013; van der Putten & Lotz, 2013). Increased expression of heat shock 70kDa protein 12A (Predicted), isoform CRA\_a; heat shock cognate 71 kDa protein, HSP 90 –  $\alpha$ , HSP 90 –  $\beta$ , and mitochondrial HSP 60 kDa were noted in the present study that are indicative of different mode and extent of the experimental subjects in response to different stressors. Current findings are in agreement with those of Yao *et al.* (2007). Oxidative modification and lowered expression of the heat shock cognate 71 in the AD brains have been noticed by Butterfield *et al.*, (2004). These observations indicate that excessive load of the oxidatively modified, ubiquinited, HNE-bound proteins disrupt the proteasomal degradation of the dysfunctional proteins resulting in increased accumulation of damaged proteins and neurodegeneration (Keller *et al.*, 2000).

Heat shock proteins (HSP) act as molecular chaperones and their role in inhibiting A $\beta$  aggregation, facilitating A $\beta$  degradation and clearance have been documented (Evans *et al.*, 2006; Peterson & Blagg, 2009). They are capable of inhibiting A $\beta$  production and aggregation in a concentration dependent way. Binding with the A $\beta$  in an ATP-dependent or –independent way, HSP 40/HSP 70 and HSP 90 can guard against A $\beta$  aggregation (Evans *et al.*, 2006; Peterson & Blagg, 2009).

Although A $\beta$ -induced expression of HSP 70 have been found to be variable, their activity has been reported to go beyond action against protein misfolding towards transportation of the misfolded proteins up to the proteasome for degradation (Lam *et al.*, 2013; Takano *et al.*, 2013).

Exogenous HSP 32, HSP 70 and HSP 90 act as the facilitators of  $A\beta$  clearance that has been supposed to be mediated through activation of the microglial phagocytosis

(Kakimura *et al.*, 2002). Their role in  $A\beta$  degradation has also been reported and the process might have involved activation of the NF- $\kappa$ B and p38 MAPK through the Tolllike receptor-4 (TLR4) pathway (Kakimura *et al.*, 2002). Degradation of the phosphorylated tau by the Hsp90 - carboxy terminus of Hsp70-interacting protein (CHIP) complex and thus lowered propensity towards NFT formation has also been observed (Dickey *et al.*, 2007). Inhibition of HSP 90 showed reduced activation of the kinases (GSK3 $\beta$ , CDK5, Akt, p35 and p25) involved in tau phosphorylation and thus lowered NFT formation and increased degradation (Luo *et al.*, 2010). Also, decreased serum level of HSP 90 has been observed in AD patients. In line with this, administration of HSP 90 inhibitors have been found to prevent A $\beta$ -induced neurotoxicity by activating the heat shock factor 1 and increasing HSP 70 level (Ansar *et al.*, 2007; Gezen-Ak *et al.*, 2013).

# b. Role of Aldehyde dehydrogenase 2 (mitochondrial) in AD

Aldehyde dehydrogenases (ALDH1, ALDH2 and ALDH3), oxidatively detoxify aldehydes into carboxylic acids and its deficiency acts as AD promoting (Kamino *et al.*, 2000). ALDH usually metabolize 4-hydroxy-2-nonenal (HNE) and upregulation of HNE and acetaldehyde are indicative of reduced ALDH activity and increased vulnerability towards AD (Kamino *et al.*, 2000; Picklo *et al.*, 2002). Specifically, aldehyde dehydrogenase 2 (mitochondrial) detoxifies aldehydes in the mitochondria whose level increases with aging and in AD. Increased expression of aldehyde dehydrogenase 2 (mitochondrial) in the AD rats correspond towards excessive OS in the AD rat brains (Picklo *et al.*, 2001). Binding of aldehyde dehydrogenase with A $\beta$  and A $\beta$ binding peptide has been ascribed as an important aspect of AD pathogenesis. Thus, inhibition of this interaction seems a potent target of AD therapeutics (Borger *et al.*, 2013; Yao *et al.*, 2011).

#### E. Proteins involved in cholesterol metabolism and transportation

## a. Apolipoproteins

In rats, apolipoprotein A-IV is the major constituent of HDL and chylomicrons and involved in various functions including cholesterol efflux, stimulation of lipoprotein lipase and cholesteryl ester transfer protein as well as anti-oxidtive and antiinflammatory effects (Duka *et al.*, 2013). Apolipoprotein A-IV was downregulated in the AD groups in the present study. Obtained findings are in line with those of Cui *et al.*, (2011) who showed that genetic extirpation of apolipoprotein A-IV aggrandizes Aβ plaque deposition, aggravates neuronal loss and impairs spatial memory in the AD mouse models. Normally, apolipoprotein A-IV functions as the Aβ clearing agent and *in vitro* studies showed that they co-localize with Aβ plaque in the brain (Cui *et al.*, 2011). Down-regulation or loss of function of this lipo-protein impairs the Aβ clearing process that ultimately leads towards Aβ deposition (Cui *et al.*, 2011).

Down-regulated expression of apolipoprotein A-I (apo A-I) was observed in the hippocampus of the AD versus control rats. Current findings are compatible with those of Liu *et al.*, (2006) who demonstrated that AD patients' hippocampi and serum possess lowered apo A-I compared with their age matched controls. Redox proteomics has identified apo A-I as one of the most promising biomarker of neurodegenearation (Keeney *et al.*, 2013). In this regard, upregulated expression of apo A-I found in the AE versus control group of the present study is indicative of neurohealth giving effect of the *G. lucidum* HWE.

In the present study, up-regulated expression of apolipoprotein E (apo E) was noticed in the AD rat groups. Apolipoprotein E is the principal carrier of cholesterol and aids in lipid transportation and injury repairment in the brain (Liu *et al.*, 2013). Individuals with its  $\epsilon$ 4 allele are at increased (about 8 fold) risk of AD pathogenesis than those with the

 $\varepsilon$ 3 allele and  $\varepsilon$ 2 allele possessing people are at rather protection against AD generation (Corder et al., 1993). Though, any specific allele has not been detected in the present study but the total apolipoprotein E, observed findings are compatible with some established knowledge that apolipoprotein E is involved in AD pathogenesis (Sizova et al., 2007). Like that of binding with the cell surface receptors, apolipoprotein E transfers lipids to the hydrophobic amyloid- $\beta$  (A $\beta$ ) peptide and aids formation of A $\beta$ , cerebral amyloid angiopathy, generation of OS, neurotoxicity and neurodegeneration in AD (Ellis et al., 1996). They also stimulate AB aggregation and participate in metabolism, neuro-signaling and neuroinflammation (Kim et al., 2009). Recently, Choi et al., (2015) had shown that brain level of Apo E and Aß are regulated by the lowdensity lipoprotein receptor (LDLR) degrading E3 ubiquitin ligase, idol. Idol had been implicated as the principal regulator of LDLR degradation and an inhibitor to the clearance of brain level of both Apo E and Aß (Choi et al., 2015). Its absence in AD transgenic mice increased brain level of LDLR as well as decreased the levels of both Apo E and Aβ as well as improved Aβ-induced neuroinflammation (Choi et al., 2015). Genome wide association studies (GWAS) also have linked Apo E with LDLR and CLU (clusterin) inter alia as the cholesterol dysregulatory factors of AD pathogenesis (Tosto & Reitz, 2013). Thus, increased expression of apolipoprotein E might have been among the most favored factors of AD pathogenesis in the current experiment. Likewise, decreased expression in the control and G. lucidum HWE fed rats bear testimony to the AD enhanced protection against AD pathogenesis in the respective rat groups.

## b. Apolipoprotein J (Clusterin)

Apolipoprotein J (Clusterin) is a glycoprotein normally expressed in different tissue but highly in the brain and with aging and AD pathogenesis. Clusterin gene, CLU, is the third most risk gene linked with late onset AD and it is responsible about 9% of AD risks (Bertram *et al.*, 2007). Single nucleotide polymorphism of clusterin gene (CLU) heightens AD risk (Bertram *et al.*, 2007). Hong *et al.*, (2013) reported its elevated expression in the hippocampus of the 5XFAD mice. Clusterin affects AD pathogenesis in various ways such as promoting A $\beta$  aggregation, neuro-inflammation, apoptosis and controlling lipid metabolism and cell cycle regulation and also through epigenetic mechanism (Yu & Tan, 2012). Binding with A $\beta$ , clusterin facilitates A $\beta$  transportation in the blood in chaperone-like mechanism and thus clusterin has been regarded as a blood biomarker of AD (Weinstein *et al.*, 2016). In this way, clusterin also acts as the A $\beta$  clearing agent (Yu & Tan, 2012). Thus, compelling conclusion is yet to be drawn about the effect of clusterin on AD and a state-of-the time explanation is that at low ratio of clusterin to A $\beta$ , clusterin acts as the A $\beta$  clearing agent but as the ratio changes, clusterin adds injury to AD pathogenesis (Yerbury *et al.*, 2007).

## c. ACAT (acyl-coA : cholesterol acyltransferase)

Elevated expression of ACAT (acyl-coA - cholesterol acyltransferase), enzyme involved in cholesterol esterification, points towards altered cholesterol metabolism in the AD rats (Chang *et al.*, 2001; Puglielli *et al.*, 2001). Impaired ACAT and cholesterol metabolism have previously been linked with increased A $\beta$  production and neurotoxicity (Puglielli *et al.*, 2001; Sizova *et al.*, 2007; Sjögren *et al.*, 2006).

## d. Fibrinogen

Fibrinogen was overexpressed in the AD rats' brain of the current study. Fibrinogen is the precursor of fibrin, the primary protein involved in blood clotting. Normally, fibrinogen circulates in plasma and can not cross the blood brain barrier (BBB). But damaged vasculature of the AD subjects allows its accumulation in the extravascular spaces that further intensifies AD pathogenesis (Cortes-Canteli *et al.*, 2012). Elevated level of fibrinogen has been linked with AD propensity. Indeed, interaction between Aβ and fibrinogen produces structurally modified fibrin oligomer that causes altered clot formation and vasculopathies (reduced cerebral blood flow, neuroinflammation and neurodegeneration) associated with AD (Ahn *et al.*, 2010; Cortes-Canteli *et al.*, 2012). Oxidative modification of the plasma isoforms of fibrinogen  $\gamma$ -chain precursor and  $\alpha$ 1antitrypsin precursor had been linked with AD pathogenesis (Choi *et al.*, 2002).

#### e. Transthyretin

Transthyretin (TTR) co-localizes A $\beta$  in the human AD brains and has been suggested to sequester A $\beta$ , prevent A $\beta$  aggregation and fibrillation and thus protective against AD (Li & Buxbaum, 2011) . Downregulation of transthyretin in the AD rats bears proof of increased A $\beta$  plaque formation. On the contrary, *G. lucidum* treatment showed boosting up effect upon TTR expression as the extent of its down-regulation was less in the AE group.

## f. Clathrin

Defected clathrin-mediated endocytosis has been reported to be associated with AD pathogenesis (Cao *et al.*, 2010; Wu & Yao, 2009). Clathrin coat consists of three heavy chains (~190 kDa) connected in a hub and each heavy chain containg an axial light chain (~25 kDa). The light chain has been mostly been implicated in AD pathogenesis (Nakamura *et al.*, 1994). They contain multiple adaptors (AP) to bind to the plasma membranes among which AP2 is clathrin-coated vesicle specific and AP 180 is synaptic vesicle specific. Increased expression of the clathrin heavy chain and isoform 2 of clathrin coat assembly protein AP180 were found in all the rat groups. Though present findings are seemingly in contrast with some others, it can not be claimed that the

clathrin-mediated endocytosis remain unaffected in the AD subjectspresent study (Cao *et al.*, 2010). May be, exogenous A $\beta$  bound to the plasma membrane needed excessive rate of clathrin-mediated endocytosis and to fulfill this demand, expression of the clathrin heavy chain and isoform 2 of clathrin coat assembly protein AP180 had to be triggered (Wu & Yao, 2009).

# F. Proteins involved in Ca<sup>2+</sup>transportaion, homeostasis and signaling

Dysregulation of  $Ca^{2+}$  metabolism and signaling has been linked with neurodegeneration and AD pathogenesis (Brawek & Garaschuk, 2014). Differential expression of the proteins involved in  $Ca^{2+}$  transportation, homeostasis and signaling such as calmodulin, CamK2a and CamK2b were noticed in the present study.

## a. Calmodulin

Calmodulin is a Ca<sup>2+</sup>- binding protein involved in Ca<sup>2+</sup>-mediated signaling and maintenance of ion channel and enzymatic (kinases and phosphatases) activities (Brawek & Garaschuk, 2014). Calmodulin is a biomarker of AD whose increased expression and its binding proteins are associated with AD pathogenesis (O'Day *et al.*, 2015). AD rats showed upregulated expression of calmodulin. Upregulated expression of calmodulin in the AD group of the present study points towards Ca<sup>2+</sup> metabolism dysregulatory state of the AD rats compared with the C and the AE rats.

## b. CamK2a

Ca<sup>2+</sup>/calmodulin-dependent serine/threonine protein kinase (Camk2a) is highly important for maintenance of the glutametargic synaptic plasticity (Zhou *et al.*, 2010). Its role in spatial learning becomes also evident from its supporting role towards NMDAR-dependent LTP in the hippocampus (Molnár, 2011). Through Ca<sup>2+</sup>/calmodulin kinase II signaling, Camk2a also regulates neurotrophin-3 and BDNF secretion from the hippocampal post-synaptic neurons (Kolarow *et al.*, 2007). In the present study, the CamK2a showed downregulated pattern compared with their control and mushroom-treated counterparts. Current findings are in line with those of Yamasaki *et al.* (2008) and Zhou *et al.* (2010). Its deficiency has been shown to hamper neuronal development and dentate gyrus formation as well as behavioral alteration (Yamasaki *et al.*, 2008). On the other hand, up-regulated expression of CamK2a has been linked with improved cognitive performance of the SMP8 mice. Its stimulatory role towards the ubiquitin-proteasome system has also been documented (Djakovic *et al.*, 2009).

## c. Annexins (Annexin-1, -5 and -6)

Annexins (1,5 and 6) have been found to be up-regulated in the AD rats. Observed increased annexin expression in the AD rats' hippocampi is compatible with those of Yamaguchi *et al.*, (2010), Sohma *et al.*, (2014) and Hondius *et al.*, (2016). Annexins are intracellular Ca<sup>2+</sup>-respondents capable of binding with membrane phospholipids and participate in membrane trafficking, endo- and exo-cytosis (Hondius *et al.*, 2016; Lizarbe *et al.*, 2013).

Aβ-induced Ca<sup>2+</sup> dysregulation (increased intracellular level) hyperactivates c-Jun Nterminal kinase (JNK), cyclin-dependent kinase 5 (CDK5), tau phosphorylation and disrupts microtubule network (Correia *et al.*, 2016; Zempel *et al.*, 2010). It leads towards mitochondrial trafficking defects that impair the normal movement of mitochondria across the microtubules and thus cause "mitochondrial traffic jam" in the AD neurons and affects neuronal functions (Correia *et al.*, 2016). Thus, intracellular Ca<sup>2+</sup>dyshomeostasis might be among different mechanisms involved in disrupted neuronal activity and corresponding impaired memory of the AD rats in the present study.

#### G. Proteins involved in signal transduction

#### a. 14-3-3 Proteins (Ywhag)

In the current proteomics experiment, up-regulated expression of the 14-3-3 proteins  $(\zeta/\delta, \theta, \eta, \gamma, \beta/\alpha)$  were found in the AD rat brains. 14-3-3 proteins constitute about 1% of total soluble proteins of the normal brain. They participate in signaling through binding with the phospho-serine containing proteins and can regulate the activities of the kinases, phosphatases and trans-membrane proteins (Foote & Zhou, 2012). Thus, they mediate diversified activities involving neuronal plasticity, neurotransmission, neurite outgrowth generation and neurogenesis (Fu et al., 2000). In AD brains, they have been found to be closely associated with tau and aid in the formation of the NFTs (Foote & Zhou, 2012). Differential expression of 14-3-3 proteins  $\beta/\alpha$ ,  $\zeta/\delta$  and  $\varepsilon$  had been observed in the temporal neocortex and other parts of the brains of the AD patients (Andreev et al., 2012; Musunuri et al., 2014). In the AD hippocampi, both intra- and extra-cellular expression of the 14-3-3 proteins have been detected. Among different isoforms, the highest immunoreactivity towards NFT has been observed for the 14-3-3 $\zeta$ (Thomas *et al.*, 2005). 14-3-3 $\zeta$  – mediated tau phosphorylation involves protein kinases such as glycogen synthase kinase-3 beta (GSK3β) (Gao et al., 2014). GSK-3β hyperactivity impairs mitochondrial intra-neuronal anterograde movement and causes "mitochondrial traffic jam" and disrupts neuronal activities (Correia et al., 2016; Zempel *et al.*, 2010). 14-3-3 $\zeta$  also binds with  $\delta$ -catenin and disrupts the formation of the adherens junction complex that compromises the neural structure and cognitive performance (Matter *et al.*, 2009). Also, interaction of  $\delta$ -catenin with pre-senilin 1 is an important stimulator of the wnt signaling and thus of AD pathogenesis and neuronal under-development (Zhou et al., 1997). Protective effect of the 14-3-3 proteins towards  $\alpha$ -synuclein driven dopaminergic neurotoxicity has been found to be mediated through

Lrrk2 (leucine-rich repeat protein kinase 2) and Syna (Abbott & Nelson, 2000; Li *et al.*, 2011).

Besides signaling, the 14-3-3 proteins participate in the regulation of stress response, redox mechanisms, cell cycle, apoptosis and cytoskeletal maintenance (Thomas *et al.,* 2005). In addition to the different brain regions, their increased expression has been observed in the CSF also (Jayaratnam *et al.,* 2008).

# b. VDAC (Voltage-dependent anion selective channel 1)

Voltage-dependent anion selective channels (VDAC1 and VDAC 2) are mitochondrial porins involved in transportation of ATP and Ca2+and in apoptotic signaling (Reddy, 2013). Its altered expression has been noticed in AD and other NDs and in mitochondrial dysfunctions. However, conflicting expression status of VDAC 1 has been reported for different AD patients and model animals (Cuadrado-Tejedor *et al.*, 2011; Manczak & Reddy, 2012; Yoo *et al.*, 2001).

## c. SLC12A5

SLC12A5 are neuronal K<sup>+</sup>/Cl<sup>-</sup> symporter involved in maintaining intra-neuronal low Cl<sup>-</sup> concentration (Cherubini *et al.*, 2012). It mediates neuronal excitotoxicity and synaptic inhibition (Gauvain *et al.*, 2011). Its increased expression is linked with the glutamate transporter Slc17a7 that actively participates in GABAergic neurotransmission (Gauvain *et al.*, 2011).

Oxidative modification and lowered expression of the excitotoxicity-linked proteins such as glutamine synthetase, glutamate transporter – excitatory amino acid transporter 2 (EAAT2) in the AD brains have been noticed by Butterfield *et al.*, (2004). A $\beta$  (1-42) induces increased 4-hydroxy-2-trans-nonenal (HNE) generation through lipid peroxidation and HNE oxidatively modify glutamate transporter EAAT2 and impair transporter function (Lauderback *et al.*, 2001; Mark *et al.*, 1997; Masliah *et al.*, 1996). Its coupled inactivity with glutamate synthetase result in accumulation of excessive glutamate outside of the neuron that stimulates NMDA receptor driven increased Ca2+influx and compromised LTP and neurodegeneration (Kim *et al.*, 2011).

## H. Proteins involved in apoptosis

Apoptosis is an important feature of neuronal and synaptic cell losses. Decreased levels of the apoptosis regulatory enzymes found in the present study are distinct hallmarks of AD pathogenesis (Chou *et al.*, 2011).

### a. Peptidyl prolyl cis-trans isomerase (Pin1)

In the AD versus AE groups, significantly down regulated pattern of the peptidyl prolyl cis-trans isomerase (Pin1) was observed. Targeted proteomic search for oxidized protein in the AD hippocampus had reported the protein peptidyl-prolyl isomerase, Pin1 to be oxidized and significantly down-regulated (Sultana *et al.*, 2006). Normally, Pin1 is involved in dephosphorylation of tau protein, cell cycle regulation and in protection against neurodegeneration (Liou *et al.*, 2003; Lu *et al.*, 1999). Pin1 acts as a chaperone and isomerizes the pSer/Thr-Pro peptide bond and thus participates in protein folding, assembly, cellular transportation and apoptosis (Butterfield *et al.*, 2006a).

Beyond glycolysis, glyceraldehyde-3-phosphate dehydrogenase (GAPDH) is also involved in neuronal apoptosis (Chuang *et al.*, 2005; Cumming & Schubert, 2005). In the cortical neurons, A $\beta$ -mediated increased aggregation of GAPDH leading towards neuronal death emanated from the chronic A $\beta$  exposure have been documented (Cumming & Schubert, 2005). Non-native GAPDH is considered as a component of the amyloid and its binding to the GAPDH reduces the enzyme's activity and to complement the reduced functioning and to fulfill energy demand, AD brains might have triggered the increased expression of GAPDH (Mazzola, J. L. & Sirover, 2001; Naletova *et al.*, 2008; Rhein *et al.*, 2009).

## b. Ppp2ca (protein phosphatase) Protein

Protein phosphatase (Ppp2ca) is involved in dephosphorylation of tau, whose hyperphosphorylation causes NFT formation (Qian *et al.*, 2010). Thus a balance occurs between protein kinase and phosphatase in the normal state (Martin *et al.*, 2013). Upregulation of Ppp2ca is indicative of increased protein phosphatase 2A activity and thus ameliorating effect towards AD pathogenesis (Manavalan *et al.*, 2013).

## I. Proteins involved in neuronal cytoskeleton maintenance

Derangement of neuronal cytoskeleton through microtubule disassembly is an important feature of neurodegeneration (McMurray, 2000). STRING analysis revealed strong networks among microtubule assembling the cytoskeletal proteins such as tubulin,  $\beta$ -actin and keratin.

## a. Tubulin

AD neurons suffer from disrupted microtubule structure and functioning (Butler *et al.*, 2007). Tubulin is the main component of microtubule and consists of dimers imparted by the alpha and beta chains. Differential expression of tubulin  $\alpha$  -1c, -4a,  $\beta$  - 2a, -2B, -3 and -5 chains were observed in the present study. Molecular function based sub-network analysis showed different tubulin chains to be clustered together and deranged in AD. Both animal and human studies have linked decreased level of  $\alpha$  and  $\beta$ 

tubulin with human AD (Pollack *et al.*, 2003; Poon *et al.*, 2006) . In AD brain,  $\beta$  tubulin becomes abnormally hyperphosphorylated and modified tubulin fails to assemble microtubules. Consequently, microtubule disassemblage leads towards cytoskeletal vulnerability (Vijayan *et al.*, 2001). Recently, microtubular disassembly has been implicated in causing "mitochondrial traffic jam" in the AD neurons as mitochondrial shifting across the "rail-road of microtubule" becomes impeded in the AD brain (Correia *et al.*, 2016)

### b. β-Actin

Normally,  $\beta$ -actin is involved in maintenance of cytoskeleton, internal cell motility, neuronal network integrity and aids in memory and learning performances. Its altered expression and oxidized form had been linked with AD pathogenesis (Castegna *et al.*, 2003). Impaired expression of actin is in agreement with the synaptic dysregulation associated with AD and age-related altered cytoskeletal structure, axonal dystrophy, reduced dendritic spines and impaired transport across membranes (Penzes & VanLeeuwen, 2011). Enhanced accumulation of actin enhances tau-governed neurotoxicity (Fulga *et al.*, 2007; Takano *et al.*, 2013).

## c. Dihydropyrimidinase-related protein 2 (DRP-2):

Dihydropyrimidinase related protein 2 (DRP2) is involved in regulation of axonal outgrowth and becomes hyperphosphorylated in NFT and its increased level is observed in AD model animals (Sizova *et al.*, 2007). Compared to the normal neurons, AD neurons possess shortened dendrites which is a characteristic of their lowered communication with neighboring neurons (Coleman & Flood, 1987). DRP2 is expressed highly in the developing brains and altered in AD (Castegna *et al.*, 2002; Lubec *et al.*, 1999). It regulates the activity of collapsin that elongates the dendrites,
increases their communicability and repairs the damaged neurons (Castegna *et al.,* 2002; Lubec *et al.,* 1999). Oxidative modification of DRP2 might cause reduced length of the dendrites and communication leading to lowered cognitive performance (Butterfield *et al.,* 2006b).

#### d. Glial fibrillary acidic protein (GFAP)

Glial fibrillary acidic protein (GFAP) provides structural support to the astrocytes and its elevated level in AD model animals and in human subjects (Sizova *et al.*, 2007; Sultana *et al.*, 2010b). Its differential and reduced expression has been reported in AD animal studies (Korolainen *et al.*, 2005; Shiozaki *et al.*, 2004).

#### e. RhoA proteins

Ras homolog gene family, member A (RhoA) proteins are involved in cytoskeleton regulatory processes such as dendrite development, axonal extension and protrusion (Fujita & Yamashita, 2014). They also stabilize the A $\beta$ -disrupted microtubules (Fujita & Yamashita, 2014). A $\beta$  increases RhoA-GTPases and decreases neuronal spine production and neural connection both in the cell lines and also in the brains of the transgenic AD models (Minjarez *et al.*, 2016; Petratos *et al.*, 2008). Inhibition of the RhoA/ROCK signaling pathway seems ameliorating towards axonal growth and cognitive decline (Fujita & Yamashita, 2014).

# f. Septin

Septins are microtubule associated, filament-forming and GTP-binding proteins that participate in dendritic spine formation and in neurotransmitter release (Ito *et al.*, 2009; Spiliotis, 2010). Like that of Musunury *et al.*, (2014), increased expression of septin -2 and -3 were noted in the AD rats' hippocampi. Its increased expression in the AD rats

might be involved in disrupting microtubular filament formation and associated cytoskeletal derangement in the AD rats (Hanai *et al.*, 2004; Marttinen *et al.*, 2015).

#### g. Cofilin

Brain cofilin activity reduces with age and in the AD subjects, it goes down aberrantly (Barone *et al.*, 2014). Transgenic mice studies also showed its lowered expression (Tilleman *et al.*, 2002). However, unlike other cytoskeletal proteins, its levels have not been found to be up-regulated in AD ameliorating cases (Barone *et al.*, 2014; Hwang *et al.*, 2015). As cofilin is a regulator of actin, increased cofilin expression points towards increased actin turnover and increased depolymerization of actin filament (Hwang *et al.*, 2015).

#### h. Dynamin

Dynamin is a neuronal GTPase capable of free entry into and release from the synaptic vesicles (Fulga *et al.*, 2007). Current finding of its decreased expression in the AD rats is consistent with those of Cao *et al.*, (2010), Watanabe *et al.* (2010) and Zhou *et al.* (2010) and Kelly *et al.*, (2005). A $\beta$ -induced depleted dynamin1 level has been found to impair memory in the AD model rats (Watanabe *et al.*, 2010). Although some other findings found its variable level, altered dynamin induced affected neurotransmitter release has commonly been correlated with AD pathogenesis (Kelly *et al.*, 2005; Takano *et al.*, 2013).

#### i. Gelsolin

Gelsolin is a member of the actin-binding proteins having antioxidative, A $\beta$  binding and fibrillation inhibitory potentiality (Chauhan *et al.*, 2008; Ji *et al.*, 2015). Its overproduction and/or administration showed A $\beta$  lowering effect and thus, gelsolin has been regarded as an AD therapeutic agent (Carro, 2010). More importantly, administration or overexpression of gelsolin results in significant reduction of amyloid load and decrease of A $\beta$  level in AD transgenic mice (Carro, 2010, Chauhan *et al.*, 2008). Down-regulated expression of gelsolin was noted in the AD rats' hippocampus that is in par with those of Manavalan *et al.* (2013). However, some studies have reported confounding results regarding the expression of gelsolin in AD brains (Güntert *et al.*, 2010; Ji *et al.*, 2009).

Oxidative modification and reduced expression of the neuronal cytoskeletal proteins such as dihydropyrimidinase-related protein 2 (DPYSL 2),  $\beta$ -actinin the AD brains have been noticed by Butterfield *et al.*, (2004).

### j. Proteins involved in anti-oxidative defense

Anti-oxidative proteins up-regulated in the mushroom fed rats were mitochondrial superoxide dismutase [Mn], superoxide dismutase [Cu-Zn], glutathione S-transferase pi and mu, glutathione peroxidase 3; peroxiredoxin-1, -2, -5, -6; catalase, isoform 2 of haptoglobin and mitochondrial stress -70 protein.

Mitochondrial superoxide dismutase coverts superoxide anion ( $O_2^-$ ) into  $H_2O_2$ and thus reduces the deleterious effect of  $O_2^-$  (Flynn & Melov, 2013). In the AD patients, its overexpression at the lymphocytic mRNA level has been detected (De Leo *et al.*, 1998). Studies upon AD mouse model has shown that deficiency of this enzyme stimulates A $\beta$  formation, plaque deposition and tau phosphorylation whereas its increased level demolishes A $\beta$  plaque deposition through diminishing the A $\beta$ -42/A $\beta$ -40 (De Leo *et al.*, 1998). Thus, to thwart the OS and to maintain synaptic plasticity, upregulation of superoxide dismutase [Cu-Zn] (Sod1), and mitochondrial superoxide dismutase [Mn] might have been achieved through feedback mechanism (Manavalan *et al.*, 2013).

Glutathione-S-transferases pi and mu (GSTP1 and GSTM1) are involved in glutathione conjugation and phase II detoxification. The stress regulator Nrf2 regulates these enzymes in response to OS and other stressors. Through protein-protein interaction, GSTP1 lowers the activity of the stress kinase JNK and Cdk5 (Sun *et al.*, 2011).

Catalase decomposes  $H_2O_2$  into  $H_2O$  and molecular oxygen and can also inhibit A $\beta$  aggregation (Luo *et al.*, 2014). On the other side, A $\beta$ -42 can thwart anti-oxidative and other functions of catalase by forming "catalase- A $\beta$ -42" complex and generate ROS. Thus, inhibitors of the "catalase- A $\beta$ -42" complex have been recommended for diminishing the OS generated by this interaction (Habib *et al.*, 2010).

Haptoglobin is another anti-oxidant protein that is either oxidized and/or downregulated in the AD patients (Cocciolo *et al.*, 2012). It also binds with the Apo-A1 of HDL and is involved in reverse cholesterol transportation (Alayash, 2011). Its direct influence on ApoE and A $\beta$  cross-talk has also been reported (Spagnuolo *et al.*, 2014).

In the current experiment, increased levels of peroxiredoxin-1, -2, -5 and -6 in the AD brains were observed that are indicative of increased OS in the AD model rats (Minjarez *et al.*, 2016; Poynton & Hampton, 2014). Peroxiredoxins are thioredoxin peroxidases capable of eliminating hydroperoxides through thioredoxin/thioredoxin reductase. Their oxidative modification has been linked with AD and other NDs (Minjarez *et al.*, 2013).

Thus, increased expression of both enzymatic and non-enzymatic anti-oxidant protein levels in the *G. lucidum* fed rats indicate the OS attenuating capacity associated with AD pathogenesis (Sultana *et al.*, 2007; Yao, Jun *et al.*, 2007).

#### 7.4.1.2 Protein-protein interaction (PPI) findings

In addition to functional, modular and pathway-related insights, PPI maps provide disease specific information. As identification of the target protein in any disease pathogenesis is an important aspect, PPI analyses shed light towards understanding the complex connectivity and identifying the protein of interest for further evaluation and management (Koh *et al.*, 2012).

Based on the analyses of the PPI networks and pathways of the differentially expressed proteins, it is obvious that AD causes a disturbed protein expression affecting the global protein-protein interactive networks and the relevant biological pathways. They have been categorized into functional framework of metabolic process, intracellular signaling cascade, signal transduction, oxidation reduction, cell communication, molecular transport, regulation of biological processes, regulation of cellular processes and apoptosis.

#### 1. **PPI among the up-regulated proteins**

#### A.

### PPI among up-regulated proteins of AD vs control group

a. **PPI network of the upregulated protein clusters in the AD vs control group** 

PPI interaction among the upregulated proteins in the AD versus control group can be divided into several network clusters (Figure 7.5). Among them, the first two consists mainly of the cytoskeletal proteins such as 14-3-3 proteins along with tubulin, actin, Ras-related proteins, cofilin, sirtuin, actin, myosin and the Ras-like proteins. The heat shock proteins (HSP) of different molecular weight and function formed another important network cluster (Figure 7.5). Calmodulin, Ca<sup>2+</sup>/calmodulin-dependent protein kinases, serine/threonine-protein phosphatase and synapsin-1 formed another functional interaction (Figure 7.5). Enzymatic proteins involved in metabolism and energy generation formed functional intearactions among themselves (Figure 7.5). There was strong interaction among the proteins (at high confidence score of 0.700). As some of the proteins had been involved in different functions, functional overlap of the proteins led them towards extended integration and interaction beyond any single class (Figure 7.5). The following figure demonstrates the identification keys used in interpretation of PPI analysis.

Node	Node size		Node color		
	⊖ Small	Large	Colored	White	
Network	Protein of	Some 3D	Query proteins	Second shell	
nodes	unknown 3D	structure is	and first shell of	of interactors	
represent	structure	known or	interactors		
proteins		predicted			

Edges	Known		Predicted		Others			
	interactions		interactions					
			$\hat{\mathbf{r}}$					P
Edges	From	Experi	Gene	Gene	Gene	Text	Co-	Protein
represent	curated	mentally	neigh	fusions	co-	mining	expressio	homo
protein-	data	determine	borhoo		occurrenc		n	logy
protein	bases	d	d		e			
association								
S								



Figure 7.5: PPI network of the up-regulated proteins of the AD vs Control group

# b. Pathway analysis of the upregulated proteins in the AD vs control group

Through KEGG (Kyoto Encyclopedia of Genes and Genomes) pathway analysis, 65 pathways were found to be significantly enriched (P<0.05) in the AD versus control group. Among them the most notable were the AD pathway (pathway ID 05010) involving the genes *Camk2a, Camk2b, Ndufs1, Ndufv2* and *Atp5o* (FDR 0.0176); the LTP pathway (pathway ID 04720) involving the genes *Calm1, Camk2a, Camk2b* and *Rap1a* (FDR 0.0037) and the neurotrophin signaling pathway (pathway ID 04722, FDR 0.0246) involving the genes *Calm1, Camk2b, Rap1a* and *Ywhae*. The P13K-Akt signaling pathway (pathway ID 04151, FDR 0.0161) involving the genes *Hsp90aa1, Ywhab, Ywhae, Ywhag, Ywhah* and *Ywhaq*; Ca<sup>2+</sup>signaling pathway (ID 04020, FDR

0.0176) with the genes *Calm1*, *Camk2a*, *Camk2b*, *Ppp3ca*, *Slc25a4* and *Slc25a5*; *cGMP-PKG* signaling pathway (ID 04022, FDR 0.0032) involving the genes *Camk2a*, *Camk2b*, *Atp1a1*, *Atp1b1* and *Slc25a5* were also among the interacted signaling pathways.

PFAM and INTERPRO protein domain analyses identified 14-3-3 (PF0024) as the most significantly enriched protein in the AD versus control group.

# B. PPI among upregulated proteins in AD vs AE group

#### a. PPI network of the upregulated protein clusters in the AD vs AE group

In the AD versus AE group, the most notable protein networks included those involved in the biological processes such as redox mechanisms, neurogenesis, neurotransmission, neuronal development and metabolism (Figure 7.6). Interacted protein networks involved in molecular functions such as SNARE binding, syntaxin binding, syntaxin-1 binding were also enriched (Figure 7.6).



Figure 7.6: PPI network of the up-regulated proteins in the AD vs AE group.

The main network of the upregulated proteins can be divided into several subsets with respect to their gene content such as sub-set 1 consisting of synaptogamin-1, syntaxin 1A, syntaxin 1B, beta soluble NSF attachment protein (*Syt1, Stx1a, Stx1b* and *Napb*), sub-set 2 involving AP2 forming proteins and clathrin heavy chain (*Ap2a2, Ap2a1, Cltc*), sub-set 3 consisting of metabolic enzymes NADH dehydrogenase, dihydrolipoyllysine-residue acetyltransferase component of pyruvate dehydrogenase complex, pyruvate dehydrogenase E1 component subunit beta and isocitrate dehydrogenase (*Ndufv2, Ndufa9, Dlat, Pdhb and Idh3a*) and those involved in signal transduction, membrane trafficking and cytoskeletal maintenance (*Ubb, Vcp, C3, Rab7a, Rab11b, Gpi* and *HK1*) (Figure 7.6). Besides the main network, two other protein-protein interaction networks involved immune-regulatory proteins (*Igg-2a, Igh-*

*la, IgkC* and *Crp*) and proteins associated with Ca<sup>2+</sup>-mediated signaling and metabolism *(Calm1, Camk2a, Camk2b* and *Ppp3ca)* (Figure 7.6).

#### b. Pathway analysis of the upregulated proteins in the AD vs AE group

Among the 21 enriched KEGG pathways, the highly notable were the synaptic vesicle cycle, LTP, AD, HIF-1 signaling, SNARE interaction in vesicular transport and those associated with glutametargic synapse.

PFAM protein domain analysis identified alpha adaptin AP2, C-terminal (pathway ID PF02296) and INTERPRO analysis showed clathrin adaptor, alpha adaptin AP2, appendage, C-terminal as the mostly enriched domain.

### A. PPI among upregulated proteins in the AE vs C group

#### a. PPI network of the upregulated protein clusters in the AE vs control group

The upregulated proteins constitute the main interaction network that can be splitted into several network sub-sets as depicted through the STRING analysis (Figure 7.7). The first sub-set consists of the genes related with microtubule assembly and cytoskeletal structure and function such as sirtuin 2 and tubulin (Figure 7.7). The proteins involved in this network are sirtuin2 and tubulin with the corresponding genes *Sirt2, Tuba1b, Tuba1c, Tubb2a, Tubb3, Tubb2b, Tuba4a, Tubb4a, Tubb4b, Tubb5* and *Tubb6* (Figure 7.7). The second network sub-set consists of the proteins linked with anti-oxidative defense (peroxidredoxins, SOD), metabolism (malate dehydrogenase, phosphoglutarate mutase, glucose 6 phosphate isomerase, pyruvate dehydrogenase) and molecular chaperone activity (polyubiquitin, T complex protein and HSP 10 KDa) (Figure 7.7). The interacted genes in this network are *Prdx2, Sod1, Mdh1, Pgm1, Gpi, Pdhb, Ubb, Cct3* and *Hspe1* (Figure 7.7).



Figure 7.7: PPI network of the up-regulated proteins in the C vs AE group.

The third sub-set consists of the cytoskeletal proteins including keratin, neurofilament heavy polypeptide, GFAP, Dsp, junction plakoglobin and carbamoyl phosphate synthase 1 (Figure 7.7). Interacted genes in this category are *Krt8, Nefh, Krt5, Krt17, Gfap, Krt14, Dsp, Jup* and *Cps1* (Figure 7.7).

Besides the main network, there were some small networks of which the first one consisted of the genes *Stx1a*, *Stx1b*, *Napb* and *Syt1* that are correspondent of the

proteins syntaxin 1a, syntax 1b, periplasmic nitrate reductase and synaptotagmin 1, respectively (Figure 7.7). The second small network incorporated histones *(Hist1h2aa, Hist1h2bo* and *Hist2h4)* and unknown protein LOC679994 (Figure 7.7).

### b. Pathway analysis of the upregulated proteins in the AE vs control group

KEGG pathway anlaysis showed gap junction, phagosome, synaptic vesicle cycle and carbon metabolism pathways as the enriched.

PFAM and INTERPRO protein domain analyses identified tubulin, tubulin/Ftsz family – C terminal domain and GTPase domains to be significantly (P<0.05) enriched.

### 2. PPI among the downregulated proteins

### A. PPI among the down-regulated proteins of the AD vs control group

Among the downregulated AD versus control group, the most notable interaction had been observed among the cytoskeletal proteins such as keratin isoforms (*Krt1, Krt2, Neflh, Krt5, Krt8, Krt10, Krt13, Krt15, Krt14, GFAP, Krt17, Krt42, Krt73*), junction plakoglobin (Jup), neurofilament heavy chain (Neflh) and glial fibrillary acidic protein (GFAP) (Figure 7.8). Downregulated anti-oxidant proteins peroxidredoxin 5, superoxide dismutase 1, protein disulfide isomerases (*Prdx5, Sod1, P4hb, Hspe1* and *Cct3*), oxoglutarate/malate carrier protein and alcohol dehydrogenase also formed a subnetwork (Figure 7.8). Besides, metabolic and ATP (*Atp5c1, Atp6v1d* and *Atp6v1e1*) generating proteins were also interacted with each other (Figure 7.8).



Figure 7.8: PPI network among the down-regulated proteins of the AD vs C group.

# B. PPI among down-regulated proteins of the AD vs AE group

In the AD versus AE group, the most notable interaction was among the downregulated proteins of anti-oxidant (SOD, GPX, peroxiredoxin, glutathione-S-transferases) activities (Figure 7.9). Besides, several sub-networks were also noted among the proteins involved in metabolism, energy generation and ribosomal structure and function related proteins (*Rpsa, Rps7, Rps2, Rpl13, Rplp0* and *Rpl14*) (Figure 7.9).



**Figure 7.9: PPI among the down-regulated proteins of the AD vs AE group.** Ribosome is the protein synthesis machinery and abnormality in its structure and protein synthesis activity highly affects AD pathomechanisms (Ferreira *et al.*, 2016). Thus, downregulation of the relevant proteins might be involved in impaired memory and learning activities of the AD rats (Ferreira *et al.*, 2016).

From the entire PPI analyses of the present study, it has been clearly understood that AD pathogenesis mostly affected the protein networks linked with neuronal cytoskeleton. Other affected networks include those associated with metabolism,  $Ca^{2+}$ dyshomeostasis, mitochondrial structure and function, signal transduction and anti-oxidant defense activities. Our findings are compatible with those of Minjarez *et al.* (2013), Zahid *et al.* (2014) and Hondius *et al.* (2016). Actually, derangement of the neurocytoskeleton is an important hallmark of AD pathogenesis and demands

considerable attraction in identifying AD therapeutics (Hwang *et al.*, 2015; Pollak *et al.*, 2003; Vickers *et al.*, 2016).

#### 7.4.1.3 Identification of functional pathway interaction through IPA

Based on ingenuity pathway analysis Knowledge Base (IPAKB), genes are transformed into relevant networks. In the network, relationships among the genes are expressed as the "edges" and genes become connected with each other only if there is any path among them in the global network. In this case, molecules from the dataset that are uploaded are called the "focus molecules". Core analysis of IPA was performed to interpret the datasets in the form of their functional networks. In the IPAKB, corresponding objects were mapped with the protein identifiers (James *et al.*, 2012) . Depending on the physical interaction (direct relationship) among the eligible proteins, IPAKB generated the networks and the score (probability value) of the networks (James *et al.*, 2012). Higher the network connectivity greater is the representation of significant biological functions of the relevant genes (James *et al.*, 2012). Statistical justification of the network connectivity is performed through measuring "p scores (-log10 p value, Fisher's exact test)" and "network score". Network score is also measured through Fisher's exact test that is based on the focus protein and biological functions and thus shows the relevancy of the analysis.

### A. Functional networks in AD vs C

IPA of the AD vs C identified 18 networks of which the top-most one had score of 90 having 93 focus molecules among 140 total selected molecules (Figure 7.10; Summary in Appendix F, Table F 11.1). The top-most network has been associated with cell death and survival, neurological diseases and psychological disorders.

The second network had score of 54 and focus molecules 64 among 140 total selected molecules (Figure 7.11). The relevant physiological system development and functions included molecular transport, cell-cell signaling and interaction, nervous system development and functions. Nervous system development and functions involved 40 molecules (p value 4.99E-02 - 3.48E-05). The other networks had score of 1 and focus molecule 1 and most were irelated to metabolism.

# Legends of the nodes and edges used in IPA

Red color is indicator of up-regulation and green color of down-regulation. Nodes and edges are indicators as follows:

0	Complex	$\mathfrak{S}$	Transcription Regulator
E	Enzyme	Ø	Transporter
Õ	Group/Complex/Other	0	Unknown
	Ion Channel	_	Direct Relationship
G	Phosphatase		Indirect Relationship



Figure 7.10: Top-scoring gene networks differentially expressed between AD and C group (top-most network 1).



Figure 7.11: Top-scoring gene networks differentially expressed between AD and C group (top-most network 2)

The top upstream regulators with corresponding p value of overlapping were PPARG (6.50E-05), NFE2L2 (3.34E-04), STAT3 (1.36E-03), CEBPA (5.25E-03), GATA4 (5.25E-03). Peroxisome proliferator activated receptors (PPARs) are the groups of nuclear hormone receptors that regulate lipid metabolism, energy production, metabolic balance between lipid and carbohydrates by acting as the lipid sensors (Kummer & Heneka, 2008). AD ameliorating effect of the non-steroidal anti-inflammatory drugs

(NSAIDs) have been associated with PPAR stimulating effect (Camacho *et al.*, 2004; Weggen *et al.*, 2001).

Neurological diseases were among the top diseases and bio-functions with 14 molecules and p value of 4.74E-02 - 5.26E-04. Among the top toxicity lists were OS (22.2%, 7.95E-11), mitochondrial dysfunction (12.7%, p 2.30E-12), LXR/RXR activation (14.4%, 5.73E-11), positive acute phase response proteins (34.6%, 2.43E-10) and FXR/RXR activation (13.0%, 9.72E-10).

The top-most up-regulated molecules with their log ratios were (THY1 18.403), ATP5F1 (18.175), GSTP1 (17.998), VAMP2 (17.922), Igh-6 (17.867), SLC1A3 (17.808), SOD2 (17.784), HSPA9 (17.354), TPPP (17.347) and RAB11B (17.254).

The top-most down-regulated molecules with their log ratios were DDT (-18.544), LOC259246 (-18.299), CYB5A (-18.203), SOD1 (-17.609), HIST1H3E (-17.581), DSG1 (-17.409), FABP1 (-17.253), HSPE1 (-17.230), LOC100911847 (-16.892) and JUP (-16.852).

Top canonical pathways with their corresponding overlap and p values were phagosome maturation (21.5%, 2.98E-19), 14-3-3-mediated signaling (16.1%, 2.37E-14), remodeling of epithelial adherens junctions (24.6%, 6.49E-14), epithelial adherens junction signaling (14.0%, 1.55E-12) and mitochondrial dysfunction (12.9%, 1.80E-12). 14-3-3-mediated signaling and the network associated affected proteins have been depicted in Figure 7.12. Besides, pathways associated with glucose metabolism (glycolysis and gluconeogenesis), TCA cycle, oxidative phosphorylation, unfolded protein response, HIPPO signaling and xenobiotic metabolism signaling were also among the notable functionally interacted ones. The HIPPO signaling is entails the protein kinase "Hippo (HPO)" and is evolutionarily conserved for developing the

mammalian nervous system. Its emerging role in AD pathogenesis has intrigued the current AD research field (Godoy *et al.*, 2014)



Figure 7.12: 14-3-3 protein-mediated signaling and associated proteins in the AD vs C group.

Consistent with our quantitative proteomics findings and interaction analysis by STRING, the proteins altered in their expression are also functionally related with each other. The most prominent proteins in this regard are those involved in nervous system development and function (YWHAE; NCAM1,NEFH,SIRT2,VSNL1; NCAM1, NEFH, SIRT2, VSNL1; NME2; UGT1A1; SEPT1; CAMK2A; 2,SEPT5,SNCA,SOD1; KNG1, TUB1A1). Proteins involved in anti-oxidative defense through free-radical scavenging (GPX1, GPX3, PRDX1, PRDX2, PRDX6, SNCA, SOD1 and SOD2) also

were functionally related. Similar trend was observed for the proteins involved in synaptic functions, cytoskeletal arrangements, microtubule-associated proteins, cellular stress response (especially the HSPs) and calcium binding. Our findings closely resemble those of Hondius *et al.*, (2016).

Besides, proteins involved in mitochondrial structure and functions were functionally closed. In the previous section under differential expression related description, we focused on the deranged expression pattern of the metabolism, energy generation and mitochondria-associated proteins. The IPA analysis tally with those and indicate that the AD mitochondria suffer from disrupted expression and functionality of the proteins involved in metabolism, energy generation, OS regulation and Ca<sup>2+</sup> homeostasis (James *et al.*, 2012). Another important aspect of neuronal mitochondria is maintenance of cellular dynamics so that mitochondrial "traffic jam" is overcome through proper translocation of mitochondria through the neuronal cytoskeleton (Correia *et al.*, 2016). Present findings of the affected proteins linked together in the AD subjects are "red signal" across the neuronal cytoskeletal "cross-roads". In line with the mitochondrial derangement, the functional interaction among the affected cytoskeleton and microtubule-associated proteins (especially tubulins) reinforce the observed jumble of the AD hippocampal proteins.

Proteins involved in metabolism of almost all the biomolecules have been affected in the AD groups. The most notable alteration was those metabolizing glucose and lipids (APOA1, HSD11B1, PHB, SERPINA1, SIRT2; Akr1c14, COMT, HSD11B1, Sult1a1; ACSL1, ALB, FABP1; ABAT, ACSL1, ALB, APOA4, APOE, CS, DLAT, DLD, F2, FABP1, KNG1, MDH1, MDH2, PDHB, RAC1, RGN, STX1A, SUCLA2). Top-ranking proteins involved in protein metabolism and interacted together were ACADM, ALDOC, ANXA5, CRP, EEF1A1, Gnmt/LOC100911564, HBA1/HBA2, HBB, HSD17B10, IVD, JUP, MAT1A, NEFL, NME2, SOD2, VCP and YWHAB. Metabolism-related proteins were linked in the peripheral nodes of the IPA.

### **B.** Functional network in AD vs AE

IPA of the AD vs AE identified 20 networks of which the top-most one had score of 75 and focus molecules 83 among 140 total selected molecules (Figure 7.13, Summary in Appendix F, Table F 11.2). The top-most network functions include cell death and survival, cell morphology and organ morphology. The second one had score of 56 and focus molecules 73 among 140 total selected molecules (Figure 7.14). The second-top network related functions include molecular transport, cell signaling, nervous system development and function categories. The third network had score of 8 and 25 focus molecules among 135 total selected molecules involved in CVD and functions, cellular and organismal development (Figure 7.15).



Figure 7.13: Top-scoring gene networks differentially expressed between AD ane AE group (top-most network 1).



Figure 7.14: Top-scoring gene networks differentially expressed between AD and AE group (top-most network 2).



Figure 7.15: Top-scoring gene networks differentially expressed between AD and AE group (top-most network 3).

The top physiological system development and functions with their corresponding p value range and involved molecules were nervous system development and function (4.99E-02 - 3.48E-05, 40), tissue morphology (3.35E-02 - 5.29E-05, 22), tissue development (4.99E-02 - 1.62E-04, 31), organ morphology (3.35E-02 - 2.52E-04, 6) and organismal development (3.35E-02 - 2.52E-04, 18).

The top upstream regulators with corresponding p value of overlapping were MCU (1.69E-05), PPARG (6.50E-05), PDIA6 (2.64E-04), GIMAP1-GIMAP5 (2.64E-04) and NFE2L2 (3.34E-04). Metabolic (p value 1.69E-02 - 1.68E-06, molecules involved 16) and neurological (p value 4.74E-02 - 5.26E-04, molecules involved 14) diseases were among the top physiological disorders affected.

The top-most up-regulated molecules with their log ratios were AP2A1 (17.321), TNR (17.234), UPB1 (17.067), HPD (16.998), IVD (16.946), SYP (16.918) and ATP6V0D1 (16.827).

The top-most down-regulated molecules with their log ratios were ACSM1 (-18.169), ALDH6A1 (-17.880), GSTA1 (-17.673), TST (-17.527), FABP1 (-17.339), GLUD1 (-17.265), CA3 (-17.190), ALDH8A1 (-17.183), RGN (-17.177) and SEPT5 (-17.092).

Proteins involved in cellular communication and signaling (CALR, F2, KNG1, KRAS; ARF1, GSTP1, HSPA8, RAP1A, RGN, SOD1, THY1; ABAT, CLU, NCAM1; RAC1, SNCA; CKB, NME2) were highly affected in the AD vs AE group and interacted together.

Another important aspect was the functional networks of the kinases (CAMK2A, CAMK2B, GSK3 $\beta$  and CDK5). These kinases are members of the post-synaptic phosphoproteome and their associated network points towards the disrupted phosphoproteome of the AD subjects (Zhou *et al.*, 2010).

Top-most canonical pathways were like those of AD vs C. Exceptional was the eNOS and melatonin signaling pathways. The eNOS signaling has been reported to be beneficial for AD as eNOS forms a feedback loop with pin1 and A $\beta$  that prevents deposition and stimulates clearance of A $\beta$  in the cerebral microvessels (Wang *et al.*,

2014). But, the process is disrupted in AD brains due especially to the aberrantly increased level of A $\beta$  when eNOS fails to perform its feed-back loop formation (Wang *et al.*, 2014).

Melatonin (N-acetyl-5-methoxytryptamine) is a pineal gland hormone whose level decreases in the aged and AD subjects (Lin *et al.*, 2013). Its neuroprotective role involves anti-oxidative, anti-A $\beta$  producing and anti-A $\beta$  fibrillation strategies (Lin *et al.*, 2013; Poeggeler *et al.*, 2001). Their regulatory roles on the protein kinases and phosphatases and protective effect upon cholinergic system have also been regarded (Lin *et al.*, 2013). The melatonin signaling network and the associated proteins in the present study highlights another therapeutic target of the damaged AD proteome (Cecon & Markus, 2011).

# C.Functional networks in AE vs C

IPA of the C vs AE identified 20 networks of which the top-most one had score of 73 and focus molecules 82 among 140 total selected molecules (Figure 7.16, Summary in Appendix F, Table F 11.3). The top-most network functions include cell death and survival, cell morphology and organ morphology. The second one had score of 56 and focus molecules 71 among 140 total selected molecules (Figure 7.17). The second-top network related functions include molecular transport, cell signaling, nervous system development and function categories. The third network had score of 8 and 25 focus molecules among 135 total selected molecules involved in CVD and functions, cellular and organismal development (Figure 7.18).



Figure 7.16: Top-scoring gene networks of the differentially expressed between C and AE group (top-most network 1).



Figure 7.17: Top-scoring gene networks of the differentially expressed between C vs AE group (top-most network 2).



Figure 7.18: Top-scoring gene networks differentially expressed between C and AE group (top-most network 3).

Nervous system development and function was the top-most feature of the physiological system development and function with 40 molecules and p values in the range of 4.99E-02 - 3.48E-05.

The top-most up-regulated molecules with their log ratios were DSG1 (17.409), LOC100911847 (16.892), JUP (16.852), MAT1A (16.707), STX1B (16.560), DSP (16.474) and STX1A (16.313).

The top-most down-regulated molecules with their log ratios were ACSM1 (-18.169), GSTA1 (-17.673), GSTM3 (-17.204), CA3 (-17.190), ALDH8A1 (-17.183), PDHA2 (-17.179), RGN (-17.177), HSPA9 (-17.175), ARF1 (-17.101) and SEPT5 (-17.092).

Mitochondrial Ca<sup>2+</sup> uniporter (MCU) was the top-most up-stream regulator. Its involvement in OS-mediated cellular damage has been confirmed recently (Liao *et al.*, 2015). Exceptionally noted canonical pathways in the C vs AE included actin cytoskeleton signaling, ROS and NOS production, NRF2 mediated OS response, RhoA signaling, Rho family GTPases, protein kinase A signaling, EIF2 signaling, ILK signaling and paxillin signaling pathways. These findings are consistent with those of STRING analysis in the sense that maintenance of neuronal cytoskeleton, mitochondrial function, metabolism proper and defense against free-radical induced damage are the top-notch concern of AD pathogenesis. Impairment in any one and/or combined system(s) poses threat towards AD vulnerability.

# 7.5 Conclusion

Up to present day, about 96 proteins have been reported to be linked with AD (Musunuri *et al.*, 2014). Most of the differentially expressed proteins identified in the present study fall within that category. The reason for not detecting direct expression of APP or A $\beta$  in the AD rats of the present study might be that A $\beta$  was infused intraventricularly and this doesnot fall under the category of expression or repression. But its effect is revealed through up- and/or – down regulation of multiple proteins in the AD subjects. Provided the same amount of protein being processed for quantification, variability in the output may be considered due to reduced accessibility of the AD proteins towards trypsin digestion as the proteins in AD subjects are much aggregated and less soluble compared to the controls. Importantly, inability to detect proteins of

target does not imply their absence in the sample; rather it is an indication of their differential expression. As the AD rats have been found to experience impaired memory and learning activities in the previous chapter and altered expression and networks (both interacting and functional pathways, as revealed by the STRING and IPA analysis, respectively) of the memory and learning related proteins in the present chapter, differential expression of the proteins might be attributed to the infused  $A\beta$  in the respective model animals. On the other hand, proteins up- and down-regulated in the control and mushroom treated animals might have been involved in improving those derangements and thus ameliorating the AD consequences as observed through memory related tests in the previous chapter. Another important feature of the present study is that some of the proteins differentially expressed in the AD model rats are related with cholesterol metabolism (such as apolipoprotein A-I, A-IV, E) and OS management (such as SOD, GPX, peroxiredoxin, glutathione -S-transferase). They were also highly interacted with each other and formed pathway-based functional networks. This interrelationship reinforces the OS and hypercholesterolemia links of AD pathogenesis. Thus, current findings suggests towards management of OS and hypercholesterolemia in addition with AD therapeutics per se. And incorporation of G. lucidum as an AD ameliorating agent seems pertinent. Differential expression of the proteins in the mushroom-fed AD rats might either be an adaptive response or protective strategy against Aβ-mediated stresses. Admittedly, the regulated proteins, identified in the current experiment but not previously been studied, warrants extensive exploration for much conclusive remarks. In this endeavor, western blot analysis of the mostly regulated proteins for validation of the current findings is the immediate future aspect of the present study.

### **CHAPTER 8: GENERAL DISCUSSION**

The present study was aimed at elucidating the anti-oxidative and cholesterol lowering capacities of the selected edible-medicinal mushrooms in AD amelioration. The selected edible mushrooms were *H. erinaceus*, *L. edodes* and *F. velutipes* while the medicinal representative was the G. lucidum. Though mushrooms have been regarded highly for their immense nutritive and medicinal values since ancient times, identification and utilization of those bio-components in human ailments have been practised recently. Contents of different bio-components enable mushrooms to provide nutritional and medicinal support in human health and diseases (Guillamón et al., 2010; Paterson, 2006; Rahman et al., 2015 and Wasser, 2011). Among various mushroom biocomponents, phenolics, polysaccharides and proteins have been ascribed for their health giving properties such as immunostimulating, anti-microbial, anti-cancer and anti-CVD (Guillamón et al., 2010; Paterson, 2006; Rahman et al., 2015 and Wasser, 2011). Although different aspects of different mushrooms have been explored worldwide, studies encompassing combined anti-oxidative, cholesterol lowering and AD amelioarating effect of edible-medicinal mushrooms are scarce. Been intrigued by this, the current comparative anti-oxidative studies of the aforementioned edible-medicinal mushrooms have been conducted followed by the identification of the bio-active components in the potent mushroom extracts and fractions as well as evaluated the effect of the most potent fraction upon cholesterol lowering, AD ameliorating and proteomics status. As activity of different bio-components varies depending on the extracting media, hydro-philic (aqueous, hot water and methanol), lipo-philic (DCM, hexane) and mixed (methanol:DCM) type of solvents were used in the present study to extract the corresponding solvent-solvent partitioned fractions and extracts (Mazzola et al., 2008).

Utilizing the mushroom fractions and extracts, both *in vitro* and *in vivo* anti-oxidative experiments have been performed in the present study. In vitro anti-oxidative tests included DPPH free radical scavenging, total phenolic assay, lipid peroxidation and LDL oxidation assays. Although, performance of the selected edible-medicinal mushrooms varied in different tests, G. lucidum showed the most potent in vitro antioxidative potentiality. Notably, the hydro-philic HWE of G. lucidim excelled in DPPH free radical scavenging and in Folin-Ciocalteu assay (chapter 3, section 3.3.1 and 3.3.2). Folin-Ciocalteu assay measured the reducing capacities of the mushroom fractions and extracts and compared to others, the observed reducing capacities were of higher values (Abdullah et al., 2011; Heleno et al., 2015; Kim et al., 2008; Mau et al., 2002; Rawat et al., 2013; Reis et al., 2012; Wong et al., 2009). On the contrary, the lipo-philic hexane fraction showed better inhibition towards lipid peroxidation and LDL oxidation (chapter 3, section 3.3.3 and 3.3.4). Subsequent identification of the bio-components in the respective extracts and fractions revealed the presence of different bio-components. Among the most notable hydrophilic bio-components were phenolics such as protocatechuic acid, p-coumaric acid, ellagic acid, cinnamic acid and triterpenoids such as ganoderic acids of different types (chapter 4, section 4.3.5, Table 4.5). Among the lipo-philic bio-components were ergosterol and fatty acids. In the present study, 11octadecanoic acid (vaccenic acid) was identified as a novel fatty acid present in the G. lucidum (chapter 4, section 4.3.1, Table 4.1). Current findings support the notion that solvents of higher polarity extract the phenolic compounds much than those of the nonpolar (Roby et al., 2013). Lipophilic bio-components present in the non-polar hexane might have imparted lipid peroxidation and LDL oxidation inhibition either becoming oxidized themselves or through quenching the free radicals (Turchi et al., 2009). Another important finding was the content of as much as 44.15% β-D-glucan in the HWE of G. lucidum (chapter 4, section 4.3.6). As the physiological properties of  $\beta$ -D-

glucan such as molecular size and structure as well as source and extracting condition affect the anti-oxidative performance, the hot water extraction process might have conferred favorable anti-oxidative arsenal to the  $\beta$ -D-glucan and the cumulative effect has been demonstrated as the enhanced anti-oxidative potency of the HWE of *G. lucidum* (Kofuji *et al.*, 2012).

As the HWE of *G. lucidum* showed better *in vitro* anti-oxidative potential, it was chosen for *in vivo* anti-oxidative and hypocholesterolemic tests. At 200 mg/kbw, this mushroom extract improved anti-oxidative defenses of the experimental animals as revealed by the increased levels of plasma catalase, glutathione peroxidase and superoxide dismutase (chapter 5, section 5.3.1,Tables 5.1 and 5.2). Present findings are in line with those of Hasnat *et al.* (2013), Jang *et al.* (2014) and Zhu *et al.* (2016). In hypocholesterolemic study, the same mushroom extract at the same dosage could reduce body weight gain and lower plasma TC, TG and LDL-C levels and increase HDL-C level (chapter 5, section 5.3.2, Figure 5.1 and Table 5.3). Similar observations have been noticed by Amin *et al.* (2011) and Chu *et al.* (2012). Importantly, feeding of the HWE of *G. lucidum* has not evoked any adverse health effects to the rats as evidenced by the organ function tests (chapter 5, section 5.3.3, Table 5.4).

For evaluating the AD ameliorating effect of the HWE of *G. lucidum*, the above mentioned dosage (200 mg/kbw) was applied to the intraventricularly  $A\beta_{1-42}$ -infuesed AD rat models and compared with the controls. AD specific tests included memory and learning related behavior tests (8-armed radial maze and novel object recognition tests) antibody and ELISA based tests and proteomics (Table 6.2). As described in the chapter 6 (section 6.3), improved memory and learning abilities have been observed in the *G. lucidum* HWE fed rats (Figures 6.4, 6.5, 6.6 and 6.7). As memory impairment is the main symptom affecting the AD subjects, its improvement is the prime target of AD

therapeutics (Ross, 2009; Davis & Couch, 2014). Thus, the observed memory enhancing effects seemed promising in proceeding with other aspects of AD in the present study. To substantiate the memory enhancing effect of the HWE of *G. lucidum*, neuro-biochemical tests had been conducted. Enhanced level of the synaptic and memory related proteins (BDNF, SNAP 25, PSD 95 and VAchT) and decreased anti- $A\beta_{(1-42)}$  oligomer and TNF $\alpha$  levels substantiated the AD ameliorating effect of the *G. lucidum* HWE (Table 6.2). Another set of evidence emanates from the TEM studies of the hippocampi of the rats. TEM studies showed that the neurons of the *G. lucidum* HWE fed rats' hippocampi possessed prolonged dendrites whereas those of the AD that possessed degenerative neurons and diffused A $\beta$  plaque like deposition (Figure 6.9).

In addition to direct abnormality with  $A\beta$ , altered expression of various proteins has been implicated in AD pathogenesis (Butterfield & Dalle, 2014). Current study also focused on that arena through proteomics of the selected rat groups (controlled, AD and the G. lucidum HWE fed AD rats). A total of 822 proteins were identified with minimum peptide 2, threshold 95.0% and peptide threshold at 0.1% FDR. Among them, 329 were differentially expressed (P<0.05) and 289 had log fold change of 1.5 and above (chapter 7, section 7.4). Number of proteins linked with AD, OS and hypercholesterolemia was 59, 20 and 12, respectively. Most of the down-regulated proteins in the AD rats were involved in neurotransmission, synaptic functions, stress response, mitochondrial oxidative phosphorylation and neuronal cytoskeleton maintenance (chapter 7, section 7.4). As a whole, number of the proteins differentially expressed in the AD rats was those involved in metabolism. PPI analyses of the significantly expressed proteins with log fold change value of 1.5 and above were showed different interacting networks among the proteins (chapter 7, section 7.4.1.2). Last of all, IPA analyses of the same proteins were performed that revealed the connectivity strength among different networks and identified the functional pathway
interactions among them (chapter 7, section 7.4.1.3). The top-most network has been associated with cell death and survival, neurological diseases and psychological disorders. The second top-most network involved physiological system development and functions included molecular transport, cell-cell signaling and interaction, nervous system development and functions. Findings of the present study closely resemble those of Musunury *et al.* (2014) and Hondius *et al.* (2016). Thus, the present study interprets the anti-oxidative and hypocholesterolemic effects of edible-medicinal mushrooms in AD amelioration. The *G. lucidum* HWE has been found as the interpretor and thus it can be regarded as an AD therapeutic agent.

## **CHAPTER 9: CONCLUSION**

Based on the ever increasing global AD burden and need of functional food-based therapeutic approach, the present study had been designed to judge the suitability of *G*. *lucidum* in ameliorating AD through anti-oxidative and hypocholesterolemic approaches. Perhaps, this is the first in type subjected to the combined anti-oxidative, cholesterol lowering and AD ameliorating effects of this and the other selected (*H. erinaceus, L. edodes* and *F. velutipes*) mushrooms.

Among the selected edible-medicinal mushrooms, the best one was screened through *in vitro* anti-oxidative tests (DPPH free radical scavenging, Folin-Ciocalteu assay, lipid peroxidation inhibition and LDL oxidation inhibition). Based on their *in vitro* anti-oxidative performances, the selected mushrooms could be ranked as *G. lucidum* > *H. erinaceus* > *L. edodes* > *F. velutipes*. Among the *in vitro* anti-oxidative outcomes, one important aspect was the identification of the higher reducing power of the mushroom fractions and extracts expressed as the gallic acid equivalents. Besides, lengthening of the lag time of CD formation by them was a unique approach in the chosen field and also for the selected mushrooms. Identification of 11-octadecanoic acid (vaccenic acid) in the *G. lucidum* was a novel finding. According to the β-D-glucan detecting kit information, presence of 44.15% β-D-glucan in the HWE of *G. lucidum* in the present study represents very close conformity to the highest amount of β-D-glucan ever detected.

*In vivo* studies showed that the hot water extract of *G. lucidum* at 200 mg/kbw could boost up the levels of plasma anti-oxidant enzymatic levels (catalase, glutathione peroxidase and superoxide dismutase), lower plasma and liver cholesterol levels without any adverse effect to the other organs. Beyond anti-oxidative studies, this outcome would be helpful for the hyperlipidemia and toxicity studies. Thus, the objectives of the

present study concerned with the anti-oxidative and cholesterol lowering performances have been fulfilled.

As oxidative stress and hypercholesterolemia have been implicated as the common causative factors of AD, the anti-oxidative and hypocholesterolemic outcomes of the *G. lucidum* HWE turned out promising in the amelioration of AD. Improved memory and learning related behavioral performances of the *G. lucidum* HWE-fed rats are the ground-breaking findings in respect of disrupted memory functioning of AD. Compared with the AD and age-matched controls, the elevated levels of the memory and learning related proteins in the *G. lucidum* HWE-fed rats substantiates the improved cognitive accomplishments highly. TEM outcomes of regular arrangement, prolonged neuronal dendrites and absence of A $\beta$  plaque-like diffused neurons in the mushroom-treated rats bear evidence to the neuroprotective feat.

Mushroom-treated AD rats' brain proteome has not been explored yet. Proteomics outputs support the already mentioned findings of the present study. The most noteworthy are the disrupted anti-oxidative proteomic defense, mitochondrial oxidative phosphorylation, metabolic enzymes, stress response and neuronal cytoskeleton. Network analyses of the significantly altered proteins also identified unique structural and functional interactions among them. A limitation of the present study was that the proteomics outcomes could not be verified through Western blot studies. This remains one of the future aspects of the current study.

Thus, through *in vitro* anti-oxidative tests, the current study screened the HWE of G. *lucidum* among the four edible-medicinal musthooms (*H. erinaceus, L. edodes, F. velutipes* and *G. lucidum*) and undertook *in vivo* anti-oxidative, cholesterol lowering and AD specific tests including proteomics. Observed results suggest the AD ameliorating effect of the hot water extract of *G. lucidum* that would be of immense importance in formulating functional food-based AD therapeutics. In this regard, studies entailing specific bio-component such as  $\beta$ -D glucan, ganoderic acid of any specific type or ergosterol could be recommended. At the same time, both acute and chronic toxicity tests of the experimental subjects are suggested.

## REFERENCES

- Abbott, L. F., & Nelson, S. B. (2000). Synaptic plasticity: taming the beast. *Nature Neuroscience*, *3*, 1178-1183.
- Abdullah, N., Ismail, S. M., Aminudin, N., Shuib, A. S., & Lau, B. F. (2011). Evaluation of selected culinary-medicinal mushrooms for antioxidant and ACE inhibitory activities. *Evidence-Based Complementary and Alternative Medicine*, 2012.
- Acay, A., Ulu, M. S., Ahsen, A., Ozkececi, G., Demir, K., Ozuguz, U., Yuksel, S., & Acarturk, G. (2014). Atherogenic index as a predictor of atherosclerosis in subjects with familial Mediterranean fever. *Medicina*, 50(6), 329-333.
- Agarwal, K., Chakarborthy, G., & Verma, S. (2012). In vitro antioxidant activity of different extracts of Ganoderma lucidum. DHR International Journal of Pharmaceutical Sciences 3(1), 48-54.
- Ahmad, R., Muniandy, S., Shukri, N. I. A., Alias, S. M. U., Hamid, A. A., Yusoff, W. M. W., Senafi, S., & Daud, F. (2014). Antioxidant properties and glucan compositions of various crude extracts from *Lentinus squarrosulus* mycelial culture. *Advances in Bioscience and Biotechnology*, 5(10), 805.
- Ahmed, M., Davis, J., Aucoin, D., Sato, T., Ahuja, S., Aimoto, S., Elliott, J. I., Van Nostrand, W. E., & Smith, S. O. (2010). Structural conversion of neurotoxic amyloid-[beta] 1-42 oligomers to fibrils. *Nature Structural & Molecular Biology*, 17(5), 561-567.
- Ahn, H. J., Zamolodchikov, D., Cortes-Canteli, M., Norris, E. H., Glickman, J. F., & Strickland, S. (2010). Alzheimer's disease peptide β-amyloid interacts with fibrinogen and induces its oligomerization. *Proceedings of the National Academy of Sciences*, 107(50), 21812-21817.
- Aida, F., Shuhaimi, M., Yazid, M., & Maaruf, A. (2009). Mushroom as a potential source of prebiotics: a review. *Trends in Food Science & Technology*, 20(11), 567-575.
- Aisen, P. S. (2002). The potential of anti-inflammatory drugs for the treatment of Alzheimer's disease. *The Lancet Neurology*, 1(5), 279-284.
- Aisen, P. S., Schafer, K. A., Grundman, M., Pfeiffer, E., Sano, M., Davis, K. L., Farlow, M. R., Jin, S., Thomas, R. G., & Thal, L. J. (2003). Effects of rofecoxib or naproxen vs placebo on Alzheimer disease progression: a randomized controlled trial. *JAMA*, 289(21), 2819-2826.
- Alayash, A. I. (2011). Haptoglobin: old protein with new functions. *Clinica Chimica Acta*, *412*(7), 493-498.
- Ali, M., Ahmed, S., Shamim, S., Kumar, R., Singh, J., & Kumar, A. (2014). Rejuvenating Consequence of *Ganoderma lucidium* on Cellular Integrity and Function of CCl4 Induced Liver of Mice. *International Journal of Current Microbiology and Applied Sciences*, 3(5), 142-147.

- Allain, C. C., Poon, L. S., Chan, C. S., Richmond, W., & Fu, P. C. (1974). Enzymatic determination of total serum cholesterol. *Clinical Chemistry*, 20(4), 470-475.
- Allman-Farinelli, M. A., Gomes, K., Favaloro, E. J., & Petocz, P. (2005). A diet rich in high-oleic-acid sunflower oil favorably alters low-density lipoprotein cholesterol, triglycerides, and factor VII coagulant activity. *Journal of the American Dietetic Association*, 105(7), 1071-1079.
- Álvarez, A., Cacabelos, R., Sanpedro, C., García-Fantini, M., & Aleixandre, M. (2007). Serum TNF-alpha levels are increased and correlate negatively with free IGF-I in Alzheimer disease. *Neurobiology of Aging*, 28(4), 533-536.
- Alzheimer's Association. (2016). Alzheimer's disease facts and figures. *Alzheimer's & Dementia* Retrieved 4, 12
- Amadoro, G., Ciotti, M. T., Costanzi, M., Cestari, V., Calissano, P., & Canu, N. (2006). NMDA receptor mediates tau-induced neurotoxicity by calpain and ERK/MAPK activation. *Proceedings of the National Academy of Sciences of the United States* of America, 103(8), 2892-2897.
- Amin, K. A., Kamel, H. H., & Eltawab, M. A. A. (2011). The relation of high fat diet, metabolic disturbances and brain oxidative dysfunction: modulation by hydroxy citric acid. *Lipids in Health and Diseases*, 10(1), 1.
- Anderson, N. L., & Anderson, N. G. (1998). Proteome and proteomics: new technologies, new concepts, and new words. *Electrophoresis*, 19(11), 1853-1861.
- Andreev, V. P., Petyuk, V. A., Brewer, H. M., Karpievitch, Y. V., Xie, F., Clarke, J., Camp, D., Smith, R. D., Lieberman, A. P., & Albin, R. L. (2012). Label-free quantitative LC–MS proteomics of Alzheimer's disease and normally aged human brains. *Journal of Proteome Research*, 11(6), 3053-3067.
- Ansar, S., Burlison, J. A., Hadden, M. K., Yu, X. M., Desino, K. E., Bean, J., Neckers, L., Audus, K. L., Michaelis, M. L., & Blagg, B. S. (2007). A non-toxic Hsp90 inhibitor protects neurons from Aβ-induced toxicity. *Bioorganic & Medicinal Chemistry Letters*, 17(7), 1984-1990.
- Ansari, M. A., & Scheff, S. W. (2010). Oxidative stress in the progression of Alzheimer disease in the frontal cortex. *Journal of Neuropathology and Experimental Neurology*, 69(2), 155.
- Anstey, K. J., Lipnicki, D. M., & Low, L.-F. (2008). Cholesterol as a risk factor for dementia and cognitive decline: a systematic review of prospective studies with meta-analysis. *The American Journal of Geriatric Psychiatry*, 16(5), 343-354.
- Antolovich, M., Prenzler, P. D., Patsalides, E., McDonald, S., & Robards, K. (2002). Methods for Testing Antioxidant Activity. *Analyst*, 127(1), 183-198.
- Apak, R., Gorinstein, S., Böhm, V., Schaich, K. M., Özyürek, M., & Güçlü, K. (2013). Methods of measurement and evaluation of natural antioxidant capacity/activity (IUPAC Technical Report). *Pure and Applied Chemistry*, 85(5), 957-998.

- Areosa, S. A., Sherriff, F., & McShane, R. (2005). Memantine for dementia. Cochrane Databse of Systematic Review, 20(3), doi:10.1002/14651858. CD14003154.pub14651854.
- Arhan, M., Öztürk, H. S., Turhan, N., Aytac, B., Güven, M. C., Olcay, E., & Durak, I. (2009). Hepatic oxidant/antioxidant status in cholesterol-fed rabbits: Effects of garlic extract. *Hepatology Research*, 39(1), 70-77.
- Atkinson, R. C., & Shiffrin, R. M. (1968). Human memory: A proposed system and its control processes. *The Psychology of Learning and Motivation*, 2, 89-195.
- Atta-Ur-Rahman, N. F., Choudhary, M. I., Malik, S., Makhmoor, T., Nur-E-Alam, M., Zareen, S., Lontsi, D., Ayafor, J., & Sondengam, B. (2001). New antioxidant and antimicrobial ellagic acid derivatives from *Pteleopsis hylodendron*. *Planta Medica*, 67(4), 335-339.
- Augustyniak, A., Bartosz, G., Čipak, A., Duburs, G., Horáková, L. U., Łuczaj, W., Majekova, M., Odysseos, A. D., Rackova, L., & Skrzydlewska, E. (2010). Natural and Synthetic Antioxidants: an updated overview. *Free Radical Research*, 44(10), 1216-1262.
- Baba, S., Natsume, M., Yasuda, A., Nakamura, Y., Tamura, T., Osakabe, N., Kanegae, M., & Kondo, K. (2007). Plasma LDL and HDL cholesterol and oxidized LDL concentrations are altered in normo-and hypercholesterolemic humans after intake of different levels of cocoa powder. *The Journal of Nutrition*, 137(6), 1436-1441.
- Babić, M., Švob Štrac, D., Mück-Šeler, D., Pivac, N., Stanić, G., Hof, P. R., & Šimić, G. (2014). Update on the core and developing cerebrospinal fluid biomarkers for Alzheimer disease. *Croatian Medical Journal*, 55(4), 347-365.
- Babincova, M., Bacova, Z., Machova, E., & Kogan, G. (2002). Antioxidant properties of carboxymethyl glucan: comparative analysis. *Journal of Medicinal Food*, *5*(2), 79-83.
- Bäckman, L., Jones, S., Berger, A. K., Laukka, E. J., & Small, B. (2004). Multiple cognitive deficits during the transition to Alzheimer's disease. *Journal of Internal Medicine*, 256(3), 195-204.
- Baldi, S., Maurich, T., Lubrano, V., Turchi, G., Bronzetti, G., Ferguson, L., & DeFlora, S. (2003). Antioxidant properties of the chalcone plicatin B. Paper presented at the Proceedings of the Eighth International Conference on Mechanisms of Antimutagenesis and Anticarcinogenesis, Pisa, Italy.
- Balducci, C., Beeg, M., Stravalaci, M., Bastone, A., Sclip, A., Biasini, E., Tapella, L., Colombo, L., Manzoni, C., & Borsello, T. (2010). Synthetic amyloid-β oligomers impair long-term memory independently of cellular prion protein. *Proceedings of the National Academy of Sciences*, 107(5), 2295-2300.
- Ballatore, C., Lee, V. M.-Y., & Trojanowski, J. Q. (2007). Tau-mediated neurodegeneration in Alzheimer's disease and related disorders. *Nature Reviews Neuroscience*, 8(9), 663-672.

- Bard, F., Cannon, C., Barbour, R., Burke, R.-L., Games, D., Grajeda, H., Guido, T., Hu, K., Huang, J., & Johnson-Wood, K. (2000). Peripherally administered antibodies against amyloid β-peptide enter the central nervous system and reduce pathology in a mouse model of Alzheimer disease. *Nature Medicine*, 6(8), 916-919.
- Barone, E., Mosser, S., & Fraering, P. C. (2014). Inactivation of brain Cofilin-1 by age, Alzheimer's disease and γ-secretase. *Biochimica et Biophysica Acta (BBA)-Molecular Basis of Disease*, 1842(12), 2500-2509.
- Barreira, J., Oliveira, M., & Ferreira, I. (2013). Optimized chromatographic analysis of ergosterol in the most appreciated wild and cultivated edible mushrooms. *Food Analytical Methods*.
- Barros, L., Dueñas, M., Ferreira, I. C., Baptista, P., & Santos-Buelga, C. (2009). Phenolic acids determination by HPLC–DAD–ESI/MS in sixteen different Portuguese wild mushrooms species. *Food and Chemical Toxicology*, 47(6), 1076-1079.
- Barros, L., Falcão, S., Baptista, P., Freire, C., Vilas-Boas, M., & Ferreira, I. C. (2008). Antioxidant activity of *Agaricus* sp. mushrooms by chemical, biochemical and electrochemical assays. *Food Chemistry*, 111(1), 61-66.
- Beel, A. J., Sakakura, M., Barrett, P. J., & Sanders, C. R. (2010). Direct binding of cholesterol to the amyloid precursor protein: An important interaction in lipid– Alzheimer's disease relationships? *Biochimica et Biophysica Acta (BBA)-Molecular and Cell Biology of Lipids*, 1801(8), 975-982.
- Benilova, I., Karran, E., & De Strooper, B. (2012). The toxic A [beta] oligomer and Alzheimer's disease: an emperor in need of clothes. *Nature Neuroscience*, 15(3), 349-357.
- Bennett, L., Sheean, P., Zabaras, D., & Head, R. (2013). Heat-stable components of wood ear mushroom, *Auricularia polytricha* (higher Basidiomycetes), inhibit *in vitro* activity of beta secretase (BACE1). International journal of medicinal *mushrooms*, 15(3).
- Berger, A., Rein, D., Kratky, E., Monnard, I., Hajjaj, H., Meirim, I., Piguet-Welsch, C., Hauser, J., Mace, K., & Niederberger, P. (2004). Cholesterol-lowering properties of *Ganoderma lucidum in vitro, ex vivo,* and in hamsters and minipigs. *Lipids in Health and Disease, 3*(1), 1.
- Bergeron, N., Phan, B. A. P., Ding, Y., Fong, A., & Krauss, R. M. (2015). Proprotein Convertase Subtilisin/Kexin type 9 inhibition a new therapeutic mechanism for reducing cardiovascular disease risk. *Circulation*, 132(17), 1648-1666.
- Berlett, B. S., & Stadtman, E. R. (1997). Protein oxidation in aging, disease, and oxidative stress. *Journal of Biological Chemistry*, 272(33), 20313-20316.
- Bermejo-Pareja, F., Benito-León, J., Vega, S., Medrano, M., Román, G., & Group, N. D. i. C. S. S. (2008). Incidence and subtypes of dementia in three elderly populations of central Spain. *Journal of the Neurological Sciences*, 264(1), 63-72.

- Bertram, L., McQueen, M. B., Mullin, K., Blacker, D., & Tanzi, R. E. (2007). Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. *Nature Genetics*, 39(1), 17-23.
- Bettcher, B. M., & Kramer, J. H. (2014). Longitudinal inflammation, cognitive decline and Alzheimer's Disease: a mini-review. *Clinical Pharmacology & Therapeutics*.
- Bevins, R. A., & Besheer, J. (2006). Object recognition in rats and mice: a one-trial non-matching-to-sample learning task to study'recognition memory'. *Nature Protocols*, *1*(3), 1306-1311.
- Bhakta, M., & Kumar, P. (2013). Mushroom polysaccharides as a potential prebiotics. *International Journal of Health Sciences and Research*, 3(8), 78.
- Bhaskar, K., Maphis, N., Xu, G., Varvel, N. H., Kokiko-Cochran, O. N., Weick, J. P., Staugaitis, S. M., Cardona, A., Ransohoff, R. M., & Herrup, K. (2014). Microglial derived tumor necrosis factor-α drives Alzheimer's disease-related neuronal cell cycle events. *Neurobiology of Disease*, 62, 273-285.
- Bhattacharyya, R., & Kovacs, D. M. (2010). ACAT inhibition and amyloid beta reduction. *Biochimica et Biophysica Acta (BBA)-Molecular and Cell Biology of Lipids*, 1801(8), 960-965.
- Binetti, G., Magni, E., Padovani, A., Cappa, S., Bianchetti, A., & Trabucchi, M. (1996). Executive dysfunction in early Alzheimer's disease. *Journal of Neurology*, *Neurosurgery & Psychiatry*, 60(1), 91-93.
- Björkhem, I. (2006). Crossing the barrier: oxysterols as cholesterol transporters and metabolic modulators in the brain. *Journal of Internal Medicine*, 260(6), 493-508.
- Blair, L. J., Zhang, B., & Dickey, C. A. (2013). Potential synergy between tau aggregation inhibitors and tau chaperone modulators. *Alzheimers Research and Therapy*, 5, 41.
- Blennow, K. (2004). Cerebrospinal fluid protein biomarkers for Alzheimer's disease. *NeuroRx*, 1(2), 213-225.
- Bobek, P., Kuniak, L., & Ozdin, L. (1993). The mushroom *Pleurotus ostreatus* reduces secretion and accelerates the fractional turnover rate of very-low-density lipoproteins in the rat. *Annals of Nutrition and Metabolism, 37*(3), 142-145.
- Bobek, P., & Ozdin, L. (1993). The mushroom *Pleurotus ostreatus* accelerates plasma very-low-density lipoprotein clearance in hypercholesterolemic rat. *Physiological research/Academia Scientiarum Bohemoslovaca*, 43(3), 205-206.
- Bogdanovic, N., Davidsson, P., Volkmann, I., Winblad, B., & Blennow, K. (2000). Growth-associated protein GAP-43 in the frontal cortex and in the hippocampus in Alzheimer's disease: an immunohistochemical and quantitative study. *Journal* of Neural Transmission, 107(4), 463-478.

- Bonsnes, R. W., & Taussky, H. H. (1945). On the colorimetric determination of creatinine by the Jaffe reaction. *Journal of Biological Chemistry*, 158(3), 581-591.
- Booth, K., Zhao, Y., Hua, Y., Wilks, K., Solomon, P., Lucas, P., Murray, E., Raje, S., & Black, R. S. (2012). Safety, tolerability and efficacy of lecozotan SR in patients with mild-to-moderate Alzheimer's disease used as monotherapy or adjunctively with a cholinesterase inhibitor. *Alzheimer's & Dementia*, 8(4), P604.
- Borger, E., Aitken, L., Du, H., Zhang, W., J Gunn-Moore, F., & Shi Du Yan, S. (2013). Is amyloid binding alcohol dehydrogenase a drug target for treating Alzheimer's disease? *Current Alzheimer Research*, 10(1), 21-29.
- Bose, M., Lambert, J. D., Ju, J., Reuhl, K. R., Shapses, S. A., & Yang, C. S. (2008). The major green tea polyphenol,(-)-epigallocatechin-3-gallate, inhibits obesity, metabolic syndrome, and fatty liver disease in high-fat-fed mice. *The Journal of Nutrition, 138*(9), 1677-1683.
- Bosetti, F., Brizzi, F., Barogi, S., Mancuso, M., Siciliano, G., Tendi, E. A., Murri, L., Rapoport, S. I., & Solaini, G. (2002). Cytochrome c oxidase and mitochondrial F 1 F 0-ATPase (ATP synthase) activities in platelets and brain from patients with Alzheimer's disease. *Neurobiology of Aging*, 23(3), 371-376.
- Botterweck, A., Verhagen, H., Goldbohm, R., Kleinjans, J., & Van den Brandt, P. (2000). Intake of butylated hydroxyanisole and butylated hydroxytoluene and stomach cancer risk: results from analyses in the Netherlands cohort study. *Food* and Chemical Toxicology, 38(7), 599-605.
- Boyd-Kimball, D., Sultana, R., Poon, H. F., Lynn, B., Casamenti, F., Pepeu, G., Klein, J., & Butterfield, D. (2005). Proteomic identification of proteins specifically oxidized by intracerebral injection of amyloid β-peptide (1–42) into rat brain: implications for Alzheimer's disease. *Neuroscience*, 132(2), 313-324.
- Brandalise, F., Cesaroni, V., Gregori, A., Repetti, M., Romano, C., Orrù, G., Botta, L., Girometta, C., Guglielminetti, M. L., & Savino, E. (2017). Dietary supplementation of *Hericium erinaceus* increases mossy fiber-CA3 hippocampal neurotransmission and recognition memory in wild-type mice. *Evidence-Based Complementary and Alternative Medicine*, 2017.
- Brawek, B., & Garaschuk, O. (2014). Network-wide dysregulation of calcium homeostasis in Alzheimer's disease. *Cell and tissue research*, 357(2), 427-438.
- Brigelius-Flohé, R., & Flohé, L. (2011). Basic principles and emerging concepts in the redox control of transcription factors. *Antioxidants & Redox Signaling*, 15(8), 2335-2381.
- Brown, T., Avenell, A., Edmunds, L. D., Moore, H., Whittaker, V., Avery, L., & Summerbell, C. (2009). Systematic review of long-term lifestyle interventions to prevent weight gain and morbidity in adults. *Obesity Reviews*, *10*(6), 627-638.
- Burnett, J. R., Wilcox, L. J., Telford, D. E., Kleinstiver, S. J., Barrett, P. H. R., Newton, R. S., & Huff, M. W. (1997). Inhibition of HMG-CoA reductase by atorvastatin

decreases both VLDL and LDL apolipoprotein B production in miniature pigs. *Arteriosclerosis, Thrombosis and Vascular Biology, 17*(11), 2589-2600.

- Burnett, J. R., Wilcox, L. J., Telford, D. E., Kleinstiver, S. J., Barrett, P. H. R., Newton, R. S., & Huff, M. W. (1999). The magnitude of decrease in hepatic very low density lipoprotein apolipoprotein B secretion is determined by the extent of 3hydroxy-3-methylglutaryl coenzyme A reductase inhibition in miniature pigs 1. *Endocrinology*, 140(11), 5293-5302.
- Burns, M., & Duff, K. (2003). Use of *in vivo* models to study the role of cholesterol in the etiology of Alzheimer's disease. *Neurochemical Research*, 28(7), 979-986.
- Burns, M., Gaynor, K., Olm, V., Mercken, M., LaFrancois, J., Wang, L., Mathews, P. M., Noble, W., Matsuoka, Y., & Duff, K. (2003). Presenilin redistribution associated with aberrant cholesterol transport enhances β-amyloid production in vivo. *The Journal of Neuroscience*, 23(13), 5645-5649.
- Bursill, C. A., Abbey, M., & Roach, P. D. (2007). A green tea extract lowers plasma cholesterol by inhibiting cholesterol synthesis and upregulating the LDL receptor in the cholesterol-fed rabbit. *Atherosclerosis*, 193(1), 86-93.
- Busciglio, J., Gabuzda, D. H., Matsudaira, P., & Yankner, B. A. (1993). Generation of beta-amyloid in the secretory pathway in neuronal and nonneuronal cells. *Proceedings of the National Academy of Sciences*, 90(5), 2092-2096.
- Busciglio, J., Pelsman, A., Wong, C., Pigino, G., Yuan, M., Mori, H., & Yankner, B. A. (2002). Altered metabolism of the amyloid β precursor protein is associated with mitochondrial dysfunction in Down's syndrome. *Neuron*, *33*(5), 677-688.
- Butler, D., Bendiske, J., Michaelis, M. L., Karanian, D. A., & Bahr, B. A. (2007). Microtubule-stabilizing agent prevents protein accumulation-induced loss of synaptic markers. *European Journal of Pharmacology*, 562(1), 20-27.
- Butterfield, D. A. (2004). Proteomics: a new approach to investigate oxidative stress in Alzheimer's disease brain. *Brain Research*, 1000(1), 1-7.
- Butterfield, D. A., Abdul, H. M., Opii, W., Newman, S. F., Joshi, G., Ansari, M. A., & Sultana, R. (2006a). Review: Pin1 in Alzheimer's disease. *Journal of neurochemistry*, 98(6), 1697-1706.
- Butterfield, D. A., & Boyd-Kimball, D. (2005). The critical role of methionine 35 in Alzheimer's amyloid β-peptide (1–42)-induced oxidative stress and neurotoxicity. *Biochimica et Biophysica Acta (BBA)-Proteins and Proteomics*, 1703(2), 149-156.
- Butterfield, D. A., Drake, J., Pocernich, C., & Castegna, A. (2001). Evidence of oxidative damage in Alzheimer's disease brain: central role for amyloid β-peptide. *Trends in Molecular Medicine*, *7*(12), 548-554.
- Butterfield, D. A., Hardas, S. S., & Lange, M. L. B. (2010). Oxidatively modified glyceraldehyde-3-phosphate dehydrogenase (GAPDH) and Alzheimer's disease:

many pathways to neurodegeneration. *Journal of Alzheimer's Disease*, 20(2), 369-393.

- Butterfield, D. A., & Lange, M. L. B. (2009). Multifunctional roles of enolase in Alzheimer's disease brain: beyond altered glucose metabolism. *Journal of Neurochemistry*, 111(4), 915-933.
- Butterfield, D. A., Perluigi, M., Reed, T., Muharib, T., Hughes, C. P., Robinson, R. A., & Sultana, R. (2012). Redox proteomics in selected neurodegenerative disorders: from its infancy to future applications. *Antioxidants & Redox Signaling*, 17(11), 1610-1655.
- Butterfield, D. A., Perluigi, M., & Sultana, R. (2006b). Oxidative stress in Alzheimer's disease brain: new insights from redox proteomics. *European Journal of Pharmacology*, 545(1), 39-50.
- Butterfield, D. A., & Sultana, R. (2007). Redox proteomics identification of oxidatively modified brain proteins in Alzheimer's disease and mild cognitive impairment: insights into the progression of this dementing disorder. *Journal of Alzheimer's Disease*, *12*(1), 61-72.
- Butterfield, D. A., Swomley, A. M., & Sultana, R. (2013). Amyloid β-peptide (1–42)induced oxidative stress in Alzheimer disease: importance in disease pathogenesis and progression. *Antioxidants & Redox signaling*, 19(8), 823-835.
- Buxbaum, J. D., Geoghagen, N., & Friedhoff, L. T. (2001). Cholesterol depletion with physiological concentrations of a statin decreases the formation of the Alzheimer amyloid Abeta peptide. *Journal of Alzheimer's disease: JAD*, 3(2), 221-229.
- Camacho, I. E., Serneels, L., Spittaels, K., Merchiers, P., Dominguez, D., & De Strooper, B. (2004). Peroxisome proliferator-activated receptor  $\gamma$  induces a clearance mechanism for the amyloid- $\beta$  peptide. *The Journal of Neuroscience*, 24(48), 10908-10917.
- Cao, Y., Xiao, Y., Ravid, R., & Guan, Z.-Z. (2010). Changed clathrin regulatory proteins in the brains of Alzheimer's disease patients and animal models. *Journal of Alzheimer's Disease*, 22(1), 329-342.
- Carluccio, M. A., Massaro, M., Bonfrate, C., Siculella, L., Maffia, M., Nicolardi, G., Distante, A., Storelli, C., & De Caterina, R. (1999). Oleic acid inhibits endothelial activation - a direct vascular antiatherogenic mechanism of a nutritional component in the Mediterranean diet. *Arteriosclerosis, Thrombosis* and Vascular Biology, 19(2), 220-228.
- Carro, E. (2010). Gelsolin as therapeutic target in Alzheimer's disease. *Expert Opinion* on Therapeutic Targets, 14(6), 585-592.
- Castegna, A., Aksenov, M., Thongboonkerd, V., Klein, J. B., Pierce, W. M., Booze, R., Markesbery, W. R., & Butterfield, D. A. (2002). Proteomic identification of oxidatively modified proteins in Alzheimer's disease brain. Part II: dihydropyrimidinase-related protein 2, α-enolase and heat shock cognate 71. *Journal of Neurochemistry*, 82(6), 1524-1532.

- Castegna, A., Thongboonkerd, V., Klein, J. B., Lynn, B., Markesbery, W. R., & Butterfield, D. A. (2003). Proteomic identification of nitrated proteins in Alzheimer's disease brain. *Journal of Neurochemistry*, 85(6), 1394-1401.
- Castellani, R., Hirai, K., Aliev, G., Drew, K. L., Nunomura, A., Takeda, A., Cash, A. D., Obrenovich, M. E., Perry, G., & Smith, M. A. (2002). Role of mitochondrial dysfunction in Alzheimer's disease. *Journal of Neuroscience Research*, 70(3), 357-360.
- Cataldo, A. M., Petanceska, S., Terio, N. B., Peterhoff, C. M., Durham, R., Mercken, M., Mehta, P. D., Buxbaum, J., Haroutunian, V., & Nixon, R. A. (2004). Aβ localization in abnormal endosomes: association with earliest Aβ elevations in AD and Down syndrome. *Neurobiology of Aging*, 25(10), 1263-1272.
- Cecon, E., & Regina, P. M. (2011). Relevance of the chronobiological and nonchronobiological actions of melatonin for enhancing therapeutic efficacy in neurodegenerative disorders. *Recent Patents on Endocrine, Metabolic & Immune Drug Discovery*, 5(2), 91-99.
- Chang, T.-Y., Chang, C. C., Lin, S., Yu, C., Li, B.-L., & Miyazaki, A. (2001). Roles of acyl-coenzyme A: cholesterol acyltransferase-1 and-2. *Current Opinion in Lipidology*, 12(3), 289-296.
- Chauhan, V., Ji, L., & Chauhan, A. (2008). Anti-amyloidogenic, anti-oxidant and antiapoptotic role of gelsolin in Alzheimer's disease. *Biogerontology*, 9(6), 381-389.
- Chen, K. H., Reese, E. A., Kim, H.-W., Rapoport, S. I., & Rao, J. S. (2011). Disturbed neurotransmitter transporter expression in Alzheimer disease brain. *Journal of Alzheimer's disease*, 26(4), 755.
- Cherny, R. A., Atwood, C. S., Xilinas, M. E., Gray, D. N., Jones, W. D., McLean, C. A., Barnham, K. J., Volitakis, I., Fraser, F. W., & Kim, Y.-S. (2001). Treatment with a copper-zinc chelator markedly and rapidly inhibits β-amyloid accumulation in Alzheimer's disease transgenic mice. *Neuron*, 30(3), 665-676.
- Cherubini, E., Le'vi, S., & Christophe, J. (2012). Role of the neuronal K-Cl cotransporter KCC2 in inhibitory and excitatory neurotransmission. *Building up the inhibitory synapse*, 31.
- Cheung, P. (2010). The nutritional and health benefits of mushrooms. *Nutrition Bulletin*, 35(4), 292-299.
- Cho, K.-H., Hong, J.-H., & Lee, K.-T. (2010). Monoacylglycerol (MAG)-oleic acid has stronger antioxidant, anti-atherosclerotic, and protein glycation inhibitory activities than MAG-palmitic acid. *Journal of Medicinal Food*, *13*(1), 99-107.
- Choi, J., Levey, A. I., Weintraub, S. T., Rees, H. D., Gearing, M., Chin, L.-S., & Li, L. (2004). Oxidative modifications and down-regulation of ubiquitin carboxylterminal hydrolase L1 associated with idiopathic Parkinson's and Alzheimer's diseases. *Journal of Biological Chemistry*, 279(13), 13256-13264.

- Chou, J. L., Shenoy, D. V., Thomas, N., Choudhary, P. K., LaFerla, F. M., Goodman, S. R., & Breen, G. A. (2011). Early dysregulation of the mitochondrial proteome in a mouse model of Alzheimer's disease. *Journal of Proteomics*, 74(4), 466-479.
- Chu, T. T., Benzie, I. F., Lam, C. W., Fok, B. S., Lee, K. K., & Tomlinson, B. (2012). Study of potential cardioprotective effects of *Ganoderma lucidum* (Lingzhi): results of a controlled human intervention trial. *British Journal of Nutrition*, 107(07), 1017-1027.
- Chuang, D.-M., Hough, C., & Senatorov, V. V. (2005). Glyceraldehyde-3-phosphate dehydrogenase, apoptosis and neurodegenerative diseases. *Annual Review of Pharmacology and Toxicology*, 45, 269-290.
- Cocciolo, A., Di Domenico, F., Coccia, R., Fiorini, A., Cai, J., Pierce, W., Mecocci, P., Butterfield, D., & Perluigi, M. (2012). Decreased expression and increased oxidation of plasma haptoglobin in Alzheimer disease: Insights from redox proteomics. *Free Radical Biology and Medicine*, 53(10), 1868-1876.
- Coleman, P. D., & Flood, D. G. (1987). Neuron numbers and dendritic extent in normal aging and Alzheimer's disease. *Neurobiology of Aging*, 8(6), 521-545.
- Combs, C. K., Karlo, J. C., Kao, S.-C., & Landreth, G. E. (2001).  $\beta$ -Amyloid stimulation of microglia and monocytes results in TNF $\alpha$ -dependent expression of inducible nitric oxide synthase and neuronal apoptosis. *The Journal of Neuroscience*, 21(4), 1179-1188.
- Corder, E., Saunders, A., Strittmatter, W., Schmechel, D., Gaskell, P., Small, G., Roses, A., Haines, J., & Pericak-Vance, M. A. (1993). Gene dose of apolipoprotein E type 4 allele and the risk of Alzheimer's disease in late onset families. *Science*, 261(5123), 921-923.
- Correia, S. C., Perry, G., & Moreira, P. I. (2016). Mitochondrial traffic jams in Alzheimer's disease-pinpointing the roadblocks. *Biochimica et Biophysica Acta* (*BBA*)-Molecular Basis of Disease.
- Cortes-Canteli, M., Zamolodchikov, D., Ahn, H. J., Strickland, S., & Norris, E. H. (2012). Fibrinogen and altered hemostasis in Alzheimer's disease. *Journal of Alzheimer's Disease*, *32*(3), 599-608.
- Cotman, C. W., & Berchtold, N. C. (2002). Exercise: a behavioral intervention to enhance brain health and plasticity. *Trends in Neurosciences*, 25(6), 295-301.
- Cottrell, B. A., Galvan, V., Banwait, S., Gorostiza, O., Lombardo, C. R., Williams, T., Schilling, B., Peel, A., Gibson, B., & Koo, E. H. (2005). A pilot proteomic study of amyloid precursor interactors in Alzheimer's disease. *Annals of Neurology*, 58(2), 277-289.
- Craig, L. A., Hong, N. S., & McDonald, R. J. (2011). Revisiting the cholinergic hypothesis in the development of Alzheimer's disease. *Neuroscience & Biobehavioral Reviews*, 35(6), 1397-1409.

- Cravatt, B. F., Simon, G. M., & Yates Iii, J. R. (2007). The biological impact of massspectrometry-based proteomics. *Nature*, 450(7172), 991-1000.
- Cuadrado-Tejedor, M., Vilariño, M., Cabodevilla, F., Del Río, J., Frechilla, D., & Pérez-Mediavilla, A. (2011). Enhanced expression of the voltage-dependent anion channel 1 (VDAC1) in Alzheimer's disease transgenic mice: an insight into the pathogenic effects of amyloid-β. *Journal of Alzheimer's Disease*, 23(2), 195-206.
- Cui, Y., Huang, M., He, Y., Zhang, S., & Luo, Y. (2011). Genetic ablation of apolipoprotein A-IV accelerates Alzheimer's disease pathogenesis in a mouse model. *The American Journal of Pathology*, 178(3), 1298-1308.
- Cumming, R. C., & Schubert, D. (2005). Amyloid-β induces disulfide bonding and aggregation of GAPDH in Alzheimer's disease. *The FASEB Journal*, 19(14), 2060-2062.
- da Silva Marineli, R., Furlan, C. P. B., & Maróstica, M. R. (2012). Antioxidant effects of the combination of conjugated linoleic acid and phytosterol supplementation in Sprague–Dawley rats. *Food Research International*, *49*(1), 487-493.
- Daker, M., Abdullah, N., Vikineswary, S., Goh, P., & Kuppusamy, U. (2008). Antioxidant from maize and maize fermented by *Marasmiellus* sp. as stabiliser of lipid-rich foods. *Food Chemistry*, 107(3), 1092-1098.
- Daleke, D. L., & Lyles, J. V. (2000). Identification and purification of aminophospholipid flippases. *Biochimica et Biophysica Acta (BBA)-Molecular* and Cell Biology of Lipids, 1486(1), 108-127.
- Dalle-Donne, I., Rossi, R., Colombo, R., Giustarini, D., & Milzani, A. (2006). Biomarkers of oxidative damage in human disease. *Clinical chemistry*, 52(4), 601-623.
- Danysz, W., & Parsons, C. G. (2012). Alzheimer's disease, β-amyloid, glutamate, NMDA receptors and memantine–searching for the connections. *British Journal of Pharmacology*, *167*(2), 324-352.
- Davis, J., & Couch, R. (2014). Strategizing the development of Alzheimer's therapeutics. Advances in Alzheimer's Disease, 2014.
- De Leo, M., Borrello, S., Passantino, M., Palazzotti, B., Mordente, A., Daniele, A., Filippini, V., Galeotti, T., & Masullo, C. (1998). Oxidative stress and overexpression of manganese superoxide dismutase in patients with Alzheimer's disease. *Neuroscience Letters*, 250(3), 173-176.
- de Sauvage Nolting, P. R., Kusters, D. M., Hutten, B. A., Kastelein, J. J., & Group, E. S. (2011). Serum bilirubin levels in familial hypercholesterolemia: a new risk marker for cardiovascular disease? *Journal of Lipid Research*, 52(9), 1755-1759.
- De Silva, D. D., Rapior, S., Sudarman, E., Stadler, M., Xu, J., Alias, S. A., & Hyde, K. D. (2013). Bioactive metabolites from macrofungi: ethnopharmacology, biological activities and chemistry. *Fungal Diversity*, 62(1), 1-40.

- De Strooper, B. (2014). Lessons from a failed  $\gamma$ -secretase Alzheimer trial. *Cell*, 159(4), 721-726.
- DeKosky, S. T. (2005). Statin therapy in the treatment of Alzheimer disease: what is the rationale? *The American Journal of Medicine*, *118*(12), 48-53.
- DeMattos, R. B., Bales, K. R., Cummins, D. J., Paul, S. M., & Holtzman, D. M. (2002). Brain to plasma amyloid-β efflux: a measure of brain amyloid burden in a mouse model of Alzheimer's disease. *Science*, 295(5563), 2264-2267.
- Dianzani, M., Barrera, G., Alvarez, S., Evelson, P., & Boveris, A. (2008). Pathology and physiology of lipid peroxidation and its carbonyl products. *Free Radical Pathophysiology*, 19-38.
- Dickey, C. A., Kamal, A., Lundgren, K., Klosak, N., Bailey, R. M., Dunmore, J., Ash, P., Shoraka, S., Zlatkovic, J., & Eckman, C. B. (2007). The high-affinity HSP90-CHIP complex recognizes and selectively degrades phosphorylated tau client proteins. *The Journal of Clinical Investigation*, 117(3), 648-658.
- Dickson, D., Golde, T., & McGowan, E. (2007). Abeta40 inhibits amyloid deposition *in vivo. Journal of Neuroscience*, *27*, 627633.
- Dietschy, J. M., & Turley, S. D. (2001). Cholesterol metabolism in the brain. *Current Opinion in Lipidology*, 12(2), 105-112.
- Díez-Vives, C., Gay, M., García-Matas, S., Comellas, F., Carrascal, M., Abian, J., Ortega-Aznar, A., Cristòfol, R., & Sanfeliu, C. (2009). Proteomic study of neuron and astrocyte cultures from senescence-accelerated mouse SAMP8 reveals degenerative changes. *Journal of Neurochemistry*, 111(4), 945-955.
- Ding, Q., Dimayuga, E., & Keller, J. N. (2007). Oxidative damage, protein synthesis, and protein degradation in Alzheimer's disease. *Current Alzheimer Research*, 4(1), 73-79.
- Djakovic, S. N., Schwarz, L. A., Barylko, B., DeMartino, G. N., & Patrick, G. N. (2009). Regulation of the proteasome by neuronal activity and calcium/calmodulin-dependent protein kinase II. *Journal of Biological Chemistry*, 284(39), 26655-26665.
- Dobiasova, M. (2006). AIP--atherogenic index of plasma as a significant predictor of cardiovascular risk: from research to practice. *Vnitrni Lekarstvi*, 52(1), 64-71.
- Dobiášová, M., & Frohlich, J. (2001). The plasma parameter log (TG/HDL-C) as an atherogenic index: correlation with lipoprotein particle size and esterification rate inapob-lipoprotein-depleted plasma (FER HDL). *Clinical Biochemistry*, 34(7), 583-588.
- Dudhgaonkar, S., Thyagarajan, A., & Sliva, D. (2009). Suppression of the inflammatory response by triterpenes isolated from the mushroom *Ganoderma lucidum*. *International Immunopharmacology*, 9(11), 1272-1280.

- Duka, A., Fotakis, P., Georgiadou, D., Kateifides, A., Tzavlaki, K., von Eckardstein, L., Stratikos, E., Kardassis, D., & Zannis, V. I. (2013). ApoA-IV promotes the biogenesis of apoA-IV-containing HDL particles with the participation of ABCA1 and LCAT. *Journal of Lipid Research*, 54(1), 107-115.
- Dumont, M., Wille, E., Stack, C., Calingasan, N. Y., Beal, M. F., & Lin, M. T. (2009). Reduction of oxidative stress, amyloid deposition, and memory deficit by manganese superoxide dismutase overexpression in a transgenic mouse model of Alzheimer's disease. *The FASEB Journal*, 23(8), 2459-2466.

Durrington, P. (2003). Dyslipidaemia. The Lancet, 362(9385), 717-731.

- Ehehalt, R., Keller, P., Haass, C., Thiele, C., & Simons, K. (2003). Amyloidogenic processing of the Alzheimer β-amyloid precursor protein depends on lipid rafts. *The Journal of Cell Biology*, 160(1), 113-123.
- Ehrlich, D., Pirchl, M., & Humpel, C. (2012). Effects of long-term moderate ethanol and cholesterol on cognition, cholinergic neurons, inflammation, and vascular impairment in rats. *Neuroscience*, 205, 154-166.
- Eichner, J. E., Dunn, S. T., Perveen, G., Thompson, D. M., Stewart, K. E., & Stroehla,
  B. C. (2002). Apolipoprotein E polymorphism and cardiovascular disease: a HuGE review. *American Journal of Epidemiology*, 155(6), 487-495.
- Eik, L.-F., Naidu, M., David, P., Wong, K.-H., Tan, Y.-S., & Sabaratnam, V. (2011). Lignosus rhinocerus (Cooke) Ryvarden: A medicinal mushroom that stimulates neurite outgrowth in PC-12 cells. Evidence-Based Complementary and Alternative Medicine, 2012.
- Einarsson, K., Ericsson, S., Ewerth, S., Reihner, E., Rudling, M., Ståhlberg, D., & Angelin, B. (1991). Bile acid sequestrants: mechanisms of action on bile acid and cholesterol metabolism. *European Journal of Clinical Pharmacology*, 40(1), S53-S58.
- El Shawi, O. E., El-Rahman, S. S. A., & El Hameed, M. A. (2015). Reishi mushroom attenuates hepatic inflammation and fibrosis induced by irradiation enhanced carbon tetrachloride in rat model. *Journal of Biosciences and Medicines, 3*(10), 24.
- Elisaf, M. (2002). Effects of fibrates on serum metabolic parameters. *Current Medical Research and Opinion*, 18(5), 269-276.
- Ellis, R., Olichney, J. M., Thal, L., Mirra, S., Morris, J., Beekly, D., & Heyman, A. (1996). Cerebral amyloid angiopathy in the brains of patients with Alzheimer's disease The CERAD experience, part XV. *Neurology*, 46(6), 1592-1596.
- Engelhart, M. J., Geerlings, M. I., Ruitenberg, A., van Swieten, J. C., Hofman, A., Witteman, J. C., & Breteler, M. M. (2002). Dietary intake of antioxidants and risk of Alzheimer disease. *JAMA*, 287(24), 3223-3229.
- Etcheberrigaray, R., Tan, M., Dewachter, I., Kuipéri, C., Van der Auwera, I., Wera, S., Qiao, L., Bank, B., Nelson, T. J., & Kozikowski, A. P. (2004). Therapeutic

effects of PKC activators in Alzheimer's disease transgenic mice. *Proceedings of the National Academy of Sciences of the United States of America*, 101(30), 11141-11146.

- Evans, C. G., Wisén, S., & Gestwicki, J. E. (2006). Heat shock proteins 70 and 90 inhibit early stages of amyloid β-(1–42) aggregation *in vitro*. *Journal of Biological Chemistry*, 281(44), 33182-33191.
- Fassbender, K., Simons, M., Bergmann, C., Stroick, M., Lütjohann, D., Keller, P., Runz, H., Kühl, S., Bertsch, T., & Von Bergmann, K. (2001). Simvastatin strongly reduces levels of Alzheimer's disease β-amyloid peptides Aβ42 and Aβ40 in vitro and in vivo. *Proceedings of the National Academy of Sciences*, 98(10), 5856-5861.
- Ferguson, L. R., Zhu, S. t., & Harris, P. J. (2005). Antioxidant and antigenotoxic effects of plant cell wall hydroxycinnamic acids in cultured HT-29 cells. *Molecular Nutrition & Food Research*, 49(6), 585-593.
- Ferreira, E., Shaw, D., & Oddo, S. (2016). Identification of learning-induced changes in protein networks in the hippocampi of a mouse model of Alzheimer's disease. *Translational Psychiatry*, 6(7), e849.
- Ferreira, S. T., & Klein, W. L. (2011). The Aβ oligomer hypothesis for synapse failure and memory loss in Alzheimer's disease. *Neurobiology of Learning and Memory*, 96(4), 529-543.
- Festa, F., Aglitti, T., Duranti, G., Ricordy, R., Perticone, P., & Cozzi, R. (2000). Strong antioxidant activity of ellagic acid in mammalian cells *in vitro* revealed by the comet assay. *Anticancer Research*, 21(6A), 3903-3908.
- Finkel, T., & Holbrook, N. J. (2000). Oxidants, oxidative stress and the biology of ageing. *Nature*, 408(6809), 239-247.
- Flynn, J. M., & Melov, S. (2013). SOD2 in mitochondrial dysfunction and neurodegeneration. *Free Radical Biology and Medicine*, 62, 4-12.
- Foote, M., & Zhou, Y. (2012). 14-3-3 proteins in neurological disorders. *International Journal of Biochemistry and Molecular Biology*, 3(2), 152-164.
- Forman, H. J., Ursini, F., & Maiorino, M. (2014). An overview of mechanisms of redox signaling. *Journal of Molecular and Cellular Cardiology*, 73, 2-9.
- Förstl, H., & Kurz, A. (1999). Clinical features of Alzheimer's disease. *European* Archives of Psychiatry and Clinical Neuroscience, 249(6), 288-290.
- Frantz, E., Menezes, H. S., Lange, K. C., Abegg, M. P., Correa, C. A., Zangalli, L., Vieira, J. L., & Zettler, C. G. (2012). The effect of maternal hypercholesterolemia on the placenta and fetal arteries in rabbits. *Acta Cirurgica Brasileira*, 27(1), 7-12.

- Frautschy, S., Hu, W., Kim, P., Miller, S., Chu, T., Harris-White, M., & Cole, G. (2001). Phenolic anti-inflammatory antioxidant reversal of Aβ-induced cognitive deficits and neuropathology. *Neurobiology of Aging*, 22(6), 993-1005.
- Freir, D. B., Holscher, C., & Herron, C. E. (2001). Blockade of long-term potentiation by β-amyloid peptides in the CA1 region of the rat hippocampus *in vivo*. *Journal of Neurophysiology*, *85*(2), 708-713.
- Friedewald, W. T., Levy, R. I., & Fredrickson, D. S. (1972). Estimation of the concentration of low-density lipoprotein cholesterol in plasma, without use of the preparative ultracentrifuge. *Clinical Chemistry*, 18(6), 499-502.
- Friguet, B. (2006). Oxidized protein degradation and repair in ageing and oxidative stress. *FEBS Letters*, 580(12), 2910-2916.
- Fu, H., Subramanian, R. R., & Masters, S. C. (2000). 14-3-3 proteins: structure, function, and regulation. Annual Review of Pharmacology and Toxicology, 40(1), 617-647.
- Fujita, Y., & Yamashita, T. (2014). Axon growth inhibition by RhoA/ROCK in the central nervous system. *Frontiers in Neuroscience*, *8*, 338.
- Fulga, T. A., Elson-Schwab, I., Khurana, V., Steinhilb, M. L., Spires, T. L., Hyman, B. T., & Feany, M. B. (2007). Abnormal bundling and accumulation of F-actin mediates tau-induced neuronal degeneration in vivo. *Nature Cell Biology*, 9(2), 139-148.
- Galani, C., & Schneider, H. (2007). Prevention and treatment of obesity with lifestyle interventions: review and meta-analysis. *International Journal of Public Health*, *52*(6), 348-359.
- Gao, X., He, Y., Gao, L.-M., Feng, J., Xie, Y., Liu, X., & Liu, L. (2014). Ser9phosphorylated GSK3β induced by 14-3-3ζ actively antagonizes cell apoptosis in a NF-κB dependent manner. *Biochemistry and Cell Biology*, 92(5), 349-356.
- Garbarino, V. R., Orr, M. E., Rodriguez, K. A., & Buffenstein, R. (2015). Mechanisms of oxidative stress resistance in the brain: lessons learned from hypoxia tolerant extremophilic vertebrates. *Archives of Biochemistry and Biophysics*, 576, 8-16.
- Gauvain, G., Chamma, I., Chevy, Q., Cabezas, C., Irinopoulou, T., Bodrug, N., Carnaud, M., Lévi, S., & Poncer, J. C. (2011). The neuronal K-Cl cotransporter KCC2 influences postsynaptic AMPA receptor content and lateral diffusion in dendritic spines. *Proceedings of the National Academy of Sciences*, 108(37), 15474-15479.
- Geerts, H. (2004). NC-531 (Neurochem). Current Opinion in Investigational Drugs (London, England: 2000), 5(1), 95-100.
- Geschwind, D. H. (2003). Tau phosphorylation, tangles, and neurodegeneration: the chicken or the egg? *Neuron*, 40(3), 457-460.

- Gezen-Ak, D., Dursun, E., Hanağası, H., Bilgiç, B., Lohman, E., Araz, Ö. S., Atasoy, İ. L., Alaylıoğlu, M., Önal, B., & Gürvit, H. (2013). BDNF, TNFα, HSP90, CFH, and IL-10 serum levels in patients with early or late onset Alzheimer's disease or mild cognitive impairment. *Journal of Alzheimer's Disease*, 37(1), 185-195.
- Gholamhoseinian, A., Shahouzehi, B., & Sharifi-Far, F. (2010). Inhibitory activity of some plant methanol extracts on 3-hydroxy-3-methylglutaryl coenzyme a reductase. *International Journal of Pharmacology*, 6(5), 705-711.
- Ghosh, A. K., Brindisi, M., & Tang, J. (2012). Developing β-secretase inhibitors for treatment of Alzheimer's disease. *Journal of Neurochemistry*, *120*(s1), 71-83.
- Giannakopoulos, P., Herrmann, F., Bussiere, T., Bouras, C., Kövari, E., Perl, D., Morrison, J., Gold, G., & Hof, P. (2003). Tangle and neuron numbers, but not amyloid load, predict cognitive status in Alzheimer's disease. *Neurology*, 60(9), 1495-1500.
- Gil-Ramírez, A., Clavijo, C., Palanisamy, M., Ruiz-Rodríguez, A., Navarro-Rubio, M., Pérez, M., Marín, F. R., Reglero, G., & Soler-Rivas, C. (2013). Screening of edible mushrooms and extraction by pressurized water (PWE) of 3-hydroxy-3methyl-glutaryl CoA reductase inhibitors. *Journal of Functional Foods*, 5(1), 244-250.
- Gil-Ramirez, A., Clavijo, C., Palanisamy, M., Soler-Rivas, C., Ruiz-Rodriguez, A., Marín, F. R., Reglero, G., & Pérez, M. (2011). *Edible mushrooms as potential sources of new hypocholesterolemic compounds*. Paper presented at the Proceedings of the 7th International Conference on Mushroom Biology and Mushroom Products.
- Gil-Ramírez, A., Clavijo, C., Palanisamy, M., Ruiz-Rodríguez, A., Navarro-Rubio, M., Pérez, M., Marín, F. R., Reglero, G., & Soler-Rivas, C. (2013). Study on the 3hydroxy-3-methyl-glutaryl CoA reductase inhibitory properties of *Agaricus bisporus* and extraction of bioactive fractions using pressurised solvent technologies. *Journal of the Science of Food and Agriculture*, 93(11), 2789-2796.
- Gille, A., Bodor, E. T., Ahmed, K., & Offermanns, S. (2008). Nicotinic acid: pharmacological effects and mechanisms of action. *Annual Review of Pharmacology and Toxicology.*, 48, 79-106.
- Gillette, M. A., & Carr, S. A. (2013). Quantitative analysis of peptides and proteins in biomedicine by targeted mass spectrometry. *Nature Methods*, 10(1), 28-34.
- Godoy, J. A., Rios, J. A., Zolezzi, J. M., Braidy, N., & Inestrosa, N. C. (2014). Signaling pathway cross talk in Alzheimer's disease. *Cell Communication and Signaling*, 12(1), 1.
- Goedert, M., & Jakes, R. (2005). Mutations causing neurodegenerative tauopathies. Biochimica et Biophysica Acta (BBA)-Molecular Basis of Disease, 1739(2), 240-250.

- Gokce, E. C., Kahveci, R., Atanur, O. M., Gürer, B., Aksoy, N., Gokce, A., Sargon, M. F., Cemil, B., Erdogan, B., & Kahveci, O. (2015). Neuroprotective effects of *Ganoderma lucidum* polysaccharides against traumatic spinal cord injury in rats. *Injury*, 46(11), 2146-2155.
- Golde, T. E., Eckman, C. B., & Younkin, S. G. (2000). Biochemical detection of Aβ isoforms: implications for pathogenesis, diagnosis, and treatment of Alzheimer's disease. *Biochimica et Biophysica Acta (BBA)-Molecular Basis of Disease*, 1502(1), 172-187.
- Gong, B., Cao, Z., Zheng, P., Vitolo, O. V., Liu, S., Staniszewski, A., Moolman, D., Zhang, H., Shelanski, M., & Arancio, O. (2006). Ubiquitin hydrolase Uch-L1 rescues β-amyloid-induced decreases in synaptic function and contextual memory. *Cell*, 126(4), 775-788.
- Görg, A., Weiss, W., & Dunn, M. J. (2004). Current two-dimensional electrophoresis technology for proteomics. *Proteomics*, 4(12), 3665-3685.
- Götz, J., Ittner, L. M., & Kins, S. (2006). Do axonal defects in tau and amyloid precursor protein transgenic animals model axonopathy in Alzheimer's disease? *Journal of Neurochemistry*, *98*(4), 993-1006.
- Granholm, A.-C., Bimonte-Nelson, H. A., Moore, A. B., Nelson, M. E., Freeman, L. R., & Sambamurti, K. (2008). Effects of a saturated fat and high cholesterol diet on memory and hippocampal morphology in the middle-aged rat. *Journal of Alzheimer's Disease*, 14(2), 133-145.
- Gray, N. E., Harris, C. J., Quinn, J. F., & Soumyanath, A. (2016). *Centella asiatica* modulates antioxidant and mitochondrial pathways and improves cognitive function in mice. *Journal of Ethnopharmacology*.
- Guillamón, E., García-Lafuente, A., Lozano, M., Rostagno, M. A., Villares, A., & Martínez, J. A. (2010). Edible mushrooms: role in the prevention of cardiovascular diseases. *Fitoterapia*, 81(7), 715-723.
- Güntert, A., Campbell, J., Saleem, M., O'Brien, D. P., Thompson, A. J., Byers, H. L., Ward, M. A., & Lovestone, S. (2010). Plasma gelsolin is decreased and correlates with rate of decline in Alzheimer's disease. *Journal of Alzheimer's Disease*, 21(2), 585-596.
- Guzmán-Ramos, K., Moreno-Castilla, P., Castro-Cruz, M., McGaugh, J. L., Martínez-Coria, H., LaFerla, F. M., & Bermúdez-Rattoni, F. (2012). Restoration of dopamine release deficits during object recognition memory acquisition attenuates cognitive impairment in a triple transgenic mice model of Alzheimer's disease. *Learning & Memory*, 19(10), 453-460.
- Habib, L. K., Lee, M. T., & Yang, J. (2010). Inhibitors of catalase-amyloid interactions protect cells from β-amyloid-induced oxidative stress and toxicity. *Journal of Biological Chemistry*, 285(50), 38933-38943.
- Hajjaj, H., Macé, C., Roberts, M., Niederberger, P., & Fay, L. B. (2005). Effect of 26oxygenosterols from *Ganoderma lucidum* and their activity as cholesterol

synthesis inhibitors. *Applied and Environmental Microbiology*, 71(7), 3653-3658.

- Halliwell, B. (1990). How to characterize a biological antioxidant. *Free Radical Research Communications*, 9(1), 1-32.
- Hanai, N., Nagata, K., Kawajiri, A., Shiromizu, T., Saitoh, N., Hasegawa, Y., Murakami, S., & Inagaki, M. (2004). Biochemical and cell biological characterization of a mammalian septin, Sept11. *FEBS Letters*, 568(1-3), 83-88.
- Hansen, R. A., Gartlehner, G., Webb, A. P., Morgan, L. C., Moore, C. G., & Jonas, D. E. (2008). Efficacy and safety of donepezil, galantamine, and rivastigmine for the treatment of Alzheimer's disease: a systematic review and meta-analysis. *Clinical Interventions in Aging*, 3(2), 211.
- Harada, J., & Sugimoto, M. (1999). Activation of caspase-3 in  $\beta$ -amyloid-induced apoptosis of cultured rat cortical neurons. *Brain Research*, 842(2), 311-323.
- Hardy, J. (2006). Alzheimer's disease: the amyloid cascade hypothesis: an update and reappraisal. *Journal of Alzheimer's Disease, 9*(3 Supplement), 151-153.
- Hardy, J., & Selkoe, D. J. (2002). The amyloid hypothesis of Alzheimer's disease: progress and problems on the road to therapeutics. *Science*, 297(5580), 353-356.
- Hardy, J. A., & Higgins, G. A. (1992). Alzheimer's disease: the amyloid cascade hypothesis. *Science*, 256(5054), 184.
- Harnafi, H., Aziz, M., & Amrani, S. (2009). Sweet basil (Ocimum basilicum L.) improves lipid metabolism in hypercholesterolemic rats. European e-Journal of Clinical Nutrition and Metabolism, 4(4), e181-e186.
- Harnafi, H., Caid, H. S., el Houda Bouanani, N., Aziz, M., & Amrani, S. (2008). Hypolipemic activity of polyphenol-rich extracts from *Ocimum basilicum* in Triton WR-1339-induced hyperlipidemic mice. *Food Chemistry*, 108(1), 205-212.
- Hashimoto, M., Hossain, S., Agdul, H., & Shido, O. (2005). Docosahexaenoic acidinduced amelioration on impairment of memory learning in amyloid  $\beta$ -infused rats relates to the decreases of amyloid  $\beta$  and cholesterol levels in detergentinsoluble membrane fractions. *Biochimica et Biophysica Acta (BBA)-Molecular and Cell Biology of Lipids, 1738*(1), 91-98.
- Hasnat, M. A., Pervin, M., & Lim, B. O. (2013). Acetylcholinesterase inhibition and *in vitro* and *in vivo* antioxidant activities of *Ganoderma lucidum* grown on germinated brown rice. *Molecules*, 18(6), 6663-6678.
- Hauser, S. P., & O Ryan, R. (2013). Impact of apolipoprotein E on Alzheimer's disease. *Current Alzheimer Research*, 10(8), 809-817.
- He, H., He, J. P., Sui, Y. J., Zhou, S. Q., & Wang, J. (2008). The hepatoprotective effects of *Ganoderma lucidum* peptides against carbon tetrachloride-induced liver injury in mice. *Journal of Food Biochemistry*, 32(5), 628-641.

- Heidrich, J. E., Contos, L. M., Hunsaker, L. A., Deck, L. M., & Vander Jagt, D. L. (2004). Inhibition of pancreatic cholesterol esterase reduces cholesterol absorption in the hamster. *BMC Pharmacology*, 4(1), 5.
- Heleno, S. A., Barros, L., Martins, A., Queiroz, M. J. R., Santos-Buelga, C., & Ferreira,
  I. C. (2012). Fruiting body, spores and *in vitro* produced mycelium of *Ganoderma lucidum* from Northeast Portugal: A comparative study of the antioxidant potential of phenolic and polysaccharidic extracts. *Food Research International*, 46(1), 135-140.
- Henderson, V. W., Paganini-Hill, A., Emanuel, C. K., Dunn, M. E., & Buckwalter, J. G. (1994). Estrogen replacement therapy in older women: comparisons between Alzheimer's disease cases and nondemented control subjects. *Archives of Neurology*, 51(9), 896-900.
- Heverin, M., Meaney, S., Lütjohann, D., Diczfalusy, U., Wahren, J., & Björkhem, I. (2005). Crossing the barrier: net flux of 27-hydroxycholesterol into the human brain. *Journal of Lipid Research*, 46(5), 1047-1052.
- Himeno, E., Ohyagi, Y., Ma, L., Nakamura, N., Miyoshi, K., Sakae, N., Motomura, K., Soejima, N., Yamasaki, R., & Hashimoto, T. (2011). Apomorphine treatment in Alzheimer mice promoting amyloid-β degradation. *Annals of Neurology*, 69(2), 248-256.
- Hippius, H., & Neundorfer, G. (2003). The discovery of Alzheimer's disease. *Dialogues in Clinical Neuroscience*, *5*, 101-108.
- Hirai, K., Aliev, G., Nunomura, A., Fujioka, H., Russell, R. L., Atwood, C. S., Johnson, A. B., Kress, Y., Vinters, H. V., & Tabaton, M. (2001). Mitochondrial abnormalities in Alzheimer's disease. *The Journal of Neuroscience*, 21(9), 3017-3023.
- Holmstrom, K., & Finkel, T. (2014). Cellular mechanisms and physiological consequences of redox-dependent signalling. *Nature Review of Molecular and Cell Biology*, 15(6), 411-421.
- Holscher, C., Gengler, S., Gault, V. A., Harriott, P., & Mallot, H. A. (2007). Soluble beta-amyloid [25–35] reversibly impairs hippocampal synaptic plasticity and spatial learning. *European Journal of Pharmacology*, *561*(1), 85-90.
- Hondius, D. C., van Nierop, P., Li, K. W., Hoozemans, J. J., van der Schors, R. C., van Haastert, E. S., van der Vies, S. M., Rozemuller, A. J., & Smit, A. B. (2016).
  Profiling the human hippocampal proteome at all pathologic stages of Alzheimer's disease. *Alzheimer's & Dementia*.
- Hong, I., Kang, T., Yoo, Y., Park, R., Lee, J., Lee, S., Kim, J., Song, B., Kim, S.-Y., & Moon, M. (2013). Quantitative proteomic analysis of the hippocampus in the 5XFAD mouse model at early stages of Alzheimer's disease pathology. *Journal* of Alzheimer's Disease, 36(2), 321-334.
- Hooijmans, C. R., & Kiliaan, A. J. (2008). Fatty acids, lipid metabolism and Alzheimer pathology. *European Journal of Pharmacology*, 585(1), 176-196.

- Hossain, S., Hashimoto, M., Choudhury, E. K., Alam, N., Hussain, S., Hasan, M., Choudhury, S. K., & Mahmud, I. (2003). Dietary mushroom (*Pleurotus* ostreatus) ameliorates atherogenic lipid in hypercholesterolaemic rats. Clinical and Experimental Pharmacology and Physiology, 30(7), 470-475.
- Huang, D., Ou, B., & Prior, R. L. (2005). The chemistry behind antioxidant capacity assays. *Journal of Agricultural and Food Chemistry*, 53(6), 1841-1856.
- Hultman K, S. S., Norris EH. (2013). The APOE ε4/ε4 genotype potentiates vascular fibrin(ogen) deposition in amyloid-laden vessels in the brains of Alzheimer's disease patients. *Journal of Cerebral blood flow and Metabolism*, 33(8), 1251-1258.
- Hunter, J. E., Zhang, J., & Kris-Etherton, P. M. (2010). Cardiovascular disease risk of dietary stearic acid compared with trans, other saturated, and unsaturated fatty acids: a systematic review. *The American Journal of Clinical Nutrition*, 91(1), 46-63.
- Hwang, J. Y., Shim, J. S., Song, M.-Y., Yim, S.-V., Lee, S. E., & Park, K.-S. (2015). Proteomic analysis reveals that the protective effects of ginsenoside Rb1 are associated with the actin cytoskeleton in  $\beta$ -amyloid-treated neuronal cells. *Journal of Ginseng Research*, 40(3), 278-284.
- Ikeda, I., Tsuda, K., Suzuki, Y., Kobayashi, M., Unno, T., Tomoyori, H., Goto, H., Kawata, Y., Imaizumi, K., & Nozawa, A. (2005). Tea catechins with a galloyl moiety suppress postprandial hypertriacylglycerolemia by delaying lymphatic transport of dietary fat in rats. *The Journal of Nutrition*, 135(2), 155-159.
- Inestrosa, N. C., Alvarez, A., Perez, C. A., Moreno, R. D., Vicente, M., Linker, C., Casanueva, O. I., Soto, C., & Garrido, J. (1996). Acetylcholinesterase accelerates assembly of amyloid-β-peptides into Alzheimer's fibrils: possible role of the peripheral site of the enzyme. *Neuron*, *16*(4), 881-891.
- Ingelsson, M., Fukumoto, H., Newell, K., Growdon, J., Hedley–Whyte, E., Frosch, M., Albert, M., Hyman, B., & Irizarry, M. (2004). Early Aβ accumulation and progressive synaptic loss, gliosis, and tangle formation in AD brain. *Neurology*, *62*(6), 925-931.
- Insull, J. W. (2006). Clinical utility of bile acid sequestrants in the treatment of dyslipidemia: a scientific review. *Southern Medical Journal*, 99(3), 257-274.
- Ito, H., Atsuzawa, K., Morishita, R., Usuda, N., Sudo, K., Iwamoto, I., Mizutani, K., Katoh-Semba, R., Nozawa, Y., & Asano, T. (2009). Sept8 controls the binding of vesicle-associated membrane protein 2 to synaptophysin. *Journal of Neurochemistry*, 108(4), 867-880.
- Izquierdo, I., & Medina, J. H. (1997). Memory formation: the sequence of biochemical events in the hippocampus and its connection to activity in other brain structures. *Neurobiology of Learning and Memory*, 68(3), 285-316.

- Izzo, C., Grillo, F., & Murador, E. (1981). Improved method for determination of highdensity-lipoprotein cholesterol I. Isolation of high-density lipoproteins by use of polyethylene glycol 6000. *Clinical Chemistry*, 27(3), 371-374.
- Jack, C. R., Albert, M. S., Knopman, D. S., McKhann, G. M., Sperling, R. A., Carrillo, M. C., Thies, B., & Phelps, C. H. (2011). Introduction to the recommendations from the National Institute on Aging-Alzheimer's Association workgroups on diagnostic guidelines for Alzheimer's disease. *Alzheimer's & Dementia*, 7(3), 257-262.
- James, P. (1997). Protein identification in the post-genome era: the rapid rise of proteomics. *Quarterly Reviews of Biophysics*, 30(04), 279-331.
- James, R., Searcy, J. L., Le Bihan, T., Martin, S. F., Gliddon, C. M., Povey, J., Deighton, R. F., Kerr, L. E., McCulloch, J., & Horsburgh, K. (2012). Proteomic analysis of mitochondria in APOE transgenic mice and in response to an ischemic challenge. *Journal of Cerebral Blood Flow & Metabolism*, 32(1), 164-176.
- Jang, S.-H., Cho, S.-w., Yoon, H.-M., Jang, K.-J., Song, C.-H., & Kim, C.-H. (2014). Hepatoprotective evaluation of *Ganoderma lucidum* pharmacopuncture: *in vivo* studies of ethanol-induced acute liver injury. *Journal of Pharmacopuncture*, 17(3), 16.
- Jayaratnam, S., Khoo, A. K. L., & Basic, D. (2008). Rapidly progressive Alzheimer's disease and elevated 14-3-3 proteins in cerebrospinal fluid. *Age and Ageing*, *37*(4), 467-469.
- Ji, L., Chauhan, A., Wegiel, J., Essa, M. M., & Chauhan, V. (2009). Gelsolin is proteolytically cleaved in the brains of individuals with Alzheimer's disease. *Journal of Alzheimer's Disease*, 18(1), 105-111.
- Ji, L., Zhao, X., & Hua, Z. (2015). Potential therapeutic implications of gelsolin in Alzheimer's Disease. *Journal of Alzheimer's Disease*, 44(1), 13-25.
- Jin, L.-W., Maezawa, I., Vincent, I., & Bird, T. (2004). Intracellular accumulation of amyloidogenic fragments of amyloid-β precursor protein in neurons with Niemann-Pick type C defects is associated with endosomal abnormalities. *The American Journal of Pathology*, *164*(3), 975-985.
- Jouvenceau, A., Hédou, G., Potier, B., Kollen, M., Dutar, P., & Mansuy, I. M. (2006). Partial inhibition of PP1 alters bidirectional synaptic plasticity in the hippocampus. *European Journal of Neuroscience*, 24(2), 564-572.
- Jung, D., Biggs, H., Erikson, J., & Ledyard, P. U. (1975). New Colorimetric reaction for end-point, continuous-flow, and kinetic measurement of urea. *Clinical Chemistry*, 21(8), 1136-1140.
- Kabir, Y., & Kimura, S. (1988). Dietary effect of *Ganoderma lucidum* mushroom on blood pressure and lipid levels in spontaneously hypertensive rats (SHR). *Journal of Nutritional Science and Vitaminology*, 34(4), 433-438.

- Kahl, R. (1984). Synthetic antioxidants: biochemical actions and interference with radiation, toxic compounds, chemical mutagens and chemical carcinogens. *Toxicology*, 33(3), 185-228.
- Kakimura, J.-I., Kitamura, Y., Takata, K., Umeki, M., Suzuki, S., Shibagaki, K., Taniguchi, T., Nomura, Y., Gebicke-Haerter, P. J., & Smith, M. A. (2002). Microglial activation and amyloid-β clearance induced by exogenous heat-shock proteins. *The FASEB Journal*, 16(6), 601-603.
- Kakio, A., Nishimoto, S.-i., Yanagisawa, K., Kozutsumi, Y., & Matsuzaki, K. (2002). Interactions of amyloid β-protein with various gangliosides in raft-like membranes: importance of GM1 ganglioside-bound form as an endogenous seed for Alzheimer amyloid. *Biochemistry*, 41(23), 7385-7390.
- Kakkar, S., & Bais, S. (2014). A review on protocatechuic acid and its pharmacological potential. *ISRN Pharmacology, 2014*.
- Kalač, P. (2009). Chemical composition and nutritional value of European species of wild growing mushrooms: A review. *Food Chemistry*, 113(1), 9-16.
- Kalogeropoulos, N., Yanni, A. E., Koutrotsios, G., & Aloupi, M. (2013). Bioactive microconstituents and antioxidant properties of wild edible mushrooms from the island of Lesvos, Greece. *Food and Chemical Toxicology*, *55*, 378-385.
- Kamanna, V. S., & Kashyap, M. L. (2000). Mechanism of action of niacin on lipoprotein metabolism. *Current Atherosclerosis Reports*, 2(1), 36-46.
- Kamanna, V. S., & Kashyap, M. L. (2008). Mechanism of action of niacin. *The American Journal of Cardiology*, 101(8), S20-S26.
- Kamenetz, F., Tomita, T., Hsieh, H., Seabrook, G., Borchelt, D., Iwatsubo, T., Sisodia, S., & Malinow, R. (2003). APP pocessing and synaptic function. *Neuron*, 37(6), 925-937.
- Kamino, K., Nagasaka, K., Imagawa, M., Yamamoto, H., Yoneda, H., Ueki, A., Kitamura, S., Namekata, K., Miki, T., & Ohta, S. (2000). Deficiency in mitochondrial aldehyde dehydrogenase increases the risk for late-onset Alzheimer's disease in the Japanese population. *Biochemical and Biophysical Research Communications*, 273(1), 192-196.
- Kanagasabapathy, G., Malek, S., Mahmood, A., Chua, K., Vikineswary, S., & Kuppusamy, U. (2013). Beta-glucan-rich extract from *Pleurotus sajor-caju* (Fr.) singer prevents obesity and oxidative stress in C57BL/6J mice fed on a high-fat diet. *Evidence-Based Complementary and Alternative Medicine*, 2013.
- Kao, P.-F., Wang, S.-H., Hung, W.-T., Liao, Y.-H., Lin, C.-M., & Yang, W.-B. (2011). Structural characterization and antioxidative activity of low-molecular-weights beta-1, 3-glucan from the residue of extracted *Ganoderma lucidum* fruiting bodies. *BioMed Research International*, 2012.

- Katayama, T., Imaizumi, K., Manabe, T., Hitomi, J., Kudo, T., & Tohyama, M. (2004). Induction of neuronal death by ER stress in Alzheimer's disease. *Journal of Chemical Neuroanatomy*, 28(1), 67-78.
- Kaya, K., Ciftci, O., Cetin, A., Tecellioğlu, M., & Başak, N. (2016). Beneficial effects of β-glucan against cisplatin side effects on the nervous system in rats 1. Acta Cirurgica Brasileira, 31(3), 198-205.
- Keawsa-Ard, S., Liawruangrath, B., Liawruangrath, S., Teerawutgulrag, A., & Pyne, S. G. (2012). Chemical constituents and antioxidant and biological activities of the essential oil from leaves of *Solanum spirale*. *Natural Product Communications*, 7(7), 955-958.
- Kedare, S. B., & Singh, R. (2011). Genesis and development of DPPH method of antioxidant assay. *Journal of Food Science and Technology*, 48(4), 412-422.
- Keeney, J. T., Swomley, A. M., Förster, S., Harris, J. L., Sultana, R., & Butterfield, D. A. (2013). Apolipoprotein A-I: Insights from redox proteomics for its role in neurodegeneration. *PROTEOMICS-Clinical Applications*, 7(1-2), 109-122.
- Keller, J. N., Hanni, K. B., & Markesbery, W. R. (2000). Impaired proteasome function in Alzheimer's disease. *Journal of Neurochemistry*, 75(1), 436-439.
- Kelly, B. L., Vassar, R., & Ferreira, A. (2005). β-Amyloid-induced Dynamin 1 Depletion in Hippocampal Neurons a potential mechanism for early cognitive decline in Alzheimer disease. *Journal of Biological Chemistry*, 280(36), 31746-31753.
- Khan, A., Gani, A., Masoodi, F., Kousar, S., Ahmad, M., & Singh, M. (2014). Antioxidant and functional properties of  $\beta$ -glucan extracted from edible mushrooms Agaricus bisporus, Pleurotus ostreatus and Coprinus atramentarius. Paper presented at the Proceedings of 8th International Conference on Mushroom Biology and Mushroom Products (ICMBMP8), New Delhi, India, 19-22 November 2014. Volume I & II.
- Kim, J., Basak, J. M., & Holtzman, D. M. (2009). The role of apolipoprotein E in Alzheimer's disease. *Neuron*, 63(3), 287-303.
- Kim, J. H., Ha, H.-C., Lee, M.-S., Kang, J.-I., Kim, H.-S., Lee, S.-Y., Pyun, K.-H., & Shim, I. (2007). Effect of Tremella fuciformis on the neurite outgrowth of PC12h cells and the improvement of memory in rats. *Biological and Pharmaceutical Bulletin*, 30(4), 708-714.
- Kim, K., Lee, S. G., Kegelman, T. P., Su, Z. Z., Das, S. K., Dash, R., Dasgupta, S., Barral, P. M., Hedvat, M., & Diaz, P. (2011). Role of excitatory amino acid transporter-2 (EAAT2) and glutamate in neurodegeneration: opportunities for developing novel therapeutics. *Journal of Cellular Physiology*, 226(10), 2484-2493.
- Kim, S.-W., Park, S.-S., Min, T.-J., & Yu, K.-H. (1999). Antioxidant activity of ergosterol peroxide (5, 8-epidioxy-5α, 8α-ergosta-6, 22E-dien-3β-ol) in *Armillariella mellea*. Bulletin of the Korean Chemical Society, 20(7), 819-823.

- Kimball, T. J., Childs, M. T., Applebaum-Bowden, D., & Sembrowich, W. L. (1983). The effect of training and diet on lipoprotein cholesterol, tissue lipoprotein lipase and hepatic triglyceride lipase in rats. *Metabolism*, 32(5), 497-503.
- Kinoshita, A., Kinoshita, M., Akiyama, H., Tomimoto, H., Akiguchi, I., Kumar, S., Noda, M., & Kimura, J. (1998). Identification of septins in neurofibrillary tangles in Alzheimer's disease. *The American journal of pathology*, 153(5), 1551-1560.
- Kivipelto, M., Helkala, E.-L., Laakso, M. P., Hanninen, T., Hallikainen, M., Alhainen, K., Iivonen, S., Mannermaa, A., Tuomilehto, J., & Nissinen, A. (2002). Apolipoprotein E ε4 allele, elevated midlife total cholesterol level, and high midlife systolic blood pressure are independent risk factors for late-life Alzheimer disease. *Annals of Internal Medicine*, 137(3), 149-155.
- Kleim, J. A., Freeman, J. H., Bruneau, R., Nolan, B. C., Cooper, N. R., Zook, A., & Walters, D. (2002). Synapse formation is associated with memory storage in the cerebellum. *Proceedings of the National Academy of Sciences*, 99(20), 13228-13231.
- Klein, S., Burke, L. E., Bray, G. A., Blair, S., Allison, D. B., Pi-Sunyer, X., Hong, Y., & Eckel, R. H. (2004). Clinical implications of obesity with specific focus on cardiovascular disease a statement for professionals from the American Heart Association Council on Nutrition, Physical Activity, and Metabolism: Endorsed by the American College of Cardiology Foundation. *Circulation*, 110(18), 2952-2967.
- Kobori, M., Yoshida, M., Ohnishi-Kameyama, M., & Shinmoto, H. (2007). Ergosterol peroxide from an edible mushroom suppresses inflammatory responses in RAW264. 7 macrophages and growth of HT29 colon adenocarcinoma cells. *British journal of pharmacology*, 150(2), 209-219.
- Kofuji, K., Aoki, A., Tsubaki, K., Konishi, M., Isobe, T., & Murata, Y. (2012). Antioxidant activity of β-glucan. *ISRN Pharmaceutics*, 2012.
- Kogan, G., Staško, A., Bauerová, K., Polovka, M., Šoltés, L., Brezová, V., Navarová, J., & Mihalová, D. (2005). Antioxidant properties of yeast  $(1 \rightarrow 3)$ - $\beta$ -D-glucan studied by electron paramagnetic resonance spectroscopy and its activity in the adjuvant arthritis. *Carbohydrate Polymers*, *61*(1), 18-28.
- Koh, G. C., Porras, P., Aranda, B., Hermjakob, H., & Orchard, S. E. (2012). Analyzing protein-protein interaction networks. *Journal of Proteome Research*, 11(4), 2014-2031.
- Koh, J.-y., Yang, L. L., & Cotman, C. W. (1990). β-Amyloid protein increases the vulnerability of cultured cortical neurons to excitotoxic damage. *Brain Research*, 533(2), 315-320.
- Kolarow, R., Brigadski, T., & Lessmann, V. (2007). Postsynaptic secretion of BDNF and NT-3 from hippocampal neurons depends on calcium–calmodulin kinase II signaling and proceeds via delayed fusion pore opening. *The Journal of Neuroscience*, 27(39), 10350-10364.

- Kontush, A., Finckh, B., Karten, B., Kohlschütter, A., & Beisiegel, U. (1996). Antioxidant and prooxidant activity of alpha-tocopherol in human plasma and low density lipoprotein. *Journal of Lipid Research*, 37(7), 1436-1448.
- Korolainen, M. A., Auriola, S., Nyman, T. A., Alafuzoff, I., & Pirttilä, T. (2005). Proteomic analysis of glial fibrillary acidic protein in Alzheimer's disease and aging brain. *Neurobiology of Disease*, 20(3), 858-870.
- Kozarski, M., Klaus, A., Niksic, M., Jakovljevic, D., Helsper, J. P., & Van Griensven, L. J. (2011). Antioxidative and immunomodulating activities of polysaccharide extracts of the medicinal mushrooms *Agaricus bisporus*, *Agaricus brasiliensis*, *Ganoderma lucidum* and *Phellinus linteus*. Food Chemistry, 129(4), 1667-1675.
- Kremer, A., Louis, J. V., Jaworski, T., & Van Leuven, F. (2011). GSK3 and Alzheimer's disease: facts and fiction. *Frontiers in Molecular Neuroscience*, 4.
- Kritchevsky, D., Tepper, S. A., Wright, S., Tso, P., & Czarnecki, S. K. (2000). Influence of conjugated linoleic acid (CLA) on establishment and progression of atherosclerosis in rabbits. *Journal of the American College of Nutrition*, 19(4), 472S-477S.
- Kummer, M. P., & Heneka, M. T. (2008). PPARs in Alzheimer's disease. PPAR research, 2008.
- L Ferreira, I., Resende, R., Ferreiro, E., C Rego, A., & F Pereira, C. (2010). Multiple defects in energy metabolism in Alzheimer's disease. *Current Drug Targets,* 11(10), 1193-1206.
- LaFerla, F. M. (2002). Calcium dyshomeostasis and intracellular signalling in Alzheimer's disease. *Nature Reviews Neuroscience*, 3(11), 862-872.
- Lai, C. S.-W., Yu, M.-S., Yuen, W.-H., So, K.-F., Zee, S.-Y., & Chang, R. C.-C. (2008). Antagonizing β-amyloid peptide neurotoxicity of the anti-aging fungus *Ganoderma lucidum. Brain research*, 1190, 215-224.
- Lai, P.-L., Naidu, M., Sabaratnam, V., Wong, K.-H., David, R. P., Kuppusamy, U. R., Abdullah, N., & Malek, S. N. A. (2013). Neurotrophic properties of the Lion's mane medicinal mushroom, *Hericium erinaceus* (Higher Basidiomycetes) from Malaysia. *International journal of medicinal mushrooms*, 15(6).
- Lakey-Beitia, J., Berrocal, R., Rao, K., & Durant, A. A. (2015). Polyphenols as therapeutic molecules in Alzheimer's disease through modulating amyloid pathways. *Molecular Neurobiology*, *51*(2), 466-479.
- Lakshmi, B., Ajith, T., Jose, N., & Janardhanan, K. (2006). Antimutagenic activity of methanolic extract of *Ganoderma lucidum* and its effect on hepatic damage caused by benzo [a] pyrene. *Journal of Ethnopharmacology*, 107(2), 297-303.
- Lam, K.-K., Cheng, P.-Y., Lee, Y.-M., Liu, Y.-P., Ding, C., Liu, W.-H., & Yen, M.-H. (2013). The role of heat shock protein 70 in the protective effect of YC-1 on heat stroke rats. *European Journal of Pharmacology*, 699(1), 67-73.

- Lanctôt, K. L., Rajaram, R. D., & Herrmann, N. (2009). Review: Therapy for Alzheimer's disease: how effective are current treatments? *Therapeutic Advances in Neurological Disorders*, 2(3), 163-180.
- Landmesser, U., Hornig, B., & Drexler, H. (1999). Endothelial dysfunction in hypercholesterolemia: mechanisms, pathophysiological importance, and therapeutic interventions. Paper presented at the Seminars in thrombosis and hemostasis.
- Landolfo, S., Zara, G., Zara, S., Budroni, M., Ciani, M., & Mannazzu, I. (2010). Oleic acid and ergosterol supplementation mitigates oxidative stress in wine strains of *Saccharomyces cerevisiae*. *International Journal of Food Microbiology*, 141(3), 229-235.
- Lapeyre, C., Delomenède, M., Bedos-Belval, F., Duran, H., Nègre-Salvayre, A., & Baltas, M. (2005). Design, synthesis, and evaluation of pharmacological properties of cinnamic derivatives as antiatherogenic agents. *Journal of Medicinal Chemistry*, 48(26), 8115-8124.
- Last, A. R., Ference, J. D., & Falleroni, J. (2011). Pharmacologic treatment of hyperlipidemia. *American Family Physician*, 84(5), 551.
- Lauderback, C. M., Hackett, J. M., Huang, F. F., Keller, J. N., Szweda, L. I., Markesbery, W. R., & Butterfield, D. A. (2001). The glial glutamate transporter, GLT-1, is oxidatively modified by 4-hydroxy-2-nonenal in the Alzheimer's disease brain: the role of Aβ1–42. *Journal of Neurochemistry*, 78(2), 413-416.
- Lecanu, L., & Papadopoulos, V. (2013). Modeling Alzheimer's disease with nontransgenic rat models. *Alzheimer's research & therapy*, 5(3), 17.
- Ledesma, M. D., & Dotti, C. G. (2006). Amyloid excess in Alzheimer's disease: What is cholesterol to be blamed for? *FEBS Letters*, *580*(23), 5525-5532.
- Lee, B., Park, J., Park, J., Shin, H.-J., Kwon, S., Yeom, M., Sur, B., Kim, S., Kim, M., & Lee, H. (2011). Cordyceps militaris improves neurite outgrowth in Neuro2a cells and reverses memory impairment in rats. Food Science and Biotechnology, 20(6), 1599-1608.
- Lee, S.-J., Liyanage, U., Bickel, P. E., Xia, W., Lansbury, P. T., & Kosik, K. S. (1998). A detergent-insoluble membrance compartment contains Aβ *in vivo*. *Nature Medicine*, 4(6), 730-734.
- Lee, T., Park, Y., & Han, Y. (2006). Effect of mycelial extract of *Clavicorona pyxidata* on the production of amyloid beta-peptide and the inhibition of endogenous beta-secretase activity *in vitro*. *Journal of Microbiology-Seoul*, 44(6), 665.
- Lefevre, G., Beljean-Leymarie, M., Beyerle, F., Bonnefont-Rousselot, D., Cristol, J., Therond, P., & Torreilles, J. (1997). *Evaluation of lipid peroxidation by measuring thiobarbituric acid reactive substances*. Paper presented at the Annales de biologie clinique.

- Leow, S.-S., Sekaran, S. D., Tan, Y., Sundram, K., & Sambanthamurthi, R. (2013). Oil palm phenolics confer neuroprotective effects involving cognitive and motor functions in mice. *Nutritional neuroscience*, 16(5), 207-217.
- Li, C., Li, Y., & Sun, H. H. (2006). New ganoderic acids, bioactive triterpenoid metabolites from the mushroom *Ganoderma lucidum*. *Natural product research*, 20(11), 985-991.
- Li, F., Calingasan, N. Y., Yu, F., Mauck, W. M., Toidze, M., Almeida, C. G., Takahashi, R. H., Carlson, G. A., Flint Beal, M., & Lin, M. T. (2004). Increased plaque burden in brains of APP mutant MnSOD heterozygous knockout mice. *Journal of Neurochemistry*, 89(5), 1308-1312.
- Li, W.-J., Nie, S.-P., Liu, X.-Z., Zhang, H., Yang, Y., Yu, Q., & Xie, M.-Y. (2012). Antimicrobial properties, antioxidant activity and cytotoxicity of ethanol-soluble acidic components from Ganoderma atrum. *Food and Chemical Toxicology*, 50(3), 689-694.
- Li, X., & Buxbaum, J. N. (2011). Transthyretin and the brain re-visited: Is neuronal synthesis of transthyretin protective in Alzheimer's disease? *Molecular Neurodegeneration*, 6(1), 1.
- Li, X., Wang, X., Chen, D., & Chen, S. (2011). Antioxidant activity and mechanism of protocatechnic acid *in vitro*. *Functional Foods in Health and Disease*, 1(7), 232-244.
- Liao, J. K., & Laufs, U. (2005). Pleiotropic effects of statins. Annual Review of Pharmacology and Toxicology, 45, 89.
- Liao, L., Cheng, D., Wang, J., Duong, D. M., Losik, T. G., Gearing, M., Rees, H. D., Lah, J. J., Levey, A. I., & Peng, J. (2004). Proteomic characterization of postmortem amyloid plaques isolated by laser capture microdissection. *Journal* of Biological Chemistry.
- Lin, W.-C., & Lin, W.-L. (2006). Ameliorative effect of *Ganoderma lucidum* on carbon tetrachloride-induced liver fibrosis in rats. *World Journal of Gastroenterology*, *12*(2), 265.
- Liu, J., Chang, L., Roselli, F., Almeida, O. F., Gao, X., Wang, X., Yew, D. T., & Wu, Y. (2010). Amyloid-β induces caspase-dependent loss of PSD-95 and synaptophysin through NMDA receptors. *Journal of Alzheimer's Disease*, 22(2), 541-556.
- Liu, Q., Smith, M. A., Avilá, J., DeBernardis, J., Kansal, M., Takeda, A., Zhu, X., Nunomura, A., Honda, K., & Moreira, P. I. (2005). Alzheimer-specific epitopes of tau represent lipid peroxidation-induced conformations. *Free Radical Biology* and Medicine, 38(6), 746-754.
- Lizarbe, M. A., Barrasa, J. I., Olmo, N., Gavilanes, F., & Turnay, J. (2013). Annexinphospholipid interactions. Functional implications. *International Journal of Molecular Sciences*, 14(2), 2652-2683.

- Lobo, V., Patil, A., Phatak, A., & Chandra, N. (2010). Free radicals, antioxidants and functional foods: Impact on human health. *Pharmacognosy Reviews*, 4(8), 118.
- Lock, A. L., Corl, B. A., Barbano, D. M., Bauman, D. E., & Ip, C. (2004). The anticarcinogenic effect of trans-11 18: 1 is dependent on its conversion to cis-9, trans-11 CLA by  $\Delta$ 9-desaturase in rats. *The Journal of Nutrition, 134*(10), 2698-2704.
- Lourenco, M. V., Clarke, J. R., Frozza, R. L., Bomfim, T. R., Forny-Germano, L., Batista, A. F., Sathler, L. B., Brito-Moreira, J., Amaral, O. B., & Silva, C. A. (2013). TNF-α mediates PKR-dependent memory impairment and brain IRS-1 inhibition induced by Alzheimer's β-amyloid oligomers in mice and monkeys. *Cell Metabolism, 18*(6), 831-843.
- Lovell, M., Robertson, J., Teesdale, W., Campbell, J., & Markesbery, W. (1998). Copper, iron and zinc in Alzheimer's disease senile plaques. *Journal of the Neurological Sciences*, 158(1), 47-52.
- Lu, B. (2003). BDNF and activity-dependent synaptic modulation. *Learning & Memory*, *10*(2), 86-98.
- Lubec, G., Nonaka, M., Krapfenbauer, K., Gratzer, M., Cairns, N., & Fountoulakis, M. (1999). Expression of the dihydropyrimidinase related protein 2 (DRP-2) in Down syndrome and Alzheimer's disease brain is downregulated at the mRNA and dysregulated at the protein level. *Journal of Neural Transmission and Supplementation*, 57, 161-177.
- Luo, J., Wärmländer, S. K., Gräslund, A., & Abrahams, J. P. (2014). Non-chaperone proteins can inhibit aggregation and cytotoxicity of Alzheimer amyloid β peptide. *Journal of Biological Chemistry*, 289(40), 27766-27775.
- Luo, W., Sun, W., Taldone, T., Rodina, A., & Chiosis, G. (2010). Heat shock protein 90 in neurodegenerative diseases. *Molecular neurodegeneration*, 5(1), 1.
- Lustbader, J. W., Cirilli, M., Lin, C., Xu, H. W., Takuma, K., Wang, N., Caspersen, C., Chen, X., Pollak, S., & Chaney, M. (2004). ABAD directly links Aß to mitochondrial toxicity in Alzheimer's Disease. *Science*, *304*(5669), 448-452.
- Lv, G., Zhao, J., Duan, J., Tang, Y., & Li, S. (2012). Comparison of sterols and fatty acids in two species of *Ganoderma*. *Chemistry Central Journal*, 6(10).
- Lynch, S. M., & Frei, B. (1993). Mechanisms of copper-and iron-dependent oxidative modification of human low density lipoprotein. *Journal of Lipid Research*, 34(10), 1745-1753.
- Maity, P., Samanta, S., Nandi, A. K., Sen, I. K., Paloi, S., Acharya, K., & Islam, S. S. (2014). Structure elucidation and antioxidant properties of a soluble β-D-glucan from mushroom Entoloma lividoalbum. *International Journal of Biological Macromolecules*, 63, 140-149.

- Manavalan, A., Mishra, M., Sze, S. K., & Heese, K. (2013). Brain-site-specific proteome changes induced by neuronal P60TRP expression. *Neurosignals*, 21(3-4), 129-149.
- Mancuso, C., Siciliano, R., Barone, E., & Preziosi, P. (2012). Natural substances and Alzheimer's disease: from preclinical studies to evidence based medicine. *Biochimica et Biophysica Acta (BBA)-Molecular Basis of Disease, 1822*(5), 616-624.
- Manczak, M., & Reddy, P. H. (2012). Abnormal interaction of VDAC1 with amyloid beta and phosphorylated tau causes mitochondrial dysfunction in Alzheimer's disease. *Human molecular genetics*, 21(23), 5131-5146.
- Mangialasche, F., Polidori, M. C., Monastero, R., Ercolani, S., Camarda, C., Cecchetti, R., & Mecocci, P. (2009). Biomarkers of oxidative and nitrosative damage in Alzheimer's disease and mild cognitive impairment. *Ageing Research Reviews*, 8(4), 285-305.
- Maqsood, S., & Benjakul, S. (2010). Comparative studies of four different phenolic compounds on *in vitro* antioxidative activity and the preventive effect on lipid oxidation of fish oil emulsion and fish mince. *Food Chemistry*, 119(1), 123-132.
- Mark, R. J., Lovell, M. A., Markesbery, W. R., Uchida, K., & Mattson, M. P. (1997). A role for 4-hydroxynonenal, an aldehydic product of lipid peroxidation, in disruption of ion homeostasis and neuronal death induced by amyloid β-peptide. *Journal of Neurochemistry*, 68(1), 255-264.
- Martin, L., Latypova, X., Wilson, C. M., Magnaudeix, A., Perrin, M.-L., & Terro, F. (2013). Tau protein phosphatases in Alzheimer's disease: the leading role of PP2A. *Ageing Research Reviews*, *12*(1), 39-49.
- Martorana, A., & Koch, G. (2013). " Is dopamine involved in Alzheimer's disease?". *Frontiers in Aging Neuroscience*, 6, 252-252.
- Marttinen, M., Kurkinen, K. M., Soininen, H., Haapasalo, A., & Hiltunen, M. (2015). Synaptic dysfunction and septin protein family members in neurodegenerative diseases. *Molecular Neurodegeneration*, 10(1), 1.
- Masliah, E., Hansen, L., Alford, M., Deteresa, R., & Mallory, M. (1996). Deficient glutamate tranport is associated with neurodegeneration in Alzheimer's disease. *Annals of Neurology*, 40(5), 759-766.
- Massaro, M., Carluccio, M., & De Caterina, R. (1999). Direct vascular antiatherogenic effects of oleic acid: a clue to the cardioprotective effects of the Mediterranean diet. *Cardiologia (Rome, Italy), 44*(6), 507-513.
- Mathew, A., Yoshida, Y., Maekawa, T., & Kumar, D. S. (2011). Alzheimer's disease: Cholesterol a menace? *Brain research bulletin*, 86(1), 1-12.
- Matter, C., Pribadi, M., Liu, X., & Trachtenberg, J. T. (2009). δ-catenin is required for the maintenance of neural structure and function in mature cortex *in vivo*. *Neuron*, *64*(3), 320-327.

- Mattila, P., Salo-Väänänen, P., Könkö, K., Aro, H., & Jalava, T. (2002). Basic composition and amino acid contents of mushrooms cultivated in Finland. *Journal of Agricultural and Food Chemistry*, 50(22), 6419-6422.
- Mattson, M. P. (2004). Pathways towards and away from Alzheimer's disease. *Nature*, 430(7000), 631-639.
- Mau, J.-L., Lin, H.-C., & Chen, C.-C. (2002a). Antioxidant properties of several medicinal mushrooms. *Journal of Agricultural and Food Chemistry*, 50(21), 6072-6077.
- Mau, J.-L., Lin, H.-C., & Song, S.-F. (2002b). Antioxidant properties of several specialty mushrooms. *Food Research International*, 35(6), 519-526.
- Mau, J.-L., Tsai, S.-Y., Tseng, Y.-H., & Huang, S.-J. (2005). Antioxidant properties of methanolic extracts from *Ganoderma tsugae*. Food Chemistry, 93(4), 641-649.
- Maurer, I., Zierz, S., & Möller, H. J. (2000). A selective defect of cytochrome c oxidase is present in brain of Alzheimer disease patients. *Neurobiology of Aging*, 21(3), 455-462.
- Mayeux, R., & Stern, Y. (2012). Epidemiology of Alzheimer disease. Cold Spring Harbor Perspectives in Medicine, 2(8), a006239.
- Mazzetti, A. P., Fiorile, M. C., Primavera, A., & Bello, M. L. (2015). Glutathione transferases and neurodegenerative diseases. *Neurochemistry International*, 82, 10-18.
- Mazzola, J. L., & Sirover, M. A. (2001). Reduction of glyceraldehyde-3-phosphate dehydrogenase activity in Alzheimer's disease and in Huntington's disease fibroblasts. *Journal of Neurochemistry*, 76(2), 442-449.
- Mazzola, P. G., Lopes, A. M., Hasmann, F. A., Jozala, A. F., Penna, T. C., Magalhaes, P. O., Rangel-Yagui, C. O., & Pessoa Jr, A. (2008). Liquid–liquid extraction of biomolecules: an overview and update of the main techniques. *Journal of chemical technology and biotechnology*, 83(2), 143-157.
- McAlpine, F. E., Lee, J.-K., Harms, A. S., Ruhn, K. A., Blurton-Jones, M., Hong, J., Das, P., Golde, T. E., LaFerla, F. M., & Oddo, S. (2009). Inhibition of soluble TNF signaling in a mouse model of Alzheimer's disease prevents pre-plaque amyloid-associated neuropathology. *Neurobiology of Disease*, 34(1), 163-177.
- McLellan, M. E., Kajdasz, S. T., Hyman, B. T., & Bacskai, B. J. (2003). In vivo imaging of reactive oxygen species specifically associated with thioflavine Spositive amyloid plaques by multiphoton microscopy. The Journal of Neuroscience, 23(6), 2212-2217.
- McMurray, C. (2000). Neurodegeneration: diseases of the cytoskeleton? *Cell Death and Differentiation*, 7(10), 861-865.
- Meaney, S., Heverin, M., Panzenboeck, U., Ekström, L., Axelsson, M., Andersson, U., Diczfalusy, U., Pikuleva, I., Wahren, J., & Sattler, W. (2007). Novel route for

elimination of brain oxysterols across the blood-brain barrier: conversion into  $7\alpha$ -hydroxy-3-oxo-4-cholestenoic acid. *Journal of Lipid Research*, 48(4), 944-951.

- Melov, S., Adlard, P. A., Morten, K., Johnson, F., Golden, T. R., Hinerfeld, D., Schilling, B., Mavros, C., Masters, C. L., & Volitakis, I. (2007). Mitochondrial oxidative stress causes hyperphosphorylation of tau. *PLoS One*, 2(6), e536-e536.
- Mikkola, T. S., & Clarkson, T. B. (2002). Estrogen replacement therapy, atherosclerosis, and vascular function. *Cardiovascular Research*, 53(3), 605-619.
- Minjarez, B., Calderón-González, K. G., Rustarazo, M. L. V., Herrera-Aguirre, M. E., Labra-Barrios, M. L., Rincon-Limas, D. E., del Pino, M. M. S., Mena, R., & Luna-Arias, J. P. (2016). Identification of proteins that are differentially expressed in brains with Alzheimer's disease using iTRAQ labeling and tandem mass spectrometry. *Journal of Proteomics*, 139, 103-121.
- Minjarez, B., Rustarazo, M., Valero, L., Sanchez del Pino, M. M., González-Robles, A., Sosa-Melgarejo, J. A., Luna-Muñoz, J., Mena, R., & Luna-Arias, J. P. (2013). Identification of polypeptides in neurofibrillary tangles and total homogenates of brains with Alzheimer's disease by tandem mass spectrometry. *Journal of Alzheimer's Disease*, 34(1), 239-262.
- Miranda, C. L., Stevens, J. F., Ivanov, V., McCall, M., Frei, B., Deinzer, M. L., & Buhler, D. R. (2000). Antioxidant and prooxidant actions of prenylated and nonprenylated chalcones and flavanones *in vitro*. *Journal of Agricultural and Food Chemistry*, 48(9), 3876-3884.
- Misgeld, T., Kerschensteiner, M., Bareyre, F. M., Burgess, R. W., & Lichtman, J. W. (2007). Imaging axonal transport of mitochondria *in vivo*. *Nature methods*, 4(7), 559-561.
- Misonou, H., Morishima-Kawashima, M., & Ihara, Y. (2000). Oxidative stress induces intracellular accumulation of amyloid  $\beta$ -protein (A $\beta$ ) in human neuroblastoma cells. *Biochemistry*, 39(23), 6951-6959.
- Mizuno, M., Yamada, K., Olariu, A., Nawa, H., & Nabeshima, T. (2000). Involvement of brain-derived neurotrophic factor in spatial memory formation and maintenance in a radial arm maze test in rats. *The Journal of Neuroscience*, 20(18), 7116-7121.
- Mizuno, T., Nakata, M., Naiki, H., Michikawa, M., Wang, R., Haass, C., & Yanagisawa, K. (1999). Cholesterol-dependent generation of a seeding amyloid β-protein in cell culture. *Journal of Biological Chemistry*, 274(21), 15110-15114.
- Molnár, E. (2011). Long-term potentiation in cultured hippocampal neurons. Seminars in Cell & Developmental Biology, 22(5), 506-513.

- Morán, M., Moreno-Lastres, D., Marín-Buera, L., Arenas, J., Martín, M. A., & Ugalde, C. (2012). Mitochondrial respiratory chain dysfunction: implications in neurodegeneration. *Free Radical Biology and Medicine*, 53(3), 595-609.
- Morgan, J., & Leake, D. (1995). Oxidation of low density lipoprotein by iron or copper at acidic pH. *Journal of Lipid Research*, 36(12), 2504-2512.
- Mori, K., Obara, Y., Hirota, M., Azumi, Y., Kinugasa, S., Inatomi, S., & Nakahata, N. (2008). Nerve growth factor-inducing activity of *Hericium erinaceus* in 1321N1 human astrocytoma cells. *Biological and Pharmaceutical Bulletin*, 31(9), 1727-1732.
- Mori, K., Obara, Y., Moriya, T., Inatomi, S., & Nakahata, N. (2011). Effects of *Hericium erinaceus* on amyloid. BETA.(25-35) peptide-induced learning and memory deficits in mice. *Biomedical Research*, 32(1), 67-72.
- Mori, L. (1978). Modified Jendrassik-Grof method for bilirubins adapted to the Abbott Bichromatic Analyzer. *Clinical Chemistry*, 24(10), 1841-1845.
- Morley, J. E., & Farr, S. A. (2014). The role of amyloid-beta in the regulation of memory. *Biochemical Pharmacology*, 88(4), 479-485.
- Morris, J. C. (2002). Challenging assumptions about Alzheimer's disease: mild cognitive impairment and the cholinergic hypothesis. *Annals of Neurology*, 51(2), 143-144.
- Moselhy, H. F., Reid, R. G., Yousef, S., & Boyle, S. P. (2013). A specific, accurate, and sensitive measure of total plasma malondialdehyde by HPLC. *Journal of Lipid Research*, *54*(3), 852-858.
- Moya-Alvarado, G., Gershoni-Emek, N., Perlson, E., & Bronfman, F. C. (2015). Neurodegeneration and Alzheimer's disease. What can proteomics tell us about the Alzheimer's brain? *Molecular & Cellular Proteomics*, mcp. R115. 053330.
- Mulnard, R. A., Cotman, C. W., Kawas, C., van Dyck, C. H., Sano, M., Doody, R., Koss, E., Pfeiffer, E., Jin, S., & Gamst, A. (2000). Estrogen replacement therapy for treatment of mild to moderate Alzheimer disease: a randomized controlled trial. JAMA, 283(8), 1007-1015.
- Munter, L.-M., Botev, A., Richter, L., Hildebrand, P. W., Althoff, V., Weise, C., Kaden, D., & Multhaup, G. (2010). Aberrant amyloid precursor protein (APP) processing in hereditary forms of Alzheimer disease caused by APP familial Alzheimer disease mutations can be rescued by mutations in the APP GxxxG motif. *Journal of Biological Chemistry*, 285(28), 21636-21643.
- Muntner, P., Coresh, J., Smith, J. C., Eckfeldt, J., & Klag, M. J. (2000). Plasma lipids and risk of developing renal dysfunction: the atherosclerosis risk in communities study. *Kidney International*, 58(1), 293-301.
- Murakami, K., Murata, N., Noda, Y., Tahara, S., Kaneko, T., Kinoshita, N., Hatsuta, H., Murayama, S., Barnham, K. J., & Irie, K. (2011). SOD1 (copper/zinc superoxide dismutase) deficiency drives amyloid β protein oligomerization and memory
loss in mouse model of Alzheimer disease. Journal of Biological Chemistry, 286(52), 44557-44568.

- Murphy, M. P., Holmgren, A., Larsson, N.-G., Halliwell, B., Chang, C. J., Kalyanaraman, B., Rhee, S. G., Thornalley, P. J., Partridge, L., & Gems, D. (2011). Unraveling the biological roles of reactive oxygen species. *Cell Metabolism*, 13(4), 361-366.
- Musunuri, S., Wetterhall, M., Ingelsson, M., Lannfelt, L., Artemenko, K., Bergquist, J., Kultima, K., & Shevchenko, G. (2014). Quantification of the brain proteome in Alzheimer's disease using multiplexed mass spectrometry. *Journal of Proteome Research*, 13(4), 2056-2068.
- Nakamura, Y., Takeda, M., Yoshimi, K., Hattori, H., Hariguchi, S., Kitajima, S., Hashimoto, S., & Nishimura, T. (1994). Involvement of clathrin light chains in the pathology of Alzheimer's disease. *Acta Neuropathologica*, 87(1), 23-31.
- Nakashima, H., Ishihara, T., Yokota, O., Terada, S., Trojanowski, J. Q., Lee, V. M.-Y., & Kuroda, S. (2004). Effects of α-tocopherol on an animal model of tauopathies. *Free Radical Biology and Medicine*, 37(2), 176-186.
- Naletova, I., Schmalhausen, E., Kharitonov, A., Katrukha, A., Saso, L., Caprioli, A., & Muronetz, V. (2008). Non-native glyceraldehyde-3-phosphate dehydrogenase can be an intrinsic component of amyloid structures. *Biochimica et Biophysica Acta (BBA)-Proteins and Proteomics*, 1784(12), 2052-2058.
- National Pharmacopoeia Committee. (2010). *Pharmacopoeia of the People's Republic* of China (Vol. 1).
- Naviaux, R. K. (2012). Oxidative shielding or oxidative stress? Journal of Pharmacology and Experimental Therapeutics, 342(3), 608-618.
- Naylor, R., Hill, A. F., & Barnham, K. J. (2008). Neurotoxicity in Alzheimer's disease: is covalently crosslinked Aβ responsible? *European Biophysics Journal*, *37*(3), 265-268.
- Neyrinck, A. M., Bindels, L. B., De Backer, F., Pachikian, B. D., Cani, P. D., & Delzenne, N. M. (2009). Dietary supplementation with chitosan derived from mushrooms changes adipocytokine profile in diet-induced obese mice, a phenomenon linked to its lipid-lowering action. *International Immunopharmacology*, 9(6), 767-773.
- Ngamukote, S., Mäkynen, K., Thilawech, T., & Adisakwattana, S. (2011). Cholesterollowering activity of the major polyphenols in grape seed. *Molecules*, 16(6), 5054-5061.
- Nielsen, H., Minthon, L., Londos, E., Blennow, K., Miranda, E., Perez, J., Crowther, D., Lomas, D., & Janciauskiene, S. (2007). Plasma and CSF serpins in Alzheimer disease and dementia with Lewy bodies. *Neurology*, 69(16), 1569-1579.

- Nieto, I. J., & Chegwin, A. C. (2008). Triterpenoids and fatty acids identified in the edible mushroom *Pleurotus sajor-caju*. *Journal of the Chilean Chemical Society*, 53(2), 1515-1517.
- Niki, E. (2009). Lipid peroxidation: physiological levels and dual biological effects. *Free Radical Biology and Medicine*, 47(5), 469-484.
- Niki, E. (2016). Oxidative stress and antioxidants: Distress or eustress? *Archives of Biochemistry and Biophysics*, 595, 19-24.
- Nikolaou, M., Parissis, J., Yilmaz, M. B., Seronde, M.-F., Kivikko, M., Laribi, S., Paugam-Burtz, C., Cai, D., Pohjanjousi, P., & Laterre, P.-F. (2012). Liver function abnormalities, clinical profile, and outcome in acute decompensated heart failure. *European Heart Journal*, ehs332.
- Nile, S. H., & Park, S. W. (2014). Total, soluble, and insoluble dietary fibre contents of wild growing edible mushrooms. *Czech Journal of Food Sciecnce*, 32(3), 302-307.
- Nishina, A., Kimura, H., Sekiguchi, A., Fukumoto, R.-h., Nakajima, S., & Furukawa, S. (2006). Lysophosphatidylethanolamine in *Grifola frondosa* as a neurotrophic activator via activation of MAPK. *Journal of lipid research*, 47(7), 1434-1443.
- Nixon, R. A. (2005). Endosome function and dysfunction in Alzheimer's disease and other neurodegenerative diseases. *Neurobiology of Aging*, *26*(3), 373-382.
- Nixon, R. A. (2007). Autophagy, amyloidogenesis and Alzheimer disease. *Journal of Cell Science*, 120(23), 4081-4091.
- Nofer, J.-R., Kehrel, B., Fobker, M., Levkau, B., Assmann, G., & von Eckardstein, A. (2002). HDL and arteriosclerosis: beyond reverse cholesterol transport. *Atherosclerosis*, 161(1), 1-16.
- Noguchi, N. (2008). Role of oxidative stress in adaptive responses in special reference to atherogenesis. *Journal of clinical biochemistry and nutrition*, 43(3), 131-138.
- Nunomura, A., Castellani, R. J., Zhu, X., Moreira, P. I., Perry, G., & Smith, M. A. (2006). Involvement of oxidative stress in Alzheimer disease. *Journal of Neuropathology & Experimental Neurology*, 65(7), 631-641.
- Nunomura, A., Perry, G., Aliev, G., Hirai, K., Takeda, A., Balraj, E. K., Jones, P. K., Ghanbari, H., Wataya, T., & Shimohama, S. (2001). Oxidative damage is the earliest event in Alzheimer disease. *Journal of Neuropathology & Experimental Neurology*, 60(8), 759-767.
- Nwozo, S. O., Orojobi, B. F., & Adaramoye, O. A. (2011). Hypolipidemic and antioxidant potentials of Xylopia aethiopica seed extract in hypercholesterolemic rats. *Journal of Medicinal Food*, 14(1-2), 114-119.
- O'Byrne, D. J., Devaraj, S., Grundy, S. M., & Jialal, I. (2002). Comparison of the antioxidant effects of Concord grape juice flavonoids α-tocopherol on markers

of oxidative stress in healthy adults. *The American Journal of Clinical Nutrition*, 76(6), 1367-1374.

- O'Day, D. H., Eshak, K., & Myre, M. A. (2015). Calmodulin binding proteins and Alzheimer's Disease. *Journal of Alzheimer's Disease*, 46(3), 553-569.
- Oda, A., Tamaoka, A., & Araki, W. (2010). Oxidative stress up-regulates presenilin 1 in lipid rafts in neuronal cells. *Journal of Neuroscience Research*, 88(5), 1137-1145.
- Oddo, S., Caccamo, A., Shepherd, J. D., Murphy, M. P., Golde, T. E., Kayed, R., Metherate, R., Mattson, M. P., Akbari, Y., & LaFerla, F. M. (2003). Tripletransgenic model of Alzheimer's disease with plaques and tangles: intracellular Aβ and synaptic dysfunction. *Neuron*, 39(3), 409-421.
- Ojaimi, J., Masters, C. L., McLean, C., Opeskin, K., McKelvie, P., & Byrne, E. (1999). Irregular distribution of cytochrome c oxidase protein subunits in aging and Alzheimer's disease. *Annals of Neurology*, *46*(4), 656-660.
- Olesen, O. F., & Dagø, L. (2000). High density lipoprotein inhibits assembly of amyloid β-peptides into fibrils. *Biochemical and Biophysical Research Communications*, 270(1), 62-66.
- Olton, D. S., & Samuelson, R. J. (1976). Remembrance of places passed: spatial memory in rats. *Journal of Experimental Psychology: Animal Behavior Processes*, 2(2), 97.
- Otunola, G. A., Oloyede, O. B., Oladiji, A. T., & Afolayan, A. A. (2010). Effects of diet-induced hypercholesterolemia on the lipid profile and some enzyme activities in female wistar rats. *African Journal of Biochemical Research*, 4(6), 149-154.
- Palacios, A., Piergiacomi, V., & Catalá, A. (2003). Antioxidant effect of conjugated linoleic acid and vitamin A during non enzymatic lipid peroxidation of rat liver microsomes and mitochondria. *Molecular and Cellular Biochemistry*, 250(1-2), 107-113.
- Palop, J. J., & Mucke, L. (2010). Amyloid-[beta]-induced neuronal dysfunction in Alzheimer's disease: from synapses toward neural networks. *Nature Neuroscience*, 13(7), 812-818.
- Paola, D., Domenicotti, C., Nitti, M., Vitali, A., Borghi, R., Cottalasso, D., Zaccheo, D., Odetti, P., Strocchi, P., & Marinari, U. M. (2000). Oxidative stress induces increase in intracellular amyloid β-protein production and selective activation of βI and βII PKCs in NT2 cells. *Biochemical and Biophysical Research Communications, 268*(2), 642-646.
- Pappolla, M., Bryant-Thomas, T., Herbert, D., Pacheco, J., Garcia, M. F., Manjon, M., Girones, X., Henry, T., Matsubara, E., & Zambon, D. (2003). Mild hypercholesterolemia is an early risk factor for the development of Alzheimer amyloid pathology. *Neurology*, 61(2), 199-205.

- Park, H.-J., Shim, H. S., Ahn, Y. H., Kim, K. S., Park, K. J., Choi, W. K., Ha, H.-C., Kang, J. I., Kim, T. S., & Yeo, I. H. (2012). *Tremella fuciformis* enhances the neurite outgrowth of PC12 cells and restores trimethyltin-induced impairment of memory in rats via activation of CREB transcription and cholinergic systems. *Behavioural brain research*, 229(1), 82-90.
- Park, H., Lee, T. H., Chang, F., Kwon, H. J., Kim, J., & Kim, H. (2013). Synthesis of ergosterol and 5, 6-Dihydroergosterol glycosides and their inhibitory activities on lipopolysaccharide-induced nitric oxide production. *Bulletin of the Korean Chemical Society*, 34(5), 1339-1344.
- Park, I.-H., Jeon, S.-Y., Lee, H.-J., Kim, S.-I., & Song, K.-S. (2004). A β-secretase (BACE1) inhibitor hispidin from the mycelial cultures of *Phellinus linteus*. *Planta medica*, 70(02), 143-146.
- Park, Y. S., Lee, H. S., Won, M. H., Lee, J. H., Lee, S. Y., & Lee, H. Y. (2002). Effect of an exo-polysaccharide from the culture broth of *Hericium erinaceus* on enhancement of growth and differentiation of rat adrenal nerve cells. *Cytotechnology*, 39(3), 155-162.
- Parvathenani, L. K., Tertyshnikova, S., Greco, C. R., Roberts, S. B., Robertson, B., & Posmantur, R. (2003). P2X7 mediates superoxide production in primary microglia and is up-regulated in a transgenic mouse model of Alzheimer's disease. *Journal of Biological Chemistry*, 278(15), 13309-13317.
- Patel, M. (2016). Targeting oxidative stress in central nervous system Disorders. *Trends* in *Pharmacological Sciences*, 37(9), 768-778.
- Paterson, R. R. M. (2006). *Ganoderma*-a therapeutic fungal biofactory. *Phytochemistry*, 67(18), 1985-2001.
- Patra, S., Patra, P., Maity, K. K., Mandal, S., Bhunia, S. K., Dey, B., Devi, K. S. P., Khatua, S., Acharya, K., & Maiti, T. K. (2013). A heteroglycan from the mycelia of *Pleurotus ostreatus*: structure determination and study of antioxidant properties. *Carbohydrate Research*, 368, 16-21.

Paxinos, G. W. (1998). The at brain in stereotaxic coordinates. San Diego Academic.

- Peluzio, M., Homem, A., Cesar, G., Azevedo, G., Amorim, R., Cara, D., Saliba, H., Vieira, E., Arantes, R., & Alvarez-Leite, J. (2001). Influences of alphatocopherol on cholesterol metabolism and fatty streak development in apolipoprotein E-deficient mice fed an atherogenic diet. *Brazilian Journal of Medical and Biological Research*, 34(12), 1539-1545.
- Penzes, P., & VanLeeuwen, J.-E. (2011). Impaired regulation of synaptic actin cytoskeleton in Alzheimer's disease. *Brain Research Reviews*, 67(1), 184-192.
- Perez-Nievas, B. G., Stein, T. D., Tai, H.-C., Dols-Icardo, O., Scotton, T. C., Barroeta-Espar, I., Fernandez-Carballo, L., de Munain, E. L., Perez, J., & Marquie, M. (2013). Dissecting phenotypic traits linked to human resilience to Alzheimer's pathology. *Brain*, awt171.

- Perk, J., De Backer, G., Gohlke, H., Graham, I., Reiner, Z., Verschuren, M., Albus, C., Benlian, P., Boysen, G., & Cifkova, R. (2012). European Association for Cardiovascular Prevention & Rehabilitation (EACPR); ESC Committee for Practice Guidelines (CPG). European Guidelines on cardiovascular disease prevention in clinical practice (version 2012). The Fifth Joint Task Force of the European Society of Cardiology and Other Societies on Cardiovascular Disease Prevention in Clinical Practice (constituted by representatives of nine societies and by invited experts). *European Heart Journal*, 33(13), 1635-1701.
- Petersen, R. C., Thomas, R. G., Grundman, M., Bennett, D., Doody, R., Ferris, S., Galasko, D., Jin, S., Kaye, J., & Levey, A. (2005). Vitamin E and donepezil for the treatment of mild cognitive impairment. *New England Journal of Medicine*, 352(23), 2379-2388.
- Peterson, L. B., & Blagg, B. S. (2009). To fold or not to fold: modulation and consequences of Hsp90 inhibition. *Future Medicinal Chemistry*, 1(2), 267-283.
- Petratos, S., Li, Q.-X., George, A. J., Hou, X., Kerr, M. L., Unabia, S. E., Hatzinisiriou, I., Maksel, D., Aguilar, M.-I., & Small, D. H. (2008). The β-amyloid protein of Alzheimer's disease increases neuronal CRMP-2 phosphorylation by a Rho-GTP mechanism. *Brain*, 131(1), 90-108.
- Peyrat-Maillard, M., Cuvelier, M.-E., & Berset, C. (2003). Antioxidant activity of phenolic compounds in 2, 2'-azobis (2-amidinopropane) dihydrochloride (AAPH)-induced oxidation: Synergistic and antagonistic effects. *Journal of the American Oil Chemists' Society*, 80(10), 1007-1012.
- Phan, C.-W., Lee, G.-S., Hong, S.-L., Wong, Y.-T., Brkljača, R., Urban, S., Malek, S. N. A., & Sabaratnam, V. (2014). *Hericium erinaceus* (Bull.: Fr) Pers. cultivated under tropical conditions: isolation of hericenones and demonstration of NGF-mediated neurite outgrowth in PC12 cells via MEK/ERK and PI3K-Akt signaling pathways. *Food & function*, 5(12), 3160-3169.
- Picklo, M. J., Montine, T. J., Amarnath, V., & Neely, M. D. (2002). Carbonyl toxicology and Alzheimer's disease. *Toxicology and Applied Pharmacology*, 184(3), 187-197.
- Picklo, M. J., Olson, S. J., Markesbery, W. R., & Montine, T. J. (2001). Expression and activities of aldo-keto oxidoreductases in Alzheimer disease. *Journal of Neuropathology & Experimental Neurology*, 60(7), 686-695.
- Pinweha, S., Wanikiat, P., Sanvarinda, Y., & Supavilai, P. (2008). The signaling cascades of *Ganoderma lucidum* extracts in stimulating non-amyloidogenic protein secretion in human neuroblastoma SH-SY5Y cell lines. *Neuroscience Letters*, 448(1), 62-66.
- Pollak, D., Cairns, N., & Lubec, G. (2003). Cytoskeleton derangement in brain of patients with Down syndrome, Alzheimer's disease and Pick's disease. *Journal of Neural Transmission Supplementum*, 67, 149-158.
- Poon, H., Castegna, A., Farr, S., Thongboonkerd, V., Lynn, B., Banks, W., Morley, J., Klein, J., & Butterfield, D. (2004). Quantitative proteomics analysis of specific

protein expression and oxidative modification in aged senescence-acceleratedprone 8 mice brain. *Neuroscience*, 126(4), 915-926.

- Poon, H. F., Calabrese, V., Calvani, M., & Butterfield, D. A. (2006). Proteomics analyses of specific protein oxidation and protein expression in aged rat brain and its modulation by L-acetylcarnitine: insights into the mechanisms of action of this proposed therapeutic agent for CNS disorders associated with oxidative stress. *Antioxidants & Redox Signaling*, 8(3-4), 381-394.
- Porat, Y., Abramowitz, A., & Gazit, E. (2006). Inhibition of amyloid fibril formation by polyphenols: structural similarity and aromatic interactions as a common inhibition mechanism. *Chemical Biology & Drug Design*, 67(1), 27-37.
- Poynton, R. A., & Hampton, M. B. (2014). Peroxiredoxins as biomarkers of oxidative stress. *Biochimica et Biophysica Acta (BBA)-General Subjects*, 1840(2), 906-912.
- Prado, V. F., Martins-Silva, C., de Castro, B. M., Lima, R. F., Barros, D. M., Amaral, E., Ramsey, A. J., Sotnikova, T. D., Ramirez, M. R., & Kim, H.-G. (2006). Mice deficient for the vesicular acetylcholine transporter are myasthenic and have deficits in object and social recognition. *Neuron*, 51(5), 601-612.
- Prince, M., Wimo, A., Guerchet, M., Ali, G., Wu, Y., & Prina, M. (2015). The global impact of dementia: an analysis of prevalence, incidence, cost and trends. *World Alzheimer Report*.
- Puglielli, L., Konopka, G., Pack-Chung, E., Ingano, L. A. M., Berezovska, O., Hyman, B. T., Chang, T. Y., Tanzi, R. E., & Kovacs, D. M. (2001). Acyl-coenzyme A: cholesterol acyltransferase modulates the generation of the amyloid β-peptide. *Nature Cell Biology*, 3(10), 905-912.
- Puglielli, L., Tanzi, R. E., & Kovacs, D. M. (2003). Alzheimer's disease: the cholesterol connection. *Nature Neuroscience*, 6(4), 345-351.
- Puzzo, D., Privitera, L., Fa, M., Staniszewski, A., Hashimoto, G., Aziz, F., Sakurai, M., Ribe, E. M., Troy, C. M., & Mercken, M. (2011). Endogenous amyloid-β is necessary for hippocampal synaptic plasticity and memory. *Annals of Neurology*, 69(5), 819-830.
- Qian, W., Shi, J., Yin, X., Iqbal, K., Grundke-Iqbal, I., Gong, C.-X., & Liu, F. (2010). PP2A regulates tau phosphorylation directly and also indirectly via activating GSK-3β. *Journal of Alzheimer's Disease, 19*(4), 1221-1229.
- Rahman, M. A., Abdullah, N., & Aminudin, N. (2014). Inhibitory effect on *in vitro* LDL oxidation and HMG Co-a reductase activity of the liquid-liquid partitioned fractions of *Hericium erinaceus* (Bull.) persoon (Lion's Mane Mushroom). *BioMed Research International*, 2014.
- Rajasekaran, M., & Kalaimagal, C. (2011). In vitro antioxidant activity of ethanolic extract of a medicinal mushroom, Ganoderma lucidum. Journal of Pharmaceutical Sciences Research, 3, 1427-1433.

- Ramachandran, H., Narasimhamurthy, K., & Raina, P. (2003). Modulation of cholesterol induced hypercholesterolemia through dietary factors in Indian desert gerbils (Meriones hurrianae). *Nutrition Research*, 23(2), 245-256.
- Rapoport, M., Dawson, H. N., Binder, L. I., Vitek, M. P., & Ferreira, A. (2002). Tau is essential to β-amyloid-induced neurotoxicity. *Proceedings of the National Academy of Sciences*, 99(9), 6364-6369.
- Rawat, A., Mohsin, M., Negi, P., Sah, A., & Singh, S. (2013). Evaluation of polyphenolic contents and antioxidant activity of wildly collected *Ganoderma lucidum* from central Himalayan hills of India. *Asian Journal of Plant Science and Research*, 3(3), 85-90.
- Recanatini, M., & Valenti, P. (2004). Acetylcholinesterase inhibitors as a starting point towards improved Alzheimer's disease therapeutics. *Current pharmaceutical design*, 10(25), 3157-3166.
- Reddy, P. H. (2011). Abnormal tau, mitochondrial dysfunction, impaired axonal transport of mitochondria, and synaptic deprivation in Alzheimer's disease. *Brain Research*, 1415, 136-148.
- Reddy, P. H. (2013). Is the mitochondrial outermembrane protein VDAC1 therapeutic target for Alzheimer's disease? *Biochimica et Biophysica Acta (BBA)-Molecular Basis of Disease, 1832*(1), 67-75.
- Reddy, P. H., & Beal, M. F. (2008). Amyloid beta, mitochondrial dysfunction and synaptic damage: implications for cognitive decline in aging and Alzheimer's disease. *Trends in Molecular Medicine*, 14(2), 45-53.
- Reed, T. T. (2011). Lipid peroxidation and neurodegenerative disease. *Free Radical Biology and Medicine*, *51*(7), 1302-1319.
- Refolo, L. M., Pappolla, M. A., LaFrancois, J., Malester, B., Schmidt, S. D., Thomas-Bryant, T., Tint, G. S., Wang, R., Mercken, M., & Petanceska, S. S. (2001). A cholesterol-lowering drug reduces β-amyloid pathology in a transgenic mouse model of Alzheimer's disease. *Neurobiology of Disease*, 8(5), 890-899.
- Refolo, L. M., Pappolla, M. A., Malester, B., LaFrancois, J., Bryant-Thomas, T., Wang,
  R., Tint, G. S., Sambamurti, K., & Duff, K. (2000). Hypercholesterolemia accelerates the Alzheimer's amyloid pathology in a transgenic mouse model. *Neurobiology of Disease*, 7(4), 321-331.
- Reis, F. S., Barros, L., Martins, A., & Ferreira, I. C. (2012). Chemical composition and nutritional value of the most widely appreciated cultivated mushrooms: an interspecies comparative study. *Food and Chemical Toxicology*, 50(2), 191-197.
- Reitman, S., & Frankel, S. (1957). A colorimetric method for the determination of serum glutamic oxalacetic and glutamic pyruvic transminases. *American Journal of Clinical Pathology*, 28(1), 56-63.
- Rhein, V., Baysang, G., Rao, S., Meier, F., Bonert, A., Müller-Spahn, F., & Eckert, A. (2009). Amyloid-beta leads to impaired cellular respiration, energy production

and mitochondrial electron chain complex activities in human neuroblastoma cells. *Cellular and Molecular Neurobiology*, 29(6-7), 1063-1071.

- Rice-Evans, C., Miller, N., & Paganga, G. (1997). Antioxidant properties of phenolic compounds. *Trends in Plant Science*, 2(4), 152-159.
- Ristow, M., & Schmeisser, S. (2011). Extending life span by increasing oxidative stress. *Free Radical Biology and Medicine*, *51*(2), 327-336.
- Roach, P. D., Balasubramaniam, S., Hirata, F., Abbey, M., Szanto, A., Simons, L. A., & Nestel, P. J. (1993). The low-density lipoprotein receptor and cholesterol synthesis are affected differently by dietary cholesterol in the rat. *Biochimica et Biophysica Acta (BBA)-Lipids and Lipid Metabolism*, 1170(2), 165-172.
- Roberts, B. R., Ryan, T. M., Bush, A. I., Masters, C. L., & Duce, J. A. (2012). The role of metallobiology and amyloid-β peptides in Alzheimer's disease. *Journal of Neurochemistry*, *120*(s1), 149-166.
- Roby, M. H. H., Sarhan, M. A., Selim, K. A.-H., & Khalel, K. I. (2013). Evaluation of antioxidant activity, total phenols and phenolic compounds in thyme (*Thymus* vulgaris L.), sage (Salvia officinalis L.), and marjoram (Origanum majorana L.) extracts. Industrial Crops and Products, 43, 827-831.
- Rodrigues, H. A., Fonseca, M. d. C., Camargo, W. L., Lima, P. M., Martinelli, P. M., Naves, L. A., Prado, V. F., Prado, M. A., & Guatimosim, C. (2013). Reduced expression of the vesicular acetylcholine transporter and neurotransmitter content affects synaptic vesicle distribution and shape in mouse neuromuscular junction. *PLoS One*, 8(11), e78342.
- Rogaev, E., Sherrington, R., Rogaeva, E., Levesque, G., Ikeda, M., Liang, Y., Chi, H., Lin, C., Holman, K., & Tsuda, T. (1995). Familial Alzheimer's disease in kindreds with missense mutations in a gene on chromosome 1 related to the Alzheimer's disease type 3 gene. *Nature*, 376(6543), 775-778.
- Rogers, J., Strohmeyer, R., Kovelowski, C., & Li, R. (2002). Microglia and inflammatory mechanisms in the clearance of amyloid  $\beta$  peptide. *Glia*, 40(2), 260-269.
- Rosenberg, T., Gal-Ben-Ari, S., Dieterich, D. C., Kreutz, M. R., Ziv, N. E., Gundelfinger, E. D., & Rosenblum, K. (2014). The roles of protein expression in synaptic plasticity and memory consolidation. *Frontiers in Molecular Neuroscience*, 7, 86.
- Ross, R. K., Paganini-Hill, A., Wan, P. C., & Pike, M. C. (2000). Effect of hormone replacement therapy on breast cancer risk: estrogen versus estrogen plus progestin. *Journal of the National Cancer Institute*, 92(4), 328-332.
- Roy, S., & Rauk, A. (2005). Alzheimer's disease and the 'ABSENT'hypothesis: mechanism for amyloid  $\beta$  endothelial and neuronal toxicity. *Medical Hypotheses*, 65(1), 123-137.

- Rudel, L. L. (1999). Atherosclerosis and conjugated linoleic acid. *British Journal of Nutrition*, 81(03), 177-179.
- Saetre, P., Jazin, E., & Emilsson, L. (2011). Age-related changes in gene expression are accelerated in Alzheimer's disease. *Synapse*, 65(9), 971-974.
- Saki, N., Saki, G., Rahim, F., & Nikakhlagh, S. (2011). Modulating effect of soy protein on serum cardiac enzymes in cholesterol-fed rats. *International Journal of Medicine and Medical Sciences*, 3(14), 390-395.
- Sando, S. B., Melquist, S., Cannon, A., Hutton, M. L., Sletvold, O., Saltvedt, I., White, L. R., Lydersen, S., & Aasly, J. O. (2008). APOE ɛ4 lowers age at onset and is a high risk factor for Alzheimer's disease; A case control study from central Norway. *BMC Neurology*, 8(1), 1.
- Sano, M., Ernesto, C., Thomas, R. G., Klauber, M. R., Schafer, K., Grundman, M., Woodbury, P., Growdon, J., Cotman, C. W., & Pfeiffer, E. (1997). A controlled trial of selegiline, alpha-tocopherol, or both as treatment for Alzheimer's disease. *New England Journal of Medicine*, 336(17), 1216-1222.
- Sanodiya, B. S., Thakur, G. S., Baghel, R. K., Prasad, G., & Bisen, P. (2009). Ganoderma lucidum: a potent pharmacological macrofungus. *Current Pharmaceutical Biotechnology*, 10(8), 717-742.
- Santacruz, K., Lewis, J., Spires, T., Paulson, J., Kotilinek, L., Ingelsson, M., Guimaraes, A., DeTure, M., Ramsden, M., & McGowan, E. (2005). Tau suppression in a neurodegenerative mouse model improves memory function. *Science*, 309(5733), 476-481.
- Santos, R. X., Correia, S. C., Zhu, X., Smith, M. A., Moreira, P. I., Castellani, R. J., Nunomura, A., & Perry, G. (2013). Mitochondrial DNA oxidative damage and repair in aging and Alzheimer's disease. *Antioxidants & Redox Signaling*, 18(18), 2444-2457.
- Sato, M., Murakami, K., Uno, M., Nakagawa, Y., Katayama, S., Akagi, K.-i., Masuda, Y., Takegoshi, K., & Irie, K. (2013). Site-specific inhibitory mechanism for amyloid β42 aggregation by catechol-type flavonoids targeting the Lys residues. *Journal of Biological Chemistry*, 288(32), 23212-23224.
- Sawa, T., Ihara, H., Ida, T., Fujii, S., Nishida, M., & Akaike, T. (2013). Formation, signaling functions, and metabolisms of nitrated cyclic nucleotide. *Nitric Oxide*, *34*, 10-18.
- Saxena, U. (2010). Alzheimer's disease amyloid hypothesis at crossroads: where do we go from here? *Expert Opinion on Therapeutic Targets*, 14(12), 1273-1277.
- Schenk, D., Hagen, M., & Seubert, P. (2004). Current progress in beta-amyloid immunotherapy. *Current Opinion in Immunology*, 16(5), 599-606.
- Schirmeister, J., Willmann, H., & Kiefer, H. (1964). Plasma creatinine as rough indicator of renal function. *Deutsche Medizinische Wochenschrift* 89, 1018.

- Schönknecht, P., Lütjohann, D., Pantel, J., Bardenheuer, H., Hartmann, T., von Bergmann, K., Beyreuther, K., & Schröder, J. (2002). Cerebrospinal fluid 24Shydroxycholesterol is increased in patients with Alzheimer's disease compared to healthy controls. *Neuroscience Letters*, 324(1), 83-85.
- Schroeder, F., Gallegos, A. M., Atshaves, B. P., Storey, S. M., McIntosh, A. L., Petrescu, A. D., Huang, H., Starodub, O., Chao, H., & Yang, H. (2001). Recent advances in membrane microdomains: rafts, caveolae, and intracellular cholesterol trafficking. *Experimental Biology and Medicine*, 226(10), 873-890.
- Schwenke, D. C., Rudel, L. L., Sorci-Thomas, M. G., & Thomas, M. J. (2002). α-Tocopherol protects against diet induced atherosclerosis in New Zealand white rabbits. *Journal of Lipid Research*, 43(11), 1927-1938.
- Seeram, N. P., Adams, L. S., Henning, S. M., Niu, Y., Zhang, Y., Nair, M. G., & Heber, D. (2005). In vitro antiproliferative, apoptotic and antioxidant activities of punicalagin, ellagic acid and a total pomegranate tannin extract are enhanced in combination with other polyphenols as found in pomegranate juice. *The Journal* of Nutritional Biochemistry, 16(6), 360-367.
- Selkoe, D. J., & Hardy, J. (2016). The amyloid hypothesis of Alzheimer's disease at 25 years. *EMBO Molecular Medicine*, 8(6), 595-608.
- Şener, G., Ekşioğlu-Demiralp, E., Çetiner, M., Ercan, F., & Yeğen, B. Ç. (2006). βglucan ameliorates methotrexate-induced oxidative organ injury via its antioxidant and immunomodulatory effects. *European Journal of Pharmacology*, 542(1), 170-178.
- Seo, D.-S., Lee, E.-N., Seo, G.-S., & Lee, J.-S. (2008). Screening and optimal extraction of a new antidementia  $\beta$ -secretase inhibitor-containing mushroom. *Mycobiology*, *36*(3), 195-197.
- Seow, S. L.-S., Naidu, M., David, P., Wong, K.-H., & Sabaratnam, V. (2013). Potentiation of neuritogenic activity of medicinal mushrooms in rat pheochromocytoma cells. *BMC complementary and alternative medicine*, 13(1), 157.
- Serbinova, E., Kagan, V., Han, D., & Packer, L. (1991). Free radical recycling and intramembrane mobility in the antioxidant properties of alpha-tocopherol and alpha-tocotrienol. *Free Radical Biology and Medicine*, *10*(5), 263-275.
- Sergeant, N., Wattez, A., Galvn-valencia, M., Ghestem, A., David, J.-P., Lemoine, J., Sautire, P.-E., Dachary, J., Mazat, J.-P., & Michalski, J.-C. (2003). Association of ATP synthase α-chain with neurofibrillary degeneration in Alzheimer's disease. *Neuroscience*, 117(2), 293-303.
- Serrano-Pozo, A., Frosch, M. P., Masliah, E., & Hyman, B. T. (2011). Neuropathological alterations in Alzheimer disease. *Cold Spring Harbor Perspectives in Medicine*, 1(1), a006189.
- Shahidi, F., & Zhong, Y. (2005). Antioxidants: regulatory status. *Bailey's Industrial Oil and Fat Products*.

- Sharma, S., Schommer, E., Feist, G., & Ghribi, O. (2008). Hypercholesterolemiainduced A $\beta$  accumulation in rabbit brain is associated with alteration in IGF-1 signaling. *Neurobiology of Disease*, 32(3), 426-432.
- Sheng, M. (2001). The postsynaptic NMDA-receptor—PSD-95 signaling complex in excitatory synapses of the brain. *Journal of Cell Science*, 114(7), 1251.
- Sheng, M., Sabatini, B. L., & Südhof, T. C. (2012). Synapses and Alzheimer's disease. *Cold Spring Harbor Perspectives in Biology*, 4(5), a005777.
- Shevchenko, G., Sjödin, M. O., Malmström, D., Wetterhall, M., & Bergquist, J. (2010). Cloud-point extraction and delipidation of porcine brain proteins in combination with bottom-up mass spectrometry approaches for proteome analysis. *Journal of Proteome Research*, 9(8), 3903-3911.
- Shevchenko, G., Wetterhall, M., Bergquist, J., Höglund, K., Andersson, L. I., & Kultima, K. (2012). Longitudinal characterization of the brain proteomes for the tg2576 amyloid mouse model using shotgun based mass spectrometry. *Journal* of Proteome Research, 11(12), 6159-6174.
- Shie, F.-S., Jin, L.-W., Cook, D. G., Leverenz, J. B., & LeBoeuf, R. C. (2002). Dietinduced hypercholesterolemia enhances brain Aβ accumulation in transgenic mice. *Neuroreport*, 13(4), 455-459.
- Shin, H. W., Kim, H., & Lee, K. J. (2015). Differences in BDNF serum levels in patients with Alzheimer's Disease and mild cognitive impairment. *Journal of Psychiatry*, 2015.
- Shineman, D. W., Basi, G. S., Bizon, J. L., Colton, C. A., Greenberg, B. D., Hollister, B. A., Lincecum, J., Leblanc, G. G., Lee, L. B. H., & Luo, F. (2011). Accelerating drug discovery for Alzheimer's disease: best practices for preclinical animal studies. *Alzheimer's research & therapy*, 3(5), 28.
- Shiozaki, A., Tsuji, T., Kohno, R., Kawamata, J., Uemura, K., Teraoka, H., & Shimohama, S. (2004). Proteome analysis of brain proteins in Alzheimer's disease: subproteomics following sequentially extracted protein preparation. *Journal of Alzheimer's Disease*, 6(3), 257-268.
- Shobab, L. A., Hsiung, G.-Y. R., & Feldman, H. H. (2005). Cholesterol in Alzheimer's disease. *The Lancet Neurology*, 4(12), 841-852.
- Siemers, E., Skinner, M., Dean, R. A., Gonzales, C., Satterwhite, J., Farlow, M., Ness, D., & May, P. C. (2005). Safety, tolerability, and changes in amyloid  $\beta$  concentrations after administration of a  $\gamma$ -secretase inhibitor in volunteers. *Clinical Neuropharmacology*, 28(3), 126-132.
- Sierra, S., Ramos, M. C., Molina, P., Esteo, C., Vázquez, J. A., & Burgos, J. S. (2010). Statins as neuroprotectants: a comparative in vitro study of lipophilicity, bloodbrain-barrier penetration, lowering of brain cholesterol, and decrease of neuron cell death. *Journal of Alzheimer's disease: JAD, 23*(2), 307-318.

- Sierra, S., Ramos, M. C., Molina, P., Esteo, C., Vázquez, J. A., & Burgos, J. S. (2011). Statins as neuroprotectants: a comparative in vitro study of lipophilicity, bloodbrain-barrier penetration, lowering of brain cholesterol, and decrease of neuron cell death. *Journal of Alzheimer's Disease*, 23(2), 307-318.
- Sies, H. (2015). Oxidative stress: a concept in redox biology and medicine. *Redox Biology*, *4*, 180-183.
- Simons, K., & Ikonen, E. (1997). Functional rafts in cell membranes. *Nature*, 387(6633), 569-572.
- Simons, M., Schwärzler, F., Lütjohann, D., Von Bergmann, K., Beyreuther, K., Dichgans, J., Wormstall, H., Hartmann, T., & Schulz, J. B. (2002). Treatment with simvastatin in normocholesterolemic patients with Alzheimer's disease: A 26-week randomized, placebo-controlled, double-blind trial. *Annals of Neurology*, 52(3), 346-350.
- Simpkins, J. W., Green, P. S., Gridley, K. E., Singh, M., de Fiebre, N. C., & Rajakumar, G. (1997). Role of estrogen replacement therapy in memory enhancement and the prevention of neuronal loss associated with Alzheimer's disease. *The American Journal of Medicine*, 103(3), 19S-25S.
- Sinzinger, H., Wolfram, R., & Peskar, B. A. (2002). Muscular side effects of statins. Journal of Cardiovascular Pharmacology, 40(2), 163-171.
- Sirtori, C. R., Galli, C., Anderson, J. W., Sirtori, E., & Arnoldi, A. (2009). Functional foods for dyslipidaemia and cardiovascular risk prevention. *Nutrition Research Reviews*, 22(02), 244-261.
- Sizova, D., Charbaut, E., Delalande, F., Poirier, F., High, A. A., Parker, F., Van Dorsselaer, A., Duchesne, M., & Diu-Hercend, A. (2007). Proteomic analysis of brain tissue from an Alzheimer's disease mouse model by two-dimensional difference gel electrophoresis. *Neurobiology of Aging*, 28(3), 357-370.
- Sjögren, M., Mielke, M., Gustafson, D., Zandi, P., & Skoog, I. (2006). Cholesterol and Alzheimer's disease—is there a relation? *Mechanisms of Ageing and Development*, 127(2), 138-147.
- Skovronsky, D. M., Moore, D. B., Milla, M. E., Doms, R. W., & Lee, V. M.-Y. (2000). Protein kinase C-dependent α-secretase competes with β-secretase for cleavage of amyloid-β precursor protein in the trans-Golgi network. *Journal of Biological Chemistry*, 275(4), 2568-2575.
- Sly, W. S., & Hu, P. Y. (1995). Human carbonic anhydrases and carbonic anhydrase deficiencies. *Annual Review of Biochemistry*, 64(1), 375-401.
- Smina, T., Mathew, J., Janardhanan, K., & Devasagayam, T. (2011). Antioxidant activity and toxicity profile of total triterpenes isolated from *Ganoderma lucidum* (Fr.) P. Karst occurring in South India. *Environmental Toxicology and Pharmacology*, 32(3), 438-446.

- Sohma, H., Imai, S.-i., Takei, N., Honda, H., Matsumoto, K., Utsumi, K., Matsuki, K., Hashimoto, E., Saito, T., & Kokai, Y. (2014). Evaluation of annexin A5 as a biomarker for Alzheimer's disease and dementia with lewy bodies. *Clinical Use* of Biomarkers for Neurodegenerative Disorders, 7.
- Sorci-Thomas, M., Prack, M. M., Dashti, N., Johnson, F., Rudel, L., & Williams, D. (1989). Differential effects of dietary fat on the tissue-specific expression of the apolipoprotein AI gene: relationship to plasma concentration of high density lipoproteins. *Journal of Lipid Research*, 30(9), 1397-1403.
- Sottero, B., Gamba, P., Gargiulo, S., Leonarduzzi, G., & Poli, G. (2009). Cholesterol oxidation products and disease: an emerging topic of interest in medicinal chemistry. *Current Medicinal Chemistry*, 16(6), 685-705.
- Spagnuolo, M. S., Maresca, B., La Marca, V., Carrizzo, A., Veronesi, C., Cupidi, C., Piccoli, T., Maletta, R. G., Bruni, A. C., & Abrescia, P. (2014). Haptoglobin interacts with apolipoprotein E and beta-amyloid and influences their crosstalk. ACS Chemical Neuroscience, 5(9), 837-847.
- Sparks, D. L., Scheff, S. W., Hunsaker, J. C., Liu, H., Landers, T., & Gross, D. R. (1994). Induction of Alzheimer-like β-amyloid immunoreactivity in the brains of rabbits with dietary cholesterol. *Experimental Neurology*, 126(1), 88-94.
- Spiliotis, E. T. (2010). Regulation of microtubule organization and functions by septin GTPases. *Cytoskeleton*, 67(6), 339-345.
- Spires-Jones, T. L., & Hyman, B. T. (2014). The intersection of amyloid beta and tau at synapses in Alzheimer's disease. *Neuron*, 82(4), 756-771.
- Squire, L. R., & Kandel, E. R. (2000). Memory: From mind to molecules: Macmillan.
- Staels, B., Dallongeville, J., Auwerx, J., Schoonjans, K., Leitersdorf, E., & Fruchart, J.-C. (1998). Mechanism of action of fibrates on lipid and lipoprotein metabolism. *Circulation*, 98(19), 2088-2093.
- Stagi, M., Dittrich, P. S., Frank, N., Iliev, A. I., Schwille, P., & Neumann, H. (2005). Breakdown of axonal synaptic vesicle precursor transport by microglial nitric oxide. *The Journal of Neuroscience*, 25(2), 352-362.
- Stampfer, M. (2006). Cardiovascular disease and Alzheimer's disease: common links. Journal of Internal Medicine, 260(3), 211-223.
- Stanimirovic, D. B., & Friedman, A. (2012). Pathophysiology of the neurovascular unit: disease cause or consequence&quest. *Journal of Cerebral Blood Flow & Metabolism*, 32(7), 1207-1221.
- Stefani, M., & Rigacci, S. (2013). Protein folding and aggregation into amyloid: the interference by natural phenolic compounds. *International Journal of Molecular Sciences*, 14(6), 12411-12457.

- Steinberg, D., Parthasarathy, S., Carew, T. E., Khoo, J. C., & Witztum, J. L. (1989). Beyond cholesterol. Modifications of low-density lipoprotein that increase its atherogenicity. *New England journal of medicine*, 321(17), 1196-1197.
- Stepanichev, M. Y., Moiseeva, Y. V., Lazareva, N., Onufriev, M., & Gulyaeva, N. (2003). Single intracerebroventricular administration of amyloid-beta (25–35) peptide induces impairment in short-term rather than long-term memory in rats. *Brain Research Bulletin*, 61(2), 197-205.
- Stepanichev, M. Y., Zdobnova, I., Zarubenko, I., Lazareva, N., & Gulyaeva, N. (2006). Studies of the effects of central administration of β-amyloid peptide (25–35): pathomorphological changes in the hippocampus and impairment of spatial memory. *Neuroscience and Behavioral Physiology*, 36(1), 101-106.
- Stocker, R., & Keaney, J. F. (2004). Role of oxidative modifications in atherosclerosis. *Physiological Reviews*, *84*(4), 1381-1478.
- Stojković, D. S., Barros, L., Calhelha, R. C., Glamočlija, J., Ćirić, A., Van Griensven, L. J., Soković, M., & Ferreira, I. C. (2014). A detailed comparative study between chemical and bioactive properties of *Ganoderma lucidum* from different origins. *International Journal of Food Sciences and Nutrition*, 65(1), 42-47.
- Stolberg, A. (2005). Ranking of memories and behavioral strategies in the radial maze. *Acta Neurobiologiae Experimentalis*, 65(1), 39-49.
- Sudheesh, N. P., Ajith, T. A., Mathew, J., Nima, N., & Janardhanan, K. K. (2012). Ganoderma lucidum protects liver mitochondrial oxidative stress and improves the activity of electron transport chain in carbon tetrachloride intoxicated rats. Hepatology Research, 42(2), 181-191.
- Sudhof, T. C. (2004). The synaptic vesicle cycle. *Annual Review of Neuroscience*, 27, 509.
- Sultana, R., Banks, W. A., & Butterfield, D. A. (2010a). Decreased levels of PSD95 and two associated proteins and increased levels of BCl2 and caspase 3 in hippocampus from subjects with amnestic mild cognitive impairment: insights into their potential roles for loss of synapses and memory, accumulation of A $\beta$ , and neurodegeneration in a prodromal stage of Alzheimer's disease. *Journal of Neuroscience Research*, 88(3), 469-477.
- Sultana, R., Boyd-Kimball, D., Cai, J., Pierce, W. M., Klein, J. B., Merchant, M., & Butterfield, D. A. (2007). Proteomics analysis of the Alzheimer's disease hippocampal proteome. *Journal of Alzheimer's Disease*, *11*(2), 153-164.
- Sultana, R., Boyd-Kimball, D., Poon, H. F., Cai, J., Pierce, W. M., Klein, J. B., Markesbery, W. R., Zhou, X. Z., Lu, K. P., & Butterfield, D. A. (2006a). Oxidative modification and down-regulation of Pin1 in Alzheimer's disease hippocampus: a redox proteomics analysis. *Neurobiology of Aging*, 27(7), 918-925.

- Sultana, R., Boyd-Kimball, D., Poon, H. F., Cai, J., Pierce, W. M., Klein, J. B., Merchant, M., Markesbery, W. R., & Butterfield, D. A. (2006b). Redox proteomics identification of oxidized proteins in Alzheimer's disease hippocampus and cerebellum: an approach to understand pathological and biochemical alterations in AD. *Neurobiology of Aging*, 27(11), 1564-1576.
- Sultana, R., Perluigi, M., Newman, S. F., Pierce, W. M., Cini, C., Coccia, R., & Butterfield, D. A. (2010b). Redox proteomic analysis of carbonylated brain proteins in mild cognitive impairment and early Alzheimer's disease. *Antioxidants & Redox Signaling*, 12(3), 327-336.
- Sultana, R., Poon, H. F., Cai, J., Pierce, W. M., Merchant, M., Klein, J. B., Markesbery, W. R., & Butterfield, D. A. (2006c). Identification of nitrated proteins in Alzheimer's disease brain using a redox proteomics approach. *Neurobiology of Disease*, 22(1), 76-87.
- Sun, K. H., Chang, K. H., Clawson, S., Ghosh, S., Mirzaei, H., Regnier, F., & Shah, K. (2011). Glutathione-S-transferase P1 is a critical regulator of Cdk5 kinase activity. *Journal of Neurochemistry*, 118(5), 902-914.
- Surh, Y.-J., Kundu, J. K., Li, M.-H., Na, H.-K., & Cha, Y.-N. (2009). Role of Nrf2mediated heme oxygenase-1 upregulation in adaptive survival response to nitrosative stress. Archives of Pharmaceutical Research, 32(8), 1163-1176.
- Sutton, M. A., & Schuman, E. M. (2006). Dendritic protein synthesis, synaptic plasticity, and memory. *Cell*, 127(1), 49-58.
- Suzanne, M., & Wands, J. R. (2008). Alzheimer's disease is type 3 diabetes—evidence reviewed. *Journal of Diabetes Science and Technology*, 2(6), 1101-1113.
- Suzuki, T., Hide, I., Ido, K., Kohsaka, S., Inoue, K., & Nakata, Y. (2004). Production and release of neuroprotective tumor necrosis factor by P2X7 receptor-activated microglia. *The Journal of Neuroscience*, 24(1), 1-7.
- Sweeney, W. A., Luedtke, J., McDonald, M. P., & Overmier, J. B. (1997). Intrahippocampal injections of exogenous β-amyloid induce postdelay errors in an eight-arm radial maze. *Neurobiology of Learning and Memory, 68*(1), 97-101.
- Swerdlow, R. H. (2007). Pathogenesis of Alzheimer's disease. *Clinical Interventions in Aging*, 2(3), 347.
- Swerdlow, R. H., & Khan, S. M. (2004). A "mitochondrial cascade hypothesis" for sporadic Alzheimer's disease. *Medical Hypotheses*, 63(1), 8-20.
- Tacnet-Delorme, P., Chevallier, S., & Arlaud, G. J. (2001).  $\beta$ -amyloid fibrils activate the C1 complex of complement under physiological conditions: evidence for a binding site for A $\beta$  on the C1q globular regions. *The Journal of Immunology*, *167*(11), 6374-6381.

- Tahara, K., Kim, H.-D., Jin, J.-J., Maxwell, J. A., Li, L., & Fukuchi, K.-i. (2006). Role of toll-like receptor signalling in Aβ uptake and clearance. *Brain*, *129*(11), 3006-3019.
- Takano, M., Yamashita, T., Nagano, K., Otani, M., Maekura, K., Kamada, H., Tsunoda, S.-i., Tsutsumi, Y., Tomiyama, T., & Mori, H. (2013). Proteomic analysis of the hippocampus in Alzheimer's disease model mice by using two-dimensional fluorescence difference in gel electrophoresis. *Neuroscience Letters*, 534, 85-89.
- Taler, V., & Phillips, N. A. (2008). Language performance in Alzheimer's disease and mild cognitive impairment: a comparative review. *Journal of Clinical and Experimental Neuropsychology*, 30(5), 501-556.
- Tamagno, E., Bardini, P., Guglielmotto, M., Danni, O., & Tabaton, M. (2006). The various aggregation states of β-amyloid 1–42 mediate different effects on oxidative stress, neurodegeneration, and BACE-1 expression. *Free Radical Biology and Medicine*, 41(2), 202-212.
- Tamaoka, A., Miyatake, F., Matsuno, S., Ishii, K., Nagase, S., Sahara, N., Ono, S., Mori, H., Wakabayashi, K., & Tsuji, S. (2000). Apolipoprotein E allele– dependent antioxidant activity in brains with Alzheimer's disease. *Neurology*, 54(12), 2319-2321.
- Tang, M.-X., Jacobs, D., Stern, Y., Marder, K., Schofield, P., Gurland, B., Andrews, H., & Mayeux, R. (1996). Effect of oestrogen during menopause on risk and age at onset of Alzheimer's disease. *The Lancet*, 348(9025), 429-432.
- Tanzi, R. E. (2012). The genetics of Alzheimer disease. Cold Spring Harbor Perspectives in Medicine, 2(10), a006296.
- Tariot, P. N., Farlow, M. R., Grossberg, G. T., Graham, S. M., McDonald, S., Gergel, I., & Group, M. S. (2004). Memantine treatment in patients with moderate to severe Alzheimer disease already receiving donepezil: a randomized controlled trial. JAMA, 291(3), 317-324.
- Teixeira, J., Gaspar, A., Garrido, E. M., Garrido, J., & Borges, F. (2013). Hydroxycinnamic acid antioxidants: an electrochemical overview. *BioMed Research International, 2013*.
- Terry, A. V., & Buccafusco, J. J. (2003). The cholinergic hypothesis of age and Alzheimer's disease-related cognitive deficits: recent challenges and their implications for novel drug development. *Journal of Pharmacology and Experimental Therapeutics*, 306(3), 821-827.
- Teyler, T., & DiScenna, P. (1987). Long-term potentiation. Annual Review of Neuroscience, 10(1), 131-161.
- Thirumangalakudi, L., Prakasam, A., Zhang, R., Bimonte-Nelson, H., Sambamurti, K., Kindy, M. S., & Bhat, N. R. (2008). High cholesterol-induced neuroinflammation and amyloid precursor protein processing correlate with loss of working memory in mice. *Journal of Neurochemistry*, 106(1), 475-485.

- Thomas, D., Guthridge, M., Woodcock, J., & Lopez, A. (2005). 9 14-3-3 Protein Signaling in Development and Growth Factor Responses. *Current Topics in Developmental Biology*, 67, 286-305.
- Tietz, N., Burtis, C., Duncan, P., Ervin, K., Petitclerc, C., Rinker, A., Shuey, D., & Zygowicz, E. (1983). A reference method for measurement of alkaline phosphatase activity in human serum. *Clinical Chemistry*, 29(5), 751-761.
- Tilleman, K., Stevens, I., Spittaels, K., Haute, C. V. d., Clerens, S., Van den Bergh, G., Geerts, H., Van Leuven, F., Vandesande, F., & Moens, L. (2002). Differential expression of brain proteins in glycogen synthase kinase-3 transgenic mice: A proteomics point of view. *Proteomics*, 2(1), 94-104.
- Todaro, L., Puricelli, L., Gioseffi, H., Pallotta, M. a. G., Lastiri, J., de Kier Joffé, E. B., Varela, M., & de Lustig, E. S. (2004). Neural cell adhesion molecule in human serum. Increased levels in dementia of the Alzheimer type. *Neurobiology of Disease*, 15(2), 387-393.
- Toomey, S., Roche, H., Fitzgerald, D., & Belton, O. (2003). Regression of preestablished atherosclerosis in the apoE-/- mouse by conjugated linoleic acid. *Biochemical Society Transactions*, 31(5), 1075-1079.
- Tosto, G., & Reitz, C. (2013). Genome-wide association studies in Alzheimer's disease: a review. *Current Neurology and Neuroscience Reports, 13*(10), 1-7.
- Townsend, M., Shankar, G. M., Mehta, T., Walsh, D. M., & Selkoe, D. J. (2006). Effects of secreted oligomers of amyloid β-protein on hippocampal synaptic plasticity: a potent role for trimers. *The Journal of Physiology*, *572*(2), 477-492.
- Toyn, J. H., & Ahlijanian, M. K. (2014). Interpreting Alzheimer's disease clinical trials in light of the effects on amyloid-β. *Alzheimers Res Ther*, *6*(2), 14.
- Tsai-Teng, T., Chin-Chu, C., Li-Ya, L., Wan-Ping, C., Chung-Kuang, L., Chien-Chang, S., Chi-Ying, H. F., Chien-Chih, C., & Shiao, Y.-J. (2016). Erinacine Aenriched Hericium erinaceus mycelium ameliorates Alzheimer's disease-related pathologies in APPswe/PS1dE9 transgenic mice. *Journal of biomedical science*, 23(1), 49.
- Tu, S., Okamoto, S.-i., Lipton, S. A., & Xu, H. (2014). Oligometric Aβ-induced synaptic dysfunction in Alzheimer's disease. *Mol Neurodegeneration*, *9*(1), 48.
- Turchi, G., Alagona, G., & Lubrano, V. (2009). Protective activity of plicatin B against human LDL oxidation induced in metal ion-dependent and-independent processes. Experimental and theoretical studies. *Phytomedicine*, 16(11), 1014-1026.
- Tyler, W. J., Alonso, M., Bramham, C. R., & Pozzo-Miller, L. D. (2002). From acquisition to consolidation: on the role of brain-derived neurotrophic factor signaling in hippocampal-dependent learning. *Learning & Memory*, 9(5), 224-237.

- Ullrich, C., Pirchl, M., & Humpel, C. (2010). Hypercholesterolemia in rats impairs the cholinergic system and leads to memory deficits. *Molecular and Cellular Neuroscience*, 45(4), 408-417.
- Umemura, T., Kodama, Y., Hioki, K., Inoue, T., Nomura, T., & Kurokawa, Y. (2001). Butylhydroxytoluene (BHT) increases susceptibility of transgenic rasH2 mice to lung carcinogenesis. *Journal of Cancer Research and Clinical Oncology*, 127(10), 583-590.
- Valko, M., Leibfritz, D., Moncol, J., Cronin, M. T., Mazur, M., & Telser, J. (2007). Free radicals and antioxidants in normal physiological functions and human disease. *The International Journal of Biochemistry & Cell Biology*, 39(1), 44-84.
- Valverde, M. E., Hernández-Pérez, T., & Paredes-López, O. (2015). Edible mushrooms: improving human health and promoting quality life. *International Journal of Microbiology*, 2015.
- van der Putten, H., & Lotz, G. P. (2013). Opportunities and challenges for molecular chaperone modulation to treat protein-conformational brain diseases. *Neurotherapeutics*, 10(3), 416-428.
- Vanhanen, M., & Soininen, H. (1998). Glucose intolerance, cognitive impairment and Alzheimer's disease. *Current Opinion in Neurology*, 11(6), 673-677.
- Vannucci, R. C., & Vannucci, S. J. (2000). *Glucose Metabolism in the Developing Brain.* Paper presented at the Seminars in perinatology.
- Vaz, J. A., Barros, L., Martins, A., Morais, J. S., Vasconcelos, M. H., & Ferreira, I. C. (2011). Phenolic profile of seventeen Portuguese wild mushrooms. *LWT-Food Science and Technology*, 44(1), 343-346.
- Vaziri, N. D. (2003). Molecular mechanisms of lipid disorders in nephrotic syndrome. *Kidney International*, 63(5), 1964-1976.
- Vickers, J., Kirkcaldie, M., Phipps, A., & King, A. (2016). Alterations in neurofilaments and the transformation of the cytoskeleton in axons may provide insight into the aberrant neuronal changes of Alzheimer's disease. *Brain Research Bulletin*, 126(3), 324-333.
- Vijayan, S., El-Akkad, E., Grundke-Iqbal, I., & Iqbal, K. (2001). A pool of β-tubulin is hyperphosphorylated at serine residues in Alzheimer disease brain. *FEBS Letters*, 509(3), 375-381.
- Visioli, F., Bordone, R., Perugini, C., Bagnati, M., Cau, C., & Bellomo, G. (2000). The kinetics of copper-induced LDL oxidation depend upon its lipid composition and antioxidant content. *Biochemical and Biophysical Research Communications*, 268(3), 818-822.
- Wachtel-Galor, S., Tomlinson, B., & Benzie, I. F. (2004). Ganoderma lucidum ('Lingzhi'), a Chinese medicinal mushroom: biomarker responses in a controlled human supplementation study. British Journal of Nutrition, 91(02), 263-269.

- Wahrle, S., Das, P., Nyborg, A. C., McLendon, C., Shoji, M., Kawarabayashi, T., Younkin, L. H., Younkin, S. G., & Golde, T. E. (2002). Cholesterol-dependent  $\gamma$ -secretase activity in buoyant cholesterol-rich membrane microdomains. *Neurobiology of Disease*, 9(1), 11-23.
- Waldemar, G., Dubois, B., Emre, M., Georges, J., McKeith, I., Rossor, M., Scheltens, P., Tariska, P., & Winblad, B. (2007). Recommendations for the diagnosis and management of Alzheimer's disease and other disorders associated with dementia: EFNS guideline. *European Journal of Neurology*, 14(1), e1-e26.
- Walsh, D. M., & Selkoe, D. J. (2007). Aβ oligomers–a decade of discovery. Journal of Neurochemistry, 101(5), 1172-1184.
- Wang, J.-Z., Zhu, W.-D., Xu, Z.-X., Du, W.-T., Zhang, H.-Y., Sun, X.-W., & Wang, X.-H. (2014). Pin1, endothelial nitric oxide synthase, and amyloid-β form a feedback signaling loop involved in the pathogenesis of Alzheimer's disease, hypertension, and cerebral amyloid angiopathy. *Medical Hypotheses*, 82(2), 145-150.
- Wang, M.-F., Chan, Y.-C., Wu, C.-L., Wong, Y.-C., Hosoda, K., & Yamamoto, S. (2004). Effects of Ganoderma on aging and learning and memory ability in senescence accelerated mice. Paper presented at the International Congress Series.
- Wang, Q., Wu, J., Rowan, M. J., & Anwyl, R. (2005). β-amyloid inhibition of long-term potentiation is mediated via tumor necrosis factor. *European Journal of Neuroscience*, 22(11), 2827-2832.
- Wang, X.-M., Yang, M., Guan, S.-H., Liu, R.-X., Xia, J.-M., Bi, K.-S., & Guo, D.-A. (2006). Quantitative determination of six major triterpenoids in *Ganoderma lucidum* and related species by high performance liquid chromatography. *Journal of Pharmaceutical and Biomedical Analysis*, 41(3), 838-844.
- Wasser, S. P. (2002). Medicinal mushrooms as a source of antitumor and immunomodulating polysaccharides. *Applied Microbiology and Biotechnology*, 60(3), 258-274.
- Wasser, S. P. (2014). Medicinal mushroom science: Current perspectives, advances, evidences, and challenges. *Biomedical Journal*, 37(6), 345.
- Watanabe, T., Iwasaki, K., Takasaki, K., Yamagata, N., Fujino, M., Nogami, A., Ii, M., Katsurabayashi, S., Mishima, K., & Fujiwara, M. (2010). Dynamin 1 depletion and memory deficits in rats treated with Aβ and cerebral ischemia. *Journal of Neuroscience Research*, 88(9), 1908-1917.
- Weggen, S., Eriksen, J. L., Das, P., Sagi, S. A., Wang, R., Pietrzik, C. U., Findlay, K. A., Smith, T. E., Murphy, M. P., & Bulter, T. (2001). A subset of NSAIDs lower amyloidogenic Aβ42 independently of cyclooxygenase activity. *Nature*, 414(6860), 212-216.
- Weinstein, G., Beiser, A. S., Preis, S. R., Courchesne, P., Chouraki, V., Levy, D., & Seshadri, S. (2016). Plasma clusterin levels and risk of dementia, Alzheimer's

disease, and stroke. Alzheimer's & Dementia: Diagnosis, Assessment & Disease Monitoring, 3, 103-109.

- Werner, M., Gabrielson, D. G., & Eastman, J. (1981). Ultramicro determination of serum triglycerides by bioluminescent assay. *Clinical Chemistry*, 27(2), 268-271.
- Winblad, B., & Jelic, V. (2004). Long-term treatment of Alzheimer disease: efficacy and safety of acetylcholinesterase inhibitors. *Alzheimer Disease & Associated Disorders*, 18, S2-S8.
- Wiseman, H. (1993). Vitamin D is a membrane antioxidant ability to inhibit irondependent lipid peroxidation in liposomes compared to cholesterol, ergosterol and tamoxifen and relevance to anticancer action. *FEBS Letters*, 326(1-3), 285-288.
- Wolf, H., Hensel, A., Arendt, T., Kivipelto, M., Winblad, B., & Gertz, H. J. (2004). Serum lipids and hippocampal volume: the link to Alzheimer's disease? *Annals of Neurology*, 56(5), 745-749.
- Wolozin, B. (2001). A fluid connection: cholesterol and Aβ. Proceedings of the National Academy of Sciences, 98(10), 5371-5373.
- Wolozin, B., Kellman, W., Ruosseau, P., Celesia, G. G., & Siegel, G. (2000). Decreased prevalence of Alzheimer disease associated with 3-hydroxy-3-methyglutaryl coenzyme A reductase inhibitors. *Archives of Neurology*, 57(10), 1439-1443.
- Wong, K. H., Sabaratnam, V., Abdullah, N., Kuppusamy, U. R., & Naidu, M. (2009). Effects of cultivation techniques and processing on antimicrobial and antioxidant activities of *Hericium erinaceus* (Bull.: Fr.) Pers. extracts. *Food Technology and Biotechnology*, 47(1), 47.
- Wood, W. G., Schroeder, F., Igbavboa, U., Avdulov, N. A., & Chochina, S. V. (2002). Brain membrane cholesterol domains, aging and amyloid beta-peptides. *Neurobiology of Aging*, 23(5), 685-694.
- Wu, F., & Yao, P. J. (2009). Clathrin-mediated endocytosis and Alzheimer's disease: an update. *Ageing Research Reviews*, 8(3), 147-149.
- Wyss-Coray, T., & Mucke, L. (2002). Inflammation in neurodegenerative disease—a double-edged sword. *Neuron*, 35(3), 419-432.
- Yamaguchi, M., Kokai, Y., Imai, S. I., Utsumi, K., Matsumoto, K., Honda, H., Mizue, Y., Momma, M., Maeda, T., & Toyomasu, S. (2010). Investigation of annexin A5 as a biomarker for Alzheimer's disease using neuronal cell culture and mouse model. *Journal of Neuroscience Research*, 88(12), 2682-2692.
- Yamamoto, K., Shimada, H., Koh, H., Ataka, S., & Miki, T. (2014). Serum levels of albumin-amyloid beta complexes are decreased in Alzheimer's disease. *Geriatrics & Gerontology International*, 14(3), 716-723.

- Yamasaki, N., Maekawa, M., Kobayashi, K., Kajii, Y., Maeda, J., Soma, M., Takao, K., Tanda, K., Ohira, K., & Toyama, K. (2008). Alpha-CaMKII deficiency causes immature dentate gyrus, a novel candidate endophenotype of psychiatric disorders. *Molecular Brain*, 1(1), 1.
- Yan, R., Fan, Q., Zhou, J., & Vassar, R. (2016). Inhibiting BACE1 to reverse synaptic dysfunctions in Alzheimer's disease. *Neuroscience & Biobehavioral Reviews*, 65, 326-340.
- Yao, J., Irwin, R. W., Zhao, L., Nilsen, J., Hamilton, R. T., & Brinton, R. D. (2009). Mitochondrial bioenergetic deficit precedes Alzheimer's pathology in female mouse model of Alzheimer's disease. *Proceedings of the National Academy of Sciences*, 106(34), 14670-14675.
- Yao, J., Taylor, M., Davey, F., Ren, Y., Aiton, J., Coote, P., Fang, F., Chen, J. X., Du Yan, S., & Gunn-Moore, F. J. (2007). Interaction of amyloid binding alcohol dehydrogenase/Aβ mediates up-regulation of peroxiredoxin II in the brains of Alzheimer's disease patients and a transgenic Alzheimer's disease mouse model. *Molecular and Cellular Neuroscience*, 35(2), 377-382.
- Ye, L., Li, J., Zhang, J., & Pan, Y. (2010). NMR characterization for polysaccharide moiety of a glycopeptide. *Fitoterapia*, 81(2), 93-96.
- Yerbury, J. J., Poon, S., Meehan, S., Thompson, B., Kumita, J. R., Dobson, C. M., & Wilson, M. R. (2007). The extracellular chaperone clusterin influences amyloid formation and toxicity by interacting with prefibrillar structures. *The FASEB Journal*, 21(10), 2312-2322.
- Yildiz, O., Can, Z., Laghari, A. Q., Şahin, H., & Malkoç, M. (2015). Wild edible mushrooms as a natural source of phenolics and antioxidants. *Journal of Food Biochemistry*, 39(2), 148-154.
- Yongpan, H., Bingjin, L., Nanhui, Y., Xianlin, L., Xianglin, K., & Jun, L. (2016). Study of *Ganoderma lucidum* three terpenoids protective effect on brain tissue of AD rat model (2010-02-05 16:53:39 ed.): http://www.yourpaper.net/article/20100205/240333.html#.V1j37tJ97IU.
- Yoo, B. C., Fountoulakis, M., Cairns, N., & Lubec, G. (2001). Changes of voltagedependent anion-selective channel proteins VDAC1 and VDAC2 brain levels in patients with Alzheimer's disease and Down Syndrome. *Electrophoresis*, 22(1), 172-179.
- Yoshida, H., & Kisugi, R. (2010). Mechanisms of LDL oxidation. *Clinica Chimica Acta*, 411(23), 1875-1882.
- Yu, J.-T., & Tan, L. (2012). The role of clusterin in Alzheimer's disease: pathways, pathogenesis, and therapy. *Molecular Neurobiology*, 45(2), 314-326.
- Zang, L.-Y., Cosma, G., Gardner, H., Shi, X., Castranova, V., & Vallyathan, V. (2000). Effect of antioxidant protection by p-coumaric acid on low-density lipoprotein cholesterol oxidation. *American Journal of Physiology-Cell Physiology*, 279(4), C954-C960.

- Zapolska-Downar, D., Siennicka, A., Kaczmarczyk, M., Kołodziej, B., & Naruszewicz, M. (2004). Butyrate inhibits cytokine-induced VCAM-1 and ICAM-1 expression in cultured endothelial cells: the role of NF-κB and PPARα. *The Journal of Nutritional Biochemistry*, 15(4), 220-228.
- Zempel, H., Thies, E., Mandelkow, E., & Mandelkow, E.-M. (2010). Aβ oligomers cause localized Ca2+ elevation, missorting of endogenous Tau into dendrites, Tau phosphorylation, and destruction of microtubules and spines. *The Journal of Neuroscience*, *30*(36), 11938-11950.
- Zetterberg, H., & Mattsson, N. (2014). Understanding the cause of sporadic Alzheimer's disease. *Expert Review of Neurotherapeutics*, 14(6), 621-630.
- Zhang, C. (2012). Natural compounds that modulate BACE1-processing of amyloidbeta precursor protein in Alzheimer's disease. *Discovery Medicine*, 14(76), 189-197.
- Zhang, L., Peng, S.-p., Han, R., Qiu, J.-h., Gu, Z.-l., & Qin, Z.-h. (2005). A lipophilic fraction of *Ganoderma lucidum* promotes PC12 differentiation. *Chinese Pharmacological Bulletin*, 21(6), 662.
- Zhang, R., Xu, S., Cai, Y., Zhou, M., Zuo, X., & Chan, P. (2011a). Ganoderma lucidum protects dopaminergic neuron degeneration through inhibition of microglial activation. Evidence-Based Complementary and Alternative Medicine, 2011.
- Zhang, X.-Q., Ip, F. C., Zhang, D.-M., Chen, L.-X., Zhang, W., Li, Y.-L., Ip, N. Y., & Ye, W.-C. (2011b). Triterpenoids with neurotrophic activity from *Ganoderma lucidum*. *Natural product research*, 25(17), 1607-1613.
- Zhang, Y., & Lee, D. H. (2011). Sink hypothesis and therapeutic strategies for attenuating Aβ levels. *The Neuroscientist*, 17(2), 163-173.
- Zhang, Y., Repa, J. J., Gauthier, K., & Mangelsdorf, D. J. (2001). Regulation of lipoprotein lipase by the oxysterol receptors, LXRα and LXRβ. *Journal of Biological Chemistry*, 276(46), 43018-43024.
- Zhao, H.-B., Lin, S.-Q., Liu, J.-H., & Lin, Z.-B. (2004). Polysaccharide extract isolated from *Ganoderma lucidum* protects rat cerebral cortical neurons from hypoxia/reoxygenation injury. *Journal of Pharmacological Sciences*, 95(2), 294-298.
- Zhao, H.-B., Wang, S.-Z., He, Q.-h., Yuan, L., Chen, A. F., & Lin, Z.-B. (2005). Ganoderma total sterol (GS) and GS 1 protect rat cerebral cortical neurons from hypoxia/reoxygenation injury. Life Sciences, 76(9), 1027-1037.
- Zhao, L., Wang, J.-L., Liu, R., Li, X.-X., Li, J.-F., & Zhang, L. (2013). Neuroprotective, anti-amyloidogenic and neurotrophic effects of apigenin in an Alzheimer's disease mouse model. *Molecules*, 18(8), 9949-9965.
- Zhou, J., Liyanage, U., Medina, M., Ho, C., Simmons, A. D., Lovett, M., & Kosik, K. S. (1997). Presenilin 1 interaction in the brain with a novel member of the Armadillo family. *Neuroreport*, 8(8), 2085-2090.

- Zhou, K., Yang, Y., Gao, L., He, G., Li, W., Tang, K., Ji, B., Zhang, M., Li, Y., & Yang, J. (2010a). NMDA receptor hypofunction induces dysfunctions of energy metabolism and semaphorin signaling in rats: a synaptic proteome study. *Schizophrenia Bulletin*, sbq132.
- Zhou, Y., Qu, Z.-q., Zeng, Y.-s., Lin, Y.-k., Li, Y., Chung, P., Wong, R., & Hägg, U. (2012). Neuroprotective effect of preadministration with *Ganoderma lucidum* spore on rat hippocampus. *Experimental and Toxicologic Pathology*, 64(7), 673-680.
- Zhou, Z.-Y., Tang, Y.-P., Xiang, J., Wua, P., Jin, H.-M., Wang, Z., Mori, M., & Cai, D.-F. (2010b). Neuroprotective effects of water-soluble *Ganoderma lucidum* polysaccharides on cerebral ischemic injury in rats. *Journal of Ethnopharmacology*, 131(1), 154-164.
- Zhu, K. X., Nie, S. P., Tan, L., Li, C., Gong, D., & Xie, M. Y. (2016). A polysaccharide from *Ganoderma atrum* improves liver function in type 2 diabetic rats via antioxidant action and short-chain fatty acids excretion. *Journal of Agricultural* and Food Chemistry, 64(9), 1938-1944.

## LIST OF PUBLICATIONS AND PAPERS PRESENTED

- Rahman, M. A., Abdullah, N., & Aminudin, N. (2014). Inhibitory effect on in vitro LDL oxidation and HMG Co-a reductase activity of the liquid-liquid partitioned fractions of *Hericium erinaceus* (Bull.) persoon (Lion's Mane Mushroom). *BioMed Research International*, 2014.
- Rahman, M. A., Abdullah, N., & Aminudin, N. (2015). Interpretation of mushroom as a common therapeutic agent for Alzheimer's disease and cardiovascular diseases. *Critical reviews in biotechnology*, 1-12.
- Rahman, M. A., Abdullah, N., & Aminudin, N. (2015). Antioxidative Effects and Inhibition of Human Low Density Lipoprotein Oxidation In Vitro of Polyphenolic Compounds in *Flammulina velutipes* (Golden Needle Mushroom). *Oxidative Medicine and Cellular Longevity*.
- Rahman, M. A., Abdullah, N., & Aminudin, N. (2016). *Lentinula edodes* (Shiitake Mushroom): an assessment of *in vitro* anti-atherosclerotic bio-functionality. *Saudi Journal of Biological Sciences*.

Hindawi Publishing Corporation BioMed Research International Volume 2014, Article ID 828149, 9 page http://dx.doi.org/10.1155/2014/828149



## Research Article

## Inhibitory Effect on *In Vitro* LDL Oxidation and HMG Co-A Reductase Activity of the Liquid-Liquid Partitioned Fractions of *Hericium erinaceus* (Bull.) Persoon (Lion's Mane Mushroom)

## Mohammad Azizur Rahman,<sup>1,2</sup> Noorlidah Abdullah,<sup>1</sup> and Norhaniza Aminudin<sup>1</sup>

Mushroom Research Centre, Institute of Biological Sciences. Faculty of Science, University of Malaya, 50603 Kuala Lumpur, Malaysia

<sup>2</sup> Department of Biochemistry and Molecular Biology, Jahangirnagar University, Savar, Dhaka 1340, Bangladesh

Correspondence should be addressed to Noorlidah Abdullah; noorlidah@um.edu.my

Received 9 February 2014; Revised 11 April 2014; Accepted 11 April 2014; Published 13 May 2014

Academic Editor: Paul Evans

Copyright © 2014 Mohammad Azizur Rahman et al. This is an open access article distributed under the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Oxidation of low-density lipoprotein (LDL) has been strongly suggested as the key factor in the pathogenesis of atherosclerosis. Mushrooms have been implicated in having preventive effects against chronic disease due especially to their antioxidant properties. In this study, *in vitro* inhibitory effect of *Hericium erinaccus* on LDL oxidation and the activity of the cholesterol biosynthetic key enzyme, 3-hydroxy-3-methyl glutaryl coenzyme A (HMG Go-A) reductase, was evaluated using five liquid-liquid solvent fractions consisting of methanol: dichloromethane (M:DCM), hexane (HEX), dichloromethane (DCM), ethyl acetate (EA), and aqueous residue (AQ). The hexane fraction showed the highest inhibition of oxidation of human LDL as reflected by the increased lag time (100 mins) for the formation of conjugated diene (CD) at 1µg/mL and decreased production (68.28%, IC<sub>50</sub> 0.73 mg/mL) of thiobarbituric acid reactive substances (TBARS) at 1 mg/mL. It also mostly inhibited (59.91%) the activity of the HMG Go-A reductase at 10 mg/mL. The GC-MS profiling of the hexane fraction identified the presence of myconutrients: *inter alia*, ergosterol and linoleic acid. Thus, hexane fraction of *Hericium erinaceus* was found to be the most potent *in vitro* inhibitor of both LDL oxidation and HMG Co-A reductase activity having therapeutic potential for the prevention of oxidative stress-mediated vascular diseases.